DISSERTATION

IFIDENTIFICATION AND CHARACTERIZATION OF LEPROSY T CELL ANTIGENS IN THE CONTEXT OF EARLY DIAGNOSIS AND CD1a RESTRICTED T CELL ACTIVATION

Submitted by

Hee Jin Kim

Department of Microbiology, Immunology and Pathology

In partial fulfillment of the requirements

For the Degree of Doctor of Philosophy

Colorado State University

Fort Collins, Colorado

Summer 2011

Doctoral Committee:

Advisor: Patrick J. Brennan

Noreen E. Reist Alan R. Schenkel Varalakshmi D. Vissa Copyright by Hee Jin Kim 2011

All Rights Reserved

ABSTRACT

IDENTIFICATION AND CHARACTERIZATION OF LEPROSY ANTIGENS IN THE CONTEXT OF EARLY DIAGNOSIS AND CD1a RESTRICTED T CELL ACTIVATION

Some of the work conducted, specifically the production of certain *M. leprae* recombinant proteins, has been published as follows: Spencer JS, <u>Kim HJ</u>, Wheat WH, Chatterjee D, Balagon MV, Cellona RV, Tan EV, Gelber R, Saunderson P, Duthie MS, Reece ST, Burman W, Belknap R, Mac Kenzie WR, Geluk A, Oskam L, Dockrell HM, Brennan PJ; on behalf of the IDEAL Consortium. Analysis of Antibody Responses to *Mycobacterium leprae* Phenolic Glycolipid I, Lipoarabinomannan, and Recombinant Proteins To Define Disease Subtype-Specific Antigenic Profiles in Leprosy. Clin Vaccine Immunol. 2011. 18:260-267.

The progression of leprosy disease highly correlates with predominant type 1/type 2 cytokines and proliferation of Th1/Th2 cells in response to *M. leprae*. Th1 immunity is thought to be protective to *M. leprae* infection but Th2 immunity is pathogenic. The primary objective of this dissertation was to identify and characterize a panel of new protein based T cell antigens, particularly those that induce Th1 immunity, in order to understand host-*M. leprae* interaction and to develop diagnostic tests for leprosy.

Cellular mediated immunity (CMI) based *in vitro* interferon-γ released assay (IGRA) to *M. leprae* specific antigens is known as the most promising diagnostic test to

detect those individuals in the early stages of *M. leprae* infection. Here, we identified two classes of antigens using advanced genomic and proteomic approaches.

Comparative bioinformatic analysis among mycobacterial genomes identified potential *M. leprae* unique proteins called "hypothetical unknowns" from the *M. leprae* genome. We performed cDNA based quantitative real time PCR to investigate the expression status of 136 putative open reading frames (ORFs) encoding hypothetical unknowns. Twenty six of *M.leprae* specific antigen candidates showed significant levels of gene expression compared to that of ESAT-6 (ML0049) which is an important T cell antigen of low abundance. Fifteen out of 26 selected antigen candidates were expressed and purified in *E. coli*. Serological analysis using the sera obtained from lepromatous leprosy patients and cavitary tuberculosis patients clearly indicated that 9 out of 26 selected proteins elicited *M.leprae* specific immune responses.

M. leprae membrane antigen (MLMA) induces a strong Th1 immune response. However, the large amount of lipomannans/lipoarabinomannans (LMs/LAMs) content hindered the further processing of this fraction and the optimal use of this fraction as diagnostic reagents in a CMI based assay. A simple sodium carbonate treatment successfully enriched membrane proteins into MLMA-SP (alkali soluble proteins of MLMA) and excluded the most hydrophobic lipids. Subsequent IFN-γ release assay (IGRA) of peripheral blood mononuclear Cells (PBMCs) from 16 mycobacterial laboratory/leprosy clinic workers revealed that MLMA-SP induced higher levels of IFN-γ secretion in the group exposed to leprosy patients, as compared to the group with environmental/professional mycobacterial exposure. Therefore, MLMA-SP and 9 of the

iii

hypothetical unknowns may be good diagnostic reagents to improve both sensitivity and specificity in detection of individuals with asymptomatic leprosy.

Bacterial lipoproteins, containing an N-acyl di-O-acylglyceryl-cysteine unit, have been known to induce strong CMI response to bacterial infections through Toll like receptor (TLR-2) signaling. CD1-restricted T cells in response to mycobacterial lipid antigens efficiently and rapidly induce cytolytic T cell response to kill intracellular pathogens. In this work, we developed simple proteomic and chemical methodologies to enrich lipoproteins of low abundance from MLMA. T cell reactivity to these lipoprotein enriched fractions combined with proteolysis provided the first evidence that bacterial lipoproteins are a new class of CD1a restricted antigens recognized by a CD1a restricted T cell line derived from skin lesions of a leprosy patient. Further proteomic and genomic analyses identified lipoproteins, ML1086, ML2095 and ML2446, as CD1a antigen candidates. However, these three lipoproteins expressed in recombinant form in M. smegmatis were unable to activate a CD1a restricted T cell line (LCD4.15) though the acylation of all recombinant proteins was biochemically verified. The possible reasons for this result are: 1) that the antigen recognized by LCD4.15 would be another lipoprotein of the least abundance making it difficult to identify using current proteomic analysis; 2) that the bacterial lipoproteins of *in vitro* grown *M. smegmatis* could present different acylation patterns from those of *in vivo* grown *M. leprae*.

The presence of acylated cysteine at the N terminus of lipoprotein is primarily predicted by bioinformatic analysis but rarely verified by biochemical analysis. Lipopeptides of mycobacterial lipoproteins were prepared by proteolysis and organic extraction, and analyzed by gas chromatography-mass spectrometry (GC-MS) and

iv

quadrapole time of flight liquid chromatography/mass spectrometry (Q-TOF LC/MS). GC-MS results revealed that the major fatty acids of the glyceryl portion of the lipoproteins are tuberculostearic acid, stearic acid, oleic acid and palmitic acid, which are also the dominant fatty acids of mycobacterial phospholipids. The N-linked fatty acid of the cysteine residue is predominantly palmitic acid. Interestingly, Q-TOF LC/MS data showed that mycobacterial lipoproteins appeared as di- and tri-acylated mature forms, but also had an array of fatty acids in various combinations. Di-acylated lipopeptides appeared more frequently than tri-acylated lipopeptides. The present study demonstrated a distinct fatty acid composition of the lipid portion of mycobacterial lipoproteins compared to other bacterial lipoproteins. Moreover, we defined the full spectrum of acylation pattern in tri- or di-acyled mycobacterial lipoproteins, the impact of which in host-bacteria responses is subject to further immunological investigation.

Here, we have identified and characterized a panel of new *M. leprae* derived protein antigens by developing novel biochemical and proteomic techniques. The knowledge and methologies presented in this dissertation will allow for further understanding of the physiology and pathogenicity of *M. leprae* in human hosts.

ACKNOWLEDGEMENTS

I would like to express my gratitude to my adviser and mentor, Dr. Patrick J. Brennan, for his full support of my graduate research as well as my scientific career for last 13 years. His encouragement and guidance helped me to complete this dissertation and steered my academic growth as a researcher. I would like to thank my graduate committee members, Dr. Noreen Reist, Dr. Alan Schenkel and Dr. Varalakshmi Vissa for their participation, guidance and productive critics regarding my graduate research. I thank Dr. Robert Modlin, Dr. Peter Sieling and Dr. John Belisle for a valuable collaboration.

I am deeply grateful to my colleagues in the Mycobacterial Research Laboratories for sharing their expertise and friendship. Especially, I thank to all members of Leprosy Research group for their friendship and support.

I really appreciate my family for their support. I thank my parents for their endless love. I am profoundly indebted to my husband, Dr. Sebabrata Mahapatra, for his encouragement, understanding, sacrifice, patience and support during these difficult times. I also want to give my special thanks to my child, Ananya, for being a happy kid in spite of Mom's absence.

vi

DEDICATION

This dissertation is dedicated to my husband, Sebabrata Mahapatra, and my child, Ananya Mahapatra, who have given their love and strength to accomplish my desire. I hope to be with Ananya one day, as she faces her own challenges and difficulties to pursue her dream.

CHAPTER 1 LITERATURE REVIEW: LEPROSY.

HISTORY, TREATMENT, EPIDERMIOLOGY, THE BACTERIUM, IMMUNE RESPONSE AND DIAGNOSIS

| <u>1.1 History</u> | 2 |
|---|-------|
| <u>1.2 The disease</u> | 4 |
| 1.2.1 Clinical features | 4 |
| 1.2.2 Nerve damage | 5 |
| <u>1.3 Diagnosis</u> | 7 |
| <u>1.4 Treatment</u> | 8 |
| 1.4.1 Early form of chemotherapy | 8 |
| 1.4.2 Multidrug Therapy | 10 |
| <u>1.5 Epidemiology</u> | 10 |
| 1.6 The Bacterium; Mycobacterium leprae | 12 |
| 1.6.1 Genome | 13 |
| 1.6.2 Proteome | 16 |
| 1.6.3 Unique cell envelope | 17 |
| 1.6.3.1 Peptidoglycan | 20 |
| 1.6.3.2 Arabinogalactan | 21 |
| 1.6.3.3 Mycolic acids | 23 |
| 1.6.3.4 Phostphatidylinositol mannosides (PIMs), lipomannans (LMs) | |
| and lipoarabinomannans (LAMs) | 24 |
| 1.6.3.5 Glycolipids: Phenolic glycolipids (PGLs) and phthiocerol dimycocero | sates |
| (PDIM);Trehalose containing glycolipids (TDMs, TMMs) | 26 |
| 1.6.3.6 Outer membrane (OM) | 29 |
| 1.6.3.7 Biosynthesis and genetics of cell wall components | 30 |
| 1.7 Post-translational modification of mycobacterial proteins | 31 |
| 1.7.1 Lipoproteins | 32 |
| 1.7.1.1 Identification | 33 |
| 1.7.1.2 Biosynthesis | 35 |
| 1.7.1.3 Transport | 37 |
| 1.7.1.4 Localization | 40 |
| 1.7.1.5 Physiological function | 41 |

| | 1.7.1.5.a | Transport; Solute binding proteins of ATP Binding Cassette | |
|-----------------|--------------------|--|----|
| | t | ransport transport systems and lipid carriers | 44 |
| | 1.7.1.5.b | Enzymes | 46 |
| | 1.7.1.5.c | Adhesion and cell invasion | 46 |
| | 1.7.1.5.d | Signaling and sensory function | 46 |
| | 1.7.1.5.e | Unknown function | 47 |
| 1.7.2 | Glycoprotein | S | 47 |
| 1.7.2 | 2.1 Identification | ation and structure | 47 |
| 1.7.2 | 2.2 Biosynth | esis and function | 49 |
| 1.7.3 | Phosphoprot | eins | 50 |
| 1.7.4 l | Methylated p | proteins | 53 |
| 1.7.5 | Pupylated pr | oteins (Pupylation: prokaryotic ubiquitin-like | |
| ŀ | protein modi | fication) | 54 |
| <u>1.8 Imm</u> | une response | 2 | 56 |
| 1.8.1 1 | Innate immu | nity | 60 |
| 1.8.2 | Immune mod | ulation | 66 |
| 1.8.3 (| CD1 restrict | ed T cells | 68 |
| <u>1.9 Host</u> | genetic infl | uences on human leprosy | 70 |
| <u>1.10 Imm</u> | une based di | agnosis | 73 |
| 1.10.1 | Skin test | | 74 |
| 1.10.2 A | Antigen iden | tification | 75 |
| 1.10.3 I | In vitro Inter | feron-γ Release Assays (IGRAs) | 80 |
| 1.10.4 | Serodiagnos | is | 81 |
| 1.11 Refe | rences | | 84 |

CHAPTER 2 GENE EXPRESSION PROFILE AND IMMUNOGICAL EVALUATION OF UNIQUE HYPOTHETICAL UNKNOWN PROTEINS OF *MYCOBACTERIUM LEPRAE* BY USING QUANTITATIVE REAL TIME-PCR

| 2.1 | Abstract 10 | | |
|-----|---|-----|--|
| 2.2 | 2 Introduction 1 | | |
| 2.3 | <u>3 Materials and methods</u> 10 | | |
| 2.3 | 1.1 Isolation of M. leprae RNA | 106 | |
| 2.3 | 2.2 Primer design for quantitative real time PCR (qRT-PCR) analysis | 107 | |
| 2.3 | 2.3 Synthesis of complementary DNA (cDNA) and qRT-PCR assay | 108 | |

| 2.3.4 | Cloning of hypothetic unknowns ORFs from M. leprae | 108 |
|---------------|---|-----|
| 2.3.5 | Purification of recombinant M. leprae proteins | 110 |
| 2.3.6 | Subjects and samples | 111 |
| 2.3.7 | Western blot analysis | 111 |
| 2.4 Results 1 | | |
| 2.4.1 | Establishment of the gene expression profile of M. leprae specific hypothetical | |
| unkn | own ORFs and selection of novel antigen candidates | 112 |
| 2.4.2 | Purification and serological reactivity | |
| of | the recombinant M. leprae specific antigens | 120 |
| <u>2.5 Di</u> | scussion | 121 |
| <u>2.6 Re</u> | ferences | 126 |

CHAPTER 3 EVALUATION OF NOVEL ANTIGENS FROM *MYCOBACTERIUM LEPRAE* CYTOSOLIC MEMBRANE FOR EARLY DIAGNOSIS

| 3.1 | Abst | <u>ract</u> | 129 |
|-----|--------|--|----------------|
| 3.2 | Intro | duction | 130 |
| 3.3 | Mate | rials and methods | 132 |
| 3. | 3.1 1 | solation of M. leprae and preparation | |
| | (| of M. leprae membrane antigen (MLMA) | 132 |
| 3 | 3.2 | Protein enrichment of the MLMA | 134 |
| 3 | 3.3 | C cell stimulation assay | 136 |
| | 3.3.3 | 1 Study subjects | 136 |
| | 3.3.3 | 2 T cell stimulation assay with MLMA derived antigens and IFN- γ Enzyme |) - |
| | | Linked ImmunoSorbent Assay (ELISA) | 136 |
| | 3.3.3 | 3 PGL-I rapid lateral flow test | 137 |
| | 3.3.3 | 4 Statistical analysis | 138 |
| 3.4 | Resu | lts | 138 |
| 3.4 | 4.1 \$ | Subfractionation of MLMA by sodium carbonate treatment | 138 |
| 3.4 | 4.2 0 | Classification of the study groups | 140 |
| 3.4 | 4.3 1 | Production of IFN- γ by PBMCs in response to MLMA derived antigens | 142 |
| 3.5 | Disc | ussion | 144 |
| 3.6 | Refe | rences | 149 |

CHAPTER 4 IDENTIFICATION AND BIOCHEMICAL DEFINITION OF NEW MYCOBACTERIUM LEPRAE ANTIGENS PRESENTED BY CD1a OF LANGERHANS CELLS

| 4.1 | 4.1 Introduction 15. | | |
|-----|----------------------|---|-----|
| 4.2 | Ma | terials and methods | 155 |
| 4 | .2.1 | T cell lines, proliferation and IFN- γ released assays | 155 |
| 4 | .2.2 | Proteinase K treatment of M. leprae membrane antigen (MLMA) | 156 |
| 4 | .2.3 | Sub-fractionation of MLMA | 157 |
| | 4.2. | 3.1 Anion exchange chromatography of MLMA-P for preparation | |
| | | of a lipoprotein enriched fraction. | 158 |
| | 4.2. | 3.2 Size Exclusion (SE) fractionation of MLMA-P | 158 |
| | 4.2. | 3.3 Preparative isoelectric Focusing (IEF) fractionation of MLMA | 159 |
| 4 | .2.4 | Limulus Amoebocyte Lysate Test, bicinchoninic acid assay | |
| | and | l SDS-PAGE analysis of protein fractions | 159 |
| 4 | .2.5 | Butanol extraction of MLMA and MLMA-P | 160 |
| 4 | .2.6 | In gel tryptic digestion and electrospray ionization-MS analysis | 160 |
| 4 | .2.7 | Analysis of lipoproteins by using a bacterial lipoprotein database | |
| | | and BLASTP algorithms | 162 |
| 4 | .2.8 | Cloning of M. leprae lipoproteins into M. smegmatis expression vector | |
| | | <i>pVV16</i> | 162 |
| 4 | .2.9 | Subcellular fractionation of M. smegmatis and detection | |
| | of t | he recombinant lipoproteins | 163 |
| 4 | .2.10 | Purification of recombinant lipoproteins | 164 |
| 4 | .2.11 | Construction of the hybrid Lpr1086 by using the signal sequence | |
| | | of M. tuberculosis LprG (Rv1411c) | 164 |
| 4.3 | Res | <u>ults</u> | 165 |
| 4 | .3.1 | Characterization of the LCD4.15 T-cell clone | 165 |
| 4 | .3.2 | Activation of LCD4.15 in response to MLMA derived protein subfractions | 167 |
| 4 | .3.3 | Identification of a CD1a restricted M. leprae specific antigen by using | |
| | | mimic Multi-Dimensional Protein Identification Technology | 169 |
| 4 | .3.4 | Bioinformatics based identification of a novel CD1a restricted antigen | 173 |
| 4 | .3.5 | Expression of putative lipoproteins | 173 |
| 4 | .3.6 | Purification, characterization and evaluation of T cell reactivity | |
| | | of recombinant Lpr1086, ML2095 and ML2446 | 175 |

| <u>4.4</u> | Discussion 177 | | |
|--------------|---|---------|--|
| <u>4.5</u> | 182 <u>182</u> | | |
| СНАР | PTER 5 STRUCTURAL ANALYSIS OF MYCOBACTERIAL LIPOPI | EPTIDES | |
| 5.1 | <u>Abstract</u> | 186 | |
| 5.2 | Introduction | 187 | |
| <u>5.3 l</u> | Materials and methods | 189 | |
| 5.3. | 1 Preparation of a hydrophobic protein fraction from Mycobacterium smeg | gmatis | |
| | through Triton X-114 phase separation | 189 | |
| 5.3.2 | 2 Proteolysis of M. smegmatis hydrophobic proteins and release | | |
| | of the ester-linked fatty acids | 190 | |
| 5.3 | 3 Purification and analysis of lipopeptides from M. smegmatis | | |
| | hydrophobic proteins | 190 | |
| 5.3.4 | 4 Quadrapole time of flight (Q-TOF) LC/MS analysis of lipopeptides | | |
| | from recombinant lipoproteins Lpr1086, ML2095 and ML2446 | 191 | |
| <u>5.4</u> | Results | 192 | |
| 5.4. | 1 Preparation and analysis of lipoproteins of M. smegmatis | 192 | |
| 5.4.2 | 2 Purification and analysis of lipopeptides | 194 | |
| 5.4 | 3 Identification of lipopeptides from recombinant Lpr1086, ML2095 | | |
| | and ML2446 by Q-TOF LC/MS | 196 | |
| <u>5.5</u> | Discussion | 198 | |
| <u>5.6</u> | References | 204 | |
| | | | |

CHAPTER 6 CONCLUSIONS AND FUTURE DIRECTIONS

| <u>6.1</u> | Leprosy diagnostic antigens | 208 |
|------------|---|-----|
| <u>6.2</u> | Identification and characterization of lipoproteins as CD1a restricted antigens | 212 |
| 6.3 | References | 218 |

List of Figures

| Figure | | Pages |
|--------|---|-------|
| | | |
| 1.1 | Disease spectrum of leprosy | 5 |
| 1.2 | Proposed model of infection of peripheral nerve by <i>M. leprae</i> . | 6 |
| 1.3 | Global prevalence and incidence of leprosy, 1985 -2009 | 12 |
| 1.4 | Schematic representation of the cell envelope of Mycobacterium spp | 19 |
| 1.5 | Structure of the repeating unit of mycobacterial peptidoglycan | 21 |
| 1.6 | Structure of mycolyl arabinogalactan and its attachment to peptidoglycan | 22 |
| 1.7 | Structures of mycolic acids | 24 |
| 1.8 | Schematic structures of mycobacterial PIM ₆ and ManLAMs | 25 |
| 1.9 | Structures of phthiocerol dimycocerosates (PDIM) | |
| | and Phenolic glycolipids (PGLs) from M.leprae | 27 |
| 1.10 | Structures of trehalose monomycolate and trehalose dimycolate | 28 |
| 1.11 | Models of the mycobacterial outer memebrane (OM) | 30 |
| 1.12 | General structure of a mycobacterial lipoprotein | 32 |
| 1.13 | The tripartite region of signal peptides recognized by either Type I | |
| | or Type II signal peptidase in Gram-positive or Gram-negative bacteria | 34 |
| 1.14 | Biosynthesis of bacterial lipoproteins | 36 |
| 1.15 | Schematic overview of the secretion (Sec) and twin-arginine translocation (TAT) | |
| | pathways | 39 |
| 1.16 | Schematic overview of phosphorylation and signal transduction | |
| | of bacterial proteins | 51 |
| 1.17 | Comparison of the ubiquitin and prokaryotic ubiquitin-like protein (Pup) | |
| | proteasome pathways | 54 |
| 1.18 | Gene arrangement of Pup and proteasome in both M. tuberculosis | |
| | and <i>M. leprae</i> | 56 |
| 1.19 | Immune response in tuberculoid leprosy (TT) lesions | 58 |
| 1.20 | Myeloid differentiation factor 88 (MyD88) dependent | |
| | Toll like receptor-2 (TLR-2) signaling pathway | 61 |
| 1.21 | Immune response in lepromatous leprosy (LL) lesions | 67 |
| 1.22 | Chemical structures of CD1 restricted mycobacterial antigens | 69 |
| 1.23 | Two-stage model of genetic susceptibility on human leprosy | 71 |
| 2.1 | Summary of the hypothetical unknowns investigated in present study | 113 |

| 2.2 | Western blot analysis of the recombinant forms of the selected hypothetical | |
|-----|---|-----|
| | unknown proteins | 120 |
| 3.1 | Experimental scheme for the fractionation | |
| | of <i>M. leprae</i> membrane antigen (MLMA) | 135 |
| 3.2 | SDS-PAGE and Western blot analysis | |
| | of subfractionated proteins of MLMA | 139 |
| 3.3 | IFN- γ production by PBMCs induced in response to PHA and PPD | 143 |
| 3.4 | IFN-γ production by PBMCs induced in response | |
| | to three of the MLMA derived subfractions | 143 |
| 4.1 | Crystal structure of murine CD1d loaded with a synthetic α -galactosylceramide | 154 |
| 4.2 | Experimental scheme for the sub-fractionation of MLMA | 157 |
| 4.3 | Proliferative response and IFN- γ production of LCD4.15 | 166 |
| 4.4 | SDS-PAGE analysis of MLMA derived fractions and IFN- γ production | |
| | by LCD4.15 induced in response to antigens | 167 |
| 4.5 | Thin layer chromatography analysis and T cell reactivity of MLMA | |
| | and MLMA-P before/after delipidation | 168 |
| 4.6 | Size exclusion (SE) fractionation of MLMA-P | 170 |
| 4.7 | Preparative Isoelectric Focusing (IEF) fractionation of MLMA | 171 |
| 4.8 | Subcellular localization of recombinant lipoproteins | |
| | by using Western blot with anti-6-Histidine tag antibody | 174 |
| 4.9 | SDS-PAGE analysis and CD1a-restricted T cell response | |
| | to recombinant ML2446, ML2095 and Lpr1086 from M. smegmatis | 176 |
| 5.1 | Proteolysis of lipoproteins and thin layer chromatography analysis | |
| | of the alkali-treated lipopeptides | 193 |
| 5.2 | Purification of lipopeptides by reverse-phase HPLC | 194 |
| 5.3 | Analysis of the fatty acids of lipopeptides by GC-MS | 195 |
| 5.4 | Amino acid analysis of lipopeptides by GC-MS | 196 |
| 5.5 | N- terminal amino acid sequence of mature ML2095, ML2446 and Lpr1086 | |
| | and map of lipopeptides observed by Q-TOF LC/MS analysis | 197 |

Lists of Tables

| Table | | Page |
|-------|---|---------|
| 1.1 | Comparison of genome features of M. leprae TN and M. tuberculosis | 14 |
| 1.2 | Functional classification of genes in M. leprae | 15 |
| 1.3 | Functional categories of M. leprae and M. tuberculosis putative lipoproteins | 42-43 |
| 1.4 | Mycobacterial phosphoproteins | 52 |
| 1.5 | TLR recognition of mycobacterial components | 65 |
| 1.6 | M. leprae antigens and their potential use in diagnostics | 77-78 |
| 2.1 | List of primers used in cloning of 24 selected hypothetical unknown ORFs | 109 |
| 2.2 | Analysis of qRT-PCR of 136 hypothetical unknown ORFs | 115-117 |
| 2.3 | Analysis of qRT-PCR of the 26 selected hypothetical unknown ORFs | 118 |
| 2.4 | Amino acid sequence similarity of the 26 selected M. leprae-specific proteins | 119 |
| 3.1 | Characteristics of participating study population | 141 |
| 4.1 | List of primers used in cloning of putative lipoprotein ORFs | 163 |
| 4.2 | List of CD1a antigen candidates indentified by proteomic | |
| | analysis of both IEF fraction 4 and SE fraction 8 | 172 |
| 4.3 | List of eight lipoprotein candidates selected by bioinformatics | |
| | and proteomics analyses | 173 |
| 5.1 | List of lipopeptides observed by Q-TOF LC/MS analysis | 198 |

List of Abbreviations

| 1,25D3 | 1,25-dihydroxyvitamin D_3 or 1,25-dihydroxycholecalciferol |
|-----------------|--|
| 2D | Two-Dimensional |
| 19 kDa/LpqH | 19 kDa M. tuberculosis Lipoprotein Antigen |
| 33 kDa/LpK | 33 kDa M. leprae Lipoprotein Antigen |
| 38 kDa/PhoS1 | 38 kDa M. tuberculosis Lipoprotein Antigen |
| 45/47-kDa/MPT32 | 45/47-kDa M. tuberculosis Glycoprotein Antigen |
| AA | Amino Acid |
| Ala, A | Alanine |
| AraLAM | Uncapped Lipoarabinomannan |
| Arg, R | Arginine |
| Asp, D | Asparagine |
| BB | Mid Borderline Leprosy |
| BCG | M. bovis Bacillus Calmette-Guérim; a vaccine for tuberculosis |
| BL | Borderline Lepromatous Leprosy |
| BSA | Bovine Serum Albumin |
| BT | Borderline Tuberculoid Leprosy |
| C _# | Carbon _{Number} |
| CD^+ | Cluster of Differentiation Restricted |
| CFP-10 | 10 kDa M. tuberculosis Culture Filtrated Proteins, ML0050/Rv3874 |
| CMI | Cell-Mediated Immunity |
| C-terminus | Carboxyl Terminus |
| CYP27b1 | 25-Hydroxy Vitamin D3-1α-Hydroxylase |
| Cys | Cysteine |
| DC-SIGN | Dendritic Cell-Specific Intracellular Adhesion Molecule-Grabbing Nonintegrin |
| ELISA | Enzyme-Linked Immunosorbent Assay |
| ERK1/2 | Extracellular Signal-Regulated Kinase (ERK), |
| ESAT-6 | Early Secretory Antigenic Target of 6 kDa, ML0049/Rv3875 |
| GC | Gas Chromatography |
| Gly, G | Glycine |
| Gln, Q | Glutamine |
| Glu, E | Glutamate |
| ННС | Household Contacts |
| HLA | Human Leukocyte Antigen |
| IEF | Isoelectric Focusing |
| Ile, I | Isoleucine |
| IFN-γ | Interferon-y |
| IGRA | IFN-γ Release Assay |
| IRAK-1 or 4 | IL-1 Receptor-Associated Kinase-1 or 4 |
| Leu, L | Leucine |

| LAM | Lipoarabinomannan |
|--------------|---|
| LepLAM | LAM of <i>M. leprae</i> |
| Lgt | Diacylglyceryl Transferase |
| LID-1 | Recombinant Fusion Protein containing ML0405 and ML2331 |
| LL | Lepromatous Leprosy |
| LM | Lipomannan |
| Lnt | N-Acyl Transferase |
| Lpr1086 | Recombinant Hybrid Protein of ML1086 and Signal Sequence of LprG Expressed in M. smegmatis (Signal sequence of LprG was used to generate acylated form of ML1086 but removed during lipid modification) |
| LprG | 26kDa Mycobacterial Lipoprotein, ML0557/Rv1411c |
| LspA | Signal Peptidase II |
| Lys, K | Lysine |
| Met, M | Methionine |
| MALDI-TOF | Matrix-Assisted Laser Desorption/Ionization-Time of Flight |
| ManLAM | Mannose-Capped LAM |
| МАРК | Mitogen – Activated Protein Kinase |
| MARCO | Macrophage Receptor with Collageneous Structure |
| MB | Multibacillary Leprosy |
| MCP | Major Cytosolic Protein |
| MHC | Major Histocompatibility Complex |
| ML#### | Sanger Identification Number of M. leprae Protein |
| ML Flow test | M. leprae Lateral Flow Test |
| MLCwA | M. leprae Cell Wall Antigens Excluding LAMs |
| MLMA | M. leprae Membrane Antigen or Membrane Fractions of M. leprae |
| MLMA-InsP | Alkali Insoluble (Sodium Carbonate Soluble) Proteins of of MLMA |
| MLMA-P | Trion X-114 Insoluble Fraction of MLMA-InsP |
| MLMA-SP | Alkali Soluble (Sodium Carbonate Soluble) Proteins of MLMA; Peripheral Membrane Protein of MLMA |
| MAD | M. teprae Soluble (Cytosol) Antigens Excluding LAMs |
| MS | Mago Spectrometry |
| | Musloid Differentiation Easter 88 |
| | Nuclear Easter Kenne Light Chain Enhancer of Activated P colle |
| NF-KB | Nuclear Factor Rappa-Light-Cham-Enhancer of Activated B cens |
| NUD | Outer Membrane |
| OM | Oner Des line France |
| OKF | Open Reading Frame |
| ramr | Pattorn Associated Molecular Pattern |
| rb DDMC | Paucibacillary Leprosy |
| PEMC | Peripheral Blood Mononuclear Cell |
| PCR | Polymerase Chain Reaction |

| PDIM | Phthiocerol Dimycocerosates |
|------------|---|
| PGL | Phenolic Glycolipids |
| Phe, F | Phenylalanine |
| PIMs | Phosphatidylinositol Mannosides |
| PPD | Purified Protein Derivative of M. tuberculosis |
| Pup | Prokaryotic Ubiquitin-Like Protein |
| Pupylation | Prokaryotic Ubiquitin-Like Protein Modification |
| qRT-PCR | Quantitative Real Time PCR |
| QTOF | Quadrapole Time of Flight |
| Rv#### | Sanger Identification Number of M. tuberculosis Protein |
| SE | Size Exclusion |
| Sec | Secretion |
| Ser, S | Serine |
| Sod | Superoxide Dismutase |
| TAT | Twin-Arginine Translocation |
| ТВ | Tuberculosis |
| TDM | Trehalose Dimycolate |
| TGF-β | Transforming Growth Factor- β |
| THP-1 | Human Acute Monocytic Leukemia Cell Line |
| Thr, T | Threonine |
| TICAM1 | Toll /Interleukin -1 Receptor -Domain-Containing Molecule 1 |
| TIRAP/MAL | TIR Associated Protein /MyD88 Adaptor Like |
| TLR | Toll Like Receptor |
| TMM | Trehalose Monomycolate |
| TNF | Tumor Necrosis Factor |
| TRAF-6 | TNF Receptor Associated Factor-6 |
| TRAM | TRIF Related Adaptor Molecule (TRAM) |
| TRIF | TIR-Domain Containing Adaptor Protein-Inducing IFN-β |
| Trp, W | Tryptophan |
| TT | Tuberculoid Leprosy |
| Tyr, Y | Tyrosine |
| Val, V | Valine |
| VDR | Vitamin D Receptor |
| WHO | World Health Organization |
| | |

CHAPTER 1 LITERATURE REVIEW: LEPROSY HISTORY, TREATMENT, EPIDERMIOLOGY, THE BACTERIUM, IMMUNE RESPONSE AND DIAGNOSIS

The primary objective of this dissertation is to identify and characterize a panel of new protein antigens that induce Th1 immunity. *Mycobacterium leprae*, the causative agents of leprosy, persist in the human host for long periods without presenting obvious clinical symptoms. Th1/Th2 immune response to M. leprae plays an important role in determining to develop leprosy in affected individuals. Most individuals exposed to the bacterium do not develop disease, but instead manifest cellular mediated immune (CMI) response to the bacterium. We hypothesized that a set of *M. leprae* antigens trigger protective Th1 immunity in host-pathogen interactions and also allow for the improved specificity and sensitivity of diagnosis to detect individuals in the early stages of *M*. <u>leprae infection.</u> To test these hypotheses, we embarked on the following two specific aims. In aim one, advanced bioinformatic and proteomic approaches enhanced the optimal selection of *M. leprae* antigens that can be used to detect CMI responses of *M. leprae* infected individuals, but not those of individuals with other mycobacterial infection or exposure. In aim two, M. leprae lipoproteins that induce protective CD1a restricted T cell immunity in leprosy skin lesions were identified and characterized by biochemical and genomic approaches combined with a T cell reactive assay.

Post-translational modification of mycobacterial proteins diversifies biological roles of proteins in host-pathogen interactions. <u>Additionally, this dissertation</u> <u>hypothesized that mycobacterial lipoproteins contain the distinct mycobacterial fatty acid</u> (tuberculostearic acid) and tri- or di-acyl glyceryl-cysteine unit at their N-terminus. To test this hypothesis, we developed the biochemical methodologies and defined the chemical structure of the lipid portion of the mycobacterial lipoproteins

1.1 History

Leprosy, caused by *Mycobacterium leprae*, predominantly appears in skin, peripheral nerves and body parts that are cooler than normal body temperature such as ear lobes, and other anterior parts of the body (127). Disease progression leads to disfiguring skin lesions, nerve damage and permanent disability (112,262). Leprosy is an ancient disease described in many archaic writings (148). The first description of this disease appears in the Ebers Papyrus, an ancient Egyptian medical document written around 1550 B.C. (148). However, the description of leprosy manifestations in these ancient medical texts is still controversial because the descriptions match other skin diseases (dermatitis, gas gangrene and fungal infections) (148). Later, in the 6th century B.C. (63,156), *Sushruta samhita* and Kautilya's *Arthashastra* in India and *Nei Ching* in the Chou Dynasty of China described the more typical clinical leprosy symptoms, which included sensory changes and deformities. In the West, leprosy first appeared in the records of ancient Greece after the army of Alexander the Great came back from India, and then in Rome in 62 B.C (168). Recent paleontological studies from ancient human skeletal

remains revealed that lepromatous leprosy had existed from 2000 B.C. in India (168,237,244).

There are two hypotheses about the origin and transmission of human leprosy. One hypothesis, proposed from paleontological analysis, is that leprosy extending from India was the origin of Western or Eastern Asian and North African leprosy (244); the other hypothesis supported from recent phylogeographical analysis of single nucleotide polymorphisms is that leprosy originated in Eastern Africa and then disseminated through Europe to the remainder of the world with continuous human migration (191).

Given the incurable deformity, unknown etiological features, misunderstandings, and abhorrence of the disease, led people to think that leprosy was a curse or punishment from the gods (38,112). Consequently, leprosy was left to be treated by priests or holy men, not physicians. Since the disease often appeared in family members, some people thought it was hereditary (112). Therefore, leprosy patients and their contacts were restricted to live in isolated leprosaria.

In 1873, the Norwegian scientist, Dr. Gerhard Henrik Armauer Hansen, discovered rod shaped bacteria, *M. leprae*, from leprous nodules under a microscope (38,112). He proved that leprosy is an infectious disease, not hereditary or the result of a divine curse (112,145,303). By using aniline dyes, Neisser was able to stain *M. leprae* in leprosy patients' tissue samples provided by Dr. Hansen and proved Dr. Hansen's theory (38,303). Therefore, leprosy is also known as Hansen's disease to honor Dr. Hansen's role in this significant discovery. In addition, leprosy is often called "Hansen's disease" by many clinicians in an attempt to protect patients from the stigmas associated with the term leprosy (38).

1.2 The disease

1.2.1 Clinical features

Leprosy presents as two contrasting spectra correlating with the type of host immune response against the pathogen and resulting in bacterial and disease progression (185). At one end of the disease spectrum, patients with tuberculoid leprosy (TT) or paucibacillary leprosy (PB) exhibit a strong CMI which is responsible for the protective immunity to *M. leprae* (194,262,272). Affected skin lesions contain few bacilli and well defined granulomas consisting of Langerhans cells, CD4⁺ and CD8⁺ T-cells (262,272).

At the opposite end of the spectrum, patients with lepromatous leprosy (LL) or multibacillary leprosy (MB) exhibit a strong humoral immune response and a high number of bacilli (175,262). Patients present numerous skin and nodular lesions on all parts of the body. The biopsies of LL patients contain a number of foamy lipid-laden macrophages with diffused abundant bacilli. According to the Ridley-Jopling classification (241), the disease spectrum is subdivided into five parts: TT, borderline tuberculoid (BT), borderline borderline (BB), borderline lepromatous (BL) to LL forms (Figure 1.1).

Immunohistochemistry and gene expression profile analysis by using RT-PCR demonstrated that the Th1 cytokines such as IL-2, IL-12 and IFN- γ , were predominantly expressed in the skin lesions of TT patients. In contrast, Th2 cytokines including IL-4, IL-5, and IL-10 predominated in the skin lesions of LL patients (186,272) (Figure 1.1; further discussed in section 1.8).



Figure 1.1 Disease spectrum of leprosy (adapted from Misch *et al.* **(183) with permission).** Clinical manifestation of leprosy is classified as TT, BT, BB, BL and LL (see above for definitions). The humoral immunity of host and bacterial load highly correlates.

1.2.2 Nerve damage

One hallmark of leprosy is damage to the peripheral nerves, causing sensory and motor impairment, with characteristic limb deformities and disability. *M. leprae*, as the causative agent of leprosy, preferentially invade and grow in the Schwann cells surrounding the axons of peripheral nerve cells. The tropism of *M. leprae* to Schwann cell is associated with the tissue specific expression of laminin-2 on the membrane of Schwann cell (183). By using a rat Schwann cell axon coculture system, Rambukkana *et al.* (234) demonstrated that the trisaccharide unit of phenolic glycolipid-I (PGL-I) on *M. leprae* binds specifically to the domain of the α 2 chain of native laminin-2 in the basal lamina of the Schwann cell. In addition, *M. leprae* laminin binding protein 21 which is heavily methylated (see section 1.7.4) also mediates the intracellular entry into Schwann cells (183). Collectively, according to Rambukkana *et al.* (233), PGL-I and the laminin binding protein 21 are the key determinants in the tropism of *M. leprae* for peripheral

nerve invasion and induce demyelination which results in neurodegenerative disease (Figure 1.2).



Figure 1.2 Proposed model of infection of peripheral nerve by *M. leprae* (adapted from Misch *et al.* (183) with permission). In non-immune-mediated nerve injury, laminin binding protein 21 (LBP21) and PGL-I in the *M. leprae* cell wall bind to the α 2 chain of laminin-2 (LAMA2) and α -dystroglycan on the Schwann cell membrane permitting entry and subsequent damage to the peripheral nerve. In immune mediated mechanism, the activated macrophages induce proinflammatory cytokines, NO, influx of immune cells and Th1 cellular immunity leading to nerve damage.

Cell surface molecules of *M. leprae* usually activate monocytes or macrophage to induce proinflammatory cytokines such as TNF, IL-12, IL-6, IL-1, IL-15 and nitric oxide (NO). Nerve biopsies from leprosy patients reveal perineural and intraneural inflammation (183). The spontaneous leprosy reactions are acute inflammatory complications considered as medical emergencies during the course of treated or untreated leprosy (262). It may cause irreversible tissue damage and nerve destruction in leprosy patients if not treated timely. Two major clinical types of leprosy reactions are known: type 1 "reversal reactions" and type 2 "erythema nodosum leprosum" (262). Type 1 reactions frequently occur in BL, BB and BT patients and are due to the sudden increases of CMI response to *M. leprae* antigens (183). However, type 2 reactions often exhibit in MB patients with poor CMI against *M. leprae* (183). These patients often experience neuritis with sensory and motor neuropathy (262). However, the triggering factors and mechanism of leprosy reactions are still unknown.

The influx of immune cells and edema via inflammatory response to *M. leprae* could cause the immune-mediated nerve injury (183). By using *in vitro* stimulation assays, this hypothesis indirectly supported that 19 kDa *M. tuberculosis* lipoprotein, which is representing mycobacterial toll like receptor-2 (TLR-2) ligand, recognized TLR-2/1 on the cultured Schwann cells and induced innate immune response (214) (Figure 1.2).

1.3 Diagnosis

Current standard diagnosis of leprosy by the World Health Organization (WHO) is based on the presence of one or more of the clinical signs as follows: 1) hypopigmented or reddish skin lesion consistent with leprosy and with definite sensory loss; and 2) thickened peripheral nerves (311). For treatment purposes, histopathological classification in leprosy diagnosis is important to determine the type and course of chemotherapy (94,312).

Major problems to leprosy diagnosis are that:1) there is no measure to detect individuals with asymptomatic *M. leprae* infection who could be healthy household contacts (HHCs) of MB patients but mostly have no known history of close contact with leprosy patients (81-83,277); 2) clinical symptoms-based diagnosis and a microscopic diagnosis by acid-fast bacilli staining is insensitive to greater than 70% of infected

patients, especially PB/TT patients (30); and 3) acid-fast bacilli staining on skin slit smear or biopsy specimen is rarely applied, nowadays, in routine diagnosis. Moreover, a microscopic diagnosis and classification by counting acid-fast bacilli in skin slit smear or biopsy is highly dependent on well-trained leprologists or laboratory technicians. Therefore, these problems often result in delayed and missed diagnosis, permitting disease progression and increasing potential disease transmission. Since multi-drug therapy (described in section 1.4.2) is capable to sterilize *M. leprae* infected individuals, early diagnosis and correct classification of leprosy are major requirements to control leprosy and eventually to prevent irreversible disability of leprosy patients.

PCR based assays targeting either *M. leprae* antigen coding sequences (18kDa/, 65 kDa/ GroEL antigens or 36kDa/Proline-rich antigen) or repeated DNA sequences, improves both sensitivity and specificity of leprosy diagnostics by detecting 95% of MB/LL patients and 55% of PB/TT patients (30,94,251). However, this advanced molecular technique cannot be used to detect preclinical leprosy due to poor resources in endemic regions. During the last decade, research towards immune based diagnostic assays has made rapid progress in early diagnosis of leprosy. Immune based diagnosis of leprosy is described in section 1.10 following a discussion of the major antigens of *M. leprae*.

1.4 Treatment

1.4.1 Early form of chemotherapy

Along with description of symptoms, the treatment of leprosy with Chaulmoogra oil was mentioned in a number of ancient medical documents (63,202). In the late 19th

century, a British surgeon serving in India introduced this plant into the West (63,202). It was extensively used on leprosy patients until the 1940s. Chaulmoogra oil consists of hydnocarpic and chaulmoogric acids. Early studies hypothesized that these compounds might inhibit the biosynthesis of bacterial cell envelope and/ or enhance the host defense mechanism (126). Jacobsen and Levy (125) showed that the growth of *M. intracellulare* was susceptible to 2 μ g of hydnocarpic acid. Subsequently, the presence of biotin in culture media attenuated this action of hydnocarpic acid suggesting that the action mechanism of hydnocarpic acid might be related with biotin metabolism of bacteria (125). However, Levy (150) demonstrated that the multiplication of *M. leprae* in mouse food pad infection model was highly susceptible to chaulmoogric acids but not to hydnocarpic acids. The action mechanism of Chaulmoogra oil in the treatment of *M. leprae* infection is still obscure.

Later in 1937, a sulfone derivative (sodium salt of p,p'-diamninodiphenyl-sulfone-N-didextrose sulfate) was synthesized and put on the market under the name Promin by Parke, Davis and Co.. This drug was investigated for the treatment of various acid-fast bacilliary infections (327). A clinical study on leprosy patients reported that Promin inhibited the progress of leprosy in a significant number of cases. This resulted in its approval as standard chemotherapy for the treatment of human leprosy (154,155,327). However, long-term dapsone monotherapy with poor compliance resulted in the failure of the treatment and consequently dapsone-resistant leprosy emerged (262,264). Additional anti-microbial agents such as rifampicin and clofazimine were then introduced to treat dapsone-resistant leprosy (262,265).

1.4.2 Multidrug Therapy

In order to resolve the emergence of drug-resistant leprosy, the WHO recommended multidrug therapy as the new standard regimen for leprosy control in 1981 (265). The initial regimen consisted of a combined daily dose of dapsone and clofazimine and a monthly dose of rifampicin and clofazimine for two years or until skin smears became negative (262). In order to reduce the economical burden in developing countries, the multidrug therapy regimen was continuously modified and the period of treatment was shortened (30). The current regimen for MB patients consists of receiving 600 mg of rifampicin once a month, 100mg of dapsone daily, 300 mg of clofazimine once a month as well as 50 mg daily for 12 months. PB patients are treated with a combination of 600 mg monthly and 100 mg of dapsone daily for six months. The patients with a single lesion are treated with single dose of rifampicin (600 mg), ofloxacin (400 mg) a minocycline (100mg) (312).

1.5 Epidemiology

The 12th, 13th and 14th centuries were believed to be the times of the highest prevalence of leprosy (112). By the end of the 18th century, quarantining measures mentioned earlier in this section successfully decreased disease incidence (38,112). Currently, leprosy has become less prevalent, but continues to be problematic (112).

The route of transmission in the human host is still unknown, but it is suspected that nasal carriage or invagination through skin wounds, is the major ports of entry for *M*. *leprae* (302). The natural reservoirs of *M. leprae* are mostly humans and, perhaps, nine-banded armadillos, *Dasypus novemcinctus* (141). Armadillos with low body temperature

(34°C) are currently used as a host for the multiplication of large quantities of *M. leprae* for research purposes (262).

The WHO in 1976 estimated the total number of leprosy cases worldwide to be over 12 million (308). The WHO-directed MDT program introduced in 1982 has resulted in a dramatic reduction of global leprosy prevalence (265). The prevalence continued to be reduced from 5,351,408 in 122 countries in 1985 to 2,361,407 in 1991 in the top 32 endemic countries (209,309) (Figure 1.3). Given the success of the multidrug therapy program, the World Health Assembly in 1991 ambitiously projected the "elimination of leprosy as a public health problem by the year 2000" to achieve a level of prevalence of less than one case per 10,000 populations (265). Since then, effective chemotherapy treatments and efforts from governmental and non-governmental organizations have made it feasible to achieve the global goal of the leprosy elimination campaign (152). In a recent report from WHO, the global registered prevalence from 141 countries/territories at the beginning of 2010 was 211,903 and the number of new cases detected during 2009 was 244,796, which was significantly reduced from 620,638 in 2002 (309,310) (Figure 1.3). These remaining numbers of cases are predominantly in India and Brazil, but also in Indonesia, the Democratic Republic of the Congo, Central African Republic, Nepal, Nigeria, Mozambique, and Tanzania (310). Although the elimination campaign with multidrug therapy efficiently sterilized the patients and removed the biological reservoir of M. leprae, the number of new cases was sustained until now indicating that quiescent leprosy is continuously transmitted in endemic regions.



Figure 1.3 Global prevalence and incidence of leprosy, 1985-2009. The prevalence of leprosy is the total number of registered cases of leprosy at December 31 of a given year. The incidence is the number of new cases from January 1 to December 31 at a given year.

1.6 The bacterium: Mycobacterium leprae

M. leprae is an aerobic, rod-shaped and non-spore forming bacillus in the *Mycobacterium* spp., belonging to the family *Mycobacteriaceae* of the order *Actinomycetales* (262,302). Due to the possession of mycolic acids in a unique cell wall structure, *Mycobacterium* spp. are positive to the acid fast stain (16). *Mycobacterium* spp. are categorized into two subgroups: fast-growing (one to two hours doubling time) and slow growing (12 to 24 hours doubling time) (55). While the fast growers, such as *M. smegmatis* and *M. phlei*, are prevalent in the environment and are mostly nonpathogenic, the slow growing species including *M. tuberculosis*, *M. bovis*, *M. avium* and *M. avium paratuberculosis*, are highly adapted to survive in a host intracellular niche and are generally considered as the pathogens of mammalian diseases (55). Unlike other slow growing mycobacterial pathogens, *M. leprae* is a noncultivable intracellular bacterium with approximately 12-14 day generation time even in immunologically

defective mice (nu-/-) (47,151). Therefore, it takes an average of three to five years after infection for presentation of clinical symptoms in humans (302). Different from other mycobacteria, *M. leprae* preferentially invade the Schwann cell of the peripheral nerve and result in irreversible neurological injury causing sensory and sensorimotor loss in leprosy patients (233). The bacilli chronically infect and propagate inside of both macrophages and Schwann cells, contributing to pathogenesis of leprosy (194,233,262) (described in section 1.2.2).

1.6.1 Genome

Cole *et al.* (47) originally sequenced the entire genome of the *M. leprae* TN strain which had been purified from skin lesions of a multi-bacillary leprosy patient from Tamil Nadu, India and was subsequently propagated in a nine-banded armadillo. The genome of *M. leprae* is 3.27 megabase (Mb) in size and encodes 1614 coding sequences. Comparison with the genome to *M. tuberculosis* (4.41 Mb) with a high coding capacity (3,959 coding sequences), indicates that *M. leprae* contains an exceptionally large number of 1133 pseudogenes which are functionally active in *M. tuberculosis* (46,47) (Table 1.1).

| Feature | M. leprae | M. tuberculosis | |
|----------------------------|-----------|-----------------|--|
| Genome size (bp) | 3,268,203 | 4,411,532 | |
| G+C (%) | 57.8 | 65.6 | |
| Protein coding (%) | 49.5 | 91.2 | |
| Unknown genes | 142 | 606 | |
| Protein-coding genes | 1,604 | 3,993 | |
| Pseudogenes | 1,133 | 6 | |
| Gene density (bp per gene) | 2,024 | 1,106 | |
| Average gene length (bp) | 1,007 | 1,008 | |

Table 1.1 Comparison of genome features of *M. leprae* TN and *M. tuberculosis* (adapted from Scollard *et al.* (262) with permission)

This massive decay of genes results in the inactivation of important metabolic pathways, more frequently in catabolic pathways than in anabolic pathways, such as regulation, detoxification, DNA repair and transport or efflux of metabolites such as amino acids (AAs), peptides, cations and anions (47,301). The functional gene redundancy which is commonly found in *M. tuberculosis* does not occur in *M. leprae*; *M. leprae* has fewer genes in almost all functional categories (47,301). Genes involved in anabolic pathways and β -oxidation of fatty acids are conserved in *M. leprae*. Similarly, intracellular survival of *M. tuberculosis* was proven to depend on host lipid degradation and the glyoxylate shunt for energy production (47,246,262). In agreement with chemical analysis of cell wall structure (see section 1.6.3), the genes encoding for the biosynthesis pathway of the cell wall in *M. leprae* are considerably conserved (301). Comparative genomic analyses across *Mycobacterium* spp. indicate that the genome of *M. leprae* shares higher homology with *Mycobacterium* spp. (47). Therefore, this sequence similarity of genes encoding important immunogenic proteins results in considerable cross-reactivity in distinguishing individuals infected by *M. leprae* from those exposed to *M. tuberculosis* or other mycobacteria (277). In order to overcome this

problem, researches have vigorously evaluated the antigenic potential of hypothetical unknown proteins in functional gene class VI that have no/ low homology in any current mycobacterial database (80,82,277) (Table 1.2; see section 1.10.2).

 Table 1.2
 Functional classification of genes in M. leprae

 (http://www.pasteur.fr/recherche/unites/Lgmb/NATURE_DATA/ML_gene_list)

| Class | Function |
|-------|--|
| Ι | Small-molecules metabolism (sugar, carbon compound, fatty acids and amino acids) |
| II | Macromolecules metabolism (polysaccharide, lipopolysaccharide, cell envelope, RNA, |
| | DNA and proteins etc) |
| III | Cell process (Transport/binding proteins, cell devision, adaptation) |
| IV | Others (virulence, inserted elements, antibiotic production and resistance etc) |
| V | Conserved hypotheticals |
| VI | Hypothetical unknowns |
| | |

By using conventional methods ($6 \times$ coverage) and advanced next generation sequencing platforms ($38 \times$ or $46 \times$ coverage), Monot *et al.* (192) recently completed the whole genome sequences of three different *M. leprae* strains including Br4923 which was isolated from a Brazilian patient, Thai53 from a Thai patient and NHDP 63 from a North American (192). This study revealed that all new three strains and the TN strain share 99.995% of sequence identity despite geographical distance. It clearly indicated that the genomes of *M. leprae* strains are highly conserved and lack polymorphisms in coding sequences (192). Additionally, 16 single nucleotide polymorphism sites which were newly found in 5 pseudogenes, verified the previous transmission patterns of early human leprosy by genotyping both extant and extinct strains of *M. leprae* worldwide (described in the section 1.1).

1.6.2 Proteome

Due to massive gene decay, *M. leprae* is incapable of independent growth, being restricted to human and certain animal models where it manages to persistently multiply in the host. Therefore, proteomic analysis of *M. leprae* has focused on understanding leprosy pathogenesis and developing tools for leprosy diagnosis.

In the early stage of proteomic analysis, the WHO Scientific Working Groups on the Immunology of Leprosy identified distinct protein antigens from extracts of *M. leprae* by using monoclonal antibodies (137). These were further screened as recombinant proteins from the early λgt11 library and identified as heat-shock proteins (70 kDa=DNAK; 65 kDa=GroEL; 12-14 kDa=10 kDa=GroES) (3,4). Further biochemical and immunological studies on large quantities of armadillo-derived *M. leprae* allowed subcellular fractionation, isolation and purification of the major native proteins including the major cytosolic protein (MCP)-I (GroES/ML0380), MCP-II (18 kDa/heat shock protein), MCP-III (Sod A/ML0072c), major membrane protein (MMP)-I (35 kDa/ML0841) and MMP-II (ML2038/bacterioferritin (BfrA)/22 kDa) (117). MMP-II in *M. leprae* was found to be a multimer with a molecular weight of 380 kDa. Analysis of the native form by using the absorption spectrum demonstrated that the 380 kDa multimer contained iron. This result indicates that MMP-II may play a crucial role in the intracellular survival of *M. leprae* by acquisition of iron (224).

However, much of this early proteomic approach highly depended on the immunogenicity of proteins, limiting identification to only immunodominant proteins. Later, the addition of 2-D gel electrophoresis technology and emerging genome sequence data allowed further identification of *M. leprae* proteins such as alkyl

hydroxide (AhpC), sulfate sufurtransferase (CysA), L7/L12 ribosomal binding protein (RplL), transcription activator in two component system (MtrA) and protein elongation factor (EF-Tu) (170).

By using advanced modern mass spectrometry technology, two consecutive studies indentified 382 proteins which were mostly enzymes for lipid degradation and biosynthesis, biosynthesis of cell envelope, lipid carriers, etc. (171,172). It suggested that mycobacterial cell wall and lipid metabolism play a major role in bacterial pathogenesis. Recently, Souza *et al.* (61) indentified 1046 proteins out of 1614 coding sequences proteins by using a high throughput proteomic approach combined with bioinformatic analysis by using a new ORF annotation algorithm. In this study, five new proteins encoded by pseudogenes were identified. However, it is not clear whether these proteins are functionally expressed in the *M. leprae* proteome.

1.6.3 Unique cell envelope

The mycobacterial cellular envelope constitutes 40% of the dry weight of the bacterium and, as for all members of the *Mycobacterium* genus, contains highly unusual, but mycobacteria typical, lipids, glycolipids, proteins and polymers (26). The unique cell wall of *Mycobacterium* spp., *M. tuberculosis* and *M. leprae*, has been intensely investigated and proven to be a major pathogenic contributor in the context of the survival of mycobacterial pathogens within phagocytic cells (55,124,246). The structures of its components have been well defined and share common characteristics with all *Mycobacterium* spp. (25,26). These structures are introduced in the following sections. Knowledge of some (e.g. the phosphatidylinositol mannosides (PIMs), lipomannans
(LMs), lipoarabinomannans (LAMs) is important in the context of the experimental work (e.g. Chapter 3); other structures are discussed more in the context of an overall perspective of the complex nature in the tubercle bacillus. This section will briefly review the basic features of the mycobacterial envelope related to pathogenic aspects and some of the typical features of *M. leprae*.

Originally, it was thought that the mycolyl arabinogalactan-peptidoglycan complex of *M. leprae* should be simpler than that of other mycobacteria due to the large scale of gene reduction or gene decay (301). However, detailed biochemical analysis does not support this concept.

The mycobacterial cell envelope consists of the plasma membrane, a phospholipid bi-layer but unique to mycobacteria and related bacteria in possessing esterified tuberculostearic acid and PIMs. Extending outward from the plasma membrane, the cellwall core consists of peptidoglycan covalently attached to arabinogalactan, which in turn is attached to the mycolic acids (25). The outer membrane (OM) is loosely associated or intercalated by the mycolic acid layer through hydrophobic interaction in the outermost part of the cell (26) but serves to form a true bilayers (Figure 1.4).



Figure 1.4 Schematic representation of the cell envelope of *Mycobacterium spp.* (adapted from Bhamidi *et al.* (15) and Niederweis *et al.* (207) with permissions). AG, arabinogalactan; PG, peptidoglycan; PM, plasma membrane.

1.6.3.1 Peptidoglycan

The peptidoglycan, a prokaryotic skeleton, maintains cell structure and resists osmotic pressure arising from the cytoplasm or external sources. It consists of the repetitive units of N-acetyl-α-D-glucosamine and modified muramic acid as a backbone (25). In the C3 position, the muramic acid is substituted with a tetrapeptide which consists of L-alanyl-D-isoglutaminyl-meso-diaminopimelyl-D-alanine (161) (Figure 1.5). Compared to *Escherichia coli*, there is extensive cross-linking between each peptide in the *Mycobacterium* spp. and provides the rigid structural integrity (25). However, the peptidoglycan of the Mycobacterium spp. is a unique mixture of N-glycolylated and Nacetylated muramic acids and the carboxylic acids of the peptide that are often amidated (164). Recent studies revealed that the N-glycoly muramic dipeptides of M. tuberculosis are more powerful than N-acetyl muramyl dipeptides in activation of proinflammatory cytokine secretion through the RIP-2 and NF-κB (Nuclear Factor Kappa-Light-Chainenhancer of Activated B cells) intracellular networks (50). Also, the genetically mutated *namH* gene of *M. smegmatis* is responsible for the conversion of the N-acetyl muramic acid to N-glycolyl-muramic acid. The absence of the N-glycolyl function in NamH mutants substantially decreased NOD2 (nucleotide-binding oligomerization domain containing 2)-mediated TNF (tumor necrosis factors) secretion by infected macrophages compared to those of wild type *M. smegmatis* (50). However, *M. leprae* encodes namH as a pseudogene and is believed to be devoid of N-glycolyl muramic acids in its mature peptidoglycan. Recently, an advanced mass spectrometry based approach finally provided the biochemical evidence for this concept by analyzing the peptidoglycan of *in* vivo derived M. leprae (163) (Figure 1.5). Another unique feature of M. leprae

peptidoglycan is the replacement of L-alanine (L-Ala) in the first position of the tetrapeptide side chain with glycine due to its *in vivo* growth conditions (70,162) (Figure 1.5). The substitution of L-Ala with D-Ala in the peptidoglycan of *Staphylococcus aureus* and *Bacillus subtilis* was demonstrated to contribute to evading the recognition by NOD2 in host cells (121). Taken together, the loss of glycolyl function in muramic acids and the substitution of L-Ala with Gly in peptidoglycan may favor *M. leprae* in host-pathogen interaction, a question yet to be addressed.



Figure 1.5 Structure of the repeating unit of mycobacterial peptidoglycan (adapted from Mahapatra *et al.* (161) with permission). Boxes indicate the differences between *M. leprae* and *M. tuberculosis.* GlcNAc, *N*-acetyl-α-D-glucosamine.

1.6.3.2 Arabinogalactan

The fundamental features of arabinogalactan are similar among all mycobacteria including *M. leprae* (70). However, early structural analyses indicated a simpler, less extended structure in the case of *M. leprae* (301). Arabinogalactan is a tripartite structure consisting of the linker unit, an alternating β -(1 \rightarrow 5) and β -(1 \rightarrow 6) linked D-galactofuran

(30 units long), and a branched D-arabinofuran, a typical mycobacterial glycopolymer (25) (Figure 1.6).



Figure 1.6 Structure of mycolyl arabinogalactan and its attachment to peptidoglycan (from Bhamidi et al. (15) with permission). Two-thirds of 31-unit long arabinans (red) attached to mycolyl residues. Three chains are present per galactan (blue) chain. The GalNH₂ or the succinyl residue (green) is attached to an internal branching point of the arabinan.

The linker unit, α -rhamnopyranosyl- $(1\rightarrow 3)$ - α -*N*-acetyl- α -D-glucosamine- $(1\rightarrow phosphate)$ of arabinogalactan is covalently attached at the C6 position of muramic acid residues in peptidoglycan through a phophodiester bond (Figure 1.6). Bhamidi *et al* (15) analyzed the arabinogalactan of *M. tuberculosis*, and revealed that three arabinan chains are present per galactan chain and two-thirds of the 31-unit long arabinan are attached to mycolyl residues. Either succinyl or glucosaminosyl groups are individually linked at an internal branching point of two-thirds of the arabinan (Figure 1.6).

1.6.3.3 Mycolic Acids

Mycolic Acids are α -alkyl, β -hydroxyl fatty acids, 70 to 90 carbons in length (Figure 1.7), forming a lipophilic protective barrier and contributing to inherent drug resistance to some drug and solute uptake. The majority of mycolic acids exists as bound esters of arabinogalactan, but is also present as extractable lipids, such as trehalose monomycolate, trehalose dimycolate in the outer membrane (OM) (see section 1.6.3.5) (Figures 1.4 and 1.10). The structure of the mycolic acids consists of C₂₀ to C₂₅ α -branches and C₅₀ to C₇₀ β - meromycolate chains containing one or two functional groups (keto, methoxy, epoxy, wax ester) and varying within *Mycobacterium* spp. (26).

In *M. leprae*, α -branch mycolic acids are shorter than that of other mycobacteria (C₂₀ vs C₂₄₋₂₆) and the β -chains do not contain the methoxy function (69). The complete loss of mycolic acids is lethal for mycobacteria but results in viable *Corynebacterium glutamicum* which otherwise shares mycobacterial cell wall features (173). This *C. glutamicum* mutant is more susceptible to hydrophilic antibiotics (173,207). In addition, genetic mutation of either modification or length of mycolic acids in mycobacteria alters the permeability of hydrophilic materials. Taken together, mycolic acids serve as a significant permeability barrier responsible for some of the special physiological feature of mycobacteria such as impermeability to uptake nutrients and drugs and slow growth (207).



Figure 1.7 Structures of mycolic acids (adapted from Takayama *et al.* **(288) with permission)**. Three major forms of mycolic acids are found in *M. leprae*, *M. tuberculosis*, *M. bovis* and *M. bovis* BCG.

1.6.3.4 Phostphatidylinositol mannosides (PIMs), lipomannans (LMs), and

lipoarabinomannans (LAMs)

The PIMs, LMs and LAMs are among the most abundant and highly conserved components in all mycobacteria. They are found in the plasma membrane or/and the OM of all *Mycobacterium* spp. and exist via non-covalent hydrophobic interactions. Their insertion into the membraneous bilayers is derived from their acylated phosphatidyl portion (26,292) (Figure 1.8.A). The phosphatidyl inositol (*sn*-glycerol-3-phospho-1-D-*myo*-inositol) unit of PIMs (PIM₁ to PIM₆) is modified with one to six mannose residues (208). The inositol group can additionally be acylated at the C3-position and contain a mannose residue at the C2-position (Figure 1.8.A). This mannose also can be acylated at the C6-position. Current research on their essentiality suggests that these molecules play an important role to maintain cell viability in all *Mycobacterium* spp. (123). LMs consist of PIM₄ with a long α (1 \rightarrow 6)-linked mannan branched with alternative α (1 \rightarrow 2)-linked mannose residues. LAMs are derived from LMs by attaching an arabinan chain at the

mannan backbone (Figure 1.8.B). The arabinan segment of LAMs consists of α (1 \rightarrow 5)-D-arabinofuranose polymers branching at internal residue to 3,5- α -D-arabinofuranose residues (208). The terminal non-reducing arabinose residues may be capped with mannosyl residues (ManLAM) or devoid of these residues (AraLAM) (208). ManLAMs are mostly found in pathogenic slow growing mycobacteria, such as *M. tuberculosis*, *M. bovis*, *M. avium*, *M. leprae*; the mannose caps consist of a single mannose, or di- or trimannosides containing α (1 \rightarrow 2) linkages (39,208) (Figure 1.8.B). LAMs are more abundant in *M. leprae* and have less mannose capping and simpler arabinans in comparison to LAMs of *M. tuberculosis* (208,292).



Figure 1.8 Schematic structures of mycobacterial PIM₆ and ManLAMs. (from Berg *et al.* (14) with permission). A) PIM₆, B) ManLAMs; Boxes indicate mannose capping: R1, palmitic acid; R2, tuberculostearic acid.

The presence of mannose caps on LAMs has been demonstrated to contribute to the successful pathogenesis of virulent mycobacteria (157,256,259,260). In addition, the number, nature, and position of the acyl residues are important in the interaction of LAM with receptors such as CD14, and TLR-4 (130). The immunomodulatory role of these components, particularly LMs and LAMs, has been extensively investigated in terms of pathogenesis (see section 1.8.2).

1.6.3.5 Glycolipids: Phenolic glycolipids (PGLs) and phthiocerol dimycocerosates (PDIM); Trehalose containing glycolipids (TDMs, TMMs)

Similar to LAM and PIMs, components of the OM including PDIM, PGL, trehalose dimycolate (TDM, also known as cord factor), trehalose monomycolate (TMM) may associate or even intercalate with the mycolic acids via hydrophobic interactions (26,54) (Figure 1.4). The PGLs, polyketide derived virulence factors, are dominantly produced in the slow growing mycobacteria including *M. leprae*, *M. kansasii*, *M. bovis*, *M. microti* and a few of the clinically isolated *M. tuberculosis* strains particularly, apparently, in the case of MDR tuberculosis (26,55). The PDIM, the lipid base of the PGLs is composed of a mixture C₃₃-C₄₁ diols with two polymethylated branched fatty acids (C₂₇-C₃₃), mycocerosic and phthioceranic acids (26,124) (Figure 1.9). PDIM itself also contributes to the pathogenesis of mycobacterial disease by protecting from antimicrobial activity of nitric oxide and suppressing the secretion of proinflammatory cytokines by activated macrophages (124). The PGL structure is generally composed of a PDIM core, a phenol and a considerably hydrophobic oligosaccharide part containing one to four deoxysugars modified with multiple O-methyl groups (55). The PGL-I, II

and III in *M. leprae* are major antigenic glycolipids. PGL-I contains the *M. leprae* specific saccharide of 3-O-methyl (Me)-rhamnose, 2,3-di-O-Me-rhamnose, and 3,6-di-O-Me-glucose attached to the phenol of the PDIM via a glycosidic bond (116) (Figure 1.9). This uniquely modified sugar induces the species specific humoral response (41) and also plays an important role in host pathogen interaction (see sections 1.2.2 and 1.10.4) (35,233,258).



Figure 1.9 Structures of phthiocerol dimycocerosates (PDIM) and Phenolic glycolipids (PGLs) from *M. leprae* (adapted from Jackson *et al.* (124) with

permission). Two of the polymethylated mycocerosic acids attach at long-chain β -diols (phthiocerol moiety) of PDIM via ester bonds. The PGL-I from *M. leprae* consists of phenolphthiocerol esterified by two polymethylated mycocerosic acids and a trisaccharide unit consisting 3-O-methyl (Me)-rhamnose, 2,3-di-O-Me-rhamnose, and 3,6-di-O-Me-glucose at the phenol groups (R=-CH₂-CH₃ or -CH₃) (116). The structure of the oligosaccharide unit of the PGL from *M. tuberculosis* is different (2,3,4-tri-O-Me-fucose-(1-3)-rhamnose-(1-3)-2-O-Me-rhamnose) (124). The oligosaccharide structures of glycolipids from environmental mycobacteria are different as well.

Trehalose containing lipids present in OM of *M. tuberculosis* including trehalose 6,6'-O-dimycolate (TDM), trehalose 6-O-monomycolate (TMM) and sulfolipids which consist of trehalose 2'-sulfate acylated with hydroxyphthioceranic, phthioceranic and saturated fatty acids (palmitic or stearic acid) (26) (Figures 1.10 and 1.22). All of these

lipids are found predominantly in the cultures of virulent *M. tuberculosis* (55) and play an important role in the pathogenesis of tuberculosis including pro-inflammatory cytokines (primarily TNF- α) production and granuloma formation of macrophages (19,124). TDM, also known as a cord factor, induces the potent inflammatory responses of macrophage via TLR-2 signaling (19). TDM and TMM were not found in *M. leprae*, though a significant numbers of leprosy patients presented seroreactivity to TDM and its precursor, TMM, derived from *M. tuberculosis* (132). Recently, Kai *et al.* (132) isolated both *M. leprae*-derived TDM and TMM from armadillo tissue infected with the bacteria. Subsequent MALDI-TOF mass analysis revealed that these trehalose based lipids of *M. leprae* possessed both α -mycolates (C₇₈) and keto- mycolates (C₈₁ or C₈₃). However, sulfolipids are absent from *M. leprae* due to deletion of a series of genes which are responsible for biosynthesis of their acyl chains (47).



Figure 1.10 Structures of trehalose monomycolate and trehalose dimycolate (adapted from Takayama *et al.* (288) with permission). In *M. leprae*, α - or/and keto-mycolates are estrified to the trehalose head groups of both TMM and TDM.

1.6.3.6 Outer membrane (OM)

The OM of *Mycobacterim* spp. consists of an asymmetric bilayer containing pore forming proteins (porins), lipoproteins and free lipids including the phthiocerolcontaining lipids, PIMs, LMs, LAM, TDM, TMM, and the diacyl- and polyacyltrehaloses (25) and, above all, the arabinogalactan-bound mycolates (Figure 1.4). Minnikin proposed the existence of a pseudo OM akin to that of Gram negative bacteria (54). OM has since then been considered as an asymmetrical lipid bilayer consisting of covalently linked mycolic acid solely comprising the inner leaflet and the extractable lipids restricted to the outer leaflet (26,54).

Recently, Hoffmann *et al.* (110) and Zuber *et al.* (328), by using cryo-electron tomography and vitreous sections, provided a new perspective on the spatial organization of the cell envelope, an altered version of the original Minnikin's model. One major change is the existence of extractable lipids in both the inner and outer leaflets of the OM leading to symmetrical lipid bilayers. The other difference is that the OM is thinner than conventional models supporting the functional accommodation of porins (110). The thickness of the OM of *Mycobacterium* spp. is 8nm (110). However, the theoretical length (around 9nm) of the meromycolate of mycolic acid in an extended conformation is too long to fit this new model (110). New findings raise the question of whether meromycolates in the hydrophobic matrix of the OM have an elongated or a folded conformation (207).

Zuber *et al.* (328) proposed a new model with a folded meromycolates in a symmetric OM, unlike that previously depicted as the Figure 1.11. Since there is no direct experimental evidence of the conformation of mycolic acids, both extended and

folded conformations of mycolic acids are equally acceptable in Figure 1.11 model of the organization of the OM of the mycobacterial envelope (207).



Figure 1.11 Models of the mycobacterial outer membrane (OM) (adapted from Niederweis *et al.* (207) with permission). A) Schematic structure of mycolic acids in the elongated and folded conformation; B) Models of the mycobacterial OM. The mycolic acids covalently attached to the arabinogalactan polymer are drawn in red and inserted either in the elongated conformation (model I, left side) or the folded conformation (model II, right side). A porin, MspA from *M. smegmatis*, is represented in blue. Open symbols of lipid headgroups indicate apolar groups; filled symbols represent polar groups. These free lipids could occur in both leaflets of the outer membrane.

1.6.3.7 Biosynthesis and genetics of cell wall components

In previous sections, we described the chemical definition of the mycobacterial cell envelope mostly in the context of the pathogenesis of mycobacterial diseases. The chemical structures of its components are very similar among all *Mycobacterium* spp. but are different from those of other bacteria and such structures are absent in host eukaryotes. Therefore, the biosynthesis of cell wall components has been long investigated in order to develop new drugs against *M. tuberculosis*. The availability of genome sequences of *Mycobacterium* spp. combined with recent advances in both genetic manipulation and

analytical techniques allow for an understanding of the basic genetics and biosynthesis of cell wall components of *M. tuberculosis* (25). Genome analyses of *M. leprae* revealed that the important genes responsible for the biosynthetic pathways of cell wall components are conserved and share high degrees of homology with orthologues of *M. tuberculosis* (27,47,301). Therefore, the genetics and biosynthesis of mycolyl arabinogalactan-peptidoglycan complex, PDIMs and LMs/LAMs/PIMs in *M. leprae* probably resemble those of *M. tuberculosis*. These topics have been reviewed by Crick *et al.* (52), Guilhot *et al.* (97), Kaur *et al.* (136), among others.

<u>1.7 Post-translational modification of mycobacterial proteins</u>

Post-translational protein modifications such as protein acylation, glycosylation, methylation, phosphorylation and ubiquitination diversify the biochemical structure of proteins. These chemical groups on mature proteins ultimately influence protein localization, stability, regulation and biological functions which cannot be predicted by the genomic DNA sequences (33)

The primary structure of certain mycobacterial proteins can be modified by acylation and phosphorylation (240,305). In addition, growing evidence suggests that mycobacteria possess protein modification machineries for glycosylation (253,299), methylation (227) and ubiquitination (33), which are typically found in eukaryotes. The exact biochemical structures of post-translationally modified proteins are largely unknown as well as their physiological or pathogenic roles. This section reviews posttranslationally modified mycobacterial proteins, primarily on lipoproteins, with emphasis on the biosynthesis, structure and pathogenic roles.

1.7.1 Lipoproteins

Bacterial lipoproteins were initially identified in *E. coli* (20,21). This archetypical bacterial lipoprotein, Braun's lipoprotein, is secreted across the plasma membrane structures and attaches to the peptidoglycan (285). Lipoproteins are acylated at the thiol group of the N-terminal cysteine (Cys) residue through a thioester linked diacylglycerol unit. An additional acyl group is linked through an amide bond to the N terminus of Cys (134) (Figure 1.12). Chemically identical lipoproteins have been found in a variety of bacteria primarily on the basis of sequence analysis (9). However, the chemical structure of few lipoproteins has been experimentally verified to date.



Figure 1.12 General structure of a mycobacterial lipoprotein (adapted from Tschumi *et. al.* (295) with permission). The shaded fatty acids reflect on the composition of phopholipids.

Bacterial lipoproteins have been long studied as a Toll like receptor-2 (TLR-2) ligand that contains conserved pathogen associated molecular patterns (PAMPs) and initiates an inflammatory response (289). The N-terminal acylated region of lipoproteins interacts with TLR-2 on the antigen presenting cells as well as various nonprofessional immune cells, and induces innate and adaptive immune responses required for host defense against microbial pathogens (2,255). In 1999, it was demonstrated that the mycobacterial lipoprotein LpqH/19 kDa strongly stimulated IL-12 induction in Th1

immunity through TLR-2 in human macrophage cell lines (28). Since then, not only LpqH/19 kDa (45,79,205,210,219), but LprG (78,114,115), LprA (222), PhoS1/38 kDa (131,236) from *M. tuberculosis* and 33 kDa (158,319) from *M. leprae* have been studied in terms of Th1 immunity and mycobacterial pathogenesis.

1.7.1.1 Identification

Identification of bacterial lipoproteins has been conducted largely by three approaches: 1) metabolically with radiolabeled fatty acids, those mostly found in phospholipids on accounting of their precursor role in lipoprotein biosynthesis; 2) disturbing lipid modification processes by mutating the signal peptide sequence of prolipoproteins which direct to the lipid modification machinery and result in a mature lipoprotein; and 3) identification of the conserved consensus sequence within the prolipoprotein signal peptides (240).

Biosynthesis of bacterial lipoproteins depends on the presence of a Type II signal peptide containing the consensus sequence $[LVI]_{-3}$ [ASTVI]_2 [GAS]_1 C₊₁ for lipid modification, commonly known as the "lipobox" (9,304) (Figure 1.13). The signal sequence of secretory proteins, including lipoproteins, consists of three parts: 1) an N-terminal region containing 3-15 amino acids (AAs) with at least two positively charged AAs; 2) a central hydrophobic region which contains 7-22 AA with predominantly hydrophobic and non-charged residues; 3) a cleavage region which distinguishes the Type II signal peptide of lipoproteins from the Type I signal peptide of non-acylated proteins (especially glycoproteins). The cleavage region of the Type II signal peptide possesses the lipobox at -3 to +1 at Cys but a Type I signal peptide contains the Ala-X-

Ala motif as the cleavage site (120,285) (Figure 1.13). These features of signal peptides have facilitated the identification of putative lipoproteins via bioinformatic analysis (9,285) (http://www.mrc-lmb.cam.ac.uk/genomes/dolop/).

| Gram (-) Bacteria | h-region | c-region | Cleavage site by either Type I or Type II signal peptidase | |
|-------------------|----------|-------------------------------------|---|--|
| | | AX A [LVI][ASTVI][GAS C | | |
| Gram (+) Bacteria | | | | |
| n-region | h-region | c-region | | |
| | | ہ [LIVMFWSTAG] [LIVMFYSTAGCQ][AG | ХА S) С | |

Figure 1.13 The tripartite region of signal peptides recognized by either Type I or Type II signal peptidase in Gram-positive or Gram-negative bacteria (adapted from Hutchings *et al.* (120) with permission). Type I signal peptidase recognizes the AXA motif at the cleavage region (c-region). Type II signal peptidase recognizes the lipobox. The N-terminal region (n-region) containing 3-15 amino acids (AAs) with at least two of positively charged AAs and the hydrophobic region (h-region) consisting of 7-22 AA of hydrophobic and non-charged residues. In the case of Gram-positive bacteria, the consensus sequence of the lipobox is more variable than in Gram-negative bacteria.

More than 2,000 bacterial lipoproteins have been identified and account for 1% to 3% of coding genes in each bacterial genome (9). The signal sequences of 90 lipoproteins from Gram-positive organisms have been extensively studied and established as a consensus Prosite profile PS51257 ([DERK]₋₆ [LIVMFWSTAG]. ₃[LIVMFYSTAGCQ]₋₂[AGS]₋₁C₊₁) (120) (Figure 1.13). This *in silco* analysis has indentified 99 and 31 putative lipoproteins in *M. tuberculosis* and *M. leprae* genome respectively (143,149,286,296) (Table 1.3).

Few mycobacterial lipoproteins have been experimentally characterized in terms of physiological function (68,142,284,295,321). Evidence of acylation in putative mycobacterial lipoproteins has mostly been provided by their appearance in the detergent phase generated from Triton X-114 phase separation (18). However, this method could not fully characterize protein acylation because lipoproteins colocalized with integral membrane proteins in the resultant detergent phase. Young *et al.* (321) metabolically labeled *M. tuberculosis* lipoproteins by using radiolabeled fatty acids. Two-dimensional (2-D) gel electrophoresis and autoradiography of the resultant proteins revealed that several proteins were acylated including 19 kDa and 38 kDa. In *M. leprae*, 33 kDa/LpK/ML0603 (149) expressed in *E coli* was studied by metabolic labeling with [¹⁴C] glycerol and globomycin treatment (a specific inhibitor of signal peptidase II, see section 1.7.1.2), leading to the accumulation of the precursor form of LpK (158).

1.7.1.2 Biosynthesis

The biosynthetic machinery of bacterial lipoproteins was identified and fully characterized in Gram-negative bacteria, especially *E. coli* (105,240,249,250,317). In Gram-positive bacteria, recombinant Braun's lipoproteins expressed in both *Bacillus subtilis* and *Staphylococcus aureus* were structurally equivalent to the three-acylated forms produced in *E. coli* (287). Therefore, it is believed that bacterial lipoproteins from Gram-positive and Gram-negative organisms are structurally similar and share similar biosynthetic machinery.

The lipoprotein biosynthetic machinery consists of three enzymes. The first enzyme is diacylglyceryl transferase (Lgt) which covalently links the diacylglycerol residue at the thiol group of Cys in the lipobox (120,249) (Figure 1.14). The orthologues of this enzyme have been identified as Rv1614 in *M. tuberculosis* and ML1274 in *M. leprae*, respectively (149,296). The second enzyme is a signal peptidase II (LspA)

(240,318). LspA/Rv1539/ ML1199 (149,296) cleaves the signal peptide resulting in a new free N-terminus of di-acylated Cys (240,318) (Figure 1.14).



Figure 1.14 Biosynthesis of bacterial lipoproteins (adapted from Hutchings *et al.* **(120) with permission).** The precursor lipoprotein synthesized in the cytoplasm translocates across the plasma membrane by either the Sec or Tat protein export system (detailed in section 1.7.1.3). First, diacylglyceryl transferase (Lgt) covalently links the diacylglycerol residue to the thiol group of Cys in the lipobox. Second, signal peptidase (LspA) cleaves the signal peptide of the diacylglyceryl prolipoprotein and N-acyl transferase (Lnt) subsequently attaches the third fatty acid to the amide group of Cys. Mature lipoproteins are either retained within the plasma membrane or transported to the outer membrane of the cell wall in Gram-negative bacteria.

The *M. tuberculosis* LspA/Rv1539 enzyme has been partially characterized (248). The growth rate of the *lspA/rv1537* deletion *M. tuberculosis* mutant in both mice and macrophage cell lines was decreased but was unaffected in *in vitro* cultured media (248). This suggests that although LspA/Rv1539 is not essential for growth and is not a good drug target, its activity might contribute to the pathogenesis of mycobacterial diseases (240,248).

The N-acyl transferase (Lnt), as a final processing enzyme (120), adds an amidelinked fatty acid (Figure 1.14). In the process of protein acylation, the acyl groups are donated from bacterial membrane phospholipids such as phosphatidylglycerol, phosphatidylethanolamine and cardiolipin (13,98). Lnt/Rv2051/ML1441 (149,296) is conserved in all mycobacteria and contains 2 domains: an Lnt domain at the N-terminus and a polyprenol monophosphomannose synthase domain at the C-terminus (240). The Rv2051 expressed in either *M. smegmatis* or *E. coli* has been demonstrated to attach *in vitro* guanosine diphosphate mannose onto endogenous mycobacterial polyprenol monophosphates acceptors, which are responsible for biosynthesis of LMs and LAMs (101,240). Recently, the Lnt function of Rv2051 was characterized by using an *lnt* deletion mutant of *M. smegmatis* (295). The subsequent MALDI-TOF analysis showed that this *lnt* deletion mutant failed to acylated the recombinant LppX but the complementation of *lnt/rv2051* in an *lnt* deletion mutant of *M. smegmatis* restored the N-acylation of LppX. Additionally, this study provided the first direct evidence that the mycobacterial lipoproteins contain the species specific tuberculostearic acid derived from phospholipids as the acyl donor (295).

1.7.1.3 Transport

Lipoproteins are exported from the cytoplasm, the site of their synthesis, across both the inner membrane and hydrophobic cell wall. In general, there are conserved protein export systems for the majority of proteins: the classic secretion (Sec) pathway and the twin-arginine translocation (TAT) pathway (204). In some bacteria, there are also specialized protein export systems that translocate specific subsets of proteins (204,242). Lipoproteins are modified and simultaneously transported through the interaction of their Type II signal peptide and protein export systems (Figures 1.13 and

1.14). The positively charged AAs in the n-region of the signal sequence attach to the negatively charged inner surface of the cytoplasmic membrane. The hydrophobic region of the signal peptide is inserted into the cytoplasmic membrane through the Sec or TAT machinery (120,177,304) (Figure 1.14).

The Sec-dependent pathway is a conserved protein export system in all procaryotes (22,204). The Sec-dependent system is composed of several proteins: SecA (ATPase) as an energy dependent motor in the cytoplasm; Sec B as a chaperone in the cytoplasm; SecYEG as a translocase complex in the membrane; and YajC. During the translational process, the signal peptide of a secretory protein binds with the signal recognition particle and forms a ternary complex with the ribosome. This complex targets a precursor protein to the Sec machinery through FtsY (204) (Figure 1.15.A). SecB maintains a preprotein in an unfolded state that is capable to translocate (Figure 1.15.B). SecA accepts a precursor protein from either chaperone, Sec B or signal recognition particle/ribosome/protein complex (200,204). SecA and the target protein associate with the SecYEG complex at the cytoplasmic site. Subsequent conformational changes of the Sec dependent machinery translocate the protein across the plasma membrane and attach the lipid onto the protein (22,204) (Figure 1.15). YajC increases the efficiency of this type of protein export (200,204) (Figure 1.15). Unlike the Sec dependent pathway, the TAT system exports the folded state of preproteins across the plasma membrane (Figure 1.15.C). The N-terminal signal sequence of proteins destined for Tat-dependent translocation typically contains a two arginine motif at the n-region and is longer than in Sec dependent translocation (204).



Figure 1.15 Schematic overview of the secretion (Sec) and twin-arginine translocation (TAT) pathways (adapted from Natale *et al.* (204) with permission). Sec dependent translocation of unfolded precursor protein, A) during and B) after translation. C) Translocation of folded precursor proteins by the Tat translocase (see above description). SRP, signal recognition particle.

Mycobacteria have two homologs of SecA: SecA1/Rv3240/ML0779 and SecA2/Rv1821/ ML2082c (22,149,242,296). As in other bacteria, SecA1 is essential and similar to well characterized SecA proteins (242). SecA2 is present in *Streptococcus gordonii*, *Streptococcus parasanguinis*, *M. tuberculosis* and *Listeria monocytogenes* which are capable to survive in hostile intracellular niches (242). SecA2 is assumed to be responsible for secretion of the following virulence factors: SodA (superoxide dismutase, Rv3846); KatG (catalase-peroxidase-peroxynitritase T, Rv1908); Acr (alpha-crystalline, Rv2031) found in *M. tuberculosis* (22,23). However, it still remains to be addressed how to distinguish SecA1 and SecA2 dependent protein export. Based on genomic analysis, *M. tuberculosis* has a functional TAT pathway consisting of TatA/Rv2094c/ ML1331, TatB/Rv1224/ ML1079, tatC/Rv2093c/ ML1332 (149,177,296). Bioinformatics analysis predicted 10-15% of putative lipoproteins as the substrates of TAT in *M. tuberculosis* (120). Currently, β-lactamase and phopholipase C were experimentally verified as substrates of TAT by using genetic mutants of the Tat components (177,238,240). In *M. leprae*, bioinformatics analysis predicted seven putative lipoproteins as TAT substrates (149).

1.7.1.4 Localization

Proteins on the surface of bacteria are major contributors to host-pathogen interaction including nutrient acquisition, adhesion, invasion and efflux of drugs (242,248). In Gram-negative bacteria, lipoproteins are exported to the outer membrane (OM) or retained in the outer leaflet of the plasma membrane (213). Some lipoproteins contain Asp_{+2} as a sorting signal, allowing it to attach to the plasma membrane (291). In the periplasm, lipoproteins are transported to the OM by the LolABCDE complex consisting of the ATP binding cassette transporters and chaperones (203,213). The lipid moiety of lipoproteins serves as an anchor to attach to the inner membrane or OM in Gram-negative or Gram-positive bacteria cell wall (203,230). Therefore, lipoproteins of Gram-positive bacteria are believed to be functional equivalents of periplamic proteins in Gram-negative bacteria (120). Localization of mycobacterial lipoproteins across this layer has rarely been defined. Proteomic analysis on subcellular fractionation of M. tuberculosis revealed that the majority of mycobacterial lipoproteins are found in the cell wall and membrane fractions (89,176). Current findings about mycobacterial OM (mentioned in section 1.6.3.6) suggest that mycobacterial lipoproteins may either associate with the OM or exist in the periplasm between OM and plasma membrane. Transport mechanisms across this layer have yet to be defined, but may be similar to the Lol transport system in *E. coli* (213,291).

1.7.1.5 Physiological Function

Mycobacterial lipoproteins are believed primarily to play an important role in maintaining cell wall integrity and are classified into 5 categories according to their physiological role (286) (Table 1.3). However, this classification heavily relies on comparative bioinformatic analysis of other bacterial genomes. Physiological functions of a few lipoproteins in *Mycobacterium* spp. have been experimentally characterized by using genetic mutants of either mycobacterial lipoproteins or the biosynthetic machinery proteins, LspA and Lnt. In the following, the physiological roles of mycobacterial lipoproteins will be briefly reviewed, focusing on those that have been experimentally verified.

| Functional category | Sub-category | M. leprae ORF /M.tuberculosis ORF | M.tuberculosis ORF without M. leprae orthologue |
|------------------------|------------------------|--------------------------------------|---|
| Transporters | | | |
| - SBPs in ATP | Iron | ML1729/Rv3044(FecB) | Rv0265c (FecB2) |
| binding cassette | Molybdenum and | ML2095/Rv0928(PstS3) | Rv0932c(PstS2), Rv0934 (PstS1), Rv1857 (ModA) |
| transporter | Phosphate | | |
| | Peptides | ML0303/Rv0411c(GlnH) | Rv3666c, Rv3759c |
| | | ML0489/Rv2585c | |
| | | ML1093/Rv1244 (LpqZ) | |
| | | ML1121/Rv1280c (OppA) | |
| | Sugars | ML1086/Rv1235 (LpqY) | Rv2833c |
| | | ML1427Rv2041c | |
| | Sulfates | ML0615/Rv2400c (SubI) | |
| | PIMs | ML0136/Rv2945 (LppX) | |
| - Lipid Carrier | DIM | ML0557/Rv1411c (LprG) | |
| | PIMs, LMs, LAMs | ML1497/Rv1166 (LpqW) | |
| Enzymes | | | |
| - Cell wall | | ML1923/Rv3593 (LpqF) | Rv0399c (LpqK), Rv0838 (LpqR), Rv2905 (LppW) |
| metabolism | | | Rv1922, Rv2864c (pbpB) |
| - Degradative process | Esterases | ML0715/Rv3298c (LpqC) | Rv0671 (LpqP) |
| | Glycosyl hydrolase | | Rv0237 (LpqI) |
| | Phosphorylase | | Rv2293c |
| | Proteinase/ peptidases | ML1339/Rv2672 | Rv0418, Rv0419 |
| | | ML1633/Rv2224c | |
| - Metabolic activities | Copper oxidase | | Rv0846c |
| | FAD-linked oxidase | | Rv2251 |
| | γ-Glutamyl transferase | | Rv2394 (ggtB) |
| | Oxidoreductases and | ML2412 /Rv0526 (dsbF) | Rv0132c (Fgd2), Rv1677 |
| | thioredoxins | | Rv3006 (LppZ) |
| | Phosphoglycerate | | Rv3390 |
| | mutase | | |

Table 1.3 Functional categories of *M. leprae* and *M. tuberculosis* putative lipoproteins (adapted from Sutcliffe and Harrington (286) with permission)

| Functional category | Sub-category | M. leprae ORF M.tuberculosis ORF | M.tuberculosis ORF without M. leprae ortholog |
|---------------------------------|---------------------------------------|--|--|
| Adhesion and cell invasion | | ML2593(LprK)/Rv0173(Mce1E) | Rv2873 (MPT83), Rv0593 (Mce2E), Rv1970 (Mce3E), Rv3495c (Mce4E) |
| Signaling and sensory functions | | ML0775/Rv3244c (LpqB) | Rv1009 (RpfB), Rv1270c, Rv1368 (LprF), Rv1690 (LprJ), Rv1911c, Rv2403c, Rv3576 |
| Unknown | Inter-related Lpp of unknown function | ML0246/Rv1016c (LpqT) ML0426/Rv2518c (LppS) ML2446/ Rv0483 (LprQ) | Rv0583c (LpqN), Rv0604 (LpqO), Rv2999 (LppY), Rv1228 (LpqX), Rv2341 (LppQ) |
| | The LppA paralog family | | Rv2543 (LppA), Rv2544 (LppB), Rv2796c (LppV) |
| | Others | ML0319/Rv3584 (LpqE) ML0603/ Rv2413c (LpK) ML0902/Rv2171 (LppM) ML1099/ Rv1252c (LprE) ML1115 /Rv1274 (LprB) ML1116 /Rv1275 (LprC) ML1315/ Rv2116 (LppK) ML1560/Rv2843 ML1966/ Rv3763 (LpqH) | Rv0179c(LprO), Rv0344c(LpqJ), Rv0381c, Rv0460, Rv0679c, Rv0847(LpqS), Rv0962c(LpqP), Rv1064c (LpqV), Rv1270c(LprA), Rv1418(LprH), Rv1541c(LprI), Rv1799 (LppT), Rv1881c(LppE), Rv1921c(LppF), Rv2046 (LppI), Rv2080(LppJ), Rv2138(LppL), Rv2270(LppN), Rv2290(LppO), Rv2330c(LppP), Rv2784c(LppU), Rv3016(LpqA), Rv3623(LpqG) |

All lipoproteins categorized by their functional motif. The function of ORF in shade has been experimentally verified. SBP, solute binding proteins.

1.7.1.5.a Transporter; Solute binding proteins of the ATP Binding Cassette transport systems and lipid carriers

The majority of lipoproteins are classified as solute binding proteins of the ATP binding cassette transport system. They import the hydrophilic substrates (phosphate, peptides, iron and sugar) into cells via membrane associated permeases in an energy dependent manner (286) (Table 1.3). A gene deletion study in a mouse infection model showed that several of the mycobacterial solute binding proteins play an important role in intracellular survival in the early stages of infection (34,254). All phosphate binding proteins, 38 kDa Rv0934/PstS-1, Rv0932/PstS-2 and Rv0928/Pst-3/ML2095 (149,296), share the conserved phosphate binding domain as *E. coli* phosphate binding proteins (36,165). In particular, 38 kDa (Rv0934/PstS-1, PhoS1, PhoS/ML0746) (149,296), a lipoglycoprotein, is abundant on the cell surface or in the cultured supernatant of *in vitro* grown *M. tuberculosis* (36). The 38kDa Rv0934/PstS-1 was experimentally shown to have phosphate binding capacity by using immunoblot analysis of *M. tuberculosis* which was cultured in phosphate starvation condition (75).

Comparative genome analysis indicated that 50% of putative mycobacterial lipoproteins are not conserved in other bacteria, but are very similar among *Mycobacterium* spp. particularly between *M. tuberculosis* and *M. leprae* (240). The mycobacterial cell envelope is highly enriched with unusual lipids. The function of these unknown lipoproteins has been speculated to be as lipid carriers for a hydrophobic precursor of those lipids on the cell envelope of mycobacteria. Several recent studies support this hypothesis, demonstrating that LppX/Rv2945/ML0136, LpqW/Rv1166/ML1497, and LprG/Rv1411c /ML0557 (149,296) transport lipid

precursors across the mycobacterial cell envelope (68,142,284). All three lipid carriers were also found in *M. leprae* proteomic analysis implying the importance of their physiological role (172). The lpqW disruption mutant and its complementation with lpqW have indicated that this lipoprotein acts as a lipid carrier of PIM₄ in the biosynthesis of polar PIMs and LAMs (142,169).

The *lppX/rv2945c* is located in a gene cluster which is involved in the synthesis and transport of the PDIM complex to the cell envelope. Disruption of this gene cluster by transposon mutagenesis attenuated *in vivo* growth of *M. tuberculosis*, but complementation with *lppX* was able to recover the growth rate of this transposon mutant. Lipid analysis of both mutants revealed that LppX is responsible for translocating PDIMs across the plasma membrane and releasing into the culture medium (284).

LprG/Rv1411c has high sequence similarity to LppX and is associated with *M*. *tuberculosis* virulence (17). It has mostly been investigated as a TLR-2 ligand to elicit protective immunity (67,78,270). Drage *et al.* (68) demonstrated that LprG plays an important role in the *in vivo* growth of *M. tuberculosis* by using a *lprG* knockout mutant. In this study, recombinant LprG/Rv1411c from *M. smegmatis* showed high binding affinity with the tri-acylated glycolipid portion of LMs, LAMs and PIMs. The crystal structure of LprG/Rv1411c with PIMs revealed that the hydrophobic pocket of LprG accommodated three acyl chains (68). The results indicate that LprG serves as a lipid carrier to assemble and to transport acylated glycolipids in cell wall biogenesis. Additionally, LMs, LAMs and PIMs bound to LprG enhanced TLR-2 sensing compared to LprG alone (68).

1.7.1.5.b Enzymes

Lipoproteins enzymes can be divided into three subcategories; cell wall metabolic enzymes, degradative enzymes and others enzymes (Table 1.3). However, the exact functions of these enzymes are still unclear (286). SodC/Rv0432/ML1925c (149,296) contribute to the survival of *M. tuberculosis* in activated macrophages, suggesting that it plays an important role in the defense against the oxidative burst produced *in vivo* (229). SodC/Rv0432 was originally assumed to be a lipoglycoprotein localized in the cellular envelope, but was later found to be only a glycoprotein lacking acylation in the culture of *in vitro* grown *M. tuberculosis* (252). However, it is unclear how SodC/Rv0432 avoids or misses the lipid modification step during protein export.

1.7.1.5.c Adhesion and cell invasion

MPB83 and Mce lipoproteins are primarily categorized in the roles of adhesion and cell invasion. Several studies, predominantly using gene knockout mutants, attempted to address the pathogenic roles of MPB83 and Mce lipoproteins in a mouse model. However, conclusion from the results are still unclear (286) (Table 1.3).

1.7.1.5.d Signaling and sensory function

LpqB/Rv3244c/ML0775 (149,296) is located downstream of the two-component signal transduction system MtrAB (see section 1.7.3) (113,286). The *M. smegmatis lpqB* defective mutant revealed that LpqB/Rv3244c interacted with the extracellular domain of MtrB and enhanced MtrA phosphorylation. This mutant also exhibited a large enhancement of susceptibility to multiple antibiotics with diversified action mechanisms.

Therefore, the LpqB in combination with MtrAB possibly affects cell division and cell homeostasis in response to environmental stress (206).

In both *M. tuberculosis* and *M. smegmatis*, LprF /Rv1368 and LprJ/Rv1690 interact and form a tertiary structure with the histidine kinase KdpD which is involved in the signal transduction pathway (296). RpfB/Rv1009 (296) is the only lipoprotein among the five resuscitation-promoting factor (Rpf) proteins of *M. tuberculosis* that stimulate dormant cells to divide (240). The inactivation of *rpfB/rv1009* significantly delayed the growth rate of *M. tuberculosis* in the persistently infected mice model, suggesting a role of RpfB/Rv1009 as a resuscitation-promoting factor (297) (Table 1.3).

1.7.1.5.e Unknown function

The physiological/pathogenic roles of most of the mycobacterial lipoproteins are unknown. These include the 19 kDa/LpqH/ Rv3763/ML1966, LprA/Rv1270c, LppA and LppB, ML0603/33 kDa (45,78,79,115,131,149,158,210,219,222,275,296) (Table 1.3). However, the immunological role of these lipoproteins (described in section 1.8.1) has been intensively investigated in the context of developing a vaccine and elucidating hostpathogen interactions.

1.7.2 Glycoproteins

1.7.2.1 Identification and structure

Glycosylation of proteins was thought to be a unique phenomenon of higher eukaryotes, an event that is either absent or very scarce in prokaryotes (153). The 38 kDa/PhoS1/Rv0934 was first characterized as a glycoprotein associated with, or covalently linked to, sugars similar to those in lipo/arabinomannan, i.e. mostly mannose (56). By using immunoblot analysis of Concanavalin A, putative mycobacterial glycoproteins were identified including the 38 kDa/Pst1/Rv0934 and the 45 kDa/MPT32/Rv1860 (74). The concept of protein glycosylation in M. smegmatis was further enhanced by using the mutated LpqH/Rv3763 which substituted Thr-Thr, the presumed glycosylation motif, to Val-Val. This mutated LpqH/Rv3763 expressed in M. smegmatis, abolished the Concanavalin A binding affinity but was unaffected in wild type (109). Eleven different *M. tuberculosis* lipoproteins were screened by a neural network (NetOglyc) analysis and expressed in *M. smegmatis* by using the 19kDa lipoprotein signal peptide (108). Concanavalin A immunoblot detected the glycosylations of eight recombinant lipoproteins including 19 kDa/LpqH/Rv3763, 38 kDa/PhoS1/Rv0934, MPT83/Rv2873, GlnG/Rv0411c, LppN/Rv2270, LppQRv2431, LprI/Rv1541c and SodC (108,296). Recombinant LprA/Rv1270c (222) and LprG/Rv1411c (270) were recently shown to be Concanavalin A-reactive as mycobacterial glycoproteins.

Direct biochemical evidence of protein glycosylation has been demonstrated for only three proteins, 45/47-kDa/MPT32/Rv1860 (64,65), SodC/Rv0432 from *M*. *tuberculosis* (252) and MPB83 from *M. bovis* (181). By mass spectrometry and Nterminal AA sequencing, two studies (64,65) showed that four of the Thr residues of 45/47-kDa/MPT32/Rv1860 were attached to α -D-mannose, α -D-mannobiose or α -Dmannotiose solely by $\alpha(1\rightarrow 2)$ linkages. Michell *et al.* (181) identified an O-glycosylated Thr-Thr doublet at the N-terminus of secreted MPB83 in *M. bovis* by site directed mutagenesis, Concanavalin A immunoblot and ESI mass spectrometry. MPB83 in *M*.

bovis predominantly contained a mannose and mannobiose via an $\alpha(1\rightarrow 3)$ linkage (181). With the same approaches, Sartain *et al.* (252) showed that SodC in *M. tuberculosis* contained six O-glycosylated residues within a 13-amino-acid region near the N-terminus of the protein.

1.7.2.2 Biosynthesis and function

Mycobacterial protein glycosylation resembles eukarytotic protein Omannosylation and mostly occurs in the Thr/Ser rich region of proteins (252). The eukaryotic O-mannosyl transferases contain an Asp-Glu motif which plays a crucial role in the catalysis of glycosylation (153). The biosynthetic mechanism of mycobacterial glycoproteins is largely unknown. Recently, VanderVen et al (299) provided evidence that the Rv1002c/ML0192 which also contained the conserved Asp-Glu motif, was involved in the first step of protein glycosylation. In this study, the Asp-Glu motif of Rv1002c was substituted to either Ala or Asp and Ala-Asp. The *in vitro* enzyme catalytic assay revealed that the wild type Rv1002c efficiently transferred radiolabeled $[^{14}C]$ mannose from guanosine diphosphate mannose to synthetic peptide compared to the substituted mutants (299). By using mass spectrometry, the glycosylation pattern analyses of recombinant hybrid secretory proteins, with/or without Type I signal peptide, showed that the Sec-dependent protein export system was required for protein Omannosylation in M. tuberculosis (299). However it is still not clear whether other glycosyltrasferases are involved in the elongation step of protein mannosylation.

Protein glycosylation plays numerous roles in eukaryotes, including protein stability, protein localization, protein folding, receptor recognition and enzymatic activity

(153). The functional roles of prokaryotic protein glycosylation are still obscure but may contribute to maintenance of bacterial shape, protection against proteolytic digestion, enzyme-substrate binding and direct interaction with the host immune system (298). Additionally, glycosylation of both 45/47-kDa/MPT32/Rv1860 and LprG/Rv1411c is required for T-cell recognition of these proteins in order to serve as Major Histocompatibility Complex-II (MHC-II) restricted epitopes (111,245,270). Glycosylated MPT32 as an adhesion molecule was capable of capturing pulmonary C-type lectin surfactant protein, which is often facilitated in immune suppression by pathogens, from crude human bronchoalveolar lavage fluid (232). The Thr to Val substitutions at the glycosylation may serve to protect against proteolysis (109).

1.7.3 Phosphoproteins

In bacteria, protein phosphorylation is a major signal transduction system to sense and adapt to environmental conditions by two conserved components, a histidine protein kinase and a response regulator protein (282). The coupled phosphorylation and dephosphorylation conveys intracellular signals to regulate cellular responses. In response of environmental stimuli, histidine protein kinase is autophosphorylated at a histidine residue of the cytoplasmic sensor domain by its conformational change. This phosphate group of histidine protein kinase is transferred to an aspartate residue of histidine protein kinase. Subsequent phosphorylation of response regulator elicits the regulation of gene expression and intracellular responses (282) (Figure 1.16.A). The *M. tuberculosis* genome encodes 11 pairs of histidine protein kinase- response regulator

known as two-component systems and a few isolated histidine protein kinases and response regulators (46) but only three histidine protein kinase-response regulator s are found in *M. leprae* (Table 1.4).



Figure 1.16 Schematic overview of phosphorylation and signal transduction of bacterial proteins. A) Two component system B) Phosphorylation system of serine/threonine protein kinases; the existence of a partner phosphatase is still unclear.

Eukaryotic-like serine/threonine protein kinases have two domains: the kinase domain at the N-terminus transmits from the environmental signal to the C-terminus region, which induces intracellular responses through a transmembrane segment, and phosphorylated substrates (46) (Figure 1.16.B). A partner phosphatase in eukaryotes detaches a phosphate group from Ser, Thr or Tyr of serine/threonine protein kinases as the primary signal transduction pathway but it is still unclear if the process occurred in prokaryotes. Compared to other bacterial genomes, mycobacterial genomes contain a relatively small number of histidine protein kinase-response regulator systems (305). Eleven of the annotated serine/threonine protein kinases in the *M. tuberculosis* genome are assumed to compensate for the paucity of histidine protein kinase-response regulator (46,305) but only four putative serine/threonine protein kinases are found in *M. leprae*(8). Recent studies revealed that serine/threonine protein kinases regulate intracellular metabolic processes including transcription, cell division, virulence and chemotaxis (96)
(Table 1.4).

| | Gene Name | ORF from M. tuberculosis | ORF from <i>M. leprae</i> | Signal/ Substrate | Function |
|----------------------------|------------------------------------|-----------------------------|------------------------------|--|---|
| Two component system | MprA/ MprB ⁽³²⁵⁾ | Rv0982/Rv0981 | ML0174/ ML0174 | ? | Maintenance of persistent infections |
| (HK/RR) | TcrS/ TcrR ⁽¹⁰⁶⁾ | Rv1032/Rv1033 | <u>ML0260/ML026</u> 1 | ? | ? |
| | RegX3/ SenX3 ⁽⁹²⁾ | Rv0490/Rv0491 | ML2440c/ML24 39 | Phosphate starvation | ? |
| | DevS/ DevR ^(122,247) | Rv3132/Rv3133 | -/- | Hypoxia, NO | Induction of dormacy |
| | MtrB/MtrA ⁽³²⁴⁾ | Rv3245/Rv3248 | ML0774/ ML0773 | ? | Cell wall Integrity/ cell division |
| STPKs | PknA (135,300) | Rv0015 | ML0017 | ?/ EmbR | Cell elongation /division |
| | | | | ?/mtFabH | Regulation of mycolic acid biosynthesis |
| | PknB ^(135,305) | Rv0014 | ML0016 | ?/ EmbR | Cell elongation /division |
| | PknE ⁽¹⁸⁸⁾ | Rv1743 | ML0992* | ? | ? |
| | PknF ^(8,190,300,305) | Rv1746 | <u>ML1266</u> | Cell growth / Rv1747 ? /mtFabH (Rv0533) | Membrane transport/ regulation of mycolic acid biosynthesis |
| | PknG ^(8,51) | Rv0410 | ML0304 | Glutamine level | Amino-acid uptake, stationary-phase metabolism |
| | PknH ^(8,189) | Rv1266 | <u>ML0992</u> * | Stress response/ EmbR, Rv0681, Rv3330 | Arabinan metabolism |

Table 1.4 Mycobacterial phosphoproteins (adapted from Av-Gay and Everett (8)with permission).

All genes in *M. tuberculosis* have been demonstrated to autophosphorylate their recombinant forms and their function has been evaluated mostly with gene deletion mutants. Underlined ORF in *M. leprae* indicates a pseudogene. HK, histidine protein kinase; RR, response regulator; STPKs, serine/threonine protein kinases, NO, nitric oxide. *: ML0992c is similar to both PknE/Rv1743 and PknH/Rv1266.

1.7.4 Methylated Proteins

The heparin-binding-hemagglutinin has been shown to bind sulfated or glycoconjugated surface proteins of nonphagocytic cells postulated to be important for extrapulmonary dissemination (179,180,225). Recombinant heparin-bindinghemagglutinin/Rv0485/ ML2454c (149,296) produced in *M. smegmatis* was found to be multiple methylated on 13 lysine residues but the *E. coli*-expressed proteins heparinbinding-hemagglutinin/Rv0485/ML2454c was not observed to be methylated (227,228). Methylated heparin-binding-hemagglutinin was well recognized by sera of tuberculosis and leprosy patients (60,266,276,326) and also induced strong T-cells responses from *M. tuberculosis* infected individuals (290).

M. leprae surface-exposed laminin-binding protein/ML1683c (149) plays an important role in invasion of Schwann cells. Soares *et al.* (276) demonstrated that the recombinant *M. leprae* laminin-binding protein interacts with the extracellular matrix and host cell surface. In this study, the C-terminus region of recombinant *M. leprae* laminin-binding protein was shown to be methylated by immunoblot analysis.

However, the mechanism and function of protein methylation in *Mycobacterium* spp. are still unknown. It has been proven that methylated heparin-binding-hemagglutinin enhanced binding affinity to epithelial glycosaminoglycans by using an anti-methyllysine antibody blocking assay (226). *E. coli* expressed recombinant heparinbinding-hemagglutinin and laminin-binding protein are susceptible to proteolytic digestion. However, the chemical methylation of these recombinant heparin-binding-hemagglutinin and laminin-binding protein by using NaBH₄ and formalin, restores the resistance to proteolysis (227).
1.7.5. Pupylated Proteins (Pupylation: prokaryotic ubiquitin-like protein modification)

Pupylation was recently characterized in *M. tuberculosis* (58). In eukaryotes, almost all proteins destined for proteasomal degradation are covalently attached to ubiquitin at lysine (Lys) residues of the C-terminus. Ubiquitin is a long immature polypeptide which is processed by ubiquitin-specific proteases and ubiquitin carboxy-terminal hydrolases. The resultant 76 AAs ubiquitin is first adenylated and activated by an E1 enzyme and ATP. The activated ubiquitin is subsequently transferred to a ubiquitin-conjugating enzyme (E2) and a protein ligase (E3), conjugating at a Lys residue of the target substrate through iso-peptide bonds (33) (Figure 1.17.A).



Figure 1.17 Comparison of the ubiquitin and prokaryotic ubiquitin-like protein (Pup) proteasome pathways (adapted from Burns and Darwin (33) with permission).

A) Ubiquitin is adenylated at the C-terminal di-glycine, followed by a series of thioesterification reactions and finally conjugated to a substrate protein. Ubiquitin is removed by deubiquitinases (DUBs) and the substrate protein is unfolded and delivered into the proteasome complex consisting core particles. B) Pup is first deamidated at the C-terminal glutamine and then conjugated to substrate proteins. Pup targets proteins to *Mycobacterium* proteasome associated ATPase (Mpa) and the proteasome complex.

E3 ligases contain various substrate binding activities that provide specificity to the ubiquitin–proteasome system. In general, polyubiquitin chains can be recognized by the regulatory complex of proteasomes, then removed and recycled by deubiquitinase for further ubiquitination (57)

Bacterial proteasomes and pupylation have been identified exclusively in Actinomycetes, including *Mycobacterium, Streptomyces, Rhodococcus, Frankia*, and experimentally verified in *Methanococcus jannaschii* by degrading model peptide substrates (57). Similar to ubiquitin, Pup (prokaryotic ubiquitin-like protein/Rv2111/ ML1321) (149,296) is attached at Lys residues of protein substrates via its C-terminus. Pup has a terminal GlyGly motif at the C-terminus followed by either glutamate (Glu) or glutamine (Gln) (Figure 1.17.B). Prior to conjugation to the Lys residue of target proteins, the Gln of Pup is deamidated to Glu via catalysis of Pup deamidase/ Rv2112/ML1320 (149,283,296). Proteasome-associated factor A/Rv2097/ML1328 (149,283,296) catalyses conjugation of Pup to proteasome-substrates complex in the presence of ATP and Pup deamidase (33,57). The mechanism and functions of proteasome in prokaryotic physiology are largely unknown.

Nitric oxide produced by activated macrophage efficiently kills invading pathogenic bacteria; however, *M. tuberculosis* can persist even in the presence of nitric oxide (57,58). The *mpa* (*Mycobacterium proteasome associated ATPase/rv2115/ml1316*) and proteasome-associated factor (*rv2097*) mutants increased *in vitro* nitric oxide sensitivity and attenuated *M. tuberculosis* virulence in infected mouse model compared to the wild type (58). Additionally, *mpa* mutants have been proven to alter the expression levels of Rv2111c, FabD and PanA compared to the wild-type strain, implicating

Mpa/Rv2115 in a protein degradation pathway (221). Genome analysis revealed that *M*. *leprae* contain Pup and all proteasome related components, including proteasome-associated facto, *mpa* and the protease core genes *prcB* and *prcA* (149) (Figure 1.18). The Pup-proteasome system is conserved in all *Mycobacterium* spp. and might be crucial to intracellular survival of *M. tuberculosis* and *M. leprae* in the presence of the immune response and nitric oxide. However, the mechanism of proteasome-associated nitric oxide tolerance remains unclear in terms of host-pathogen interaction.



Figure 1.18 Gene arrangement of Pup and proteasome in both *M. tuberculosis* and *M. leprae* (adapted from Darwin *et al.* (57) with permission). Genes of the bacterial proteasome core consist of a β -subunit gene (*prcB*) and an α -subunit (*prcA*). Dop, deamidase of Pup; pafA, proteasome-associated factor.

1.8 Immune response

The immune response to *M. leprae*, like other pathogens, is divided into the innate and adaptive immune response. Adaptive immune response to *M. leprae* infection is thought to be mediated by $CD4^+$ T-cells, which play a critical role in both protective and pathogenic immune response (185). Differentiation of $CD4^+$ T cells is initiated by the innate immune system, which provides costimulatory molecules to allow for activation and proliferation of T cells in response to specific antigens (53). Subsequently, activated

CD4⁺ T cells produce the polarized cytokines (Th1 or Th2) in response to antigens presented by Major Histocompatibility Class II (MHC-II) complexes on antigen presenting cells. Th1 cells secrete IL-2 and IFN- γ which activate cellular mediated immunity and cytotoxicity T cells (53). In contrast, Th2 cells produce IL-4 and IL-5 mostly activating a humoral response.

More than 90-95% of *M. leprae*-exposed individuals are resistant to disease, while developing a strong immune response to the bacteria (3,81). As previously mentioned (see section 1.2.1), tuberculoid leprosy (TT) patients exhibit strong cellular mediated immunity (CMI) to *M. leprae* antigens and can localize the infection to a few affected lesions. On the other hand, lepromatous leprosy (LL) patients present a weak CMI and higher antibody levels to *M. leprae* antigens, having higher bacterial loads in diffused skin lesions. Therefore, the CMI response is thought to represent protective immunity to *M. leprae* infection, while humoral immune response is considered to favor the multiplication of bacteria in human hosts (summarized in Figures 1.19 and 1.21). Since there is no available facile whole animal model to study *M. leprae* infection in the laboratory, the role of Th1/Th2 immunity in *M. leprae* infection has been indirectly evaluated by *in vitro* stimulation of peripheral blood mononuclear cells (PBMCs) or T cell clones derived from leprosy patients, and comparative immunohistology between TT and LL skin lesions.



Figure 1.19 Immune response in tuberculoid leprosy (TT) lesions. In TT lesions, several pattern recognition receptors on phagocytes recognize *M. leprae* and initiate innate immunity via a MyD88 dependent signaling pathway or unknown mechanisms (mostly shown in the left side of the figure). Particularly, VitD dependent antimicrobial activity via TLR signaling induces autophagy to control intracellular *M. leprae* (illustrated in blue and described in section 1.7.2). In response to *M. leprae*, activated antigen presenting cells (APCs) produce high levels of proinflammatory cytokines and upregulate the surface expression of MHC-I, II, CD1s instructing adaptive Th1 immunity (illustrated at the right upper corner). Th1 cytokines (IL-2 and IFN- γ) produced by T cells elicit the recruitment of the cellular immune components and enhance the antimicrobial activity and antigen presentation of APCs (shown in green). VDR, vitamin D receptor; GM-CSF, granulocyte-macrophage colony stimulating factor.

Th1 cytokines, IL-2 and IFN- γ , are predominantly found in skin lesions of TT patients. Early immunohistochemical analysis of T-cell phenotype showed that the ratio of CD4⁺ versus CD8⁺T cells in TT lesions was 2:1. CD4⁺ T helper cells and CD8⁺T-cells were also more abundant in skin lesions of TT patients than in those of LL patients (186). MHC class II-restricted CD4⁺ T-cell clones from TT patients were shown to secrete high levels of IFN- γ (216). Additionally, granulysin-expressing CD4⁺ T cells frequently appeared in TT skin lesions (211). Later, immunohistochemistry study combined with RT-PCR analysis showed that the expression of IL-12 in TT lesions was 10-fold greater than LL lesions (272). Subsequently, recombinant IL-12 (rIL-12) successfully induced the proliferation of type 1 CD4⁺ T cell clones from TT patients but failed to stimulate that of type 2 CD8⁺ T cell clones from LL patients (274).

In contrast, Th2 cytokines, IL-4 and IL-5, and IL-10 highly expressed in LL lesions and CD8⁺ T cells were more frequently observed than CD4 ⁺ T cells by immunohistochemical analysis of these lesions (186). The phenotypes of both T cell subsets were found to be equivalent to those of a naïve subset or a suppressor subset of T cells. CD8⁺ T cell clones from these patients secreted high levels of IL-4 (186). Sieling *et al.* (274) showed that the presence of IL-4 and IL-10 inhibited IL-12 secretion by monocytes upon *in vitro* stimulation with *M. leprae*. This result suggested that Th2 cytokines representing Th2 immunity in LL lesions could play an important role in T cell anergy, favoring *M. leprae* dissemination in LL lesions. Several hypotheses exist to explain the T cell anergy in LL lesions including the genetic predisposition of human host and the immune modulation of *M. leprae* (see sections 1.8.2 and 1.9). The efficient innate immunity observed in TT skin lesions successfully controls *M. leprae* infection. It

suggests that the failure of innate immunity allows *M. leprae* survival in phagocytes and subsequently delays the initiation of adaptive immunity, contributing to the pathogenesis of leprosy.

Collectively, the progression of leprosy disease highly correlates with the polarized Th1/Th2 cytokines and T cell proliferation in response to *M. leprae*. In conjunction with the clinical features of leprosy (see section 1.2.1), growing evidence indicates that human genetic polymorphisms involved in the innate or adaptive immune response strongly influence the disease outcome (see section 1.9).

1.8.1 Innate immunity

Innate response is initiated through interaction between the pathogen associated molecular patterns (PAMPs) on the surface of pathogens and pattern recognition receptors such as TLRs, C type lectin receptors, NOD and NOD like receptors on or within host phagocytes (289). Among several pattern recognition receptors, TLR-2 plays an important role in immune recognition of *M. leprae* as well as other mycobacteria (128). The heterodimer TLR-2/1 is triggered by tri-acylated lipoproteins while heterodimer TLR-2/6 is activated by di-acylated lipoproteins (289) (Figure 1.20).

Bacterial lipoproteins activate pro-inflammatory signals via TLR-2 and induce Th1 immunity (2,255). TLRs, as integral membrane glycoproteins, share a conserved intracellular Toll /interleukin-1 receptor domain in the cytoplasmic region. The extracellular leucine rich repeats domain of TLRs recognizes PAMPs and is responsible for the signal transduction pathway in innate immune system (289) (Figure 1.20). The leucine rich repeats domain of TLRs is sensitized by a heterodimer with either TLR-1 for the tri-acylated form or TLR-6 for the di-acylated form (2) (Figure 1.20). X-ray crystallography of TLR-2 and TLR-1/ or -6 revealed that extracellular M-shaped structures of the TLR-2/1 and the TLR-2/6 heterodimer interacted with lipid portion of their ligands (289).



Figure 1.20 Myeloid differentiation factor 88 (MyD88) dependent toll like receptor-2 (TLR-2) signaling pathway (adapted from Takeuchi and Akira (289) with

permission). Either TLR-2/1 or TLR-2/6 recognizes the bacterial lipoproteins and recruits MyD88 by bridging with TIRAP/MAL. MyD88 interacts with IRAK-4. IRAK-4 is activated and hyperphosphorylates IRAK-1/IRAK-2. IRAKs dissociate from MyD88 and interact with TRAF-6 (=E3 ubiquitin protein ligase). TRAF-6 catalyzes Lys 63-linked polyubiquitin on itself and generates an unconjugated polyubiquitin chain activating Ubc13, Uev1A and TGF-β-activated kinase-1 (TAK-1) complex. Subsequent TAK-1 binding protein-2 (TAB-2)-TAB-3 complex induce the phosphorylation of the IKK (IκB kinase subunit)-γ/NF-κB essential modulator (NEMO) and MAPKs. The activation of the IKK complex (IKK-α, IKK-β and IKKγ) leads to the translocation NF-KB to the nucleus by phosphorylation and degradation of IκB. Mitogen–activated protein kinase (MAPKs) including ERK-1/2, p38, and JNK (c-Jun N-terminal kinase) activate activator protein-1 (AP-1) to induce the production of proinflammatory cytokines and antimicrobial activity.

TLR signaling pathways are intracellularly mediated through five Toll /interleukin-1 receptor domain domains containing adaptors, including myeloid differentiation factor 88 (MyD88), Toll /interleukin-1 receptor domain associated protein /MyD88 adaptor like (TIRAP/MAL), Toll /interleukin-1 receptor domain containing adaptor protein-inducing IFN- β /or Toll /interleukin-1 receptor domain-containing molecule 1, TRIF related adaptor molecule and sterile- α and armadillo motif-containing protein. While TLR-4 signaling is mediated by both MyD88 dependent and independent pathways, TLR-2 signaling relies solely on the MyD88 dependent pathway (2). MyD88 by association with IL-1 receptor-associated kinase -1 (IRAK-1) and IRAK-4 triggers mitogen–activated protein kinase, leading to the activation of NF- κ B and activator protein-1as transcription factors (220,289).

As TLR-2 ligands, mycobacterial lipoproteins such as *M. tuberculosis* 19 kDa/LpqH/Rv3793 (45,79,205,210,219), LprG (78,114,115), LprA (222), PhoS1 (131,236) and *M. leprae* 33 kDa/LpK/ML0603 (158,319) can mostly activate three different types of cellular mediated immunity (CMI) on antigen-presenting cells such as macrophages and dendritic cells: 1) increase cytokine production (IL-1, IL-12, IL-18, IFN- γ and TNF- α) (143,144); 2) enhance the antimicrobial activity produced reactive nitrogen intermediates, reactive oxygen intermediates and antimicrobial peptides (28,193,320); and 3) induce the expression of MHC-I, II and CD1 on the cell surface and then enhance adaptive immunity (102,104,195) (Figure 1.19). The N-terminal acylated moiety of mycobacterial lipoproteins is mostly responsible for activation of innate immunity and has been demonstrated using signal peptide deletion mutants of LprG/Rv1411c and LprA/Rv1270 (114,222) or synthetic lipopeptides derived from 19 kDa/LpqH/Rv3793 or 33 kDa/LpK/ML0603 (143).

Additionally, several mycobacterial lipoproteins were investigated as vaccine candidates to induce efficient adaptive immunity, particularly in *M. tuberculosis* infections (131,235,236,270). *M. tuberculosis* 19 kDa/LpqH and 38 kDa/PhoS1 induce antibody production and strong memory CD4⁺ T cells and CD8⁺ cytotoxic T cells from human and mice (131,235). Recently, Sieling *et al.* (270) demonstrated that both recombinant LprG/Rv1411c and LprG/ML0557 were recognized by CD4⁺ MHC-II restricted T cell clones from a leprosy lesion and patient blood.

Pathogenic mycobacteria actively inhibit, subvert or modulate the antigen presentation by suppressing the expression of MHC-I, II and CD1 molecules delaying adaptive immunity (88,102,119,128,139,140,210,219). Differential TLR-2 signaling modulates downstream MAPK signaling pathways contributing to the pathogenesis of mycobacterial disease by the regulation of antimycobacterial activities including apoptosis and MHC-II antigen presentation (128). Prolonged *in vitro* exposure of murine and human macrophages to LpqH/Rv3763, LprG/1411c and LprA/Rv1270c, inhibited the expression of MHC class II molecule (78,131,210,222). Pennini *et al.* (223) demonstrated that LpqH/Rv3763 inhibited the expression of MHC-II transcription activator or the IFN-γ dependent histone acetylation, which control MHC-II transcription activator promoter, in murine macrophage RAW264.7 cell line but not in TLR-2 -/- mice. Immunoblot analysis of both LpqH/Rv3763-stimulated macrophages by using antiphospho-38 and extracellular signal-regulated kinase (ERK) antibodies showed that the subsequent regulation of MHC-II was mediated by the p38 or ERK pathways (223). However, this finding needs to be carefully investigated because the MAPKs cascade is redundantly shared with other pattern recognition receptors particularly C-type lectin receptors. Several biochemical studies demonstrated the glycosylation of 19kDa/LpqH/Rv3763 (109), 38kDa/PhoS1/Rv0934(74), LprA/Rv1270c (222) and LprG/1411c (270). The glycosylated MPT32, as an adhesion molecule, was capable to bind the C-type lectin on phagocytic cells (see section 1.7.2.2) (232) implying that glycosylation of mycobacterial lipoproteins could synergistically contribute to this type of immune evasion response. Additionally, glycolipids copurified with recombinant lipoproteins can modulate the immune response which was recently demonstrated using recombinant LprG (68). Since mycobacterial lipoproteins used in previous immunological studies have never been structurally characterized, further studies need to clarify the immune mechanisms in mycobacterial infection by using structurally characterized mycobacterial lipoproteins.

In *M. leprae* infection, immunohistochemical analysis of leprosy lesions showed that TLR-2 and TLR-1 were more strongly expressed in TT lesions than LL lesions (143,194) suggesting that immune recognition via TLR-2 signaling plays an important role in the development of protective immunity against *M. leprae* infection.

In *M. tuberculosis* infection, LMs (130), LAMs (107,130,313), trehalose dimycolate (19), and the 19 kDa lipoprotein (28,147) activated TLR-2 on macrophages with or without the other cell surface receptors, TLR-1, CD14, scavenger receptor A, dectin-1 and macrophage receptor with collageneous structure, to produce the subsequent proinflammatory cytokines (TNF- α , IFN- γ , IL-12) and antimicrobial activity (Table 1.5) (Figure 1.19). As previously described, the structures of these mycobacterial PAMPs are

similar to those of *M. leprae* with some structural modifications. It remains to be addressed whether the structural differences of PAMPs between *M. leprae* and other pathogenic mycobacteria affect the TLR-2 signaling pathway in terms of the pathogenesis of leprosy.

Table 1.5 TLR recognition of mycobacterial components (adapted from Jo *et al.*(128) with permission)

| Mycobacterial components | TLR | Species |
|---|--------------|---|
| 33 kDa lipoprotein ^(143,319) | TLR-2 | M. leprae |
| MMP-II ⁽¹⁶⁶⁾ | TLR-2 | M. leprae |
| 19 kDa lipoprotein/LpqH ^(28,210) | TLR-2 | M. tuberculosis, M. bovis |
| 38kDa lipoprotein/PhoS1 ⁽¹³¹⁾ | TLR-2, TLR-4 | M. tuberculosis |
| LprA ⁽²²²⁾ | TLR-2 | M. tuberculosis, M. bovis |
| LprG ⁽⁷⁸⁾ | TLR-2 | M. tuberculosis, M. bovis |
| LMs ⁽²³¹⁾ | TLR-2 | M. tuberculosis, M. bovis |
| (212) | TLR-2 | |
| AraLAMs ⁽³¹³⁾ | | M. smegmatis |
| PIMs ⁽⁹⁰⁾ | TLR-2 | M. tuberculosis, M. bovis, M. smegmatis |
| Trehalose dimycolate ⁽¹⁹⁾ | TLR-2 | M. tuberculosis |

A recent major research advance in understanding mycobacterial innate immunity was that the vitamin D pathway plays a key role in antimicrobial mechanisms. Vitamin D receptor is intracellularly activated by 1,25D3, an active form of VitD. The 25-hydroxy vitamin D3-1α-hydroxylase (CYP27b1) is responsible for catalyzing 25D (an inactive form of vitamin D) into 1,25D3 (an active form of vitamin D) (323) (Figures 1.12; shown in blue). Analysis of gene expression profiles showed that Vitamin D receptor, CYP27b1 and cathelicidin were highly induced in skin lesions of TT patients (194). In response to *M. tuberculosis*, upregulated 1,25D3 and Vitamin D receptor highly induced the expression of both cathelicidin and defensin and subsequently activated autophagy, which is destructive to intracellular pathogens (323). Recently, Shin *et al.* (267) revealed that the 19 kDa/LpqH from *M. tuberculosis* causes the upregulation of

autophagy, NO, and antimicrobial peptides. Therefore, VitD antimicrobial activity via TLR-2 signaling plays an important role in the control of mycobacterial infections.

1.8.2 Immune modulation

Mycobacteria enter into phagocytes through receptor mediated phagocytosis. These receptors responsible for entry are the complement receptors and C-type lectin receptors, including mannose receptor (CD206), DC-SIGN (CD209, dendritic cellspecific intracellular adhesion molecule-grabbing nonintegrin) and scavenger receptors. The receptors recognize the carbohydrate containing molecules in mycobaterial envelopes, such as ManLAMs, PGL-I, glycoproteins (262). These molecules can modulate the intracellular signaling cascades of the host and suppress CMI, favoring the intracellular survival of mycobacteria (10,102,246). This immune modulation is thought to be an important mechanism of T cell anergy which appears in LL skin lesions (Figure 1.21).

M. leprae enter macrophages through interaction of complement (C3), complement receptors on phagocytes and PGL-I of *M. leprae* (257,258) suppressing the IL-2 dependent pathway of Th1 cell activation (37,178,258). ManLAMs interact with mannose receptors and DC-SIGNs on macrophages, dampening antimicrobial activity by modulating the production of cytokines and nitric oxide (130,262). Although different from macrophages, DC-SIGN on dendritic cells binds ManLAMs to uptake mycobacteria and suppresse dendritic cells maturation (208). The virulent mycobacteria in phagosomes actively produce ManLAMs which inhibit the Ca²⁺ dependent recruitment of type 3 phospatidyl inositol 3- kinase into the phagosomes (130).



Figure 1.21 Immune response in lepromatous leprosy (LL) lesions. Some PAMPs of *M. leprae*, particularly carbohydrate-containing cell surface molecules, enhance phagocytosis of *M. leprae* by interacting with host pattern recognition receptors, primarily complement receptors (CRs) and C-type lectin receptors (CLRs) Purple boxes indicate the immune modulation of ManLAMs as major virulence factors. *M. leprae* in the phagosome of antigen presenting cell (APCs) actively produce ManLAMs which inhibit phagolysosome maturation, resulting in the suppression of dendritic cells maturation. This event allows escaping *M. leprae* antigens to be presented and to induce Th 2 immunity in LL lesions (illustrated in red). The resulting Th 2 cytokines suppress the antimicrobial activity of APCs but also deposit host oxidized low-density lipoproteins (LDLs) in *M. leprae* infected phagocytes, favoring *M. leprae*.

This action depletes phosphatidyl inositol 3-phosphate in the membrane of phagosomes and inhibits normal phagolysosomal maturation (130) (Figure 1.21). Subsequently, it allows mycobacteria to avoid killing and the presentation of mycobacterial antigens to T cells.

1.8.3 CD1 restricted T cells

The CD1 family of MHC-class-I-like glycoprotein (CD1a, CD1b, CD1c, CD1d and CD1e) presents both foreign and self lipids to T cells (11,29,199). In antigen presentation by CD1, the alkyl chain(s) of an amphipathic lipid antigen is (are) inserted into the hydrophobic and invariant antigen binding pocket of CD1 but the hydrophilic head of the lipid antigen is exposed out of this binding pocket (11). Nonpolymorphic CD1 molecules present many mycobacterial nonpeptide antigens including LM, PIM, LAM (59,268) mycolic acids (146,196), didehydroxy-mycobactin (198), mannosyl β -1phosphomycoketides (174,197) and sulfolipid Ac₂SGL (91) (Figures 1.7, 1.8 and 1.22) and activate clonal CD1⁺ T cells. Importantly, the immune response to these nonpeptide antigens has been more frequently observed in patients infected by *M. tuberculosis* or *M. leprae* compared to healthy individuals with positive tuberculosis skin tests (59,91).



Figure 1.22 Chemical structures of CD1 restricted mycobacterial antigens (adapted from Barral and Brenner (11) with permission) A) CD1a antigens. B) CD1b antigens. C) CD1c antigens. n1=17, n2=12, n3=14, n4=24.

CD1 restricted T-cells activated by mycobacterial antigens contribute to developing protective immunity in mycobacterial infection by releasing high levels of IFN- γ (269), but also by proliferation of cytolytic T cells (12,194). These cytolytic T cells efficiently kill cells infected by live virulent *M. tuberculosis* (281). The cytolytic activity of CD1⁺ T cells in mycobacterial infections is categorized into two types: 1) CD4⁻CD8⁻ CD1 restricted T cells kill the mycobacteria-infected cells by the induction of apoptosis, but do not kill intracellular *M. tuberculosis*. LAMs and glucose monomycolate are presented to CD1b⁺ T cell lines derived from leprosy patients (196,268); 2) CD8⁺ CD1 restricted T cells lyse the infected cells via the expression of granulysin and kill approximately 50% of intracellular bacteria (280,281). Most CD1 antigens are recognized by these types of T cell lines. Recently, a CD4⁺ CD1 restricted T cell clone was found in *M. tuberculosis* infection and recognized glycerol monomycolates but its cytolytic activity was not characterized yet (146).

Higher levels of CD1⁺dendritic cells are found in the skin lesions of TT patients compared to those in LL skin lesions (144,271,273). Langerins, as a C-type lectin receptor, are highly expressed on the surface of these CD1⁺dendritic cells which are known as Langerhans cells (273). Langerhans cells in skin lesions of TT patients coexpressed CD1a and langerin, and efficiently presented non-peptide antigens which were assumed to be the mycolic acids and arabinogalactan complex (273). However, the structure of antigens presented by CD1a has not been defined.

After human blood monocytes were infected with *M. tuberculosis*, flow cytometric analysis of these cells showed that lipid antigen presentation by CD1b restricted dendritic cells was detected within 24 hr. However, peptide antigen presentation of MHC-II restricted dendritic cells was found to be delayed or absent after four days though the dendritic cells were fully mature (104). Therefore, CD1 restricted T cell immunity plays an important role in controlling pathogenic mycobacteria in the early stages of infection.

<u>1.9 Host genetic influences on human leprosy</u>

Several human genetic studies have provided evidence of linkage between genes of the immune response on chromosome 6, 9, 11, 12, 13, 15, 16, 17 and 20, and leprosy susceptibility (3,95,262). Based on results of complex segregation and linkage analysis, genetic susceptibility was found to influence the immune response in leprosy as a two stage model (3,262). At stage 1, the group1 genes (Parkin and Parkin co regulated gene (PARK2/PACRG), lymphotoxin A and natural resistance-associated macrophage protein 1) contribute to establish innate immunity eventually leading to leprosy resistance in 90-95% of *M. leprae* exposed individuals (Figure 1.23). At stage 2, 5-10% of susceptible individuals developed TT (CMI response), BT, BL and LL (humoral immune response) depending on group 2 genes such as those for TNF- α , TLR-1/2, IFN- γ , IFN- γ receptor, IL-10, MHC-I, MHC-II, transporter associated with antigen processing (TAP), IL-12 and its receptor (IL12RB1) and Vitamin D receptor (VDR) (3,262) (Figure 1.23).



Figure 1.23 Two-stage model of genetic susceptibility on human leprosy. Upon *M. leprae* exposure, group 1 genes result in either susceptibility or resistance by establishing innate immunity. Group 2 genes in 5-10% of infected individuals determine the clinical presentation of leprosy by affecting acquired immunity.

One of the most important finding in leprosy research is that loci within *PARK2* and a coregulated gene, *PACRG*, located on chromosome 6 q25-q27 are correlated with the development of human leprosy, despite the fact that leprosy is clearly an infectious disease (182). Both genes are liked to susceptibility to Parkinson disease. PARK2 codes for E3 ubiquitin–protein ligase that is involved in the delivery of polyubiquitinated proteins to the proteasome complex in other neurodegenerative diseases (95) but the

function of PACRG is unknown. Both genes are strongly expressed in Schwann cells and monocyte derived macrophages (182). Single nucleotide polymorphisms in the 5' regulatory region shared by both *PARK2* and *PACRG*s, were found to be associated with susceptibility to general leprosy by genome scans of both a Vietnamese patient population and a Brazilian population. Additionally, two of the single nucleotide polymorphisms (PARK2_e01, position 697; PARK2_e01, position-2599) associated with leprosy susceptibility (182). However, these single nucleotide polymorphisms in an Indian population revealed different effects in the regulation of genetic susceptibility to leprosy (95,261); the role of these genes in *M. leprae* infection is still not clear.

Several studies showed that TLR-1 and 2 genes contribute to the development of leprosy. Recent genetic polymorphism studies support that TLR-1 T to G nucleotide substitution at 1,805 single nucleotide polymorphism of TLR-1 (I602S), is associated with a decreased incidence of leprosy among white Europeans (129,316). Significantly, this study provided the first experimental evidence to directly evaluate the biological function of mutated TLR-1 in conjunction with gene susceptibility to leprosy. A common single nucleotide polymorphism, I602S, within TLR-1, leads to aberrant trafficking of TLR-1 to the cell surface and diminishes responses of blood monocytes to bacterial lipopeptides (129).

In general, genetic polymorphisms could potentially affect clinical outcomes of leprosy. However, the great degree of genetic diversity within and among populations and complexity of molecular interactions in individuals interfere in efforts to analyze the potential risk of such human genetic factors for leprosy (95).

<u>1.10 Immune based diagnosis</u>

As previously discussed in section 1.2.1, all leprosy patients characteristically exhibit an immune response to *M. leprae* antigens; tuberculoid leprosy (TT) or paucibacillary leprosy (PB) patients predominantly present CMI to bacterial antigens but lepromatous leprosy (LL) or multibacillary leprosy (MB) patients exhibit humoral immune response. Therefore, a substantial amount of research focused on developing immune based diagnostic tests for the diagnosis of leprosy (6,7,24,32,41,66,71-73,80-82,84-87,187,277,306).

House hold contacts (HHCs) with close contact to MB patients, develop strong CMI and delayed type hypersensitivity to *M. leprae* antigens but do not present clinical symptoms of leprosy (85). Therefore, CMI based diagnostic tests have been extensively investigated to detect *M. leprae* infection in the early stages and also to distinguish it from infections due to other mycobacterial exposure/infection.

As previously described, most individuals exposed to *M. leprae* do not develop the disease, but raise an effective CMI response to *M. leprae* (3,81,85). A CMI based assay to *M. leprae* specific antigens is thought to be the most promising diagnostic tool to detect asymptomatic leprosy in the early stages of *M. leprae* infection (24,84,85). This immune response to *M. leprae* antigens has proven to be predominantly dependent on the presence of memory T-cells secreting IFN- γ and IL-2 as the "inducer" of CMI (218). Therefore, CMI based diagnosis has focused on evaluating the ability of memory T cells to produce IFN- γ in response to a panel of *M. leprae* specific antigens as a consequence of mycobaterial infection and exposure (49,201,217). In order to be used routinely in clinical practice, this type of diagnostic assay has been developed to measure either local

delayed type hypersensitivity *in vivo* or the IFN-γ level released *in vitro* to *M. leprae* specific antigens (24,84).

1.10.1 Skin test

delayed type hypersensitivity is manifested by infiltration of T cells and monocytes into the affected tissues (217). It occurs 24-72 hr after an immunized or infected individual is challenged by microbial antigens (218). In 1890, the tuberculin skin test was developed for measuring local delayed type hypersensitivity reflecting past or recent exposure to *M. tuberculosis* using crude *M. tuberculosis* antigens (purified protein derivatives of *M. tuberculosis*; PPD) (218).

In 1919, Mitsuda revealed that volunteers with or without leprosy developed a local granulomatous induration to injected Lepromin-H 3-4 weeks later. This Lepromin-H had been prepared by autoclaving suspensions of macerated nodules from untreated LL patients and evoked the reaction known as the Mitsuda reaction (184). Later, it was observed that PB/TT patients and HHCs showed an early response to Lepromin-H between 48-72 hr, after injection (76). This Fernandez reaction represents a delayed type hypersensitivity response to *M. leprae* soluble antigens similar to PPD of *M. tuberculosis* (24). However, a substantial amount of healthy individuals in endemic region produced a false positive response to the Lepromin H (24,262). Thus, the lepromin test is only used for classification and prognosis of leprosy (24,262,277).

Two more recent skin test reagents, Convits' and Rees' antigens, were prepared by fractionation of armadillo derived *M. leprae* to produce a new type of the leprosy skin test antigen (24). Efficacy of both antigens on human subjects was extensively studied in

different geographical endemic countries. It was proven that both antigens induced positive reactions in BT and TT patients but were negative in LL patients (48,99). However, both Convits' and Rees' antigens induced variable skin test responses in HHCs and endemic healthy population (100).

A major advantage of the skin test is that it costs little to prepare and does not require any laboratory infrastructure to perform the diagnostic test in poor resource settings where leprosy is most prevalent. (217,218). In order to improve the specificity of the leprosy skin test, fractionation of native whole cell lysate of *M. leprae* generated two subcellular fractions, cell wall antigens without LAMs (MLCwA) and soluble antigens devoid of LAMs (MLSA-LAM); Both MLCwA and MLSA-LAM had been subjected to Triton X-114 phase separation to deplete the large amount of LAMs responsible for T-cell cross-reactivity and suppression of their immune response (24). An *in vitro* stimulation study on peripheral blood mononuclear cells (PBMCs) showed that both MLCwA and MLSA-LAM induced strong CMI response in leprosy patients and HHCs but little cross-reactivity in PPD-positive healthy individuals in leprosy endemic regions (167,306). These products had completed Phase II trials in Nepal in 2010. Data analysis now on the way will reveal the diagnostic potential of these antigens.

1.10.2 Antigen identification

The major focus of research in the development of leprosy diagnosis was on studies to identify *M. leprae* specific antigens, those stimulating both B- and T-cell responses. Identification of *M. leprae* antigens, particularly protein based antigens, has been accomplished using two different approaches: 1) immunological and biochemical

analysis of *M. leprae* derived molecules and; 2) *in silico* analysis of mycobacterial genomes (a postgenomic approach). Table 1.6 summarizes all antigens which have been evaluated by both CMI and serological assay formats.

The first immunological and biochemical approach to antigen identification focused on the characterization of the dominant *M. leprae* protein antigens, those which are abundantly expressed in *M. leprae*. They include GroES (243), MCP-II (322), Sod A (322), major membrane proteins-I (MMP-I) (294,315) and MMP-II (224) as well as the glycolipid antigens, LAMs, PIMs and PGL-I (117) (see section 1.6.2). However, the challenge encountered with these abundant *M. leprae* protein antigens is that many of these antigens share over 80% of amino acid sequence identity/similarity with orthologues in other mycobacteria resulting in low specificity of the corresponding diagnostic assays (66). Although *M. avium* contains an orthologue of MMP-I (294,315), MMP-I elicited strong T-cell proliferation from peripheral blood mononuclear cells (PBMCs) of TT patients (263,294). MMP-II shares higher homology with orthologues among mycobacteria but this protein is low abundance in other mycobacteria and elicited strong seropositivity in leprosy patients but negligible reactivity in *tuberculosis* (TB) patients (62,160,166,212,263).

Completion of whole genome sequencing of *Mycobacterium* spp. was anticipated to resolve the low specificity of both TB and leprosy diagnostic tests (6,7,66,73,81,278,279). A post genomic approach was utilized to identify TB antigens. An early antigenic target was the 6 kDa (ESAT-6) secretory protein. It appears in culture filtrated proteins of *M. tuberculosis* and is encoded by genes in the region of deletion-1 which is deleted from all BCG vaccine strains (4,5,307).

| <i>M. leprae</i> antigens | Homology with other mycobacteria | Immune responses | Types of test to be used | Observation |
|--|--|-----------------------------|-----------------------------|---|
| MLSA-LAM | high | CMI | Skin test | Highly associated with abundant immunogens (7, 65, 45, 35, 28, 18 and 10kDa), <i>M. leprae</i> lysates and PPD. ^(167,306) |
| MLCwA | high | CMI | Skin test | Highly associated with abundant immunogens (7, 65, 45, 35, 28, 18 and 10kDa), <i>M. leprae</i> lysates and PPD. ^(167,306) |
| GroES | 89.9% with <i>M. tuberculosis</i> | Humoral immunity | ELISA | High titers of IgG1 in all disease spectra. ⁽¹¹⁸⁾ |
| p25-40 derived from <i>M. leprae</i> GroES | 86.6% with <i>M. tuberculosis</i> | CMI | IGRA | Specifically detected 5-80% of PB and HHCs. ^(42,43,138) |
| MMP-I/ | 92.2% with | Humoral | ELISA/ | Specifically detected 97.5% of MB. ⁽²⁹³⁾ |
| 35 kDa/ML0841 | M. avium | immunity | IGRA | <i>M. smegmatis</i> -derived 35-kDa induced strong T cell immunity in the majority of PB and HHCs but not in MB. ^{(294)} |
| p60-76, p132-151, p206-224 and p267- 286 derived from MMP-I | 92.2%, 84%, 94% and 89% with <i>M.</i> <i>avium</i> , respectively | Humoral immunity | ELISA | Specifically detected <i>M. leprae</i> exposure 30% for PB, 48% for MB, 22% for HHCs. ⁽⁴⁴⁾ |
| p 206-2242 derived from MMP-I | 94% with <i>M. avium</i> | CMI | IGRA | Specifically detected <i>M. leprae</i> exposure 63% for PB, 66% for HHCs. ⁽³¹⁴⁾ |
| MMP-II | 90.1% with <i>M. tuberculosis</i> | Humoral immunity | ELISA (IgG) | Elicited humoral responses predominantly in leprosy patients but negligible responses in TB patients. ^(112,114) Detected 39-48% of PB and 85% of MB. Antibody titer decreased 50% to 18% after chemotherapy. ^(103,133,159) |
| ESAT-6 | 38% with <i>M. tuberculosis</i> | CMI /humoral immunity | IGRA | Induced a strong CMI from PBMCs of leprosy patients as well as <i>tuberculosis</i> patients and TB HHCs. ^(86,279) |
| CFP-10 | 40% with <i>M. tuberculosis</i> | CMI /humoral immunity | IGRA | Cross-reactive with <i>M. tuberculosis</i> . ^(87,278) |

 Table 1.6 M. leprae antigens and their potential use in diagnostics (adapted from Goulart and Goulart (94) with permission)

| <i>M. leprae</i> antigens | Homology with other mycobacteria | Immune responses | Types of test to be used | Observation |
|--|--|---------------------|-----------------------------|---|
| ML0308, ML2498 | No | CMI/ Humoral | ELISA IGRA | Detected TT and LL patients. ⁽⁷⁾ |
| Ep1-0308 derived from ML0308 | No | CMI | IGRA | Proposed a simple robust whole blood test for early diagnosis of leprosy. ⁽⁶⁾ |
| ML0678, ML0757, ML2177, ML2244, ML2498 | No ^a | Humoral immunity | IGRA | Strongly recognized circulating antibodies in MB patients. ⁽⁷⁾ |
| ML0008, ML0126, ML1057, ML2567 | No | CMI | IGRA | Showed IFN-γ responses only in paucibacillary patients. ⁽²⁷⁷⁾ |
| Peptides:p67, p68 and p69 from ML1419; p75 from ML2177; p91 from ML2347; and p92 from ML2452 | No | CMI | IGRA | Detected 60-100% of PB patients and 50-100% of HHCs. ⁽²⁷⁷⁾ |
| ML0576, ML1989, ML1990, ML2283 and ML2567 | No | | IGRA | Significantly inducted IFN- γ and detected 94% of negative PGL-1 patients. ML2283 is most specific antigen to detect <i>M. leprae</i> exposure/infection. ^(82,83) |
| Peptides (ML1989 p11, ML1990 p7, ML2283 p5, and ML2567 p3) | No | CMI | IGRA | Detected all PB patients (n=6) and 13 out of 14 HHCs. ⁽⁸³⁾ |
| LAMs PGL-1 ND-O-BSA NT-P-BSA | No | Humoral immunity | ELISA/ ML flow test | Detected 74–100% of MB and 13.6% of PB patients. ^(31,32,93,215) Used for operational classification. ^(40,41) |
| ML0405, ML2331 | 38% & 82% with M. tuberculosis | Humoral immunity | ELISA (IgG) | Specifically identified LL/borderline lepromatous (BL) patients Detect 84-88% of MB and 16-20% of PB. ⁽²³⁹⁾ (71,73) |
| LID-1, fusion construct of ML405 and ML2331 | N/A | Humoral immunity | ELISA (IgG) | Detected 92% of MB and 48% of PB patients. ^(71,73) |

a, ML0757 showed 77% with *M. tuberculosis* CDC at 2010 search; NA, not shown in literature. ML flow test, *M. leprae* lateral flow test; IGRA, *in vitro* interferon- γ release assay; ELISA, enzyme-linked immunosorbent assay:

In vitro IFN-γ release assay (IGRA) of PBMCs to ESAT-6 specifically differentiates TB patients from healthy individuals who were immunized with the BCG vaccine or exposed to environmental mycobacteria (218,307).

M. leprae ESAT-6/ML0049 and CFP-10/ML0050 show only 38% and 40% of AA sequence identity with orthologues of *M. tuberculosis* (149,296). Consequently, these antigens generated no cross-reactive CMI or humoral responses in mice immunized with either *M. leprae* or *M. tuberculosis* (278,279). However, both *M. leprae* ESAT-6 and CFP-10 induced a strong CMI from PBMCs of leprosy patients as well as TB patients and HHCs of TB patients in endemic regions (86,87).

Comparative bioinformatic analysis of the genomes of *Mycobacterium* spp. allowed the identification of *M. leprae* unique proteins into a group called as "hypothetical unknowns" (functional class VI) (47,149,277). Several hypothetical unknowns containing T-cell epitopes restricted to major HLA (human leukocyte antigen)-DR alleles were identified as potential antigen candidates and were evaluated in leprosy and control populations of various endemic regions by using IGRA (6,7,80,81,85,277). Geluk *et al.* (80) revealed that the selected hypothetical unknowns proteins, ML0576, ML1989, and ML1990, ML2283 and ML2567 induced significant levels of IFN- γ secretion from PBMCs of PB patients and HHCs, but not from most endemic healthy individuals and TB patients. However, a few of endemic controls and TB patients responded to these antigens due to previous exposure to *M. tuberculosis* or environmental mycobacteria in endemic settings.

In order to enhance the specificity of IGRA, Geluk *et al* (81) and Spencer *et al*. (277) generated several peptide antigens derived from hypothetical unknowns.

Subsequently, IGRA of PBMCs from PB patients and HHCs revealed that these peptide antigens can detect 60%-100% of the PB patients and 50-100% of the leprosy HHC group. Overall, these peptides enhanced the specificity of IGRAs which could distinguish leprosy patients or HHCs from individuals exposed to other mycobacteria but the inherent low sensitivity of these tests made it difficult to detect individuals with preclinical leprosy from healthy populations.

Additionally, ML0405 and ML2331 were identified from the *M. leprae* genomic expression library using sera of untreated LL patients (239). ML0405, ML2331 and LID-1, a fusion protein of both proteins, were capable of detecting antibodies in sera of LL and BL patients in various geographical regions (48,50).

1.10.3 In vitro Interferon-y Release Assays (IGRAs)

In mycobacterial infections, IFN- γ is a potent pro-inflammatory mediator to induce both memory CD4⁺ T cells and CD8⁺ T cells (201,217). Since IFN- γ is a stable cytokine and robustly induced upon stimulation, the level of IFN- γ secretion by PBMCs is facilitated to measure the induction levels of CMI to mycobacterial antigens as an outcome of mycobaterial infection (49,201,217).

As previously mentioned, abundant protein antigens of *M. leprae* as well as *M. tuberculosis* share a great deal of sequence similarity with orthologues among other mycobacteria. The antigens with this type of sequence similarity when applied to IGRAs generated high sensitivity but poor specificity particularly in endemic regions where most individuals are exposed to environmental mycobacteria and are vaccinated with BCG (82,218,263). A new type of IGRA emerged in the case of TB diagnosis which has been

developed commercially and is available as the QuantiFERON-TB or QuantiFERON-CMI tests. This test achieved impressive specificity for TB diagnosis by using overlapping peptide antigens derived from *M. tuberculosis* ESAT-6 (77,201,218).

Quanti-FERON types IGRAs have been adapted to develop CMI-based assays for early diagnosis of preclinical leprosy (6,7,80,82,85,277). As previously described, IGRAs based on hypothetical unknown proteins or their peptides, can detect individuals exposed to or infected by *M. leprae*, especially HHCs with MB or PB patients (6,7,80,82,85,277). Recent studies in five different geographical regions (Brazil, Nepal, Bangladesh, Pakistan and Ethiopia) revealed that IGRA of the peptides derived from hypothetical unknowns (see section 1.10.2) specifically discriminated *M. leprae*-exposed healthy individuals from endemic healthy populations but showed low sensitivity to detect the *M. leprae* exposure (82,84). Therefore, further searches for *M. leprae* specific antigens should continue to enhance both sensitivity and specificity of leprosy diagnosis in endemic regions.

1.10.4 Serodiagnosis

Leprosy serodiagnosis is capable of detecting 75-100% MB patients but show very poor performance in the case of HHCs and PB patients where only 15-40% are positive (1,73,263). Antibody levels in leprosy serodiagnosis are often used as a surrogate marker to measure the bacterial load in skin lesions of leprosy patients. Serological assays to *M. leprae* specific PGL-I and MMP-I detected more than 95% and 89% of MB patients and 33% and 40% of PB patients respectively, and showed no response in the case of TB patients or endemic healthy volunteers (31,32,41).

Seroreactivity of leprosy patients to MMP-I decreased after effective chemotherapy (314). Serodiagnosis of leprosy has been used to classify leprosy patients in order to assess the efficacy of chemotherapy and to predict relapse (263).

Serology based antibody detection is the most feasible diagnostic format to apply in leprosy endemic regions (49,72). Currently, the only available field friendly test for diagnosis of leprosy is the detection of circulating anti-PGL-I antibodies in patents' sera (72,73,83). The terminal trisaccharide unit of PGL-I, which is responsible for *M. leprae* specificity in serodiagnosis, is conjugated to bovine serum albumin (BSA) and human serum albumin (HSA) via either an octyl linker (NDO-BSA/HSA) or a phenyl linker (NT-P-BSA) (32,215). The resulting semi-synthetic glycoconjugates are incorporated into user friendly lateral flow tests (known as ML flow test) (32).

Recently, several studies revealed the potential of leprosy serodiagnosis to identify individuals with the risk of developing leprosy (72,93,103,133). A five-year monitoring study of 1,396 of HHCs using simple ML flow test identified 10.4% of HHCs who are test positive and subsequently developed leprosy symptoms. The disease development of HHCs with positive ML flow tests occurred six times higher than that of individuals with negative ML flow tests (93). However, this ML flow test often exhibits a higher background response making it difficult to distinguish the true antibody response from nonspecific antibody binding by detecting anti-PGL-I IgM antibody.

In order to overcome this problem, Maeda *et al* (159) demonstrated that seroreactivity of recombinant MMP-II in patients' sera was able to detect 39-48 % PB patients as well as 85% MB patients in non-endemic (Japan) and endemic regions (Indonesia and Vietnam). After chemotherapy, anti-MMP-II antibody titers in sera of leprosy patients decrease by 18-50%. Subsequently, seropositivity of recombinant MMP-II could predict disease progression of leprosy patients and HHCs similar to that of PGL-I (103,133). The fusion construct of ML0405 and ML2331, LID-1, was also proven to detect individuals who were in preclinical leprosy but presented clinical symptom, 6-8 month later (73).

1.11 References

- Agis, F., P. Schlich, J. L. Cartel, C. Guidi, and M. A. Bach. 1988. Use of anti-*M. leprae* phenolic glycolipid-I antibody detection for early diagnosis and prognosis of leprosy. Int.J.Lepr.Other Mycobact.Dis. 56:527-535.
- 2. Akira, S. 2003. Toll-like receptor signaling. J.Biol.Chem. 278:38105-38108.
- 3. Alter, A., A. Grant, L. Abel, A. Alcais, and E. Schurr. 2011. Leprosy as a genetic disease. Mamm.Genome. 22:19-31.
- 4. Andersen, P., A. B. Andersen, A. L. Sorensen, and S. Nagai. 1995. Recall of long-lived immunity to *Mycobacterium tuberculosis* infection in mice. J.Immunol. **154**:3359-3372.
- Andersen, P., M. E. Munk, J. M. Pollock, and T. M. Doherty. 2000. Specific immune-based diagnosis of tuberculosis. Lancet 356:1099-1104.
- Araoz, R., N. Honore, S. Banu, C. Demangel, Y. Cissoko, C. Arama, M. K. Uddin, S. K. Hadi, M. Monot, S. N. Cho, B. Ji, P. J. Brennan, S. Sow, and S. T. Cole. 2006. Towards an immunodiagnostic test for leprosy. Microbes.Infect. 8:2270-2276.
- Araoz, R., N. Honore, S. Cho, J. P. Kim, S. N. Cho, M. Monot, C. Demangel, P. J. Brennan, and S. T. Cole. 2006. Antigen discovery: a postgenomic approach to leprosy diagnosis. Infect.Immun. 74:175-182.
- 8. Av-Gay, Y. and M. Everett. 2000. The eukaryotic-like Ser/Thr protein kinases of *Mycobacterium tuberculosis*. Trends Microbiol. 8:238-244.
- Babu, M. M., M. L. Priya, A. T. Selvan, M. Madera, J. Gough, L. Aravind, and K. Sankaran. 2006. A database of bacterial lipoproteins (DOLOP) with functional assignments to predicted lipoproteins. J.Bacteriol. 188:2761-2773.
- 10. Baena, A. and S. A. Porcelli. 2009. Evasion and subversion of antigen presentation by *Mycobacterium tuberculosis*. Tissue Antigens 74:189-204..
- 11. **Barral, D. C. and M. B. Brenner**. 2007. CD1 antigen presentation: how it works. Nat.Rev.Immunol. **7**:929-941.
- 12. Beckman, E. M., S. A. Porcelli, C. T. Morita, S. M. Behar, S. T. Furlong, and M. B. Brenner. 1994. Recognition of a lipid antigen by CD1-restricted ab+ T cells. Nature **372**:691-694.
- 13. Belisle, J. T., M. E. Brandt, J. D. Radolf, and M. V. Norgard. 1994. Fatty acids of *Treponema pallidum* and *Borrelia burgdorferi* lipoproteins. J.Bacteriol. 176:2151-2157.
- Berg, S., D. Kaur, M. Jackson, and P. J. Brennan. 2007. The glycosyltransferases of *Mycobacterium tuberculosis*-roles in the synthesis of arabinogalactan, lipoarabinomannan, and other glycoconjugates. Glycobiology 17:35-56R.
- Bhamidi, S., M. S. Scherman, C. D. Rithner, J. E. Prenni, D. Chatterjee, K. H. Khoo, and M. R. McNeil. 2008. The identification and location of succinyl residues and the characterization of the interior arabinan region allow for a model of the complete primary structure of *Mycobacterium tuberculosis* mycolyl arabinogalactan. J.Biol.Chem. 283:12992-13000.
- 16. Bhatt, A., N. Fujiwara, K. Bhatt, S. S. Gurcha, L. Kremer, B. Chen, J. Chan, S. A. Porcelli, K. Kobayashi, G. S. Besra, and W. R. Jacobs, Jr. 2007. Deletion of kasB in *Mycobacterium tuberculosis* causes loss of acid-fastness and subclinical latent tuberculosis in immunocompetent mice. Proc.Natl.Acad.Sci U.S.A 104:5157-5162.
- Bigi, F., A. Gioffre, L. Klepp, M. P. Santangelo, A. Alito, K. Caimi, V. Meikle, M. Zumarraga, O. Taboga, M. I. Romano, and A. Cataldi. 2004. The knockout of the lprG-Rv1410 operon produces strong attenuation of *Mycobacterium tuberculosis*. Microbes.Infect. 6:182-187.
- 18. **Bordier, C.** 1981. Phase separation of integral membrane proteins in Triton X-114 solution. J.Biol.Chem. **256**:1604-1607.
- 19. Bowdish, D. M., K. Sakamoto, M. J. Kim, M. Kroos, S. Mukhopadhyay, C. A. Leifer, K. Tryggvason, S. Gordon, and D. G. Russell. 2009. MARCO, TLR2, and CD14 are required for

macrophage cytokine responses to mycobacterial trehalose dimycolate and *Mycobacterium tuberculosis*. PLoS.Pathog. **5**:1-14

- 20. **Braun, V.** 1975. Covalent lipoprotein from the outer membrane of *Escherichia coli*. Biochim.Biophys.Acta **415**:335-377.
- 21. **Braun, V. and K. Rehn**. 1969. Chemical characterization, spatial distribution and function of a lipoprotein (murein-lipoprotein) of the *E. coli* cell wall. The specific effect of trypsin on the membrane structure. Eur.J.Biochem. **10**:426-438.
- 22. Braunstein, M., B. J. Espinosa, J. Chan, J. T. Belisle, and W. R. Jacobs, Jr. 2003. SecA2 functions in the secretion of superoxide dismutase A and in the virulence of *Mycobacterium tuberculosis*. Mol.Microbiol. **48**:453-464.
- 23. Braunstein, M., I. V. Griffin TJ, J. I. Kriakov, S. T. Friedman, N. D. Grindley, and W. R. Jacobs, Jr. 2000. Identification of genes encoding exported *Mycobacterium tuberculosis* proteins using a Tn552'phoA *in vitro* transposition system. J.Bacteriol. **182**:2732-2740.
- 24. Brennan, P. J. 2000. Skin test development in leprosy: progress with first-generation skin test antigens, and an approach to the second generation. Lepr.Rev. 71 Suppl:S50-S54.
- 25. Brennan, P. J. and D. C. Crick. 2007. The cell-wall core of *Mycobacterium tuberculosis* in the context of drug discovery. Curr.Top.Med Chem. **7**:475-488.
- Brennan, P. J. and H. Nikaido. 1995. The envelope of mycobacteria. Annu.Rev.Biochem. 64:29-63.
- Brennan, P. J. and V. D. Vissa. 2001. Genomic evidence for the retention of the essential mycobacterial cell wall in the otherwise defective *Mycobacterium leprae*. Lepr.Rev. 72:415-428.
- Brightbill, H. D., D. H. Libraty, S. R. Krutzik, R. B. Yang, J. T. Belisle, J. R. Bleharski, M. Maitland, M. V. Norgard, S. E. Plevy, S. T. Smale, P. J. Brennan, B. R. Bloom, P. J. Godowski, and R. L. Modlin. 1999. Host defense mechanisms triggered by microbial lipoproteins through toll-like receptors. Science 285:732-736.
- 29. **Brigl, M. and M. B. Brenner**. 2004. CD1: antigen presentation and T cell function. Annu.Rev.Immunol. **22**:817-890..
- 30. Britton, W. J. and D. N. Lockwood. 2004. Leprosy. Lancet 363:1209-1219.
- Buhrer, S. S., H. L. Smits, G. C. Gussenhoven, C. W. van Ingen, and P. R. Klatser. 1998. A simple dipstick assay for the detection of antibodies to phenolic glycolipid-I of *Mycobacterium leprae*. Am.J.Trop.Med Hyg. 58:133-136.
- Buhrer-Sekula, S., H. L. Smits, G. C. Gussenhoven, L. J. van, S. Amador, T. Fujiwara, P. R. Klatser, and L. Oskam. 2003. Simple and fast lateral flow test for classification of leprosy patients and identification of contacts with high risk of developing leprosy. J.Clin.Microbiol. 41:1991-1995.
- 33. Burns, K. E. and K. H. Darwin. 2010. Pupylation versus ubiquitylation: tagging for proteasomedependent degradation. Cell Microbiol. **12**:424-431.
- Camacho, L. R., D. Ensergueix, E. Perez, B. Gicquel, and C. Guilhot. 1999. Identification of a virulence gene cluster of *Mycobacterium tuberculosis* by signature-tagged transposon mutagenesis. Mol.Microbiol. 34:257-267.
- Chan, J., T. Fujiwara, P. Brennan, M. McNeil, S. J. Turco, J. C. Sibille, M. Snapper, P. Aisen, and B. R. Bloom. 1989. Microbial glycolipids: possible virulence factors that scavenge oxygen radicals. Proc.Natl.Acad.Sci U.S.A 86:2453-2457.
- Chang, Z., A. Choudhary, R. Lathigra, and F. A. Quiocho. 1994. The immunodominant 38kDa lipoprotein antigen of *Mycobacterium tuberculosis* is a phosphate-binding protein. J.Biol.Chem. 269:1956-1958.
- Charlab, R., E. N. Sarno, D. Chatterjee, and M. C. Pessolani. 2001. Effect of unique *Mycobacterium leprae* phenolic glycolipid-I (PGL-I) on tumour necrosis factor production by human mononuclear cells. Lepr.Rev. 72:63-69.

- 38. Charles Davis, MD PhD. http://www.medicinenet.com/leprosy/article.htm.
- Chatterjee, D., K. Lowell, B. Rivoire, M. R. McNeil, and P. J. Brennan. 1992. Lipoarabinomannan of *Mycobacterium tuberculosis*. Capping with mannosyl residues in some strains. J.Biol.Chem. 267:6234-6239.
- Cho, S. N., R. V. Cellona, L. G. Villahermosa, T. T. Fajardo, Jr., M. V. Balagon, R. M. Abalos, E. V. Tan, G. P. Walsh, J. D. Kim, and P. J. Brennan. 2001. Detection of phenolic glycolipid I of *Mycobacterium leprae* in sera from leprosy patients before and after start of multidrug therapy. Clin.Diagn.Lab Immunol. 8:138-142.
- Cho, S. N., D. L. Yanagihara, S. W. Hunter, R. H. Gelber, and P. J. Brennan. 1983. Serological specificity of phenolic glycolipid I from *Mycobacterium leprae* and use in serodiagnosis of leprosy. Infect.Immun. 41:1077-1083.
- 42. Chua-Intra, B., J. Ivanyi, A. Hills, J. Thole, C. Moreno, and H. M. Vordermeier. 1998. Predominant recognition of species-specific determinants of the GroES homologues from *Mycobacterium leprae* and *M. tuberculosis*. Immunology **93**:64-72.
- 43. Chua-Intra, B., S. Peerapakorn, N. Davey, S. Jurcevic, M. Busson, H. M. Vordermeier, C. Pirayavaraporn, and J. Ivanyi. 1998. T-cell recognition of mycobacterial GroES peptides in Thai leprosy patients and contacts. Infect.Immun. 66:4903-4909.
- 44. Chua-Intra, B., S. Wattanapokayakit, S. Srisungngam, T. Srisungngam, K. Mahotarn, P. J. Brennan, and J. Ivanyi. 2003. T-cell recognition of peptides from the *Mycobacterium leprae* 35 kDa protein in Thai leprosy patients, healthy contacts, and non-contacts. Immunol.Lett. **88**:71-76.
- 45. Ciaramella, A., A. Martino, R. Cicconi, V. Colizzi, and M. Fraziano. 2000. Mycobacterial 19kDa lipoprotein mediates *Mycobacterium tuberculosis*-induced apoptosis in monocytes/macrophages at early stages of infection. Cell Death.Differ. **7**:1270-1272.
- 46. Cole, S. T., R. Brosch, J. Parkhill, T. Garnier, C. Churcher, D. Harris, S. V. Gordon, K. Eiglmeier, S. Gas, C. E. Barry, III, F. Tekaia, K. Badcock, D. Basham, D. Brown, T. Chillingworth, R. Connor, R. Davies, K. Devlin, T. Feltwell, S. Gentles, N. Hamlin, S. Holroyd, T. Hornsby, K. Jagels, A. Krogh, J. McLean, S. Moule, L. Murphy, K. Oliver, J. Osborne, M. A. Quail, M. A. Rajandream, J. Rogers, S. Rutter, K. Seeger, J. Skelton, R. Squares, S. Squares, J. E. Sulston, K. Taylor, S. Whitehead, and B. G. Barrell. 1998. Deciphering the biology of *Mycobacterium tuberculosis* from the complete genome sequence. Nature 393:537-544.
- Cole, S. T., K. Eiglmeier, J. Parkhill, K. D. James, N. R. Thomson, P. R. Wheeler, N. Honore, T. Garnier, C. Churcher, D. Harris, K. Mungall, D. Basham, D. Brown, T. Chillingworth, R. Connor, R. M. Davies, K. Devlin, S. Duthoy, T. Feltwell, A. Fraser, N. Hamlin, S. Holroyd, T. Hornsby, K. Jagels, C. Lacroix, J. Maclean, S. Moule, L. Murphy, K. Oliver, M. A. Quail, M. A. Rajandream, K. M. Rutherford, S. Rutter, K. Seeger, S. Simon, M. Simmonds, J. Skelton, R. Squares, S. Squares, K. Stevens, K. Taylor, S. Whitehead, J. R. Woodward, and B. G. Barrell. 2001. Massive gene decay in the leprosy bacillus. Nature 409:1007-1011.
- Convit, J., C. Sampson, M. Zuniga, P. G. Smith, J. Plata, J. Silva, J. Molina, M. E. Pinardi, B. R. Bloom, and A. Salgado. 1992. Immunoprophylactic trial with combined *Mycobacterium leprae*/BCG vaccine against leprosy: preliminary results. Lancet 339:446-450.
- 49. Corstjens, P. L., M. Zuiderwijk, H. J. Tanke, van der Ploeg-van Schip JJ, T. H. Ottenhoff, and A. Geluk. 2008. A user-friendly, highly sensitive assay to detect the IFN-gamma secretion by T cells. Clin.Biochem. **41**:440-444.
- Coulombe, F., M. Divangahi, F. Veyrier, L. L. de, J. L. Gleason, Y. Yang, M. A. Kelliher, A. K. Pandey, C. M. Sassetti, M. B. Reed, and M. A. Behr. 2009. Increased NOD2-mediated recognition of *N*-glycolyl muramyl dipeptide. J.Exp.Med 206:1709-1716.
- 51. Cowley, S., M. Ko, N. Pick, R. Chow, K. J. Downing, B. G. Gordhan, J. C. Betts, V. Mizrahi, D. A. Smith, R. W. Stokes, and Y. Av-Gay. 2004. The *Mycobacterium tuberculosis* protein

serine/threonine kinase PknG is linked to cellular glutamate/glutamine levels and is important for growth *in vivo*. Mol.Microbiol. **52**:1691-1702.

- Crick, D. C., L. Quadri, and P. J. Brennan. 2008. Biochemistry of the cell envelope of *Mycobacterium tuberculosis*, p. 1-19. *In*: S. H. E. Kaufmann and E. Rubin (eds.), Handbook of Tuberculosis: Moleucular biology and Biochemstry. WIELY-VCH Verlag GmbH & Co, Weinheim.
- 53. D'Elios, M. and P. G. Del. 1998. Th1/Th2 balance in human disease. Transplant.Proc. **30**:2373-2377.
- 54. **Daffe, M.** 2008. The architecture of envelope, p. 3-11. *In*: M. Daffe and J. M. Reyrat (eds.), The Mycobacterial cell envelope. ASM press.
- 55. **Daffe, M. and P. Draper**. 1998. The envelope layers of mycobacteria with reference to their pathogenicity. Adv.Microb.Physiol **39**:131-203.
- Daniel, T. M., N. J. Gonchoroff, J. A. Katzmann, and G. R. Olds. 1984. Specificity of *Mycobacterium tuberculosis* antigen 5 determined with mouse monoclonal antibodies. Infect.Immun. 45:52-55.
- 57. **Darwin, K. H.** 2009. Prokaryotic ubiquitin-like protein (Pup), proteasomes and pathogenesis. Nat.Rev.Microbiol. **7**:485-491.
- Darwin, K. H., S. Ehrt, J. C. Gutierrez-Ramos, N. Weich, and C. F. Nathan. 2003. The proteasome of *Mycobacterium tuberculosis* is required for resistance to nitric oxide. Science 302:1963-1966.
- de la Salle, H., S. Mariotti, C. Angenieux, M. Gilleron, L. F. Garcia-Alles, D. Malm, T. Berg, S. Paoletti, B. Maitre, L. Mourey, J. Salamero, J. P. Cazenave, D. Hanau, L. Mori, G. Puzo, and L. G. De. 2005. Assistance of microbial glycolipid antigen processing by CD1e. Science 310:1321-1324.
- de Lima, C. S., M. A. Marques, A. S. Debrie, E. C. Almeida, C. A. Silva, P. J. Brennan, E. N. Sarno, F. D. Menozzi, and M. C. Pessolani. 2009. Heparin-binding hemagglutinin (HBHA) of *Mycobacterium leprae* is expressed during infection and enhances bacterial adherence to epithelial cells. FEMS Microbiol.Lett. 292:162-169.
- 61. **de Souza, G. A., T. Softeland, C. J. Koehler, B. Thiede, and H. G. Wiker**. 2009. Validating divergent ORF annotation of the *Mycobacterium leprae* genome through a full translation data set and peptide identification by tandem mass spectrometry. Proteomics. **9**:3233-3243.
- 62. **Deshpande, R. G., M. B. Khan, D. A. Bhat, and R. G. Navalkar**. 1995. Immunoaffinity chromatographic isolation of a high molecular weight seroreactive protein from *Mycobacterium leprae* cell sonicate. FEMS Immunol.Med Microbiol. **11**:163-169.
- 63. DHARMENDRA. 1947. Leprosy in ancient Indian medicine. Int.J.Lepr. 15:424-430.
- 64. **Dobos, K. M., K. H. Khoo, K. M. Swiderek, P. J. Brennan, and J. T. Belisle**. 1996. Definition of the full extent of glycosylation of the 45-kilodalton glycoprotein of *Mycobacterium tuberculosis*. J.Bacteriol. **178**:2498-2506.
- Dobos, K. M., K. Swiderek, K. H. Khoo, P. J. Brennan, and J. T. Belisle. 1995. Evidence for glycosylation sites on the 45-kilodalton glycoprotein of *Mycobacterium tuberculosis*. Infect.Immun. 63:2846-2853.
- 66. Dockrell, H. M., S. Brahmbhatt, B. D. Robertson, S. Britton, U. Fruth, N. Gebre, M. Hunegnaw, R. Hussain, R. Manandhar, L. Murillo, M. C. Pessolani, P. Roche, J. L. Salgado, E. Sampaio, F. Shahid, J. E. Thole, and D. B. Young. 2000. A postgenomic approach to identification of *Mycobacterium leprae*-specific peptides as T-cell reagents. Infect.Immun. 68:5846-5855.
- Drage, M. G., N. D. Pecora, A. G. Hise, M. Febbraio, R. L. Silverstein, D. T. Golenbock, W. H. Boom, and C. V. Harding. 2009. TLR2 and its co-receptors determine responses of macrophages and dendritic cells to lipoproteins of *Mycobacterium tuberculosis*. Cell Immunol. 258:29-37.

- Drage, M. G., H. C. Tsai, N. D. Pecora, T. Y. Cheng, A. R. Arida, S. Shukla, R. E. Rojas, C. Seshadri, D. B. Moody, W. H. Boom, J. C. Sacchettini, and C. V. Harding. 2010. *Mycobacterium tuberculosis* lipoprotein LprG (Rv1411c) binds triacylated glycolipid agonists of Toll-like receptor 2. Nat.Struct.Mol.Biol. 17:1088-1095.
- 69. **Draper, P., G. Dobson, D. E. Minnikin, and S. M. Minnikin**. 1982. The mycolic acids of *Mycobacterium leprae* harvested from experimentally infected nine-banded armadillos. Ann Microbiol.(Paris) **133**:39-47.
- Draper, P., O. Kandler, and A. Darbre. 1987. Peptidoglycan and arabinogalactan of Mycobacterium leprae. J.Gen.Microbiol. 133:1187-1194.
- 71. Duthie, M. S., W. Goto, G. C. Ireton, S. T. Reece, L. P. Cardoso, C. M. Martelli, M. M. Stefani, M. Nakatani, R. C. de Jesus, E. M. Netto, M. V. Balagon, E. Tan, R. H. Gelber, Y. Maeda, M. Makino, D. Hoft, and S. G. Reed. 2007. Use of protein antigens for early serological diagnosis of leprosy. Clin.Vaccine Immunol. 14:1400-1408.
- 72. Duthie, M. S., M. N. Hay, C. Z. Morales, L. Carter, R. Mohamath, L. Ito, L. K. Oyafuso, M. I. Manini, M. V. Balagon, E. V. Tan, P. R. Saunderson, S. G. Reed, and D. Carter. 2010. Rational design and evaluation of a multiepitope chimeric fusion protein with the potential for leprosy diagnosis. Clin.Vaccine Immunol. 17:298-303.
- 73. Duthie, M. S., G. C. Ireton, G. V. Kanaujia, W. Goto, H. Liang, A. Bhatia, J. M. Busceti, M. MacDonald, K. D. Neupane, C. Ranjit, B. R. Sapkota, M. Balagon, J. Esfandiari, D. Carter, and S. G. Reed. 2008. Selection of antigens and development of prototype tests for point-of-care leprosy diagnosis. Clin.Vaccine Immunol. 15:1590-1597.
- 74. **Espitia**, C., I. Cervera, R. Gonzalez, and R. Mancilla. 1989. A 38-kD *Mycobacterium tuberculosis* antigen associated with infection. Its isolation and serologic evaluation. Clin.Exp.Immunol. **77**:373-377.
- 75. **Espitia, C., M. Elinos, R. Hernandez-Pando, and R. Mancilla**. 1992. Phosphate starvation enhances expression of the immunodominant 38-kilodalton protein antigen of *Mycobacterium tuberculosis*: demonstration by immunogold electron microscopy. Infect.Immun. **60**:2998-3001.
- 76. Fernandez, J. M. M. 1940. The early reaction induced by lepromin. Int.J.Lepr. 8:1-14.
- 77. Ferrara, G., M. Losi, R. D'Amico, P. Roversi, R. Piro, M. Meacci, B. Meccugni, I. M. Dori, A. Andreani, B. M. Bergamini, C. Mussini, F. Rumpianesi, L. M. Fabbri, and L. Richeldi. 2006. Use in routine clinical practice of two commercial blood tests for diagnosis of infection with *Mycobacterium tuberculosis*: a prospective study. Lancet **367**:1328-1334.
- Gehring, A. J., K. M. Dobos, J. T. Belisle, C. V. Harding, and W. H. Boom. 2004. *Mycobacterium tuberculosis* LprG (Rv1411c): a novel TLR-2 ligand that inhibits human macrophage class II MHC antigen processing. J.Immunol. 173:2660-2668.
- Gehring, A. J., R. E. Rojas, D. H. Canaday, D. L. Lakey, C. V. Harding, and W. H. Boom. 2003. The *Mycobacterium tuberculosis* 19-kilodalton lipoprotein inhibits gamma interferonregulated HLA-DR and Fc gamma R1 on human macrophages through Toll-like receptor 2. Infect.Immun. **71**:4487-4497.
- Geluk, A., M. R. Klein, K. L. Franken, K. E. van Meijgaarden, B. Wieles, K. C. Pereira, S. Buhrer-Sekula, P. R. Klatser, P. J. Brennan, J. S. Spencer, D. L. Williams, M. C. Pessolani, E. P. Sampaio, and T. H. Ottenhoff. 2005. Postgenomic approach to identify novel *Mycobacterium leprae* antigens with potential to improve immunodiagnosis of infection. Infect.Immun. 73:5636-5644.
- 81. Geluk, A. and T. H. Ottenhoff. 2006. HLA and leprosy in the pre and postgenomic eras. Hum.Immunol. 67:439-445.
- 82. Geluk, A., J. S. Spencer, K. Bobosha, M. C. Pessolani, G. M. Pereira, S. Banu, N. Honore, S. T. Reece, M. MacDonald, B. R. Sapkota, C. Ranjit, K. L. Franken, M. Zewdie, A. Aseffa, R. Hussain, M. M. Stefani, S. N. Cho, L. Oskam, P. J. Brennan, and H. M. Dockrell. 2009. From genome-based *in silico* predictions to *ex vivo* verification of leprosy diagnosis. Clin.Vaccine Immunol. 16:352-359.

- 83. Geluk, A., J. van der Ploeg, R. O. Teles, K. L. Franken, C. Prins, J. W. Drijfhout, E. N. Sarno, E. P. Sampaio, and T. H. Ottenhoff. 2008. Rational combination of peptides derived from different *Mycobacterium leprae* proteins improves sensitivity for immunodiagnosis of *M. leprae* infection. Clin.Vaccine Immunol. 15:522-533.
- 84. Geluk, A., van der Ploeg-van Schip JJ, K. E. van Meijgaarden, S. Commandeur, J. W. Drijfhout, W. E. Benckhuijsen, K. L. Franken, B. Naafs, and T. H. Ottenhoff. 2010. Enhancing sensitivity of detection of immune responses to *Mycobacterium leprae* peptides in whole-blood assays. Clin.Vaccine Immunol. 17:993-1004.
- 85. Geluk, A., P. J. van der, R. O. Teles, K. L. Franken, C. Prins, J. W. Drijfhout, E. N. Sarno, E. P. Sampaio, and T. H. Ottenhoff. 2008. Rational combination of peptides derived from different *Mycobacterium leprae* proteins improves sensitivity for immunodiagnosis of *M. leprae* infection. Clin.Vaccine Immunol. 15:522-533.
- 86. Geluk, A., K. E. van Meijgaarden, K. L. Franken, Y. W. Subronto, B. Wieles, S. M. Arend, E. P. Sampaio, B. T. de, W. R. Faber, B. Naafs, and T. H. Ottenhoff. 2002. Identification and characterization of the ESAT-6 homologue of *Mycobacterium leprae* and T-cell cross-reactivity with *Mycobacterium tuberculosis*. Infect.Immun. **70**:2544-2548.
- 87. Geluk, A., K. E. van Meijgaarden, K. L. Franken, B. Wieles, S. M. Arend, W. R. Faber, B. Naafs, and T. H. Ottenhoff. 2004. Immunological crossreactivity of the *Mycobacterium leprae* CFP-10 with its homologue in *Mycobacterium tuberculosis*. Scand.J.Immunol. **59**:66-70.
- 88. Gercken, J., J. Pryjma, M. Ernst, and H. D. Flad. 1994. Defective antigen presentation by *Mycobacterium tuberculosis*-infected monocytes. Infect.Immun. **62**:3472-3478.
- 89. Gibbons, H. S., F. Wolschendorf, M. Abshire, M. Niederweis, and M. Braunstein. 2007. Identification of two *Mycobacterium smegmatis* lipoproteins exported by a SecA2-dependent pathway. J.Bacteriol. **189**:5090-5100.
- 90. Gilleron, M., V. F. Quesniaux, and G. Puzo. 2003. Acylation state of the phosphatidylinositol hexamannosides from *Mycobacterium bovis* bacillus Calmette Guerin and *Mycobacterium tuberculosis* H37Rv and its implication in Toll-like receptor response. J.Biol.Chem. **278**:29880-29889.
- Gilleron, M., S. Stenger, Z. Mazorra, F. Wittke, S. Mariotti, G. Bohmer, J. Prandi, L. Mori, G. Puzo, and L. G. De. 2004. Diacylated sulfoglycolipids are novel mycobacterial antigens stimulating CD1-restricted T cells during infection with *Mycobacterium tuberculosis*. J.Exp.Med 199:649-659.
- Glover, R. T., J. Kriakov, S. J. Garforth, A. D. Baughn, and W. R. Jacobs, Jr. 2007. The twocomponent regulatory system senX3-regX3 regulates phosphate-dependent gene expression in *Mycobacterium smegmatis*. J.Bacteriol. 189:5495-5503.
- 93. Goulart, I. M., D. O. Bernardes Souza, C. R. Marques, V. L. Pimenta, M. A. Goncalves, and L. R. Goulart. 2008. Risk and protective factors for leprosy development determined by epidemiological surveillance of household contacts. Clin.Vaccine Immunol. 15:101-105.
- Goulart, I. M. and L. R. Goulart. 2008. Leprosy: diagnostic and control challenges for a worldwide disease. Arch.Dermatol Res. 300:269-290.
- 95. Goulart, L. R. and I. M. Goulart. 2009. Leprosy pathogenetic background: a review and lessons from other mycobacterial diseases. Arch.Dermatol Res. **301**:123-137.
- 96. Greenstein, A. E., C. Grundner, N. Echols, L. M. Gay, T. N. Lombana, C. A. Miecskowski, K. E. Pullen, P. Y. Sung, and T. Alber. 2005. Structure/function studies of Ser/Thr and Tyr protein phosphorylation in *Mycobacterium tuberculosis*. J.Mol.Microbiol.Biotechnol. 9:167-181.
- Guilhot, C. and M. Daffe. 2008. Polyketides and Polyketide-Containing Glycolipids of *Mycobacterium tuberculosis*: Structure, Biosynthesis and Biological Activities, p. 21-51. *In*: S. H. E. Kaufmann and E. Rubin (eds.), Handbook of Tuberculosis: Moleucular biology and Biochemstry. WIELY-VCH Verlag GmbH & Co, Weinheim.
- 98. Gupta, S. D., W. Dowhan, and H. C. Wu. 1991. Phosphatidylethanolamine is not essential for the *N*-acylation of apolipoprotein in *Escherichia coli*. J.Biol.Chem. **266**:9983-9986.
- 99. Gupte, M. D. and D. S. Anantharaman. 1988. Use of soluble antigens in leprosy epidemiology. Lepr.Rev. 59:329-335.
- Gupte, M. D., D. S. Anantharaman, B. Nagaraju, S. Kannan, and R. S. Vallishayee. 1990. Experiences with *Mycobacterium leprae* soluble antigens in a leprosy endemic population. Lepr.Rev. 61:132-144.
- 101. Gurcha, S. S., A. R. Baulard, L. Kremer, C. Locht, D. B. Moody, W. Muhlecker, C. E. Costello, D. C. Crick, P. J. Brennan, and G. S. Besra. 2002. Ppm1, a novel polyprenol monophosphomannose synthase from *Mycobacterium tuberculosis*. Biochem.J. 365:441-450.
- 102. Harding, C. V. and W. H. Boom. 2010. Regulation of antigen presentation by *Mycobacterium tuberculosis*: a role for Toll-like receptors. Nat.Rev.Microbiol. 8:296-307.
- 103. Hatta, M., M. Makino, Ratnawati, Mashudi, Yadi, M. Sabir, N. Tandirogang, L. M. Rusyati, M. Kai, Y. Fukutomi, Y. Miyamoto, T. Mukai, and Y. Maeda. 2009. Detection of serum antibodies to *M. leprae* major membrane protein-II in leprosy patients from Indonesia. Lepr.Rev. 80:402-409.
- 104. Hava, D. L., N. van der Wel, N. Cohen, C. C. Dascher, D. Houben, L. Leon, S. Agarwal, M. Sugita, Z. M. van, S. C. Kent, H. Shams, P. J. Peters, and M. B. Brenner. 2008. Evasion of peptide, but not lipid antigen presentation, through pathogen-induced dendritic cell maturation. Proc.Natl.Acad.Sci U.S.A 105:11281-11286.
- 105. Hayashi, S. and H. C. Wu. 1990. Lipoproteins in bacteria. J.Bioenerg.Biomembr. 22:451-471.
- Haydel, S. E., N. E. Dunlap, and W. H. Benjamin, Jr. 1999. *In vitro* evidence of twocomponent system phosphorylation between the *Mycobacterium tuberculosis* TrcR/TrcS proteins. Microb.Pathog. 26:195-206.
- 107. Heldwein, K. A. and M. J. Fenton. 2002. The role of Toll-like receptors in immunity against mycobacterial infection. Microbes.Infect. 4:937-944.
- Herrmann, J. L., R. Delahay, A. Gallagher, B. Robertson, and D. Young. 2000. Analysis of post-translational modification of mycobacterial proteins using a cassette expression system. FEBS Lett. 473:358-362.
- Herrmann, J. L., P. O'Gaora, A. Gallagher, J. E. Thole, and D. B. Young. 1996. Bacterial glycoproteins: a link between glycosylation and proteolytic cleavage of a 19 kDa antigen from *Mycobacterium tuberculosis*. EMBO J. 15:3547-3554.
- Hoffmann, C., A. Leis, M. Niederweis, J. M. Plitzko, and H. Engelhardt. 2008. Disclosure of the mycobacterial outer membrane: cryo-electron tomography and vitreous sections reveal the lipid bilayer structure. Proc.Natl.Acad.Sci U.S.A 105:3963-3967.
- 111. Horn, C., A. Namane, P. Pescher, M. Riviere, F. Romain, G. Puzo, O. Barzu, and G. Marchal. 1999. Decreased capacity of recombinant 45/47-kDa molecules (Apa) of *Mycobacterium tuberculosis* to stimulate T lymphocyte responses related to changes in their mannosylation pattern. J.Biol.Chem. 274:32023-32030.
- 112. **Hortez, P. J.** 2008. Forgotten people, forgotten diseases; The neglected tropical disease and their impact on global health and development. ASM presss.
- 113. Hoskisson, P. A. and M. I. Hutchings. 2006. MtrAB-LpqB: a conserved three-component system in actinobacteria? Trends Microbiol. 14:444-449.
- 114. Hovav, A. H., L. Davidovitch, G. Nussbaum, J. Mullerad, Y. Fishman, and H. Bercovier. 2004. Mitogenicity of the recombinant mycobacterial 27-kilodalton lipoprotein is not connected to its antiprotective effect. Infect.Immun. 72:3383-3390.
- 115. Hovav, A. H., J. Mullerad, L. Davidovitch, Y. Fishman, F. Bigi, A. Cataldi, and H. Bercovier. 2003. The *Mycobacterium tuberculosis* recombinant 27-kilodalton lipoprotein induces a strong Th1-type immune response deleterious to protection. Infect.Immun. **71**:3146-3154.
- Hunter, S. W., T. Fujiwara, and P. J. Brennan. 1982. Structure and antigenicity of the major specific glycolipid antigen of *Mycobacterium leprae*. J.Biol.Chem. 257:15072-15078.

- Hunter, S. W., B. Rivoire, V. Mehra, B. R. Bloom, and P. J. Brennan. 1990. The major native proteins of the leprosy bacillus. J.Biol.Chem. 265:14065-14068.
- 118. Hussain, R., F. Shahid, S. Zafar, M. Dojki, and H. M. Dockrell. 2004. Immune profiling of leprosy and tuberculosis patients to 15-mer peptides of *Mycobacterium leprae* and *M. tuberculosis* GroES in a BCG vaccinated area: implications for development of vaccine and diagnostic reagents. Immunology 111:462-471.
- Hussain, S., B. S. Zwilling, and W. P. Lafuse. 1999. *Mycobacterium avium* infection of mouse macrophages inhibits IFN-gamma Janus kinase-STAT signaling and gene induction by downregulation of the IFN-gamma receptor. J.Immunol. 163:2041-2048.
- Hutchings, M. I., T. Palmer, D. J. Harrington, and I. C. Sutcliffe. 2009. Lipoprotein biogenesis in Gram-positive bacteria: knowing when to hold 'em, knowing when to fold 'em. Trends Microbiol. 17:13-21.
- 121. Inohara, N., Y. Ogura, A. Fontalba, O. Gutierrez, F. Pons, J. Crespo, K. Fukase, S. Inamura, S. Kusumoto, M. Hashimoto, S. J. Foster, A. P. Moran, J. L. Fernandez-Luna, and G. Nunez. 2003. Host recognition of bacterial muramyl dipeptide mediated through NOD2. Implications for Crohn's disease. J.Biol.Chem. 278:5509-5512.
- Ioanoviciu, A., E. T. Yukl, P. Moenne-Loccoz, and P. R. de Montellano. 2007. DevS, a hemecontaining two-component oxygen sensor of *Mycobacterium tuberculosis*. Biochemistry 46:4250-4260.
- Jackson, M., D. C. Crick, and P. J. Brennan. 2000. Phosphatidylinositol is an essential phospholipid of mycobacteria. J.Biol.Chem. 275:30092-30099.
- 124. Jackson, M., G. Stadthagen, and B. Gicquel. 2007. Long-chain multiple methyl-branched fatty acid-containing lipids of *Mycobacterium tuberculosis*: biosynthesis, transport, regulation and biological activities. Tuberculosis.(Edinb.) 87:78-86.
- 125. **Jacobsen, P. L. and L. Levy**. 1973. Mechanism by which hydnocarpic acid inhibits mycobacterial multiplication. Antimicrob.Agents Chemother. **3**:373-379.
- Jacobsen, P. L., H. Ng, and L. Levy. 1973. The susceptibility of Mycobacteria to hydnocarpic acid. Am.Rev.Respir.Dis. 107:1022-1029.
- 127. Jacobson, R. R. and J. L. Krahenbuhl. 1999. Leprosy. Lancet 353:655-660.
- Jo, E. K., C. S. Yang, C. H. Choi, and C. V. Harding. 2007. Intracellular signalling cascades regulating innate immune responses to Mycobacteria: branching out from Toll-like receptors. Cell Microbiol. 9:1087-1098.
- 129. Johnson, C. M., E. A. Lyle, K. O. Omueti, V. A. Stepensky, O. Yegin, E. Alpsoy, L. Hamann, R. R. Schumann, and R. I. Tapping. 2007. Cutting edge: A common polymorphism impairs cell surface trafficking and functional responses of TLR1 but protects against leprosy. J.Immunol. 178:7520-7524.
- 130. Jozefowski, S., A. Sobota, and K. Kwiatkowska. 2008. How *Mycobacterium tuberculosis* subverts host immune responses. Bioessays **30**:943-954.
 - 131. Jung, S. B., C. S. Yang, J. S. Lee, A. R. Shin, S. S. Jung, J. W. Son, C. V. Harding, H. J. Kim, J. K. Park, T. H. Paik, C. H. Song, and E. K. Jo. 2006. The mycobacterial 38-kilodalton glycolipoprotein antigen activates the mitogen-activated protein kinase pathway and release of proinflammatory cytokines through Toll-like receptors 2 and 4 in human monocytes. Infect.Immun. 74:2686-2696.
 - Kai, M., Y. Fujita, Y. Maeda, N. Nakata, S. Izumi, I. Yano, and M. Makino. 2007. Identification of trehalose dimycolate (cord factor) in *Mycobacterium leprae*. FEBS Lett. 581:3345-3350.
 - 133. Kai, M., N. H. Nguyen Phuc, T. H. Hoang Thi, A. H. Nguyen, Y. Fukutomi, Y. Maeda, Y. Miyamoto, T. Mukai, T. Fujiwara, T. T. Nguyen, and M. Makino. 2008. Serological diagnosis of leprosy in patients in vietnam by enzyme-linked immunosorbent assay with *Mycobacterium leprae*-derived major membrane protein II. Clin.Vaccine Immunol. 15:1755-1759.

- Kamalakkannan, S., V. Murugan, M. V. Jagannadham, R. Nagaraj, and K. Sankaran. 2004. Bacterial lipid modification of proteins for novel protein engineering applications. Protein Eng Des Sel 17:721-729.
- 135. Kang, C. M., D. W. Abbott, S. T. Park, C. C. Dascher, L. C. Cantley, and R. N. Husson. 2005. The *Mycobacterium tuberculosis* serine/threonine kinases PknA and PknB: substrate identification and regulation of cell shape. Genes Dev. **19**:1692-1704.
- 136. Kaur, D., M. E. Guerin, H. Skovierova, P. J. Brennan, and M. Jackson. 2009. Chapter 2: Biogenesis of the cell wall and other glycoconjugates of *Mycobacterium tuberculosis*. Adv.Appl.Microbiol. 69:23-78.
- 137. Khanolkar-Young, S., A. H. Kolk, A. B. Andersen, J. Bennedsen, P. J. Brennan, B. Rivoire, S. Kuijper, K. P. McAdam, C. Abe, H. V. Batra, and . 1992. Results of the third immunology of leprosy/immunology of tuberculosis antimycobacterial monoclonal antibody workshop. Infect.Immun. 60:3925-3927.
- 138. Kim, J., A. Sette, S. Rodda, S. Southwood, P. A. Sieling, V. Mehra, J. D. Ohmen, J. Oliveros, E. Appella, Y. Higashimoto, T. H. Rea, B. R. Bloom, and R. L. Modlin. 1997. Determinants of T cell reactivity to the *Mycobacterium leprae* GroES homologue. J.Immunol. 159:335-343.
- 139. Kincaid, E. Z. and J. D. Ernst. 2003. *Mycobacterium tuberculosis* exerts gene-selective inhibition of transcriptional responses to IFN-gamma without inhibiting STAT1 function. J.Immunol. **171**:2042-2049.
- Kincaid, E. Z., A. J. Wolf, L. Desvignes, S. Mahapatra, D. C. Crick, P. J. Brennan, M. S. Pavelka, Jr., and J. D. Ernst. 2007. Codominance of TLR2-dependent and TLR2-independent modulation of MHC class II in *Mycobacterium tuberculosis* infection *in vivo*. J.Immunol. 179:3187-3195.
- Kirchheimer, W. F., E. E. Storrs, and C. H. Binford. 1972. Attempts to establish the Armadillo (*Dasypus novemcinctus*) as a model for the study of leprosy. II. Histopathologic and bacteriologic post-mortem findings in lepromatoid leprosy in the Armadillo. Int.J.Lepr.Other Mycobact.Dis. 40:229-242.
- 142. Kovacevic, S., D. Anderson, Y. S. Morita, J. Patterson, R. Haites, B. N. McMillan, R. Coppel, M. J. McConville, and H. Billman-Jacobe. 2006. Identification of a novel protein with a role in lipoarabinomannan biosynthesis in mycobacteria. J.Biol.Chem. 281:9011-9017.
- 143. Krutzik, S. R., M. T. Ochoa, P. A. Sieling, S. Uematsu, Y. W. Ng, A. Legaspi, P. T. Liu, S. T. Cole, P. J. Godowski, Y. Maeda, E. N. Sarno, M. V. Norgard, P. J. Brennan, S. Akira, T. H. Rea, and R. L. Modlin. 2003. Activation and regulation of Toll-like receptors 2 and 1 in human leprosy. Nat.Med 9:525-532.
- 144. Krutzik, S. R., B. Tan, H. Li, M. T. Ochoa, P. T. Liu, S. E. Sharfstein, T. G. Graeber, P. A. Sieling, Y. J. Liu, T. H. Rea, B. R. Bloom, and R. L. Modlin. 2005. TLR activation triggers the rapid differentiation of monocytes into macrophages and dendritic cells. Nat.Med 11:653-660.
- 145. Larsen, O. 1973. Gerhard Henrik Armauer Hansen seen through his own eyes. A review of his memoirs. Int.J.Lepr.Other Mycobact.Dis. **41**:208-214.
- 146. Layre, E., A. Collmann, M. Bastian, S. Mariotti, J. Czaplicki, J. Prandi, L. Mori, S. Stenger, L. G. De, G. Puzo, and M. Gilleron. 2009. Mycolic acids constitute a scaffold for mycobacterial lipid antigens stimulating CD1-restricted T cells. Chem.Biol. 16:82-92.
- Lee, H. M., J. M. Yuk, D. M. Shin, and E. K. Jo. 2009. Dectin-1 is inducible and plays an essential role for mycobacteria-induced innate immune responses in airway epithelial cells. J.Clin.Immunol. 29:795-805.
- 148. LENDRUM, F. C. 1952. The name leprosy. Am.J.Trop.Med.Hyg. 1:999-1008.
- 149. Leproma. http://genolist.pasteur.fr/Leproma/
- 150. Levy, L. 1975. The activity of chaulmoogra acids against *Mycobacterium leprae*. Am.Rev.Respir.Dis. 111:703-705.

- 151. Levy, L. 1976. Studies of the mouse foot pad technique for cultivation of *Mycobacterium leprae* doubling time during logarithmic multiplication. Lepr.Rev. 47:103-106.
- 152. Lockwood, D. N. and S. Suneetha. 2005. Leprosy: too complex a disease for a simple elimination paradigm. Bull.World Health Organ 83:230-235.
- 153. Lommel, M. and S. Strahl. 2009. Protein O-mannosylation: conserved from bacteria to humans. Glycobiology **19**:816-828.
- 154. **Lowe, J.** 1950. Treatment of leprosy with diamino-diphenyl sulphone by mouth. Lancet 1:145-150.
- 155. Lowe, J. 1951. Diaminodiphenylsulphone in the treatment of leprosy. Lancet 1:18-21.
- Lowenstein, E. J. 2004. Paleodermatoses: lessons learned from mummies. J.Am.Acad.Dermatol. 50:919-936.
- 157. Maeda, N., J. Nigou, J. L. Herrmann, M. Jackson, A. Amara, P. H. Lagrange, G. Puzo, B. Gicquel, and O. Neyrolles. 2003. The cell surface receptor DC-SIGN discriminates between *Mycobacterium* species through selective recognition of the mannose caps on lipoarabinomannan. J.Biol.Chem. 278:5513-5516.
- 158. Maeda, Y., M. Makino, D. C. Crick, S. Mahapatra, S. Srisungnam, T. Takii, Y. Kashiwabara, and P. J. Brennan. 2002. Novel 33-kilodalton lipoprotein from *Mycobacterium leprae*. Infect.Immun. 70:4106-4111.
- 159. Maeda, Y., T. Mukai, M. Kai, Y. Fukutomi, H. Nomaguchi, C. Abe, K. Kobayashi, S. Kitada, R. Maekura, I. Yano, N. Ishii, T. Mori, and M. Makino. 2007. Evaluation of major membrane protein-II as a tool for serodiagnosis of leprosy. FEMS Microbiol.Lett. 272:202-205.
- Maeda, Y., T. Mukai, J. Spencer, and M. Makino. 2005. Identification of an Immunomodulating Agent from *Mycobacterium leprae*. Infect.Immun. 73:2744-2750.
- 161. Mahapatra, S., J. Basu, P. J. Brennan, and D. C. Crick. 2005. Structure, biosynthesis, and genetics of the mycolic acid-arabinogalactan-peptidoglycan complex., p. 275-285. *In*: S. T. Cole, K. D. Eisenach, D. N. McMurray, and Jr. W. R. Jacobs (eds.). ASM Press, Washington, D.C.
- Mahapatra, S., D. C. Crick, and P. J. Brennan. 2000. Comparison of the UDP-Nacetylmuramate:L-alanine ligase enzymes from *Mycobacterium tuberculosis* and *Mycobacterium leprae*. J.Bacteriol. 182:6827-6830.
- Mahapatra, S., D. C. Crick, M. R. McNeil, and P. J. Brennan. 2008. Unique structural features of the peptidoglycan of *Mycobacterium leprae*. J.Bacteriol. 190:655-661.
- 164. **Mahapatra, S., H. Scherman, P. J. Brennan, and D. C. Crick**. 2005. *N*-Glycolylation of the nucleotide precursors of peptidoglycan biosynthesis of *Mycobacterium* spp. is altered by drug treatment. J.Bacteriol. **187**:2341-2347.
- 165. Mahenthiralingam, E., B. I. Marklund, L. A. Brooks, D. A. Smith, G. J. Bancroft, and R. W. Stokes. 1998. Site-directed mutagenesis of the 19-kilodalton lipoprotein antigen reveals No essential role for the protein in the growth and virulence of *Mycobacterium intracellulare*. Infect.Immun. 66:3626-3634.
- Makino, M., Y. Maeda, and N. Ishii. 2005. Immunostimulatory activity of major membrane protein-II from *Mycobacterium leprae*. Cell Immunol. 233:53-60.
- Manandhar, R., J. W. LeMaster, C. R. Butlin, P. J. Brennan, and P. W. Roche. 2000. Interferon-gamma responses to candidate leprosy skin-test reagents detect exposure to leprosy in an endemic population. Int.J.Lepr.Other Mycobact.Dis. 68:40-48.
- Mark, S. 2002. Alexander the Great, seafaring, and the spread of leprosy. J.Hist Med.Allied Sci. 57:285-311.
- 169. Marland, Z., T. Beddoe, L. Zaker-Tabrizi, I. S. Lucet, R. Brammananth, J. C. Whisstock, M. C. Wilce, R. L. Coppel, P. K. Crellin, and J. Rossjohn. 2006. Hijacking of a substrate-binding protein scaffold for use in mycobacterial cell wall biosynthesis. J.Mol.Biol. 359:983-997.

- Marques, M. A., S. Chitale, P. J. Brennan, and M. C. Pessolani. 1998. Mapping and identification of the major cell wall-associated components of *Mycobacterium leprae*. Infect.Immun. 66:2625-2631.
- 171. Marques, M. A., B. J. Espinosa, E. K. Xavier da Silveira, M. C. Pessolani, A. Chapeaurouge, J. Perales, K. M. Dobos, J. T. Belisle, J. S. Spencer, and P. J. Brennan. 2004. Continued proteomic analysis of *Mycobacterium leprae* subcellular fractions. Proteomics. 4:2942-2953.
- 172. Marques, M. A., A. G. Neves-Ferreira, E. K. da Silveira, R. H. Valente, A. Chapeaurouge, J. Perales, B. R. da Silva, K. M. Dobos, J. S. Spencer, P. J. Brennan, and M. C. Pessolani. 2008. Deciphering the proteomic profile of *Mycobacterium leprae* cell envelope. Proteomics. 8:2477-2491.
- 173. **Marrakchi, H., F. Bardou, M. A. Laneelle, and M. Daffe**. 2008. A Comprehensive Overview of Mycolic Acid Structure and Biosynthesis, p. 41-62. *In*: M. Daffe and J. M. Reyrat (eds.), The Mycobacterial Cell Envelope. ASM Press, Washington D.C.
- 174. Matsunaga, I., A. Bhatt, D. C. Young, T. Y. Cheng, S. J. Eyles, G. S. Besra, V. Briken, S. A. Porcelli, C. E. Costello, W. R. Jacobs, Jr., and D. B. Moody. 2004. *Mycobacterium tuberculosis* pks12 produces a novel polyketide presented by CD1c to T cells. J.Exp.Med **200**:1559-1569.
- 175. Mattos, K. A., H. D'Avila, L. S. Rodrigues, V. G. Oliveira, E. N. Sarno, G. C. Atella, G. M. Pereira, P. T. Bozza, and M. C. Pessolani. 2010. Lipid droplet formation in leprosy: Toll-like receptor-regulated organelles involved in eicosanoid formation and *Mycobacterium leprae* pathogenesis. J.Leukoc.Biol. 87:371-384.
- 176. Mawuenyega, K. G., C. V. Forst, K. M. Dobos, J. T. Belisle, J. Chen, E. M. Bradbury, A. R. Bradbury, and X. Chen. 2005. *Mycobacterium tuberculosis* functional network analysis by global subcellular protein profiling. Mol.Biol.Cell **16**:396-404.
- 177. **McDonough, J. A., K. E. Hacker, A. R. Flores, M. S. Pavelka, Jr., and M. Braunstein**. 2005. The twin-arginine translocation pathway of *Mycobacterium smegmatis* is functional and required for the export of mycobacterial beta-lactamases. J.Bacteriol. **187**:7667-7679.
- 178. Mehra, V., P. J. Brennan, E. Rada, J. Convit, and B. R. Bloom. 1984. Lymphocyte suppression in leprosy induced by unique *M. leprae* glycolipid. Nature **308**:194-196.
- 179. **Menozzi, F. D., R. Bischoff, E. Fort, M. J. Brennan, and C. Locht**. 1998. Molecular characterization of the mycobacterial heparin-binding hemagglutinin, a mycobacterial adhesin. Proc.Natl.Acad.Sci U.S.A **95**:12625-12630.
- Menozzi, F. D., J. H. Rouse, M. Alavi, M. Laude-Sharp, J. Muller, R. Bischoff, M. J. Brennan, and C. Locht. 1996. Identification of a heparin-binding hemagglutinin present in mycobacteria. J.Exp.Med 184:993-1001.
- 181. Michell, S. L., A. O. Whelan, P. R. Wheeler, M. Panico, R. L. Easton, A. T. Etienne, S. M. Haslam, A. Dell, H. R. Morris, A. J. Reason, J. L. Herrmann, D. B. Young, and R. G. Hewinson. 2003. The MPB83 antigen from *Mycobacterium bovis* contains O-linked mannose and (1-->3)-mannobiose moieties. J.Biol.Chem. 278:16423-16432.
- 182. Mira, M. T., A. Alcais, V. T. Nguyen, M. O. Moraes, F. C. Di, H. T. Vu, C. P. Mai, T. H. Nguyen, N. B. Nguyen, X. K. Pham, E. N. Sarno, A. Alter, A. Montpetit, M. E. Moraes, J. R. Moraes, C. Dore, C. J. Gallant, P. Lepage, A. Verner, d. van, V, T. J. Hudson, L. Abel, and E. Schurr. 2004. Susceptibility to leprosy is associated with PARK2 and PACRG. Nature 427:636-640.
- Misch, E. A., W. R. Berrington, J. C. Vary, Jr., and T. R. Hawn. 2010. Leprosy and the human genome. Microbiol.Mol.Biol.Rev. 74:589-620.
- Mitsuda, K. 1919. On the value of a skin reaction to a suspension of leprosy nodules. Jpn.J.Dermatol Urol 19:697-708.
- 185. Modlin, R. L. 1994. Th1-Th2 paradigm: insights from leprosy. J.Invest Dermatol. 102:828-832.

- 186. Modlin, R. L., J. Melancon-Kaplan, S. M. Young, C. Pirmez, H. Kino, J. Convit, T. H. Rea, and B. R. Bloom. 1988. Learning from lesions: patterns of tissue inflammation in leprosy. Proc.Natl.Acad.Sci U.S.A 85:1213-1217.
- Moet, F. J., R. P. Schuring, D. Pahan, L. Oskam, and J. H. Richardus. 2008. The prevalence of previously undiagnosed leprosy in the general population of northwest bangladesh. PLoS.Negl.Trop.Dis. 2:1-4.
- 188. Molle, V., C. Girard-Blanc, L. Kremer, P. Doublet, A. J. Cozzone, and J. F. Prost. 2003. Protein PknE, a novel transmembrane eukaryotic-like serine/threonine kinase from *Mycobacterium tuberculosis*. Biochem.Biophys.Res.Commun. **308**:820-825.
- 189. Molle, V., L. Kremer, C. Girard-Blanc, G. S. Besra, A. J. Cozzone, and J. F. Prost. 2003. An FHA phosphoprotein recognition domain mediates protein EmbR phosphorylation by PknH, a Ser/Thr protein kinase from *Mycobacterium tuberculosis*. Biochemistry 42:15300-15309.
- 190. **Molle, V., D. Soulat, J. M. Jault, C. Grangeasse, A. J. Cozzone, and J. F. Prost**. 2004. Two FHA domains on an ABC transporter, Rv1747, mediate its phosphorylation by PknF, a Ser/Thr protein kinase from *Mycobacterium tuberculosis*. FEMS Microbiol.Lett. **234**:215-223.
- 191. Monot, M., N. Honore, T. Garnier, R. Araoz, J. Y. Coppee, C. Lacroix, S. Sow, J. S. Spencer, R. W. Truman, D. L. Williams, R. Gelber, M. Virmond, B. Flageul, S. N. Cho, B. Ji, A. Paniz-Mondolfi, J. Convit, S. Young, P. E. Fine, V. Rasolofo, P. J. Brennan, and S. T. Cole. 2005. On the origin of leprosy. Science 308:1040-1042.
- 192. Monot, M., N. Honore, T. Garnier, N. Zidane, D. Sherafi, A. Paniz-Mondolfi, M. Matsuoka, G. M. Taylor, H. D. Donoghue, A. Bouwman, S. Mays, C. Watson, D. Lockwood, A. Khamesipour, Y. Dowlati, S. Jianping, T. H. Rea, L. Vera-Cabrera, M. M. Stefani, S. Banu, M. MacDonald, B. R. Sapkota, J. S. Spencer, J. Thomas, K. Harshman, P. Singh, P. Busso, A. Gattiker, J. Rougemont, P. J. Brennan, and S. T. Cole. 2009. Comparative genomic and phylogeographic analysis of *Mycobacterium leprae*. Nat.Genet. **41**:1282-1289.
- 193. Montoya, D., D. Cruz, R. M. Teles, D. J. Lee, M. T. Ochoa, S. R. Krutzik, R. Chun, M. Schenk, X. Zhang, B. G. Ferguson, A. E. Burdick, E. N. Sarno, T. H. Rea, M. Hewison, J. S. Adams, G. Cheng, and R. L. Modlin. 2009. Divergence of macrophage phagocytic and antimicrobial programs in leprosy. Cell Host.Microbe 6:343-353.
- 194. Montoya, D. and R. L. Modlin. 2010. Learning from leprosy: insight into the human innate immune response. Adv.Immunol. 105:1-24.
- 195. Moody, D. B. 2006. TLR gateways to CD1 function. Nat.Immunol. 7:811-817.
- 196. Moody, D. B., B. B. Reinhold, M. R. Guy, E. M. Beckman, D. E. Frederique, S. T. Furlong, S. Ye, V. N. Reinhold, P. A. Sieling, R. L. Modlin, G. S. Besra, and S. A. Porcelli. 1997. Structural requirements for glycolipid antigen recognition by CD1b-restricted T cells. Science 278:283-286.
- 197. Moody, D. B., T. Ulrichs, W. Muhlecker, D. C. Young, S. S. Gurcha, E. Grant, J. P. Rosat, M. B. Brenner, C. E. Costello, G. S. Besra, and S. A. Porcelli. 2000. CD1c-mediated T-cell recognition of isoprenoid glycolipids in *Mycobacterium tuberculosis* infection. Nature 404:884-888.
- 198. Moody, D. B., D. C. Young, T. Y. Cheng, J. P. Rosat, C. Roura-Mir, P. B. O'Connor, D. M. Zajonc, A. Walz, M. J. Miller, S. B. Levery, I. A. Wilson, C. E. Costello, and M. B. Brenner. 2004. T cell activation by lipopeptide antigens. Science 303:527-531.
- Moody, D. B., D. M. Zajonc, and I. A. Wilson. 2005. Anatomy of CD1-lipid antigen complexes. Nat.Rev.Immunol. 5:387-399.
- 200. Mori, H. and K. Ito. 2001. The Sec protein-translocation pathway. Trends Microbiol. 9:494-500.
- 201. Mori, T., M. Sakatani, F. Yamagishi, T. Takashima, Y. Kawabe, K. Nagao, E. Shigeto, N. Harada, S. Mitarai, M. Okada, K. Suzuki, Y. Inoue, K. Tsuyuguchi, Y. Sasaki, G. H. Mazurek, and I. Tsuyuguchi. 2004. Specific detection of *M. tuberculosis* infection: an interferon-gamma-based assay using new antigens. Am.J.Respir.Crit Care Med 170:59-64.

- 202. Mouat, F. J. 1854. Notes on native remdeis No1: the Chaulmoogra. Ind Ann Med Sci 1:646-652.
- 203. Narita, S. and H. Tokuda. 2006. An ABC transporter mediating the membrane detachment of bacterial lipoproteins depending on their sorting signals. FEBS Lett. **580**:1164-1170.
- Natale, P., T. Bruser, and A. J. Driessen. 2008. Sec- and Tat-mediated protein secretion across the bacterial cytoplasmic membrane--distinct translocases and mechanisms. Biochim.Biophys.Acta 1778:1735-1756.
- Neufert, C., R. K. Pai, E. H. Noss, M. Berger, W. H. Boom, and C. V. Harding. 2001. *Mycobacterium tuberculosis* 19-kDa lipoprotein promotes neutrophil activation. J.Immunol. 167:1542-1549.
- Nguyen, H. T., K. A. Wolff, R. H. Cartabuke, S. Ogwang, and L. Nguyen. 2010. A lipoprotein modulates activity of the MtrAB two-component system to provide intrinsic multidrug resistance, cytokinetic control and cell wall homeostasis in Mycobacterium. Mol.Microbiol. 76:348-364.
- 207. Niederweis, M., O. Danilchanka, J. Huff, C. Hoffmann, and H. Engelhardt. 2010. Mycobacterial outer membranes: in search of proteins. Trends Microbiol. **18**:109-116.
- 208. Nigou, J., M. Gilleron, and G. Puzo. 2003. Lipoarabinomannans: from structure to biosynthesis. Biochimie 85:153-166.
- 209. Noordeen, S. K. 1991. A look at world leprosy. Lepr.Rev. 62:72-86.
- Noss, E. H., R. K. Pai, T. J. Sellati, J. D. Radolf, J. Belisle, D. T. Golenbock, W. H. Boom, and C. V. Harding. 2001. Toll-like receptor 2-dependent inhibition of macrophage class II MHC expression and antigen processing by 19-kDa lipoprotein of *Mycobacterium tuberculosis*. J.Immunol. 167:910-918.
- 211. Ochoa, M. T., S. Stenger, P. A. Sieling, S. Thoma-Uszynski, S. Sabet, S. Cho, A. M. Krensky, M. Rollinghoff, S. E. Nunes, A. E. Burdick, T. H. Rea, and R. L. Modlin. 2001. T-cell release of granulysin contributes to host defense in leprosy. Nat.Med 7:174-179.
- 212. Ohyama, H., F. Nishimura, M. Meguro, S. Takashiba, Y. Murayama, and S. Matsushita. 2002. Counter-antigen presentation: fibroblasts produce cytokines by signalling through HLA class II molecules without inducing T-cell proliferation. Cytokine 17:175-181.
- Okuda, S. and H. Tokuda. 2009. Model of mouth-to-mouth transfer of bacterial lipoproteins through inner membrane LolC, periplasmic LolA, and outer membrane LolB. Proc.Natl.Acad.Sci U.S.A 106:5877-5882.
- 214. Oliveira, R. B., M. T. Ochoa, P. A. Sieling, T. H. Rea, A. Rambukkana, E. N. Sarno, and R. L. Modlin. 2003. Expression of Toll-like receptor 2 on human Schwann cells: a mechanism of nerve damage in leprosy. Infect.Immun. 71:1427-1433.
- Oskam, L., E. Slim, and S. Buhrer-Sekula. 2003. Serology: recent developments, strengths, limitations and prospects: a state of the art overview. Lepr.Rev. 74:196-205.
- Ottenhoff, T. H., S. Neuteboom, D. G. Elfrink, and R. R. de Vries. 1986. Molecular localization and polymorphism of HLA class II restriction determinants defined by *Mycobacterium leprae*-reactive helper T cell clones from leprosy patients. J.Exp.Med. 164:1923-1939.
- 217. **P Murphy, P. T. M. W.** 2007. Janeway's Immunobiology 7th ed.
- 218. Pai, M., L. W. Riley, and J. M. Colford, Jr. 2004. Interferon-gamma assays in the immunodiagnosis of tuberculosis: a systematic review. Lancet Infect.Dis. 4:761-776.
- 219. Pai, R. K., M. E. Pennini, A. A. Tobian, D. H. Canaday, W. H. Boom, and C. V. Harding. 2004. Prolonged toll-like receptor signaling by *Mycobacterium tuberculosis* and its 19-kilodalton lipoprotein inhibits gamma interferon-induced regulation of selected genes in macrophages. Infect.Immun. **72**:6603-6614.
- 220. Palsson-McDermott, E. M. and L. A. O'Neill. 2007. Building an immune system from nine domains. Biochem.Soc.Trans. 35:1437-1444..

- Pearce, M. J., P. Arora, R. A. Festa, S. M. Butler-Wu, R. S. Gokhale, and K. H. Darwin. 2006. Identification of substrates of the *Mycobacterium tuberculosis* proteasome. EMBO J. 25:5423-5432.
- 222. Pecora, N. D., A. J. Gehring, D. H. Canaday, W. H. Boom, and C. V. Harding. 2006. *Mycobacterium tuberculosis* LprA is a lipoprotein agonist of TLR2 that regulates innate immunity and APC function. J.Immunol. 177:422-429.
- 223. Pennini, M. E., R. K. Pai, D. C. Schultz, W. H. Boom, and C. V. Harding. 2006. *Mycobacterium tuberculosis* 19-kDa lipoprotein inhibits IFN-gamma-induced chromatin remodeling of MHC2TA by TLR2 and MAPK signaling. J.Immunol. 176:4323-4330.
- 224. Pessolani, M. C., D. R. Smith, B. Rivoire, J. McCormick, S. A. Hefta, S. T. Cole, and P. J. Brennan. 1994. Purification, characterization, gene sequence, and significance of a bacterioferritin from *Mycobacterium leprae*. J.Exp.Med. 180:319-327.
- 225. Pethe, K., S. Alonso, F. Biet, G. Delogu, M. J. Brennan, C. Locht, and F. D. Menozzi. 2001. The heparin-binding haemagglutinin of *M. tuberculosis* is required for extrapulmonary dissemination. Nature **412**:190-194.
- 226. Pethe, K., M. Aumercier, E. Fort, C. Gatot, C. Locht, and F. D. Menozzi. 2000. Characterization of the heparin-binding site of the mycobacterial heparin-binding hemagglutinin adhesin. J.Biol.Chem. 275:14273-14280.
- 227. Pethe, K., P. Bifani, H. Drobecq, C. Sergheraert, A. S. Debrie, C. Locht, and F. D. Menozzi. 2002. Mycobacterial heparin-binding hemagglutinin and laminin-binding protein share antigenic methyllysines that confer resistance to proteolysis. Proc.Natl.Acad.Sci U.S.A 99:10759-10764.
- 228. Pethe, K., V. Puech, M. Daffe, C. Josenhans, H. Drobecq, C. Locht, and F. D. Menozzi. 2001. *Mycobacterium smegmatis* laminin-binding glycoprotein shares epitopes with *Mycobacterium tuberculosis* heparin-binding haemagglutinin. Mol.Microbiol. **39**:89-99.
- 229. Piddington, D. L., F. C. Fang, T. Laessig, A. M. Cooper, I. M. Orme, and N. A. Buchmeier. 2001. Cu,Zn superoxide dismutase of *Mycobacterium tuberculosis* contributes to survival in activated macrophages that are generating an oxidative burst. Infect.Immun. 69:4980-4987.
- Pugsley, A. P. 1993. The complete general secretory pathway in gram-negative bacteria. Microbiol.Rev. 57:50-108.
- 231. Quesniaux, V., C. Fremond, M. Jacobs, S. Parida, D. Nicolle, V. Yeremeev, F. Bihl, F. Erard, T. Botha, M. Drennan, M. N. Soler, B. M. Le, B. Schnyder, and B. Ryffel. 2004. Toll-like receptor pathways in the immune responses to mycobacteria. Microbes.Infect. 6:946-959.
- 232. Ragas, A., L. Roussel, G. Puzo, and M. Riviere. 2007. The *Mycobacterium tuberculosis* cellsurface glycoprotein apa as a potential adhesin to colonize target cells via the innate immune system pulmonary C-type lectin surfactant protein A. J.Biol.Chem. 282:5133-5142..
- 233. **Rambukkana, A.** 2010. Usage of signaling in neurodegeneration and regeneration of peripheral nerves by leprosy bacteria. Prog.Neurobiol. **91**:102-107.
- 234. Rambukkana, A., G. Zanazzi, N. Tapinos, and J. L. Salzer. 2002. Contact-dependent demyelination by *Mycobacterium leprae* in the absence of immune cells. Science **296**:927-931.
- 235. Rao, V., N. Dhar, H. Shakila, R. Singh, A. Khera, R. Jain, M. Naseema, C. N. Paramasivan, P. R. Narayanan, V. D. Ramanathan, and A. K. Tyagi. 2005. Increased expression of *Mycobacterium tuberculosis* 19 kDa lipoprotein obliterates the protective efficacy of BCG by polarizing host immune responses to the Th2 subtype. Scand.J.Immunol. **61**:410-417.
- Rao, V., N. Dhar, and A. K. Tyagi. 2003. Modulation of host immune responses by overexpression of immunodominant antigens of *Mycobacterium tuberculosis* in bacille Calmette-Guerin. Scand.J.Immunol. 58:449-461.
- 237. Rawcliffe, C. 2006. Leprosy in medieval england. Boydell Press.
- 238. Raynaud, C., C. Guilhot, J. Rauzier, Y. Bordat, V. Pelicic, R. Manganelli, I. Smith, B. Gicquel, and M. Jackson. 2002. Phospholipases C are involved in the virulence of *Mycobacterium tuberculosis*. Mol.Microbiol. 45:203-217.

- 239. Reece, S. T., G. Ireton, R. Mohamath, J. Guderian, W. Goto, R. Gelber, N. Groathouse, J. Spencer, P. Brennan, and S. G. Reed. 2006. ML0405 and ML2331 are antigens of *Mycobacterium leprae* with potential for diagnosis of leprosy. Clin.Vaccine Immunol. 13:333-340.
- Rezwan, M., T. Grau, A. Tschumi, and P. Sander. 2007. Lipoprotein synthesis in mycobacteria. Microbiology 153:652-658.
- Ridley, D. S. and W. H. Jopling. 1966. Classification of leprosy according to immunity. A fivegroup system. Int.J.Lepr.Other Mycobact.Dis. 34:255-273.
- Rigel, N. W. and M. Braunstein. 2008. A new twist on an old pathway-accessory Sec systems. Mol.Microbiol. 69:291-302.
- Rivoire, B., M. C. Pessolani, C. M. Bozic, S. W. Hunter, S. A. Hefta, V. Mehra, and P. J. Brennan. 1994. Chemical definition, cloning, and expression of the major protein of the leprosy bacillus. Infect.Immun. 62:2417-2425.
- 244. Robbins, G., V. M. Tripathy, V. N. Misra, R. K. Mohanty, V. S. Shinde, K. M. Gray, and M. D. Schug. 2009. Ancient skeletal evidence for leprosy in India (2000 B.C.). PLoS.One. 4:1-8.
- 245. Romain, F., C. Horn, P. Pescher, A. Namane, M. Riviere, G. Puzo, O. Barzu, and G. Marchal. 1999. Deglycosylation of the 45/47-kilodalton antigen complex of *Mycobacterium tuberculosis* decreases its capacity to elicit *in vivo* or *in vitro* cellular immune responses. Infect.Immun. 67:5567-5572.
- 246. **Russell, D. G.** 2001. *Mycobacterium tuberculosis*: here today, and here tomorrow. Nat.Rev.Mol.Cell Biol. **2**:569-577.
- 247. Saini, D. K., V. Malhotra, D. Dey, N. Pant, T. K. Das, and J. S. Tyagi. 2004. DevR-DevS is a bona fide two-component system of *Mycobacterium tuberculosis* that is hypoxia-responsive in the absence of the DNA-binding domain of DevR. Microbiology **150**:865-875.
- 248. Sander, P., M. Rezwan, B. Walker, S. K. Rampini, R. M. Kroppenstedt, S. Ehlers, C. Keller, J. R. Keeble, M. Hagemeier, M. J. Colston, B. Springer, and E. C. Bottger. 2004. Lipoprotein processing is required for virulence of *Mycobacterium tuberculosis*. Mol.Microbiol. 52:1543-1552.
- Sankaran, K., S. D. Gupta, and H. C. Wu. 1995. Modification of bacterial lipoproteins. Methods Enzymol. 250:683-697.
- 250. Sankaran, K. and H. C. Wu. 1994. Lipid modification of bacterial prolipoprotein. Transfer of diacylglyceryl moiety from phosphatidylglycerol. J.Biol.Chem. 269:19701-19706.
- 251. Santos, A. R., A. B. De Miranda, E. N. Sarno, P. N. Suffys, and W. M. Degrave. 1993. Use of PCR-mediated amplification of *Mycobacterium leprae* DNA in different types of clinical samples for the diagnosis of leprosy. J.Med Microbiol. **39**:298-304.
- 252. Sartain, M. J. and J. T. Belisle. 2009. N-Terminal clustering of the O-glycosylation sites in the *Mycobacterium tuberculosis* lipoprotein SodC. Glycobiology **19**:38-51.
- 253. Sartain, M. J., R. A. Slayden, K. K. Singh, S. Laal, and J. T. Belisle. 2006. Disease state differentiation and identification of tuberculosis biomarkers via native antigen array profiling. Mol.Cell Proteomics. 5:2102-2113.
- 254. Sassetti, C. M. and E. J. Rubin. 2003. Genetic requirements for mycobacterial survival during infection. Proc.Natl.Acad.Sci U.S.A 100:12989-12994.
- Schenk, M., J. T. Belisle, and R. L. Modlin. 2009. TLR2 looks at lipoproteins. Immunity. 31:847-849.
- Schlesinger, L. S. 1993. Macrophage phagocytosis of virulent but not attenuated strains of *Mycobacterium tuberculosis* is mediated by mannose receptors in addition to complement receptors. J.Immunol. 150:2920-2930.
- 257. Schlesinger, L. S. and M. A. Horwitz. 1990. Phagocytosis of leprosy bacilli is mediated by complement receptors CR1 and CR3 on human monocytes and complement component C3 in serum. J.Clin.Invest **85**:1304-1314.

- 258. Schlesinger, L. S. and M. A. Horwitz. 1991. Phagocytosis of *Mycobacterium leprae* by human monocyte-derived macrophages is mediated by complement receptors CR1 (CD35), CR3 (CD11b/CD18), and CR4 (CD11c/CD18) and IFN-gamma activation inhibits complement receptor function and phagocytosis of this bacterium. J.Immunol. 147:1983-1994.
- Schlesinger, L. S., S. R. Hull, and T. M. Kaufman. 1994. Binding of the terminal mannosyl units of lipoarabinomannan from a virulent strain of *Mycobacterium tuberculosis* to human macrophages. J.Immunol. 152:4070-4079.
- Schlesinger, L. S., T. M. Kaufman, S. Iyer, S. R. Hull, and L. K. Marchiando. 1996. Differences in mannose receptor-mediated uptake of lipoarabinomannan from virulent and attenuated strains of *Mycobacterium tuberculosis* by human macrophages. J.Immunol. 157:4568-4575.
- Schurr, E., A. Alcais, M. Singh, N. Mehra, and L. Abel. 2007. Mycobacterial infections: PARK2 and PACRG associations in leprosy. Tissue Antigens 69 Suppl 1:231-233.
- Scollard, D. M., L. B. Adams, T. P. Gillis, J. L. Krahenbuhl, R. W. Truman, and D. L. Williams. 2006. The continuing challenges of leprosy. Clin.Microbiol.Rev. 19:338-381.
- 263. Sekar, B. 2007. Recent advances in immunodiagnosis of leprosy. Indian J.Lepr. 79:85-106.
- 264. Shepard, C. C., R. J. Rees, L. Levy, S. R. Pattyn, J. Baohong, and E. C. Dela Cruz. 1986. Susceptibility of strains of *Mycobacterium leprae* isolated prior to 1977 from patients with previously untreated lepromatous leprosy. Int.J.Lepr.Other Mycobact.Dis. 54:11-15.
- 265. **Shetty, V. P.** 2010. Challenges facing the control of leprosy in the Indian context. Ann.Acad.Med.Singapore **39**:1-3.
- 266. Shin, A. R., K. S. Lee, J. S. Lee, S. Y. Kim, C. H. Song, S. B. Jung, C. S. Yang, E. K. Jo, J. K. Park, T. H. Paik, and H. J. Kim. 2006. *Mycobacterium tuberculosis* HBHA protein reacts strongly with the serum immunoglobulin M of tuberculosis patients. Clin.Vaccine Immunol. 13:869-875.
- 267. Shin, D. M., J. M. Yuk, H. M. Lee, S. H. Lee, J. W. Son, C. V. Harding, J. M. Kim, R. L. Modlin, and E. K. Jo. 2010. Mycobacterial lipoprotein activates autophagy via TLR2/1/CD14 and a functional vitamin D receptor signalling. Cell Microbiol. 12:1648-1665.
- 268. Sieling, P. A., D. Chatterjee, S. A. Porcelli, T. I. Prigozy, R. J. Mazzaccaro, T. Soriano, B. R. Bloom, M. B. Brenner, M. Kronenberg, P. J. Brennan, and . 1995. CD1-restricted T cell recognition of microbial lipoglycan antigens. Science 269:227-230.
- Sieling, P. A., D. Chatterjee, S. A. Porcelli, T. I. Prigozy, T. Soriano, M. B. Brenner, M. Kronenberg, P. J. Brennan, and R. L. Modlin. 1995. CD1-restricted T cell recognition of microbial lipoglycans. Science 269:227-230.
- 270. Sieling, P. A., P. J. Hill, K. M. Dobos, K. Brookman, A. M. Kuhlman, M. Fabri, S. R. Krutzik, T. H. Rea, D. G. Heaslip, J. T. Belisle, and R. L. Modlin. 2008. Conserved mycobacterial lipoglycoproteins activate TLR2 but also require glycosylation for MHC class II-restricted T cell activation. J.Immunol. 180:5833-5842.
- 271. Sieling, P. A., D. Jullien, M. Dahlem, T. F. Tedder, T. H. Rea, R. L. Modlin, and S. A. Porcelli. 1999. CD1 expression by dendritic cells in human leprosy lesions: correlation with effective host immunity. J.Immunol. 162:1851-1858.
- 272. Sieling, P. A. and R. L. Modlin. 1994. Cytokine patterns at the site of mycobacterial infection. Immunobiology 191:378-387.
- 273. Sieling, P. A., J. B. Torrelles, S. Stenger, W. Chung, A. E. Burdick, T. H. Rea, P. J. Brennan, J. T. Belisle, S. A. Porcelli, and R. L. Modlin. 2005. The human CD1-restricted T cell repertoire is limited to cross-reactive antigens: implications for host responses against immunologically related pathogens. J.Immunol. 174:2637-2644.
- 274. Sieling, P. A., X. H. Wang, M. K. Gately, J. L. Oliveros, T. McHugh, P. F. Barnes, S. F. Wolf, L. Golkar, M. Yamamura, Y. Yogi, and . 1994. IL-12 regulates T helper type 1 cytokine responses in human infectious disease. J.Immunol. 153:3639-3647.

- Silbaq, F. S., S. N. Cho, S. T. Cole, and P. J. Brennan. 1998. Characterization of a 34-kilodalton protein of *Mycobacterium leprae* that is isologous to the immunodominant 34-kilodalton antigen of *Mycobacterium paratuberculosis*. Infect.Immun. 66:5576-5579.
- 276. Soares de, L. C., L. Zulianello, M. A. Marques, H. Kim, M. I. Portugal, S. L. Antunes, F. D. Menozzi, T. H. Ottenhoff, P. J. Brennan, and M. C. Pessolani. 2005. Mapping the lamininbinding and adhesive domain of the cell surface-associated Hlp/LBP protein from *Mycobacterium leprae*. Microbes.Infect. 7:1097-1109.
- 277. Spencer, J. S., H. M. Dockrell, H. J. Kim, M. A. Marques, D. L. Williams, M. V. Martins, M. L. Martins, M. C. Lima, E. N. Sarno, G. M. Pereira, H. Matos, L. S. Fonseca, E. P. Sampaio, T. H. Ottenhoff, A. Geluk, S. N. Cho, N. G. Stoker, S. T. Cole, P. J. Brennan, and M. C. Pessolani. 2005. Identification of specific proteins and peptides in *Mycobacterium leprae* suitable for the selective diagnosis of leprosy. J.Immunol. 175:7930-7938.
- 278. Spencer, J. S., H. J. Kim, A. M. Marques, M. Gonzalez-Juarerro, M. C. Lima, V. D. Vissa, R. W. Truman, M. L. Gennaro, S. N. Cho, S. T. Cole, and P. J. Brennan. 2004. Comparative analysis of B- and T-cell epitopes of *Mycobacterium leprae* and *Mycobacterium tuberculosis* culture filtrate protein 10. Infect.Immun. 72:3161-3170.
- 279. Spencer, J. S., M. A. Marques, M. C. Lima, A. P. Junqueira-Kipnis, B. C. Gregory, R. W. Truman, and P. J. Brennan. 2002. Antigenic specificity of the *Mycobacterium leprae* homologue of ESAT-6. Infect.Immun. **70**:1010-1013.
- 280. Stenger, S., D. A. Hanson, R. Teitelbaum, P. Dewan, K. R. Niazi, C. J. Froelich, T. Ganz, S. Thoma-Uszynski, A. Melian, C. Bogdan, S. A. Porcelli, B. R. Bloom, A. M. Krensky, and R. L. Modlin. 1998. An antimicrobial activity of cytolytic T cells mediated by granulysin. Science 282:121-125.
- Stenger, S., R. J. Mazzaccaro, K. Uyemura, S. Cho, P. F. Barnes, J. P. Rosat, A. Sette, M. B. Brenner, S. A. Porcelli, B. R. Bloom, and R. L. Modlin. 1997. Differential effects of cytolytic T cell subsets on intracellular infection. Science 276:1684-1687.
- 282. Stock, A. M., V. L. Robinson, and P. N. Goudreau. 2000. Two-component signal transduction. Annu.Rev.Biochem. 69:183-215.
- 283. Striebel, F., F. Imkamp, M. Sutter, M. Steiner, A. Mamedov, and E. Weber-Ban. 2009. Bacterial ubiquitin-like modifier Pup is deamidated and conjugated to substrates by distinct but homologous enzymes. Nat.Struct.Mol.Biol. 16:647-651.
- 284. Sulzenbacher, G., S. Canaan, Y. Bordat, O. Neyrolles, G. Stadthagen, V. Roig-Zamboni, J. Rauzier, D. Maurin, F. Laval, M. Daffe, C. Cambillau, B. Gicquel, Y. Bourne, and M. Jackson. 2006. LppX is a lipoprotein required for the translocation of phthiocerol dimycocerosates to the surface of *Mycobacterium tuberculosis*. EMBO J. 25:1436-1444.
- 285. Sutcliffe, I. C. and D. J. Harrington. 2002. Pattern searches for the identification of putative lipoprotein genes in Gram-positive bacterial genomes. Microbiology 148:2065-2077.
- Sutcliffe, I. C. and D. J. Harrington. 2004. Lipoproteins of *Mycobacterium tuberculosis*: an abundant and functionally diverse class of cell envelope components. FEMS Microbiol.Rev. 28:645-659.
- Sutcliffe, I. C. and R. R. Russell. 1995. Lipoproteins of gram-positive bacteria. J.Bacteriol. 177:1123-1128.
- Takayama, K., C. Wang, and G. S. Besra. 2005. Pathway to synthesis and processing of mycolic acids in *Mycobacterium tuberculosis*. Clin.Microbiol.Rev. 18:81-101.
- Takeuchi, O. and S. Akira. 2010. Pattern recognition receptors and inflammation. Cell 140:805-820.
- 290. Temmerman, S., K. Pethe, M. Parra, S. Alonso, C. Rouanet, T. Pickett, A. Drowart, A. S. Debrie, G. Delogu, F. D. Menozzi, C. Sergheraert, M. J. Brennan, F. Mascart, and C. Locht. 2004. Methylation-dependent T cell immunity to *Mycobacterium tuberculosis* heparin-binding hemagglutinin. Nat.Med 10:935-941.

- 291. **Tokuda, H. and S. Matsuyama**. 2004. Sorting of lipoproteins to the outer membrane in *E. coli*. Biochim.Biophys.Acta **1693**:5-13.
- 292. Torrelles, J. B., K. H. Khoo, P. A. Sieling, R. L. Modlin, N. Zhang, A. M. Marques, A. Treumann, C. D. Rithner, P. J. Brennan, and D. Chatterjee. 2004. Truncated structural variants of lipoarabinomannan in *Mycobacterium leprae* and an ethambutol-resistant strain of *Mycobacterium tuberculosis*. J.Biol.Chem. 279:41227-41239.
- 293. **Triccas, J. A., P. W. Roche, and W. J. Britton**. 1998. Specific serological diagnosis of leprosy with a recombinant *Mycobacterium leprae* protein purified from a rapidly growing mycobacterial host. J.Clin.Microbiol. **36**:2363-2365.
- 294. **Triccas, J. A., P. W. Roche, N. Winter, C. G. Feng, C. R. Butlin, and W. J. Britton**. 1996. A 35-kilodalton protein is a major target of the human immune response to *Mycobacterium leprae*. Infect.Immun. **64**:5171-5177.
- Tschumi, A., C. Nai, Y. Auchli, P. Hunziker, P. Gehrig, P. Keller, T. Grau, and P. Sander. 2009. Identification of apolipoprotein N-acyltransferase (Lnt) in mycobacteria. J.Biol.Chem. 284:27146-27156.
- 296. Tuberculolist. http://genolist.pasteur.fr/TubercuList/
- 297. Tufariello, J. M., K. Mi, J. Xu, Y. C. Manabe, A. K. Kesavan, J. Drumm, K. Tanaka, W. R. Jacobs, Jr., and J. Chan. 2006. Deletion of the *Mycobacterium tuberculosis* resuscitation-promoting factor Rv1009 gene results in delayed reactivation from chronic tuberculosis. Infect.Immun. 74:2985-2995.
- 298. Upreti, R. K., M. Kumar, and V. Shankar. 2003. Bacterial glycoproteins: functions, biosynthesis and applications. Proteomics. 3:363-379.
- 299. VanderVen, B. C., J. D. Harder, D. C. Crick, and J. T. Belisle. 2005. Export-mediated assembly of mycobacterial glycoproteins parallels eukaryotic pathways. Science **309**:941-943.
- 300. Veyron-Churlet, R., V. Molle, R. C. Taylor, A. K. Brown, G. S. Besra, I. Zanella-Cleon, K. Futterer, and L. Kremer. 2009. The *Mycobacterium tuberculosis* beta-ketoacyl-acyl carrier protein synthase III activity is inhibited by phosphorylation on a single threonine residue. J.Biol.Chem. 284:6414-6424.
- Vissa, V. D. and P. J. Brennan. 2001. The genome of *Mycobacterium leprae*: a minimal mycobacterial gene set. Genome Biol. 2:REVIEWS1023.
- 302. Vissa, V. D. and P. J. Brennan. 2002. Impact of the *Mycobacterium leprae* genomoe sequence on Leprosy research, p. 85-117. *In*: From; Genomics of GC-rich Gram Positive Bacteria. Caister Academic Press, Wymondham, UK.
- 303. Vogelsang, T. 1978. Gerhard Henrik Armauer Hansen 1841-1912. The discoverer of the leprosy bacillus. His life and his work. Int.J.Lepr.Other Mycobact.Dis. 46:257-332.
- 304. **von, H. G.** 1989. The structure of signal peptides from bacterial lipoproteins. Protein Eng **2**:531-534.
- 305. Wehenkel, A., M. Bellinzoni, M. Grana, R. Duran, A. Villarino, P. Fernandez, G. Andre-Leroux, P. England, H. Takiff, C. Cervenansky, S. T. Cole, and P. M. Alzari. 2008. Mycobacterial Ser/Thr protein kinases and phosphatases: physiological roles and therapeutic potential. Biochim.Biophys.Acta 1784:193-202.
- 306. Weir, R. E., P. J. Brennan, C. R. Butlin, and H. M. Dockrell. 1999. Use of a whole blood assay to evaluate *in vitro* T cell responses to new leprosy skin test antigens in leprosy patients and healthy subjects. Clin.Exp.Immunol. 116:263-269.
- 307. Weir, R. E., P. E. Fine, S. Floyd, S. Stenson, C. Stanley, K. Branson, W. J. Britton, K. Huygen, M. Singh, G. Black, and H. M. Dockrell. 2008. Comparison of IFN-gamma responses to mycobacterial antigens as markers of response to BCG vaccination. *Tuberculosis*.(Edinb.) 88:31-38.
- 308. WHO. WHO Techinical repost series, No 607. 7-48. 1977. 'World Health Organ Tech Rep Ser.
- 309. WHO. 2002. Global Leprosy Situation. Weekly Epid.Record 77:1-8.

- 310. WHO. 2010. Global leprosy situation. Weekly Epid.Record 85:337-348.
- 311. WHO. http://www.who.int/lep/diagnosis/en/index.html.
- 312. WHO. http://www.who.int/lep/mdt/regimens/en/index.html.
- 313. Wieland, C. W., S. Knapp, S. Florquin, A. F. de Vos, K. Takeda, S. Akira, D. T. Golenbock, A. Verbon, and T. van der Poll. 2004. Non-mannose-capped lipoarabinomannan induces lung inflammation via toll-like receptor 2. Am.J.Respir.Crit Care Med 170:1367-1374.
- 314. Wilkinson, R. J., K. A. Wilkinson, S. Jurcevic, A. Hills, S. Sinha, U. Sengupta, D. N. Lockwood, K. Katoch, D. Altman, and J. Ivanyi. 1999. Specificity and function of immunogenic peptides from the 35-kilodalton protein of *Mycobacterium leprae*. Infect.Immun. 67:1501-1504.
- 315. Winter, N., J. A. Triccas, B. Rivoire, M. C. Pessolani, K. Eiglmeier, E. M. Lim, S. W. Hunter, P. J. Brennan, and W. J. Britton. 1995. Characterization of the gene encoding the immunodominant 35 kDa protein of *Mycobacterium leprae*. Mol.Microbiol. 16:865-876.
- 316. Wong, S. H., S. Gochhait, D. Malhotra, F. H. Pettersson, Y. Y. Teo, C. C. Khor, A. Rautanen, S. J. Chapman, T. C. Mills, A. Srivastava, A. Rudko, M. B. Freidin, V. P. Puzyrev, S. Ali, S. Aggarwal, R. Chopra, B. S. Reddy, V. K. Garg, S. Roy, S. Meisner, S. K. Hazra, B. Saha, S. Floyd, B. J. Keating, C. Kim, B. P. Fairfax, J. C. Knight, P. C. Hill, R. A. Adegbola, H. Hakonarson, P. E. Fine, R. M. Pitchappan, R. N. Bamezai, A. V. Hill, and F. O. Vannberg. 2010. Leprosy and the adaptation of human toll-like receptor 1. PLoS.Pathog. 6:1-9.
- Wu, H. C. and J. J. Lin. 1976. *Escherichia coli* mutants altered in murein lipoprotein. J.Bacteriol. 126:147-156.
- 318. Yamagata, H., N. Taguchi, K. Daishima, and S. Mizushima. 1983. Genetic characterization of a gene for prolipoprotein signal peptidase in *Escherichia coli*. Mol.Gen.Genet. **192**:10-14.
- 319. Yamashita, Y., Y. Maeda, F. Takeshita, P. J. Brennan, and M. Makino. 2004. Role of the polypeptide region of a 33kDa mycobacterial lipoprotein for efficient IL-12 production. Cell Immunol. 229:13-20..
- 320. Yang, C. S., J. M. Yuk, and E. K. Jo. 2009. The role of nitric oxide in mycobacterial infections. Immune.Netw. 9:46-52.
- 321. Young, D. B. and T. R. Garbe. 1991. Lipoprotein antigens of *Mycobacterium tuberculosis*. Res.Microbiol. **142**:55-65.
- 322. Young, D. B., S. H. Kaufmann, P. W. Hermans, and J. E. Thole. 1992. Mycobacterial protein antigens: a compilation. Mol.Microbiol. 6:133-145.
- 323. Yuk, J. M., D. M. Shin, H. M. Lee, C. S. Yang, H. S. Jin, K. K. Kim, Z. W. Lee, S. H. Lee, J. M. Kim, and E. K. Jo. 2009. Vitamin D3 induces autophagy in human monocytes/macrophages via cathelicidin. Cell Host.Microbe 6:231-243.
- 324. Zahrt, T. C. and V. Deretic. 2000. An essential two-component signal transduction system in *Mycobacterium tuberculosis*. J.Bacteriol. **182**:3832-3838.
- Zahrt, T. C., C. Wozniak, D. Jones, and A. Trevett. 2003. Functional analysis of the *Mycobacterium tuberculosis* MprAB two-component signal transduction system. Infect.Immun. 71:6962-6970.
- 326. Zanetti, S., A. Bua, G. Delogu, C. Pusceddu, M. Mura, F. Saba, P. Pirina, C. Garzelli, C. Vertuccio, L. A. Sechi, and G. Fadda. 2005. Patients with pulmonary tuberculosis develop a strong humoral response against methylated heparin-binding hemagglutinin. Clin.Diagn.Lab Immunol. 12:1135-1138.
- 327. Zhu, Y. I. and M. J. Stiller. 2001. Dapsone and sulfones in dermatology: overview and update. J.Am.Acad.Dermatol. 45:420-434.
- 328. Zuber, B., M. Chami, C. Houssin, J. Dubochet, G. Griffiths, and M. Daffe. 2008. Direct visualization of the outer membrane of mycobacteria and corynebacteria in their native state. J.Bacteriol. **190**:5672-5680.

CHAPTER 2 GENE EXPRESSION PROFILE AND IMMUNOLOGICAL EVALUATION OF UNIQUE HYPOTHETICAL UNKNOWN PROTEINS OF *MYCOBACTERIUM LEPRAE* BY USING QUANTITATIVE REAL TIME-PCR

From a manuscript in preparation by **Hee Jin Kim**, Kalyani Prithiviraj, Nathan Groathouse, John S. Spencer and Patrick J. Brennan. Department of Microbiology, Immunology and Pathology, Colorado State University, Fort Collins, Colorado.

2.1 Abstract

Cellular mediated immunity (CMI) based *in vitro* interferon-γ released assay (IGRA) to *M. leprae* specific antigens is known as the most promising diagnostic test to detect those individuals in the early stages of *M. leprae* infection. However, diagnosis of leprosy is a major obstacle toward disease control, and has been compromised in the past by the lack of specific markers. Comparative bioinformatic analysis among mycobacterial genomes identified potential *M. leprae* unique proteins called "hypothetical unknowns" from the *M. leprae* genome. Due to the massive gene decay and the prevalence of pseudogenes, it is unclear whether any of these proteins are expressed or are immunologically relevant. Here, we performed cDNA based quantitative real time PCR to investigate the expression status of 136 putative open reading frames (ORFs) encoding hypothetical unknowns. Twenty six *M. leprae* specific antigen candidates showed significant levels of gene expression compared to that of ESAT-6 (ML0049) which is an important T cell antigen of low abundance in *M. leprae*.

Fifteen out of 26 selected antigen candidates were expressed in *E. coli* and purified. The seroreactivity to these recombinant proteins using the pooled sera of lepromatous leprosy patients and cavitary tuberculosis patients revealed that 10 out of 15 recombinant hypothetical unknowns elicited *M. leprae* specific immune responses. Therefore, 9 of the recombinant hypothetical unknowns may be good diagnostic reagents to improve both sensitivity and specificity in the detection of individuals with asymptomatic leprosy.

2.2 Introduction

The diagnosis of leprosy is solely based on clinical symptoms, requiring the presence of a neurologic deficit and skin lesions (4,39), but is only specific to patients with progressive disease. More than 70% of infected patients are acid-fast bacilli staining negative and do not present the analgesic skin lesions, especially paucibacillary/ tuberculoid leprosy (PB/TT) patients (4) (see section 1.3). Since the presence of skin lesions in these patients is variable, their clinical symptoms are not sufficient to specifically diagnose leprosy (4). These problems are accentuated in the diagnosis of individuals with subclinical *M. leprae* infection including leprosy house hold contacts (HHCs) who have close contact with leprosy patients and may also serve as important sources of ongoing leprosy transmission (4,9,29,39).

Several *M. leprae* antigens were identified and evaluated for their diagnostic potential by serological or CMI based tests (1-3,5,6,8-11,13-19,30,31,33,37) (see section 1.10.2). Until now, serological tests by a single *M. leprae* specific antigen, phenolic glycolipid-I (PGL-I), successfully detects circulating antibody in sera of multibacillary/lepromatous leprosy (MB/LL) patients. However, this test fails to detect PB/TT patients and HHCs who usually present low seroreactivity but strong CMI

response to mycobacterial antigens (11,30). Both *in vitro* IFN- γ release assays (IGRAs) and a simple delayed type hypersensitivity skin test have been developed to detect individuals in the early stages of leprosy using *M. leprae* protein based antigenic fractions such as *M. leprae* cell wall antigens without LAMs (MLCwA), *M. leprae* soluble antigen without LAMs (MLSA-LAM) and the major individual immunogenic proteins (3,3,13,26,29,38). However, a major obstacle of the application of the IGRA to the major *M. leprae* protein fractions or antigens is that most of these share a great deal of homology among other mycobacteria causing undesirable cross-reactivity from individuals who were vaccinated with BCG and/or exposed to *M tuberculosis* and/or non-tuberculosis mycobacteria.

Comparative genomic analysis between *M. leprae* and other mycobacteria has identified up to 142 hypothetical unknowns as *M. leprae* specific proteins (7,14,30) (described in section 1.10.2). During the last decade, either recombinant proteins or synthetic peptides originating from these ORFs, and which contain T cell epitopes restricted via major HLA (human leukocyte antigen)-DR alleles, have been studied as *M. leprae* specific antigens (1,2,13-16,30). CMI based IGRA to *M. leprae*-specific antigens can distinguish individuals exposed to/infected by *M. leprae* from endemic healthy controls in Brazilian population (13,14,16,30). However, the levels of IFN- γ secretion to these antigens, particularly their peptides, were found to be too low to distinguish some individuals exposed to *M. leprae* from healthy individuals in different geographical regions (15,30), suggesting that the low response to these peptide antigens could misdiagnose individuals with preclinical leprosy. Low sensitivity of current IGRA raises a question of whether any of the hypothetical unknowns are expressed, because

approximately 50% of genes encoding for functional proteins, which were found in other mycobacteria, are deleted or inactivated in the *M. leprae* genome (7,36).

The aim of this study was to identify *M. leprae*-specific proteins that could induce high levels of IFN-γ secretion by peripheral blood monocyte cells (PBMCs) from individuals with asymptomatic *M. leprae* infection/ exposure, as well as TT/PB patients. In order to achieve this goal, we performed cDNA based quantitative real time PCR (qRT-PCR) to investigate the expression status of 136 *M. leprae* specific ORFs (hypothetical unknowns) and selected twenty six of the promising antigen candidates which showed detectable gene expression levels for recombinant protein production. Subsequent serological analysis using sera of LL patients and cavitary tuberculosis (TB) patients evaluated the immunological potentials of all new recombinant antigen candidates.

2.3 Materials and methods

2.3.1 Isolation of M. leprae RNA

M. leprae Thai-53 (10.2 mg, 2×10^{10} cells) provided by the Leprosy Contract at Colorado State University (CSU) were washed with 10 mM Tris-HCl (pH 8.0), 1mM EDTA buffer and pelleted by centrifugation at 600 × g for 10 min. The bacteria were subjected to a single tube homogenization/RNA extraction protocol according to Williams *et al.* (40). Briefly, bacteria were resuspended with 1 ml of TRIzol (Invitrogen Life Technologies, Calsbad , CA) and were mechanically lysed in a vial containing lysing matrix B (MP Biomedical LLC, Solon, OH) using a Fast Prep-24 (MP biomedical LLC, Solon, OH). The resultant homogenate solution was added with 200 µl of chloroform-

isoamyl alcohol (24:1, v/v) and subjected to centrifugation at 27,000 × g for 20 min. Nucleic acids in the aqueous phase were precipitated by adding 100 μ l of 3 M Na-acetate (pH 5.2) and 500 μ l of isopropanol and followed by incubation at -20 °C for 1 hr. Total RNA was recovered by centrifugation at 27,000 × g for 30 min at 4 °C. According to the manufacturer's instructions, the Turbo DNA free kit (Ambion, Austin, TX) was used to remove the DNA contaminants in the RNA solution prior to cDNA synthesis.

2.3.2 Primer design for quantitative real time PCR (qRT-PCR) analysis

All DNA sequences of 136 hypothetical unknown ORFs in functional class VI (7) (http://www.pasteur.fr/recherche/unites/Lgmb/NATURE_DATA/ML_gene_list) were obtained from the *M. leprae* genome database, Leproma (22). The OLIGO6 Primer Analysis Software (Molecular Biology Insights Inc, Cascade, CO) was used to design the specific primer set to each target gene of hypothetical unknown ORFs. ML2244, ML2249, ML2567, ML2151, ML0567 and ML0678 were excluded. The genes encoding ML2249 and ML2151 were too small to design proper primers. Gene expression levels of ML2567, ML2244 ML0678 and ML0567 were already studied by us or collaborators (13,30). In order to enhance the efficiency of qRT-PCR, the primer set to each target gene was designed to produce a PCR product of 200-400 bp in size. The specificity of each primers set to template was analyzed by comparing the genomes analyses of *M. tuberculosis* (35), *M. avium, M. bovis* BCG and *M. smegmatis* TIGR Microbial Database (http://cmr.jcvi.org/cgi-bin/CMR/CmrHomePage.cgi) through a BLAST search.

2.3.3 Synthesis of complementary DNA (cDNA) and qRT-PCR assay

Total RNA transcripts of *M. leprae* Thai-53 were converted to cDNA using the SuperScript III First Strand Synthesis System kit (Invitrogen Life Technologies, Calsbad, CA) with random hexamers according to the manufacturer's instructions. Total 2.5 µg of cDNA were obtained. The qRT-PCR assay of each gene was performed with *M. leprae* cDNA (10ng of cDNA, equivalent to 8×10^7 cells) and a dilution series of quantified M. *leprae* genomic DNAs (10 fg, 50 fg, 100 fg, 500 fg, 1 pg, 10 pg, 100 pg and 1 ng per microliter) for relative quantification of cDNA by using the Platinum® SYBR® Green qPCR SuperMix-UDG (Invitrogen Life Technologies, Calsbad, CA) on the iCycler Real Time PCR Machine (BioRad Laboratories, Hercules, CA). All PCRs with a volume of 25 µl were performed in 96-well optical-grade PCR plates (BioRad Laboratories, Hercules, CA) in triplicate. The primer sets of ML0380, ML2038 and ESAT-6, which were previously proven to be expressed in M. leprae (30,42), were used to optimize the PCR conditions. An initial DNA denaturation step at 95 °C for 5 min was followed by 30 cycles of denaturation at 95 °C for 30 sec, primer annealing at 64 °C for 20 sec and primer extension at 72 °C for 45 sec, with a final extension step at 72 °C for 5 min. The iCycler iQ Software (BioRad Laboratories, Hercules, CA) was used to analyze the qRT-PCR assays with a correlation co-efficiency (≥ 0.95) and PCR efficiency (85%-105%).

2.3.4 Cloning of hypothetical unknowns ORFs from M. leprae

To express 24 of the selected novel antigen candidates in *E. coli*, genes encoding these hypothetical unknowns were PCR amplified from *M. leprae* Thai 53 genomic DNA

using the *rTth* DNA polymerase XL (Applied Biosystems, Carlsbad, CA). PCR amplification was performed using primer sets which include the *NdeI* and *HindIII* sites, respectively (underlined), specific to the upstream and downstream of the open reading frame as listed in Table 2.1). Each of the target genes was amplified using a Touchdown PCR. This method had a high initial annealing temperature at 64 °C that decreased by an additional 1 °C in each of the first 7 cycles, followed by 25 cycles at 58 °C. PCR products were directly digested with restriction enzymes and cloned into the expression vector pET 29a (+) (Novagen, Madison, WI), which contained the coding sequence for a 6-Histidine tag at the C-terminus of expressed proteins to facilitate the purification of recombinant proteins. The DNA sequences of all recombinant clones were confirmed by automated nucleotide sequencing at the Proteomics and Metabolomics Facilities (PMF), Colorado State University.

| Name | Sequence | Name | Sequence |
|--------------|---|----------------|---|
| ML0023 forwa | rd AA <u>CATATG</u> CCGACCAACCCCATTTC | ML0953 forward | AA <u>CATATG</u> GCGATTTCGGCCGTTATGGCTG |
| ML0023 reven | se AA <u>AAGCTT</u> AGTCCCCCTTCGCCCACT | ML0953 reverse | AA <u>AAGCTT</u> GCGTGCGTCGTCGCAGTGAT |
| ML0070 forwa | rd AA <u>CATATG</u> CTGACCGGGGTGACGAAC | ML0959 forward | AA <u>CATATG</u> GCTCGCACTCTGCGATGTT |
| ML0070 reven | se AA <u>AAGCTT</u> CTCCCTGTGGGGACTGTT | ML0959 reverse | TAAAGCTTAGAAGACGCACTCGGGCTG |
| ML0121 forwa | rd AA <u>CATATG</u> GTTTGGCACATGACACATTCG | ML1010 forward | AC <u>CATATG</u> CGCGGCAGTCAACACAAGG |
| ML0121 rever | se AA <u>AAGCTT</u> GGAGGAGTCCAGAATGTCTTT | ML1010 reverse | A <u>AAGCTT</u> CAGCCTAGCGACGTTGCTG |
| ML0141 forwa | rd AACATATGCGTCTCAAAGTCATTGACAGC | ML1384 forward | AACATATGACGGACGGCATCCAGCAGT |
| ML0141 rever | se AAAAGCTTGCCAACGTCTATATGCCGCT | ML1384 reverse | AA <u>AAGCTT</u> GTCGCCCGGACCAACAC |
| ML0188 forwa | IND AACATATGGATCCGATCAATTCGCCCA | ML1575 forward | AACATATGATGCCCTGGGGATCGCCTG |
| ML0188 rever | se AAAAGCTTCGCAATATATGTCCGCAA | ML1575 reverse | AAAAGCTTCAACCCAGCAGGTCCACCA |
| ML0217 forwa | rd AA <u>CATATG</u> GGTGTCGGTCGCCCATGTG | ML1949 forward | AA <u>CATATG</u> TTCTGGGTTGGGGTACGCG |
| ML0217 reven | se AA <u>AAGCTT</u> AGCACTTAGCTCATCAATG | ML1949 reverse | AA <u>AAGCTT</u> CTACTGGCGTCCTGGAATTT |
| ML0448 forwa | rd AA <u>CATATG</u> CAGGCCTGGTACGTATXTAG | ML2044 forward | AA <u>CATATG</u> GAAACGATCGTGCTCCCCGA |
| ML0448 rever | se CC <u>AAGCTT</u> CTATCTCATAATTACGTGTT | ML2044 reverse | AA <u>AAGCTT</u> CGACACCACCGACCACTACA |
| ML0527 forwa | rd TA <u>CATATG</u> GTTTACCTGGAAAACCATGG | ML2307 forward | AA <u>CATATG</u> GCAAAAACACGACCGGCC |
| ML0527 rever | se TC <u>AAGCTT</u> TGTCTTGCCCGCCGCTTCG | ML2307 reverse | AA <u>AAGCTT</u> GCTAACATTGCGGTGCTTGC |
| ML0588 forwa | rd AG <u>CATATG</u> GTGATTGGCAGCCTTGACATT | ML2313 forward | AA <u>CATATG</u> GCCAGTGCTTCTCAAACACA |
| ML0588 rever | se ACAAGCTTCCAGTGTCCCGGCAGCAT | ML2313 reverse | AT <u>CTCGAG</u> GTTCTCGCCGAGGATGC |
| ML0614 forwa | rd AA <u>CATATG</u> TGGAAGGCACGGACAGCATT | ML2651 forward | AA <u>CATATG</u> ATCTTGAGCTCGATATGGCT |
| ML0614 rever | se AA <u>AACGTT</u> CCCGCGCCGAATCA | ML2651 reverse | AA <u>AAGCTT</u> GCGCCAGATAGGCAGGATTT |
| ML0755 forwa | rd AA <u>CATATG</u> GCACTGCGCGAGGCCACCAA | ML2630 forward | AAA <u>CATATG</u> CAGCATCAACCCCACCAC |
| ML0755 reven | se AA <u>AAGCTT</u> TTAGCGTGACCTCCTCGCAC | ML2630 reverse | TT <u>AAGCTT</u> ATGCGTCGCTGCTGCTTT |
| ML0920 forwa | rd AC <u>CATATG</u> AGCACACTGCACAAGGTCAA | ML2666 forward | AA <u>CATATG</u> GCGTGGCTGGAGCAGTGGAA |
| ML0920 Lowe | r AA <u>AAGCTT</u> TTGGTAGGCGTAAAAGCCGGTT | ML2666 reverse | AA <u>AAGCTT</u> GCCAACGTCTATATGCCGC |

Table 2.1 List of primers used in cloning of 24 selected hypothetical unknown ORFs.

The sequences recognized by restriction enzymes are underlined; CATATG is specific for *Nde*I and AAGCTT for *Hind*III

2.3.5 Purification of recombinant M. leprae proteins

The plasmids containing the novel antigen candidates were introduced into the *Escherichia coli* expression host BL21 Star (DE3) pLysS (Invitrogen Life Technologies, Carlsbad, CA) according to manufacturer's instruction. The transformants were grown to the log phase (optical density (O.D.) at 600 nm of 0.5) at 37 °C in Luria-Bertani (LB) broth with 50 µg/ml kanamycin. Expression of recombinant proteins was induced by adding 0.2-0.5 mM IPTG. The cells were cultured at 25 °C for overnight. The cultured cells were harvested by centrifugation at 4°C and frozen at -70 °C. The cells were resuspended in lysis buffer (50 mM Tris-HCl pH 8.0, 150 mM NaCl) containing 10 μ g/ml of DNase and 10 μ g/ml of RNase, with a protease inhibitor cocktail (P8340, Sigma, St. Louis, MO) and 20 µM PMSF (phenylmethanesulfonyl fluorid, Sigma, St. Louis, MO), and were disrupted by intermittent probe sonication with a Soniprep 150 sonicator (Sanyo MSE, London, UK) for 10 min. The lysates were centrifuged at $5,000 \times$ g for 5 min to remove unbroken cells and the supernatants were centrifuged for 30 min at $27,000 \times g$ at 4 °C. Supernatants were subjected to a Ni-NTA agarose column chromatography (Qiagen, Valencia, CA). The column was washed with $20 \times$ column volumes of the same buffer. Recombinant proteins were eluted with $1 \times \text{column}$ volume of lysis buffer containing stepwise increments of imidazole (5, 10, 20, 50, 100, 150 and 200 mM) (31). All recombinant hypothetical unknowns were found in the 50-200 mM imidazole fractions. Only 15 out of the 24 novel candidates were expressed and purified for further serological analyses.

2.3.6 Subjects and samples

Leprosy patients diagnosed at the Leonard Wood Memorial Center for Leprosy Research, Cebu, the Philippines were included in this study. Leprosy patients (n=70) were classified according to the Ridley-Jopling classification system (28) based on bacterial index, histological and clinical observations carried out by experienced leprologists and a leprosy pathologist. All leprosy patient sera were collected at the time of initial diagnosis prior to beginning multidrug therapy. Serum samples from cavitary TB patients (n=30) were from a cohort of newly diagnosed TB patients from the Tuberculosis Trials Consortium Study Group 22. Eleven of the patients were sputum smear negative, while nineteen were smear positive (ranging from 1+ to 4+, too many to count). The sera were provided by Dr. William Mac Kenzie from a serum bank repository from the Centers for Disease Control in Atlanta, Georgia. Serum samples from all sources, made anonymous and coded to protect donor identities, were obtained with informed consent and/or with permission from the local ethics committee or institutional review boards of the relevant countries and institutions involved (32). Sera were pooled from eight randomly selected lepromatous patients whose bacterial index was 6. Five sera of cavitary TB patients were pooled to investigate their cross-reactivity to *M. leprae* recombinant antigens.

2.3.7 Western blot analysis

Protein quantities of all subcellular fractions were measured by the bicinchoninic acid assay reagent (Pierce, Rockford, IL). Each novel recombinant antigen candidate (0.5 μ g/lane) was subjected to electrophoresis in 15% SDS-PAGE gel and transferred onto

nitrocellulose membrane. Blots were blocked with blocking buffer (3% BSA in PBS/0.05% Tween80) for 2 hr and were then probed with both diluted sera (1:5,000 dilutions for the leprosy sera and 1:1000 for TB sera). Blots were performed by probing with secondary anti-human IgG alkaline phosphatase (Sigma, St. Louis, MO) and developed by using 5-bromo-4-chloro-3-indolyl phosphate/nitro blue tretrazolium (Sigma, St. Louis, MO) (33).

2.4 Results

2.4.1 Establishment of the gene expression profile of M. leprae specific hypothetical unknown ORFs and selection of novel antigen candidates

Previously, the global analysis of *M. leprae* transcripts using microarray revealed that a few of the hypothetical unknown genes were expressed during infection (42). Although this global analysis has provided valuable information in bacterial pathogenesis studies, it is not sufficient to evaluate the functional status of hypothetical unknowns that were usually expressed at low levels.

In order to verify the gene expression level of all hypothetical unknowns (class VI), we applied real time PCR assays using a set of target-specific primers and SYBR green I as fluorescent probes to determine the relative quantification of hypothetical unknown ORFs in cDNAs synthesized from total *M. leprae* RNAs. SYBR green I binds to dsDNA generated with each progressive cycle of the PCR and emits a fluorescent signal; fluorescence generated is quantitatively measured to track the amplification of DNA in PCR. There is a quantitative relationship between the amount of starting template and the amount of the PCR product at an exponential phase of the PCR (25).

We established the gene expression profile of 136 hypothetical unknown ORFs which had not been previously studied in our laboratory. This target based gene expression analysis revealed that a majority of the hypothetical unknown ORFs (60%) expressed less than 100fg of mRNA (Figure 2.1 and Table 2.2).



Figure 2.1 Summary of the hypothetical unknowns investigated in the present study.

The transcript of EAST-6/ML0049, which has proved to be an important T-cell antigen (30), was expressed at low level in both nude-mouse derived *M. leprae* and *M. leprae* recovered from MB patients (42). The gene expression levels of the screened 26 hypothetical unknown genes were found to be in the range of 1 pg~500 fg, which is similar to that of ESAT-6 (Table 2.3). Of 26 hypothetical unknowns, the immunogenicity of ML0573 and ML0574 was already evaluated in endemic regions (13).

Therefore, 24 hypothetical unknowns were cloned in order to express in *E. coli* (Figure 2.1).

The class VI hypothetical unknown proteins consisting of 142 theoretical ORFs had no orthologues among any mycobacterial genome databases (7,30) at the time that this work was started. Subsequent advances in next generation DNA sequencing technology has rapidly completed a great number of microbial genome sequences and the advanced bioinformatics analysis methodology has further refined the annotation of *M*. *leprae* ORFs on the previous genome sequences (41). In a reflection of this trend, the current list of functional class VI

(http://www.pasteur.fr/recherche/unites/Lgmb/NATURE_DATA/ML_gene_list) has been adjusted continuously but has retained the majority of genes from the previous list of genes. In order to evaluate whether these class VI hypothetical unknown proteins have homology with other genes expressed in other human pathogens, the sequence similarity of 26 *M. leprae*-specific proteins were examined by BLASTP

(<u>http://www.ncbi.nlm.nih.gov/BLAST</u>). A few genes contained orthologues among other pathogens (data not shown) but appeared as hypothetical genes.

Based on e-values, 26 new antigen candidates have been categorized into 4 subclasses. In BLASTP analysis, the similar protein sequence to these candidates (21 of 26) are less than 100 AAs with high gap score and its identities was less than 25%, indicating that the selected antigen candidates cannot have orthologues in other pathogen and are considered to be *M. leprae* specific (12,27).

| 14510 20 | | Ct valu | e of geno | | Δ | -jpot | eneur | ummu | Ct value | Level of gene |
|----------|------|------------|--------------------------|------------|-------|-------|-------|-------|--------------|------------------|
| ORF | 1ng | 100g | 10pg | 1pg | 500fg | 100fg | 50fg | 10fg | of cDNA | expression |
| ML0009 | 0 | no. | signal | 10 | 0 | 0 | 0 | 5 | | |
| ML0840 | | no | signal | | | | | | | |
| MI 0939 | | no | signal | | | | | | | |
| MI 2045 | | no | signal | | | | | | | |
| MI 2669 | | no | signal | | | | | | | |
| MI 0464 | | 22.3 | 25.8 | 29.5 | 32 | 33.5 | 34 | 36.4 | 36.7 | <10fa |
| MI 0926 | | 22.1 | 2010 | 30 | 32 | 35.9 | 36.6 | 41 | N/A | <10fg |
| MI 1148 | 18.6 | 22.1 | 25.2 | 28.9 | 29.4 | 32.8 | 33.2 | 35.4 | 35.2 | 10fg |
| MI 0473 | | 21.4 | 24.4 | 28.4 | 30.7 | 33.4 | 34 | 36.4 | 34.8 | 10fg< <50fg |
| MI 0218 | | 33.2 | 36.7 | 36.3 | 38.1 | 38.7 | 39 | 42 7 | 40.1 | 10fg< <50fg |
| ML0291 | | 21.4 | 24.6 | 28.4 | 29.8 | 31.2 | 33 | 34.5 | 34.3 | 10fg< <50fg |
| ML0293 | | 21.2 | 24.8 | 28.8 | 30.4 | 31.1 | 32.7 | 36.8 | 34 | 10fg< <50fg |
| ML0938 | | 21.9 | | 28.9 | 30 | 31.3 | 32.2 | 34 | 33.4 | 10fg< <50fg |
| ML1001 | | 23.4 | 27.3 | 32.6 | 34.5 | | 36 | 38.7 | 36.5 | 10fg< < 50fg |
| ML0325 | | 25.6 | 27 | 32 | 33.1 | 34.7 | 30 | 37.9 | 35.4 | 10fg< <100fg |
| ML0927 | | 26.4 | 30.8 | 33.9 | 34,2 | 37.5 | | 41 | 38.3 | 10fg< <100fg |
| ML0152 | | 22.1 | 26 | 29.5 | 30.4 | 32.5 | 34.1 | | 35.6 | < 50fg |
| ML0162 | | 23.7 | 27.9 | 32.4 | 40.6 | 35 | 36.1 | | 37 | <50fg |
| ML0470 | 17.4 | 21 | 24.6 | 28.7 | 29.7 | 31.7 | 32.1 | | 36.4 | <50fg |
| MI 0659 | | 29 | 34 | 39.5 | 2011 | 41.6 | 42.8 | | N/A | <50fg |
| MI 1057 | | 20 | 0. | 0010 | | | 38.9 | | 39.6 | <50fg |
| MI 1210 | 19.5 | 23.5 | 27 4 | 32 | 32.5 | 35.8 | 38.8 | | N/A | <50fg |
| MI 1605 | 18.6 | 23.3 | 26.3 | 30.3 | 31.4 | 34 | 0010 | | N/A | <50fg |
| MI 1915 | 21.1 | 24.9 | 29 | 32.4 | 33.6 | 35.7 | 36.6 | | 37 | <50fg |
| MI 1976 | 20.4 | 24.2 | 27.9 | 31.9 | 32.5 | 34.5 | 35.6 | | N/A | <50fg |
| MI 1979 | 19.4 | 24 | 27.6 | 32.5 | 33 | 34.9 | 35.3 | | 36.9 | <50fg |
| MI 1989 | 19.6 | 23.6 | 27.7 | 31.8 | 34 | 36 | 00.0 | | 36.5 | <50fg |
| ML 2013 | 19.7 | 23.8 | 27.9 | 31.1 | 33.2 | 35 | | | N/A | <50fg |
| MI 2346 | 10.7 | 20.0 | 21.0 | 33.5 | 34.1 | 36.1 | 36.5 | N/A | 38 | <50fg |
| MI 2476 | 18.8 | 23.8 | 26.7 | 30.2 | 31.2 | 33.6 | 33.7 | 1.077 | 34.2 | <50fg |
| MI 2478 | 18.3 | 23.8 | 26.9 | 31.4 | 33.1 | 35.7 | 35 | | N/A | <50fg |
| MI 2491 | 24 | 29.5 | 33.5 | 37.9 | 39.5 | N/A | 41.5 | | 41.9 | <50fg |
| MI 2562 | 20.5 | 24.7 | 29.3 | 32.1 | 33.2 | 35 | 36.1 | | 37.1 | <50fg |
| ML 2603 | 20.0 | 24.7 | 35.6 | 30 | 40.5 | 41 Q | 42.6 | | N/A | <50fg |
| MI 2629 | 18.7 | 24.6 | 28.7 | 33.5 | 33.1 | 37 | 36.9 | | N/A | <50fg |
| MI 0024 | 10.7 | 24.6 | 28.3 | 31.6 | n/a | 37.6 | 39 | n/a | 38.1 | 50fg |
| ML0664 | | 24.0 | 26.0 | 30.4 | n/a | 34.1 | 34 3 | n/a | 34.5 | 50fg |
| ML0777 | | 22.4 | 26 | 29.5 | 30.8 | 32.8 | 35 | | 35.5 | 50fg |
| MI 1188 | 18.5 | 22.5 | 26.1 | 30 | 30.5 | 32.9 | 34.3 | | 34.5 | 50fg |
| MI 1189 | 10.0 | 22.0 | 27.9 | 32.1 | 33 | 36.1 | 37 | | 36.8 | 50fg |
| MI 1292 | 17 7 | 20.7 | 25.1 | 20 | 30.2 | 32 | 33 | | 33.6 | 50fg |
| MI 1420 | 19 | 22.9 | 26.5 | 20 30 1 | 31.9 | 34.1 | 34 7 | | 39.8 | 50fg |
| MI 1761 | 18 1 | <u></u> .J | 25.8 | 20.1 | 30.1 | 31 9 | 32.9 | | 32.6 | 50fg |
| MI 2001 | 19 | 23.4 | 20.0 | 31.2 | 31.8 | 32 | 34.2 | | 34.2 | 50fg |
| MI 2170 | 23.5 | 20.4 | 21 31 0 | 36.1 | 36.8 | 38 5 | 30 3 | 41 2 | 39.4 | 50fg |
| MI 2172 | 20.0 | 20.5 | 31.7 | 33.7 | 36 | 38.5 | 30 | 43 | 39.4 | 50fg |
| MI 2497 | 20.9 | 26.7 | 30.3 | 34.8 | 36 | 39 | 40 5 | -10 | 40.4 | 50fg |
| MI 0025 | 20.9 | 20.7 | 30.3 | 37 1 | 38 | 30 1 | 40.0 | | 40.4 11 1 | 50fg< <100fg |
| MI 0126 | | | 55 | 327 | 33 | 31 F | 76.2 | n/a | 35.6 | 50 fg < 100 fg |
| MI 0202 | | 20.8 | 24 4 | 27 7 | 28.7 | 31 0 | 32.6 | 34.6 | 31 | 50 m < 100 m |
| MI 0360 | | 20.0 | 2 7.4 26.1 | 30.6 | 20.7 | 33.4 | 36 1 | 54.0 | 34.5 | 50 m < 100 m |
| MI 0863 | | 20.0 | 20.1 | 20.0 | 31 | 32 | 34 | | 33.2 | 50 fg < 100 fg |
| ML0005 | | 22.1 | 25 0 | 23 20 7 | 30 | 32 B | 35.2 | | 34 7 | 50 m < 100 m |
| WIL0301 | | 20 | 20.3 | 23.1 | 52 | 02.0 | 00.0 | | 54.7 | July < 1001g |

 Table 2.2 Analysis of qRT-PCR of 136 hypothetical unknown ORFs.

| ORF 1ng 100pg 10pg 1pg 500fg 100fg 50fg 10fg of cDNA ML1018 18.7 22.4 25.7 27.9 30.2 31.6 34 32.5 | expression 50fg< <100fg |
|---|----------------------------|
| ML1018 18.7 22.4 25.7 27.9 30.2 31.6 34 32.5 | 50fa< <100fa |
| | |
| ML1243 23.9 29 33.2 38.5 39.1 40.8 42.5 42.3 | 50fg< <100fg |
| ML1275 18.8 22.9 26.7 31.3 31.7 33.4 36 34.1 | 50fg< <100fg |
| ML1523 18.6 22.4 26.7 29.8 30.8 33.4 34.7 34.5 | 50fg< <100fg |
| ML1602 18 21.9 25.4 29.2 30.2 31.5 34.3 36.5 33.2 | 50fg< <100fg |
| ML1763 18.9 26.9 30.5 31 33.5 36 34 | 50fg< <100fg |
| ML1788 18.8 22.5 26 30.2 31.2 33.2 34.8 39 34 | 50fg< <100fg |
| ML1829 17.8 22.5 26.3 29.8 30.5 31 32.7 32.5 | 50fg< <100fg |
| ML2201 18.9 23.3 27.1 31 31.2 32.3 32.8 32.6 | 50fg< <100fg |
| ML2452 19 24 27.9 32 32.6 34.6 35.2 | <100fg |
| ML2265 17.6 21.6 25.1 29 29.4 N/A N/A 31.3 | <100fg |
| ML2178 21.4 25.9 31.2 34.6 35.9 37 N/A | <100fg |
| ML1119 20.3 24.3 29.2 33.2 33.2 36 N/A | < 100fg |
| ML1517 21.4 25.9 30.9 34.2 36.4 38.8 | <100fg |
| ML1604 17.9 22 25.5 29.6 30.2 32.8 34.2 | <100fg |
| ML1717 18.8 26.2 30.6 31.1 33.3 35 | <100fg |
| ML1821 18 22.4 25.7 29.7 33.6 34.4 40.7 | <100fg |
| ML1011 18 21.8 25 28.7 30.2 32.4 32.8 35 32 | 100fg |
| ML1106 19 22.8 26.4 29.9 31.3 33.6 38.8 33.3 | 100fg |
| ML1344c 17.5 21.4 24.7 28.5 29.3 31.8 N/A 34.3 | 100fg |
| ML2158 18.9 24.1 27.6 31.5 32.7 33.2 35 33.6 | 100fg |
| ML0394 34.8 n/a 38.1 n/a 39.4 38.5 | 100fg |
| ML0663 23.2 26.2 31 31.9 33 32.9 | 100fg |
| ML0949 22.6 31 32.1 35 36.1 36.8 | 100fg |
| ML2264 18.8 23.6 27.3 32 33.3 35.4 36.2 35.5 | 100fg |
| ML2468 18 22.1 25.2 29.5 30 32.5 34.4 32.1 | 100fg |
| ML0950 23.8 28 31.1 32 33.6 35.7 | <100fg |
| ML0964 25.3 29.5 32.8 33.4 35 36.6 | < 100fg |
| ML0950 23.8 28 31.1 32 33.6 35.7 | <100fg |
| ML0472 21.3 25.3 28.2 31.9 32.8 34.4 32 | 100fg< <500fg |
| ML0656 22.4 25.8 30 30.8 34.3 35 33.6 | 100fg< <500fg |
| ML0679 18.5 22.4 25.9 29.5 30.5 33.4 33.8 31 | 100fg< <500fg |
| ML0947 21.9 29.1 30.7 31.8 32.5 33.5 39 32 | 100fg< <500fg |
| ML1186 18.5 22.3 25.7 29.3 31 35.1 34.5 | 100fg< <500fg |
| ML1294 17.2 21.4 24.6 28.1 29.3 31.7 33.2 31 | 100fg< <500fg |
| ML1572 21.5 25.8 29.7 30.4 32.6 33.5 31.7 | 100fg< <500fg |
| ML1601 18 22.2 25.3 29.2 31 34.3 32.1 | 100fg< <500fg |
| ML1603 18.1 22 25.7 28.7 29.4 32.1 31 | 100fg< <500fg |
| ML1793 18.7 21.9 26.5 30.8 31 32.6 34.5 31.8 | 100fg< <500fg |
| ML1796 18.3 22.8 26 30.4 32.2 32.8 33.6 32.5 | 100fg< <500fg |
| ML1928 17.5 21.8 25 28.4 29.1 31.8 32.7 30.9 | 100fg< <500fg |
| ML1972 19.3 23.4 26.6 30.6 32.1 34.4 35.5 33.3 | 100fg< <500fg |
| ML2035 22.5 26.3 29.6 33.3 33.8 36 35 | 100fg< <500fg |
| ML2176 19.2 23.9 28.4 31.5 32.5 34.2 38.7 39.8 32.5 | 100fg< <500fg |
| ML2252 19.9 24.3 27.2 32 32.8 36 37.9 41.1 35.5 | 100fg< <500fg |
| ML2288 19.4 24 27 31.8 32.7 36.2 34.5 | 100fg< <500fg |
| ML2379 18.8 24.3 27.7 32 32.4 36.7 37 33.7 | 100fg< <500fg |
| ML2621 19.6 25.3 29.5 33.7 36 37.9 39 36.8 | 100fg< <500fg |
| ML0963 23.4 29.8 32.7 35 N/A | <500fg |
| ML2283 17.7 22.7 25.6 29.7 31.1 34.9 31 | 500fg |
| ML2284 19.2 24 31.6 32 34.3 35.2 34.2 | 500fg |
| ML0023 24.6 26.9 38.9 37.1 37.9 38.6 n/a 37.4 | 500fg |
| ML0121 23.4 26.9 31.8 32.9 33.5 36.2 31.8 | 500fg |

| | | Ct value of genomic DNA | | | | | | | | Level of gene |
|--------------|-------------|-------------------------|------|------|-------|-------|------|------|---------|---------------|
| ORF | 1ng | 100pg | 10pg | 1pg | 500fg | 100fg | 50fg | 10fg | of cDNA | expression |
| ML0127 | | 25.3 | 29.9 | 32.7 | 33.2 | 34.3 | n/a | n/a | 33.5 | 500fg |
| ML0265 | | 29.3 | 32.1 | | 35 | 39.1 | 42.7 | | 35.4 | 500fg |
| ML0757 | | 23 | 26.7 | 30 | 31.8 | 33.1 | 35.4 | | 31.7 | 500fg |
| ML0928 | | 28.8 | 34 | 39 | 40 | 42 | | | 40.3 | 500fg |
| ML1445 | 19 | 22.7 | 26.3 | 30 | 32 | 34 | 36 | | 32.4 | 500fg |
| ML0448 | | 21.7 | 25.2 | 29.1 | 33.3 | 34.2 | | | 31.4 | 500fg< <1pg |
| ML0573 | | 21.6 | 24.7 | 28.8 | 33.4 | 34.8 | 34 | 41 | 32 | 500fg< <1pg |
| ML0755 | | 23.2 | 26.7 | 31 | 31.8 | 33.1 | 35.4 | | 31.7 | 500fg< <1pg |
| ML0920 | | 23.8 | 27.4 | 31.1 | 32.5 | 35 | | | 31.5 | 500fg< <1pg |
| ML0953 | | 27 | 29.5 | 33.8 | 35.1 | | | 39.2 | 34.1 | 500fg< <1pg |
| ML1949 | 19.6 | 23.7 | 27.1 | 31.4 | 32.9 | 34.1 | | | 32 | 500fg< <1pg |
| ML2630 | 17.3 | 22.8 | 26.5 | 30.9 | 31.6 | 32.9 | N/A | | 31.3 | 500fg< <1pg |
| ML0141 | | | 36 | 37.6 | 39.2 | 40.6 | 42.7 | n/a | 36.5 | 1pg |
| ML0188 | | 23.9 | 29.1 | | 35.4 | 36.4 | 36.8 | n/a | 34.6 | 1pg |
| ML0217 | | 23.8 | 28.4 | 33.9 | 36.9 | | 37.1 | 40.4 | 32 | 1pg |
| ML0070 | | 25.2 | 30.7 | 33.6 | 33.5 | 34 | 34.2 | 35.4 | 33.6 | 1pg |
| ML0574 | | 23.1 | 27 | 30.5 | 31 | 32 | 33.2 | | 30.9 | 1pg |
| ML0588 | | 22.8 | 26 | 30.5 | 31.8 | 32.7 | 34.8 | | 30.1 | 1pg |
| ML0614 | 20.8 | 25 | 29.1 | 32.4 | 36 | 37.5 | | | 32 | 1pg |
| ML0959 | | 24 | 29 | 32.5 | 33.4 | 36.2 | | | 32.6 | 1pg |
| ML1010 | 18.4 | 22.4 | 25.2 | 28.7 | 30.2 | 32.2 | 32.9 | | 28.5 | 1pg |
| ML1384 | 18.8 | 22.7 | 25.2 | 30.4 | 31 | 34.8 | 35.2 | | 30.7 | 1pg |
| ML1575 | 26.4 | 31.1 | 34.4 | 37.8 | 38.9 | 41.4 | | | 37.9 | 1pg |
| ML2044 | 21 | 25.7 | 30 | 33.7 | 34.7 | 37 | 39.1 | | 33.9 | 1pg |
| ML2307 | 19.9 | 24.6 | 27.2 | 31.8 | 33.2 | 34.8 | 35.2 | | 31.5 | 1pg |
| ML2313 | 20.9 | 26 | 30.5 | 34.3 | 35.8 | 38.7 | | | 32.3 | 1pg |
| ML2651 | 19.3 | 23.1 | 27.3 | 30.9 | 32 | 35.7 | N/A | | 31.1 | 1pg |
| ML2666 | 19.5 | 25.1 | 28 | 32 | 32.3 | 34.7 | 35.6 | | 32.1 | 1pg |
| ML0527 | | 22.4 | 25.4 | 29.2 | 30.2 | 32.4 | 33.2 | | 28.9 | > 1pg |
| Positive cor | trol for gR | T-PCR | | | | | | | | |
| ESAT-6 | | 22 | 26 | 29.4 | 31.7 | 32.4 | 35.2 | | 29 | ≈1pg |
| ML0380 | | 28 | 31.5 | 36.7 | | | | | 32.1 | 1pg< <10pg |
| ML2038 | | 20.2 | 24 | 27.5 | 31 | 32.9 | 34 | | 26.2 | 1pg< <10pg |

The Ct (cycle threshold) is the number of cycles at which the fluorescent signal generated in PCR just exceeds the background fluorescent level (threshold) where it reaches at an exponential phase of the PCR (<u>http://www3.appliedbiosystems.com/cms/groups/mcb_marketing/documents/generaldocument/cms_042502.pdf</u>). The expression level of each gene was determined by comparing to results of qRT-PCR assay generated using a series of genomic DNA standard and were from 8×10^7 of *M. leprae*.

| | | Ct value | e of geno | mic DNA | Ct value | Level of | | | |
|--------|-------|----------|-----------|---------|----------|----------|------|---------|-----------------|
| ORF | 100pg | 10pg | 1pg | 500fg | 100fg | 50fg | 10fg | of cDNA | gene expression |
| ML0023 | 24.6 | 26.9 | 38.9 | 37.1 | 37.9 | 38.6 | n/a | 37.4 | 500fg |
| ML0070 | 25.2 | 30.7 | 33.6 | 33.5 | 34 | 34.2 | 35.4 | 33.6 | 1pg |
| ML0121 | 23.4 | 26.9 | 31.8 | 32.9 | 33.5 | 36.2 | | 31.8 | 500fg |
| ML0448 | 21.7 | 25.2 | 29.1 | 33.3 | 34.2 | | | 31.4 | 500fg< <1pg |
| ML0573 | 21.6 | 24.7 | 28.8 | 33.4 | 34.8 | 34 | 41 | 32 | 500fg< <1pg |
| ML0755 | 23.2 | 26.7 | 31 | 31.8 | 33.1 | 35.4 | | 31.7 | 500fg< <1pg |
| ML0920 | 23.8 | 27.4 | 31.1 | 32.5 | 35 | | | 31.5 | 500fg< <1pg |
| ML0953 | 27 | 29.5 | 33.8 | 35.1 | | | 39.2 | 34.1 | 500fg< <1pg |
| ML1949 | 23.7 | 27.1 | 31.4 | 32.9 | 34.1 | | | 32 | 500fg< <1pg |
| ML2630 | 22.8 | 26.5 | 30.9 | 31.6 | 32.9 | n/a | | 31.3 | 500fg< <1pg |
| ML0141 | | 36 | 37.6 | 39.2 | 40.6 | 42.7 | n/a | 36.5 | 1pg |
| ML0188 | 23.9 | 29.1 | | 35.4 | 36.4 | 36.8 | n/a | 34.6 | 1pg |
| ML0217 | 23.8 | 28.4 | 33.9 | 36.9 | | 37.1 | 40.4 | 32 | 1pg |
| ML0574 | 23.1 | 27 | 30.5 | 31 | 32 | 33.2 | | 30.9 | 1pg |
| ML0588 | 22.8 | 26 | 30.5 | 31.8 | 32.7 | 34.8 | | 30.1 | 1pg |
| ML0614 | 25 | 29.1 | 32.4 | 36 | 37.5 | | | 32 | 1pg |
| ML0959 | 24 | 29 | 32.5 | 33.4 | 36.2 | | | 32.6 | 1pg |
| ML1010 | 22.4 | 25.2 | 28.7 | 30.2 | 32.2 | 32.9 | | 28.5 | 1pg |
| ML1384 | 22.7 | 25.2 | 30.4 | 31 | 34.8 | 35.2 | | 30.7 | 1pg |
| ML1575 | 31.1 | 34.4 | 37.8 | 38.9 | 41.4 | | | 37.9 | 1pg |
| ML2044 | 25.7 | 30 | 33.7 | 34.7 | 37 | 39.1 | | 33.9 | 1pg |
| ML2307 | 24.6 | 27.2 | 31.8 | 33.2 | 34.8 | 35.2 | | 31.5 | 1pg |
| ML2313 | 26 | 30.5 | 34.3 | 35.8 | 38.7 | | | 32.3 | 1pg |
| ML2666 | 25.1 | 28 | 32 | 32.3 | 34.7 | 35.6 | | 32.1 | 1pg |
| ML0527 | 22.4 | 25.4 | 29.2 | 30.2 | 32.4 | 33.2 | | 28.9 | > 1pg |

Table 2.3 Analysis of qRT-PCR of the 26 selected hypothetical unknown ORFs.

The transcription levels of 26 hypothetical unknown ORFs were similar to that of ML0049, an immunologically important antigen expressed at fairly low levels.

However, five selected antigens which are ML0614 in subclass VI.c and all of subclass VI.d appeared to contain orthologues in other *Mycobacterium* spp. by only sequence analysis but their physiological role and expression status are not experimentally verified yet (22). Therefore, these five candidates and are currently not considered as the hypothetical unknowns (22).

| Subclass | ORF | Molecular Mass (kDa) | Sequence Similarity | Protein with mximal similarity ¹ (Identities (%), Ratio of coverage ² , maximal stretch ³ or gap in sequence alignment ⁴) |
|----------|---------|----------------------------|------------------------|--|
| VI.a | ML0070 | 9.1 | none | N/A ⁴ |
| | ML0141 | 9.3 | none | N/A |
| | ML0448 | 10 | none | N/A |
| | ML0527 | 8.7 | none | N/A |
| | ML0574 | 11.4 | none | N/A |
| | ML0959 | 13.6 | none | N/A |
| | ML1010 | 8.4 | none | N/A |
| | ML1384 | 12.2 | none | N/A |
| | ML1575 | 11.1 | none | N/A |
| | ML2044 | 7.9 | none | N/A |
| | ML2651 | 11.6 | none | N/A |
| | ML2666 | 8.8 | none | N/A |
| VI.b | ML0121 | 9.6 | 3e-08 | Hypothetical proteins of <i>M. paratuberculosis</i> |
| | ML0023 | 11.7 | 3e-06 | Transposage for IS3514B of <i>Propionbacterium</i> freudenreichii |
| | ML0188 | 9.2 | 3e-08 | (31%, 36/401, 7) Conserved hypothetical unknown of M. paratuberculosis (46%, 36/230, 7) |
| | ML0217 | 8.4 | 1e-06 | Transposage of <i>Rhodococcus jostii</i> (98%, 80/346, 80) |
| | ML0573 | 9.5 | 1e-05 | Hypothetical protein of <i>M. kansasii</i> (41%, 30/282, 4AAs) |
| | ML0588 | 8.5 | 3e-04 | PPE protein of <i>M. tuberculosis</i> (55%, 21/539, 4) |
| | ML0953 | 8.6 | 1e-06 | Putative nucleic acid binding protein of <i>M. kansasii</i> (85%, 54/131, 13) |
| VI.c | ML0614* | 10.2 | 3e-17 | Hypothetical proteins of <i>M. tuberculosis</i> T46 (66%, 43/67, 4AAs) |
| | ML2630 | 13 | 7e-10 | conserved membrane protein of <i>M.</i> tuberculosis T92, (54%, 36/309, 5) |
| | ML1949 | 12.5 | 5e-17 | Hypothetical proteins of <i>M. kansasii</i> (54%, 41/483, 8) |
| VI.d | ML0755* | 9.6 | 1e-22 | Hydrolase of <i>M. kansasii</i> (69%, 50/183, 12AAs) |
| - | ML0920* | 6.3 | 1e-102 | Hypothetical protein of <i>M. tuberculosis</i> T17 (83%, 182/218, 8 gaps) |
| | ML2307* | 9.3 | 4e-52 | Transcription factor WhiB4 of <i>M.</i> |
| | ML2313* | 21.8 | 3e-71 | Transcriptional regulator (PadR) of <i>M. avium</i> 104 (67%, 163/242, 50 gaps) |

Table 2.4 Amino acids (AAs) sequence similarity of the 26 selected *M. leprae*-specific proteins.

All ORFs in subclass VI.a appear no homology. *M. leprae* proteins with BLASTP values between 1×10^{-4} to 1×10^{-8} are considered to have low homology (Subclass VI.b). Proteins with e-values between 1×10^{-8} and 1×10^{-20} have moderate homology (Subclass VI.c) and less than 1×10^{-20} have high homology (Subclass VI.c). 1; Most similar one of several proteins found in BLASTP. 2; Number of matched AAs of the *M. leprae* hypothetical unknown to number AAs of most similar in sequence alignment. 3: Number of AAs found in longest aligned sequence. 4: Number of gap AAs indicate different sequence in sequence alignment. *: ORFs are currently found to have orthologues and not considered as hypothetical unknown.

2.4.2 Purification and serological reactivity of the recombinant M. leprae specific antigens

Fifteen of the 24 new candidates were successfully expressed as 6 Histidine tag fusion proteins in the *E. coli*-T7 promoter driven vector system and purified by immobilized metal affinity chromatography (Figure 2.2.A). Since ML0573 and ML0574 were already studied by other group (13), we exclude them in the present study. In order to evaluate the immunogenicity of these novel antigens, we performed Western blot analysis using sera of both LL and cavitary TB patients (Figure 2.2.B and C).





The seroreactivity to all recombinant hypothetical unknowns revealed that 10 of the 15 recombinant antigens were recognized by circulating antibodies in the pooled sera from LL patients (Figure 2.2.B). Western blot analysis using pooled sera of 5 cavitary TB patients detected a single recombinant antigen out of the 15 proteins, ML2307, which had high homology with orthologues of other bacteria (Table 2.4 and Figure 2.2.C). Although all ORFs of subclass VI.d showed higher homology with the orthologues of *M*. *tuberculosis* than other candidate antigens (Table 2.4), sera of cavitary TB patients did not react to ML0755, ML0920, and ML2313 in this analysis (Figure 2.2.C). ML0141, ML0588, ML1384, ML2651 in subclass VI.a and ML1949 in subclass VI.c did not induce the seropositivity of LL or TB patients (Figure 2.2.B and C). Therefore, ML0121, ML0188, ML0448, ML0527, ML0755, ML0953, ML1384, ML2044, ML2313 and ML2666 induced *M. leprae* specific immune response in LL patients.

2.5 Discussion

The overall objective of this work was to identify the *M. leprae* specific antigens that can improve the sensitivity and specificity of CMI-based *in vitro* IFN- γ release assays (IGRAs), and to detect individuals exposed to/ infected by *M. leprae*.

Previously, 31 hypothetical unknown ORFs and hundreds of their peptides were identified as novel antigens from the 142 hypothetical unknown ORFs and their antigenic potentials were evaluated in order that the antigens are used for early diagnosis of leprosy by using the IGRA format (1,2,13,15,30). These *M. leprae* specific antigens were shown to be valuable diagnostic reagents in identifying those infected by *M. leprae* specifically, occupational contacts or HHCs of MB or PB patients (13,14,16,30). However, all

proteins and peptides were recognized by a significant number of endemic controls, but also these *M. leprae*-derived peptides showed considerable variation in reactivity to racially/ geographically different study populations (15,16). Most of the problems with these IGRAs originated from the low levels of CMI to hypothetical unknown proteins and hypothetical-unknown-derived peptides.

The genome of *M. leprae* contains exceptionally large number of pseudogenes (7) but also many hypothetical unknowns. These hypothetical unknowns (class VI; see Table 1.2) are low molecular weight proteins (13,30). Therefore, it was speculated that ORFs of hypothetical unknowns are not genuine ORFs or are barely expressed and available to T cell recognition causing the low sensitivity of IGRAs. Here, we established the functional status of 136 *M. leprae* unique proteins (class VI) to enhance the screening process of novel antigens by using target based qRT-PCR assays (Table 2.2). Of the class VI genes, twenty six of the hypothetical unknown ORFs were selected as being actively expressed with high positive PCR signals (>500 fg of cDNA) (Table 2.3). Only two of the 26 selected ORFs have already been studied or are being evaluated as T cell antigens according to the literature (ML0573 and ML0574) (1,2,13,16,30).

Two initial studies by Dr. Geluk and colleagues (13,16) showed that five of the recombinant hypothetical unknowns (ML0126, ML1420, ML1989, ML2283 and ML2346) and twenty two of their peptides induced the *M. leprae* specific T cell response from PB patients or HHCs in Brazilian population. In order to apply for the clinical use, IGRA validated whether five of the most promising recombinant hypothetical unknowns and their peptides can induce IFN-γ secretion from PB patients or HHCs in various leprosy endemic areas in Brazil, Bangladesh, Nepal, Pakistan and Ethiopia (15). The

ML2283 recombinant protein and ML2283-derived peptides were recognized by 20% and 25% of PB patients and HHCs but only detected by 6 % of TB patients (n=50) suggesting that ML2283 and ML2283-derived peptide induced the most *M. leprae* specific T cell response (15). However, the levels of IFN-γ production to other antigens were mostly found in the range of 100 pg/ml to 200 pg/ml, which are close to cutoff (100 pg/ml),and detect considerable numbers of endemic controls (30 %) and TB patients (20 % to 40 %) due to the low sensitivity of IGRA. All Brazilian subjects exhibited no T cell response to any of peptide antigens and the ML1420-derived peptide induced *M. leprae* specific T cell response only in the Pakistan population (15) suggesting that low sensitivity of IGRA to peptide antigens caused the variable reactivity in racially/geographically different study populations (15).

In the present study, the gene expression level of ML2283 was found to be 500fg which was relatively high compared to the other hypothetical unknowns. The ORFs of ML0126, ML1420, ML1989 and ML2346 which induced variable responses among three populations, were expressed in the lower amount of 100 fg (Table 2.2). The literature and our study showed that the low sensitivity of IGRA to current peptide antigens may be attributable to low antigen availability and poor T cell recognition of the parent proteins by the host immune system.

A number of well known *M. leprae* abundant antigens (major membrane protein-I (MMP-I), MMP-II, GroES, ESAT-6/ML0049 and CFP10/ML0050) induce strong cellular mediated and humoral immune responses in PB and MB patients respectively (20,23,24,31,33,34). Araoz *et al* (1,2) showed that a few of the recombinant hypothetical unknown proteins can be recognized by circulating antibodies in a few sera of MB

patients but also induced the *M. leprae* specific T cell response in PB and HHCs. Surprisingly, our study showed that the majority of the selected antigen candidates were detected by circulating antibodies in sera of LL patients suggesting that these antigens can be processed and accommodated on the Major Histocompatibility Complex (MHC) molecule to present to T-helper cells (Figure 2.2.B). Although all cavitary TB patients presented high bacterial loads, the seroreacivities of these patients were undetectable to most of the antigen candidates by Western blot analysis (Figure 2.2.C).

Recently, several serological studies to abundant antigens (PGL-I, LID-1 and MMP-II) showed that 9-10 % of HHCs exhibited the seropositivities to these antigens and 10 % of individuals with seropositivity developed the clinical leprosy within 6-8 months (5,9,21). Due to low antibody titers, 15-30% of PB patients can be detected by ELISA to the abundant antigens but the level of ELISA (close to cutoff) in these patients causes the high false positivity rates ($\geq 10\%$) (29). Therefore, it is still questionable that a stand-alone serological test is used to detect individuals with asymptomatic leprosy. Recent studies in our laboratory (32) examined the reactivity patterns of patient sera to PGL-I, lipoarabinomann (LAM), and six recombinant proteins of abundant protein antigens (ML1877, ML0841, ML2028, ML2038, ML0380, and ML0050) by Western blot analysis and ELISA. This analysis showed that antibody response patterns highly correlate with a particular disease state (32), suggesting that these serological test lead to better patient management. Since 9 selected antigens induced the M. leprae specific antigens in LL patients, it would be useful to determine the disease states and allow for a more critical assessment of the form of disease within the leprosy spectrum and for understanding of leprosy pathogenesis.

Therefore, these novel recombinant antigens have considerable potential to improve the sensitivity and specificity of CMI-based diagnosis. The CMI response to all of these purified recombinant antigens is now being evaluated against PBMCs of leprosy patients and HHCs in endemic regions. In the future, the identification of *M. leprae*specific peptides from these novel candidates that are restricted to all major HLA-DR types and specifically activate the maximal IFN- γ in PBMCs from PB patients and HHCs will allow improvements in the sensitivity and specificity of IGRA for early diagnosis of leprosy.
2.6 References

- Araoz, R., N. Honore, S. Banu, C. Demangel, Y. Cissoko, C. Arama, M. K. Uddin, S. K. Hadi, M. Monot, S. N. Cho, B. Ji, P. J. Brennan, S. Sow, and S. T. Cole. 2006. Towards an immunodiagnostic test for leprosy. Microbes.Infect. 8:2270-2276.
- Araoz, R., N. Honore, S. Cho, J. P. Kim, S. N. Cho, M. Monot, C. Demangel, P. J. Brennan, and S. T. Cole. 2006. Antigen discovery: a postgenomic approach to leprosy diagnosis. Infect.Immun. 74:175-182.
- 3. Brennan, P. J. 2000. Skin test development in leprosy: progress with first-generation skin test antigens, and an approach to the second generation. Lepr.Rev. 71 Suppl:S50-S54.
- 4. Britton, W. J. and D. N. Lockwood. 2004. Leprosy. Lancet 363:1209-1219.
- Buhrer-Sekula, S., H. L. Smits, G. C. Gussenhoven, L. J. van, S. Amador, T. Fujiwara, P. R. Klatser, and L. Oskam. 2003. Simple and fast lateral flow test for classification of leprosy patients and identification of contacts with high risk of developing leprosy. J.Clin.Microbiol. 41:1991-1995.
- Cho, S. N., D. L. Yanagihara, S. W. Hunter, R. H. Gelber, and P. J. Brennan. 1983. Serological specificity of phenolic glycolipid I from *Mycobacterium leprae* and use in serodiagnosis of leprosy. Infect.Immun. 41:1077-1083.
- Cole, S. T., K. Eiglmeier, J. Parkhill, K. D. James, N. R. Thomson, P. R. Wheeler, N. Honore, T. Garnier, C. Churcher, D. Harris, K. Mungall, D. Basham, D. Brown, T. Chillingworth, R. Connor, R. M. Davies, K. Devlin, S. Duthoy, T. Feltwell, A. Fraser, N. Hamlin, S. Holroyd, T. Hornsby, K. Jagels, C. Lacroix, J. Maclean, S. Moule, L. Murphy, K. Oliver, M. A. Quail, M. A. Rajandream, K. M. Rutherford, S. Rutter, K. Seeger, S. Simon, M. Simmonds, J. Skelton, R. Squares, S. Squares, K. Stevens, K. Taylor, S. Whitehead, J. R. Woodward, and B. G. Barrell. 2001. Massive gene decay in the leprosy bacillus. Nature 409:1007-1011..
- Dockrell, H. M., S. Brahmbhatt, B. D. Robertson, S. Britton, U. Fruth, N. Gebre, M. Hunegnaw, R. Hussain, R. Manandhar, L. Murillo, M. C. Pessolani, P. Roche, J. L. Salgado, E. Sampaio, F. Shahid, J. E. Thole, and D. B. Young. 2000. A postgenomic approach to identification of *Mycobacterium leprae*-specific peptides as T-cell reagents. Infect.Immun. 68:5846-5855.
- Duthie, M. S., W. Goto, G. C. Ireton, S. T. Reece, L. P. Cardoso, C. M. Martelli, M. M. Stefani, M. Nakatani, R. C. de Jesus, E. M. Netto, M. V. Balagon, E. Tan, R. H. Gelber, Y. Maeda, M. Makino, D. Hoft, and S. G. Reed. 2007. Use of protein antigens for early serological diagnosis of leprosy. Clin.Vaccine Immunol. 14:1400-1408.
- Duthie, M. S., M. N. Hay, C. Z. Morales, L. Carter, R. Mohamath, L. Ito, L. K. Oyafuso, M. I. Manini, M. V. Balagon, E. V. Tan, P. R. Saunderson, S. G. Reed, and D. Carter. 2010. Rational design and evaluation of a multiepitope chimeric fusion protein with the potential for leprosy diagnosis. Clin.Vaccine Immunol. 17:298-303.
- Duthie, M. S., G. C. Ireton, G. V. Kanaujia, W. Goto, H. Liang, A. Bhatia, J. M. Busceti, M. MacDonald, K. D. Neupane, C. Ranjit, B. R. Sapkota, M. Balagon, J. Esfandiari, D. Carter, and S. G. Reed. 2008. Selection of antigens and development of prototype tests for point-of-care leprosy diagnosis. Clin. Vaccine Immunol. 15:1590-1597.
- 12. Eddy, S. R. 2008. A probabilistic model of local sequence alignment that simplifies statistical significance estimation. PLoS.Comput.Biol. 4:e1000069..
- Geluk, A., M. R. Klein, K. L. Franken, K. E. van Meijgaarden, B. Wieles, K. C. Pereira, S. Buhrer-Sekula, P. R. Klatser, P. J. Brennan, J. S. Spencer, D. L. Williams, M. C. Pessolani, E. P. Sampaio, and T. H. Ottenhoff. 2005. Postgenomic approach to identify novel *Mycobacterium leprae* antigens with potential to improve immunodiagnosis of infection. Infect.Immun. 73:5636-5644.
- 14. Geluk, A. and T. H. Ottenhoff. 2006. HLA and leprosy in the pre and postgenomic eras. Hum.Immunol. 67:439-445.

- 15. Geluk, A., J. S. Spencer, K. Bobosha, M. C. Pessolani, G. M. Pereira, S. Banu, N. Honore, S. T. Reece, M. MacDonald, B. R. Sapkota, C. Ranjit, K. L. Franken, M. Zewdie, A. Aseffa, R. Hussain, M. M. Stefani, S. N. Cho, L. Oskam, P. J. Brennan, and H. M. Dockrell. 2009. From genome-based *in silico* predictions to *ex vivo* verification of leprosy diagnosis. Clin.Vaccine Immunol. 16:352-359.
- Geluk, A., J. van der Ploeg, R. O. Teles, K. L. Franken, C. Prins, J. W. Drijfhout, E. N. Sarno, E. P. Sampaio, and T. H. Ottenhoff. 2008. Rational combination of peptides derived from different Mycobacterium leprae proteins improves sensitivity for immunodiagnosis of *M. leprae* infection. Clin.Vaccine Immunol. 15:522-533.
- Geluk, A., van der Ploeg-van Schip JJ, K. E. van Meijgaarden, S. Commandeur, J. W. Drijfhout, W. E. Benckhuijsen, K. L. Franken, B. Naafs, and T. H. Ottenhoff. 2010. Enhancing sensitivity of detection of immune responses to *Mycobacterium leprae* peptides in whole-blood assays. Clin.Vaccine Immunol. 17:993-1004.
- Geluk, A., K. E. van Meijgaarden, K. L. Franken, Y. W. Subronto, B. Wieles, S. M. Arend, E. P. Sampaio, B. T. de, W. R. Faber, B. Naafs, and T. H. Ottenhoff. 2002. Identification and characterization of the ESAT-6 homologue of *Mycobacterium leprae* and T-cell cross-reactivity with *Mycobacterium tuberculosis*. Infect.Immun. 70:2544-2548.
- Geluk, A., K. E. van Meijgaarden, K. L. Franken, B. Wieles, S. M. Arend, W. R. Faber, B. Naafs, and T. H. Ottenhoff. 2004. Immunological crossreactivity of the *Mycobacterium leprae* CFP-10 with its homologue in *Mycobacterium tuberculosis*. Scand.J.Immunol. 59:66-70. doi:1358 [pii].
- Hussain, R., F. Shahid, S. Zafar, M. Dojki, and H. M. Dockrell. 2004. Immune profiling of leprosy and tuberculosis patients to 15-mer peptides of *Mycobacterium leprae* and *M. tuberculosis* GroES in a BCG vaccinated area: implications for development of vaccine and diagnostic reagents. Immunology 111:462-471.
- Kai, M., N. H. Nguyen Phuc, T. H. Hoang Thi, A. H. Nguyen, Y. Fukutomi, Y. Maeda, Y. Miyamoto, T. Mukai, T. Fujiwara, T. T. Nguyen, and M. Makino. 2008. Serological diagnosis of leprosy in patients in vietnam by enzyme-linked immunosorbent assay with *Mycobacterium leprae*-derived major membrane protein II. Clin.Vaccine Immunol. 15:1755-1759.
- 22. Leproma. http://genolist.pasteur.fr/Leproma/
- Maeda, Y., T. Mukai, M. Kai, Y. Fukutomi, H. Nomaguchi, C. Abe, K. Kobayashi, S. Kitada, R. Maekura, I. Yano, N. Ishii, T. Mori, and M. Makino. 2007. Evaluation of major membrane protein-II as a tool for serodiagnosis of leprosy. FEMS Microbiol.Lett. 272:202-205.
- 24. Makino, M., Y. Maeda, and N. Ishii. 2005. Immunostimulatory activity of major membrane protein-II from *Mycobacterium leprae*. Cell Immunol. **233**:53-60.
- 25. Malinen, E., A. Kassinen, T. Rinttila, and A. Palva. 2003. Comparison of real-time PCR with SYBR Green I or 5'-nuclease assays and dot-blot hybridization with rDNA-targeted oligonucleotide probes in quantification of selected faecal bacteria. Microbiology 149:269-277.
- 26. Manandhar, R., J. W. LeMaster, C. R. Butlin, P. J. Brennan, and P. W. Roche. 2000. Interferon-gamma responses to candidate leprosy skin-test reagents detect exposure to leprosy in an endemic population. Int.J.Lepr.Other Mycobact.Dis. **68**:40-48.
- 27. Newberg, L. A. 2008. Significance of gapped sequence alignments. J.Comput.Biol. 15:1187-1194.
- Ridley, D. S. and W. H. Jopling. 1966. Classification of leprosy according to immunity. A fivegroup system. Int.J.Lepr.Other Mycobact.Dis. 34:255-273.
- 29. Sekar, B. 2007. Recent advances in immunodiagnosis of leprosy. Indian J.Lepr. 79:85-106.
- Spencer, J. S., H. M. Dockrell, H. J. Kim, M. A. Marques, D. L. Williams, M. V. Martins, M. L. Martins, M. C. Lima, E. N. Sarno, G. M. Pereira, H. Matos, L. S. Fonseca, E. P. Sampaio, T. H. Ottenhoff, A. Geluk, S. N. Cho, N. G. Stoker, S. T. Cole, P. J. Brennan, and M. C. Pessolani. 2005. Identification of specific proteins and peptides in *Mycobacterium leprae* suitable for the selective diagnosis of leprosy. J.Immunol. 175:7930-7938.

- Spencer, J. S., H. J. Kim, A. M. Marques, M. Gonzalez-Juarerro, M. C. Lima, V. D. Vissa, R. W. Truman, M. L. Gennaro, S. N. Cho, S. T. Cole, and P. J. Brennan. 2004. Comparative analysis of B- and T-cell epitopes of *Mycobacterium leprae* and *Mycobacterium tuberculosis* culture filtrate protein 10. Infect.Immun. 72:3161-3170.
- 32. Spencer, J. S., H. J. Kim, W. H. Wheat, D. Chatterjee, M. V. Balagon, R. V. Cellona, E. V. Tan, R. Gelber, P. Saunderson, M. S. Duthie, S. T. Reece, W. Burman, R. Belknap, W. R. Mac Kenzie, A. Geluk, L. Oskam, H. M. Dockrell, and P. J. Brennan. 2011. Analysis of Antibody Responses to *Mycobacterium leprae* Phenolic Glycolipid I, Lipoarabinomannan, and Recombinant Proteins To Define Disease Subtype-Specific Antigenic Profiles in Leprosy. Clin.Vaccine Immunol. 18:260-267.
- Spencer, J. S., M. A. Marques, M. C. Lima, A. P. Junqueira-Kipnis, B. C. Gregory, R. W. Truman, and P. J. Brennan. 2002. Antigenic specificity of the *Mycobacterium leprae* homologue of ESAT-6. Infect.Immun. 70:1010-1013.
- Triccas, J. A., P. W. Roche, N. Winter, C. G. Feng, C. R. Butlin, and W. J. Britton. 1996. A 35-kilodalton protein is a major target of the human immune response to *Mycobacterium leprae*. Infect.Immun. 64:5171-5177.
- 35. Tuberculolist. http://genolist.pasteur.fr/TubercuList/
- 36. Vissa, V. D. and P. J. Brennan. 2001. The genome of *Mycobacterium leprae*: a minimal mycobacterial gene set. Genome Biol. 2:REVIEWS1023.
- Weir, R. E., P. J. Brennan, C. R. Butlin, and H. M. Dockrell. 1999. Use of a whole blood assay to evaluate *in vitro* T cell responses to new leprosy skin test antigens in leprosy patients and healthy subjects. Clin.Exp.Immunol. 116:263-269.
- Weir, R. E., P. E. Fine, S. Floyd, S. Stenson, C. Stanley, K. Branson, W. J. Britton, K. Huygen, M. Singh, G. Black, and H. M. Dockrell. 2008. Comparison of IFN-gamma responses to mycobacterial antigens as markers of response to BCG vaccination. Tuberculosis.(Edinb.) 88:31-38.
- 39. WHO. http://www.who.int/lep/diagnosis/en/index.html. 2010.
- Williams, D. L., S. Oby-Robinson, T. L. Pittman, and D. M. Scollard. 2003. Purification of *Mycobacterium leprae* RNA for gene expression analysis from leprosy biopsy specimens. Biotechniques 35:534-1.
- Williams, D. L., R. A. Slayden, A. Amin, A. N. Martinez, T. L. Pittman, A. Mira, A. Mitra, V. Nagaraja, N. E. Morrison, M. Moraes, and T. P. Gillis. 2009. Implications of high level pseudogene transcription in *Mycobacterium leprae*. BMC.Genomics 10:397.
- 42. Williams, D. L., M. Torrero, P. R. Wheeler, R. W. Truman, M. Yoder, N. Morrison, W. R. Bishai, and T. P. Gillis. 2004. Biological implications of *Mycobacterium leprae* gene expression during infection. J.Mol.Microbiol.Biotechnol. **8**:58-72.

CHAPTER 3 EVALUATION OF NOVEL ANTIGENS FROM *MYCOBACTERIUM LEPRAE* CYTOSOLIC MEMBRANE FOR THE EARLY DIAGNOSIS OF LEPROSY

From a manuscript form in preparation by **Hee Jin Kim**, Chaman Ranjit, Alan R. Schenkel and Patrick J. Brennan. Department of Microbiology, Immunology and Pathology, Colorado State University, Fort Collins, Colorado.

3.1 Abstract

The great needs towards successful control of leprosy are tools to detect *Mycobacterium leprae* infected individuals. The assessment of cellular mediated immune (CMI) responses to *M. leprae* specific antigens is thought to be the most promising diagnostic platform for the early detection of leprosy. Until now, two classes of novel antigens have been applied to CMI based diagnosis of leprosy: *M. leprae* subcellular fractions, MLSA-LAM and MLCwA, generated by proteomic approach; and protein/peptide antigens, selected by comparative genomic and bioinformatic analysis. However, CMI tests based on protein antigens are still problematic in detection of *M. leprae* infection/exposure in endemic regions due to low sensitivity to individual recombinant protein/peptide antigens and low specificity to *M. leprae* native protein antigens.

M. leprae membrane antigen (MLMA) induces a strong CMI response in tuberculoid leprosy patients. However, the large amount of lipomannans (LMs) and

lipoarabinomannans (LAMs) hinders the further processing of this fraction and the optimal use of this fraction as a diagnostic reagent in CMI based assays. Here, we developed a simple sodium carbonate treatment and successfully enriched membrane proteins into an alkali soluble membrane protein fraction (MLMA-SP) excluding the most hydrophobic lipids. Subsequent *in vitro* IFN- γ release assays (IGRA) of peripheral blood monocyte cells (PBMCs) from 16 mycobacterial laboratory/leprosy clinic workers revealed that MLMA-SP induced higher levels of IFN- γ secretion in the group exposed to leprosy patients, as compared to the group with environmental/occupational mycobacterial exposure. Therefore, MLMA-SP may contain immunogens to allow improvements in both sensitivity and specificity for the early diagnosis of leprosy.

3.2 Introduction

Since the 1980's, our laboratories have examined the native *M. leprae* proteins from armadillo derived *M. leprae* to encourage antigen discovery for diagnosis of leprosy (27,34,35,43-45). Several major antigens had been identified from subcellular compartments and their immunological potentials were evaluated toward the development of immune based diagnosis, particularly CMI-based diagnostic tests (see section 1.10). Two promising *M. leprae*-native-protein-rich antigenic fractions had been translated into clinical products, the so called "bench to bedside research". MLSA-LAM and MLCwA (5) have just completed Phase II clinical trials as delayed type hypersensitivity skin test antigens to be used for diagnosis of human leprosy in routine clinical practice. However, major obstacles remain in the identification of *M. leprae* native antigens which are: 1) the inability to cultivate *M. leprae in vitro* makes it extremely difficult to produce sufficient quantities of refined reagents; and 2) the sharing of *M. leprae* antigens among other mycobacteria. These problems have been overcome using postgenomic approaches to generate either recombinant proteins or peptides derived from *M. leprae* specific hypothetical unknowns (previously described in section 1.10.2). *In vitro* IFN- γ release assays (IGRAs) to these new antigens can diagnose individuals exposed to/or infected by *M. leprae* (2,3,17-19,21,51). However, the low level of IFN- γ production to peptide antigens make difficult to detect the *M. leprae* infected individuals in geographically/racially different endemic populations (20). Therefore, it is necessary to continue the identification of *M. leprae* specific antigens to greatly enhance sensitivity and specificity of IGRA for early diagnosis.

Either whole *M. leprae* or *M. leprae* whole cell lysates in infected dendritic cells suppressed surface expression of the antigen presenting molecules (MHC-I and II) and co-stimulatory molecules (CD80, CD83) as well as IL-12 and TNF- α (29,39). However, MLMA induced the expression of MHC-II, CD86 and CD83 on dendritic cells and stimulated IL-12 p70 production from dendritic cells (29,30). The MLMA also induced IFN- γ production from naïve and memory CD4⁺ T cells and CD8⁺ T cells by stimulating dendritic cells (29). These data suggest that MLMA contains highly immunogenic molecules which could enhance the sensitivity of current IGRA formatted diagnosis by efficiently activating antigen presenting cells and being presented to T cells.

Major concerns in studying MLMA as diagnostic reagents are that most *M. leprae* proteins share high sequence similarity with orthologues in other mycobacteria which can

raise undesirable cross-reactivity due to infection with other mycobacteria (8,9,36,50). However, the massive gene decay and reduction of *M. leprae* could create a distinct native membraneous protein profile which could achieve different antigenic profiles from those of other mycobacteria. Earlier studies in our laboratories also showed that the proteome of MLMA by gel and MS analyses appeared to be unique (25,27,35). MLMA was predominated by two major membrane protein I (MMP-I) and MMP-II (27). Since MMP-I and MMP-II were absent or of low abundance in other mycobacteria, these two antigens had already been shown to induce *M. leprae*-specific T cell response in paucibacillary/tuberculoid leprosy (PB/TT) patients and humoral response in multibacillary/lepromatous leprosy (MB/LL) patients (11,30,32,41,47,52). However, MLMA also harbors large quantities of lipomannans (LMs) and lipoarabinomannans (LAMs) which cause immune modulation and cross-reactivity with other mycobacteria (5,29). CMI assays based on the proteins from MLMA have never been fully explored.

The aim of our work was to develop and evaluate the immunogenic proteins from MLMA which could improve the sensitivity and specificity of CMI based assay to detect individuals with asymptomatic leprosy in endemic regions. In this study, we developed the procedure to enrich the membrane associated proteins from MLMA and evaluated the resultant MLMA-SP by IGRA in individuals professionally exposed to leprosy patients.

3.3 Materials and methods

3.3.1 Isolation of M. leprae and preparation of M. leprae membrane antigen (MLMA)

M. leprae Thai-53 was isolated from armadillo spleens and livers by the 'Draper protocol' (48) with modifications. Briefly, the *M. leprae* infected armadillo tissues (25g),

invariably spleen or liver, provided by Dr. Truman from Louisiana State University, were homogenized in a sterilized chamber with 75 ml of homogenization buffer (150 mM EDTA, pH7.2) at 4 °C for 3 min by using the Omni-mixer (Omni International, Kennesaw, GA). The homogenate was then centrifuged at $5,000 \times \text{g}$ for 5 min at 4 °C to remove residual non-homogenized tissue. The pellet was rehomogenized with 50 ml of homogenization buffer for 3 min. The resulting supernatants from both homogenization steps were combined and then centrifuged at $10,000 \times g$ for 10 min at 4 °C. The pellet was resuspended with 200 ml of an extraction buffer (0.1M NaOH, 10 mM EDTA) to extract out residual tissue pigments, incubated with stirring for 2 hr at 4°C and centrifuged at $10,000 \times g$ for 30 min at 4 °C. The bacteria-tissue pellet was washed, resuspended with 150 ml of buffered water (0.1 M Tris-HCl, pH 7.2, 0.1% Tween80) and digested for 3 hr in collagenase (20 mg per gram of infected tissue; Sigma, St. Louis, MO) at 37 °C. Subsequent digestion was performed with trypsin and alpha-chymotrypsin (20mg per gram of infected tissue; Sigma, St. Louis, MO) at 37 °C for 2 hr. The bacterial pellet was collected by centrifugation at $10,000 \times g$ for 30 min at 4 °C, resuspended in PEG/DEX buffer (6% polyethylene glycol (PEG), 8% dextran, 10 mM KH₂PO₄, 10mM NaCl) and mixed by vortexing. The resulting PEG/DEX-bacteria mixture was then placed into a 500-ml separatory funnel and subjected to 2-phase partition over approximately 1 hr at 25 °C. The residual tissue-pigments were in the lower dextran phase, while the M. leprae bacilli were in the upper PEG phase. The resulting PEGbacteria solution was then centrifuged at $27,000 \times g$ for 30 min at 4°C. The bacteria pellet was washed and resuspended with buffered water. The quantity of M. leprae was evaluated by reading an O.D. at 540nm corresponding to 0.365 mg/ml of bacteria (49).

The bacteria (200 mg, obtained from approximately 116 g of *M. leprae* infected armadillo tissues) were washed twice with 5 ml of lysis buffer (50 mM Tris- buffer pH 7.4, 10 mM magnesium chloride) by centrifugation at 3,000 × g for 10 min. The cells were resuspended in the lysis buffer with a protease inhibitor cocktail (P8340; Sigma, St. Louis, MO), then disrupted by intermittent probe sonication (MSE Sonic prep150, MSE-Sanyo, Palisades Park, NJ) for 30 cycles (60 sec bursts/90 sec of cooling on ice). The whole cell sonicate was digested with 10 μ g/ml of DNase and 10 μ g/ml of RNase for 30 min at 4 °C and centrifuged at 27,000 × g for 30 min. The supernatant was recentrifuged at 100,000 × g for 3 hr to provide the insoluble pellet fraction, the MLMA (34).

3.3.2 Protein enrichment of the MLMA

The MLMA was resuspended to a protein concentration of 1mg/ml with ice-cold 100 mM Na₂CO₃, pH 11.4 (16). The resulting suspension was stirred for 30 min at 4 °C and then centrifuged at 100,000 × g (3 hr, 4 °C). The released proteins, which are mostly basic peripheral membrane proteins (16), were found in the supernatant (MLMA-SP). The alkali insoluble proteins of MLMA (MLMA-InsP) in the pellet were resuspended with lysis buffer. Both MLMA-SP and MLMA-InsP were dialyzed against 10 mM ammonium bicarbonate and lyophilized and kept at -70 °C prior to use for T cell simulation assays. The LPS content of all antigens as measured by the Limulus amoebocyte lysate test (BioWhittaker, Walkersville, MD) was less than 2 ng/mg of protein. For protein analysis, the MLMA-InsP was subjected to ice-cold 1% Triton X-114 solubilization and then to ultracentrifugation (100,000 × g, for 3 hr). The pellet (P) fraction called MLMA-P was utilized for the identification of CD1a restricted antigens

(see Chapter 4). The resulting supernatant was applied to phase separation and generated the aqueous phase and detergent phase; the detergent phase contained the integral plasma membrane proteins and lipoproteins (4). Figure 3.1 summarizes the experimental procedure.



Figure 3.1 Experimental scheme for the fractionation of *M. leprae* membrane antigens (MLMA).

Both resulting aqueous and detergent phases, generated from Triton X-114 phase separation, were subjected to cold acetone precipitation to remove the detergents for gel analysis. Protein quantities of all subcellular fractions were measured by the bicinchoninic acid assay reagent. All fractions were analyzed by SDS-PAGE. A periodic acid oxidation and silver staining step visualized PIMs, LMs and LAMs, and the protein components, respectively (27). Identical gels were transferred onto nitrocellulose membranes for Western blot analyses by the procedure previously described in section 4.2.7. Briefly, blots were blocked and then probed with monoclonal antibody CS38 to detect MMP-I and mouse polyclonal anti-MMP-II antibody to detect MMP-II. Both antibodies were provided by the NIH Leprosy Contract, CSU. Blots were probed by secondary anti-mouse IgG alkaline phosphatase and developed by 5-bromo-4-chloro-3-indolyl phosphate/nitro blue tretrazolium (Sigma, St. Louis, MO).

3.3.3 T cell stimulation assay

3.3.3.1 Study subjects

Sixteen healthy volunteers were recruited at the Mycobacterial Research Laboratories in CSU. All volunteers were employees or collaborators in the laboratories and many of these individuals had worked extensively with mycobacteria and/or cared for leprosy patients. A questionnaire was used to assess the exposure level of all volunteers to nonviable laboratory strain of mycobacteria (*M. smegmatis*, *M. bovis* BCG, *M. tuberculosis* H37Rv) or *M. leprae*. Informed consent forms had been authorized by all volunteers prior to blood donation. All tests and procedure were approved by the Institutional Review Board Committee at CSU.

3.3.3.2 T cell stimulation assay with MLMA derived antigens and IFN-γ Enzyme-Linked ImmunoSorbent Assay (ELISA).

Twenty milliliter of blood from each subject was drawn by venipuncture, heparinized and PBMCs were isolated by Ficoll (GE Healthcare, Pittsburgh, PA) density centrifugation according to Spencer *et al.* (51). PBMCs were resuspended in AIM-V

medium (Invitrogen Life Technologies, Carlsbad, CA) supplemented with 100 U/ml penicillin, 100 µg/ml streptomycin, and 2 mM L-glutamine (Invitrogen Life Technologies, Carlsbad, CA). PBMCs were plated at 2×10^6 cells/ml in 96 well round bottom plates (Costar Corporation, Cambridge, MA) in triplicate and stimulated with each of the membrane subfractions (2 µg/ml such as MLMA, MLMA-SP, MLMA-InsP), or mitogen PHA (0.5 µg/ml; Sigma, St. Louis, MO), or PPD (1 µg/ml; Mycos Research, Fort Collins, CO), or media alone as negative controls. After five days of culture at 37°C in a humidified 5% CO₂ atmosphere, supernatants were collected and stored immediately at -70 °C until further analysis. The supernatants were tested for IFN-γ levels by using a IFN-γ ELISA Kit (UCytech, Utrecht, the Netherlands). All values of released IFN-γ on stimulation were subtracted by the value of IFN-γ released by media alone negative control.

3.3.3.3 PGL-I rapid lateral flow test

The PGL-I Rapid Antibody Test kits (Standard Diagnostics Company, Kyonggido, Korea) based on synthetic ND-O-BSA were provided by the Leprosy Contract, CSU (courtesy of Dr. Sang Nae Cho, Yonsei University, Seoul, Korea) and were utilized to detect circulating anti-PGL-I antibody in the sera of all participants. After the isolation of PBMCs, 5 μ l of plasma were diluted with 45 μ l of the running buffer provided in the kit and a total 50 μ l was applied onto the window of the kit. The test was read after 10 min. Positivity was determined by appearance of both test line and control line while negativity was the appearance of the single control line (7).

3.3.3.4 Statistical analysis

Differences in IFN-γ levels between antigens tested and test groups were analyzed with the one-tailed Turkey-Kramer test by JMP 8.0 Statistical Discovery Software (SAS Institute Inc., Cary, NC).

3.4 Results

3.4.1 Subfractionation of MLMA by sodium carbonate treatment

In the identification of native protein antigens, chromatographic approaches have been used to prefractionate the native protein pools and to enrich certain proteins in isolated fractions (15). However, this proteomic approach as applied to membrane fractions is difficult due to complications with both extraction and solubilization of amalgamated hydrophobic proteins (15) and it becomes uneconomical when using *M*. *leprae* membrane proteins due to the paucity of membrane fraction, i.e. MLMA versus MLCwA and MLSA. In the case of *M. leprae*, the membrane fractions also contain large amounts of LAMs and LMs rendering this task particularly difficult. Therefore this study focused on a simple biochemical approach accompanied by a cellular based assay to indentify immunodominant proteins in *M. leprae* membranes.

The sodium carbonate treatment converts the closed vesicles of membrane lipids to open membrane sheets, and peripheral proteins are released in soluble form (16). This simple method combined with ultracentrifugation successfully released the alkali soluble peripheral membrane proteins with a great reduction of LAMs, LMs and PIMs from MLMA (Figure 3.2.A and B).



Figure 3.2 SDS-PAGE and Western blot analysis of subfractionated proteins of MLMA. A) Protein pattern of MLMA derived fractions was visualized by silver staining. B) LMs, LAMs, PIMs in identical gels were revealed as diffuse bands with average masses of 30, 20 and 8 kDa respectively by sliver staining combined with sodium periodate. C) Western blot pattern obtained with CS-38 (anti-MMP-I monoclonal antibody). D) Western blot pattern obtained with anti MMP-II polyclonal antibody; 5 µg of each protein fraction was run in each lane. Lane 1 is MLMA; lane 2, MLMA-SP; lane 3, MLMA-P; lane 4, the TritonX-114 detergent phase proteins from MLMA-InsP; lane 5, all of the Triton X-114 aqueous phase proteins from MLMA-InsP.

After dialysis of the alkali soluble protein in the supernatant, the MLMA-SP contained most of the membrane proteins (Figure 3.2). The two dominant major membrane proteins, MMP-I (ML0841, 35 kDa) and MMP-II (ML2038, bacterioferritin, 18kDa), were found in MLMA-SP by gel and Western blot analyses. Subsequently, Triton X-114 separation of the alkali insoluble proteins (MLMA-InsP) showed that the integral plasma membrane proteins were enriched into the Triton X-114 detergent phase along with most of the LAM, LM, and PIMs in membrane subcellular fractions (Figure

3.2.A and B). The integral membrane proteins from MLMA made up approximately 4% of the total MLMA but were masked by the enormous amount of lipoglycans (LMs and LAMs). The Triton X-114 aqueous phase was devoid of proteins according to protein quantification assay (bicinchoninic acid protein assay). Since MMP-I and MMP-II were proven to be peripheral proteins, Western blot analysis showed that Triton X-114 detergent phase contained a minimal amount of MMP-I and no MMP-II (Figure 3.2.C and D).

3.4.2 Classification of the study groups

In order to evaluate the immunogenicity of the new *M. leprae* subfractions, MLMA-SP and MLMA-InsP were evaluated by measuring IFN-γ production of PBMCs from a group of healthy laboratory workers some of whom may have or have had a history of exposure to leprosy/tuberculosis patients or the respective bacteria. This experiment is a presage for future applications to populations in leprosy endemic regions and the only type of experiment achieved within our own laboratory setting. Also the levels of exposure to *M. leprae* or other mycobacteria was determined as a result of the PGL-I rapid lateral flow test (also known as ML flow test; see section 1.10.4) and previous TB skin test or BCG vaccination and questionnaires (Table 3.1) regarding periods of working experience with leprosy patients.

| | NonExp group (4) | labMycoExp group (7) | LepExp group (5) |
|--|-----------------------|-----------------------|--------------------------|
| Duration of Exposure to Leprosy Patient | N/A ^a | N/A | one-15years ^b |
| Duration of Exposure To Native Mycobacterial Products ^c | Less than 4 months | 3 years- 40 years | N/A |
| Positivity of TB skin test | 0/4 | 6/7 | 2/5 |
| No. BCG Vaccination | 0 | one or 3 ^d | one |
| Endemic Site born/lived | 0/4 | 6/7 | 5/5 |
| Positivity to PGL-I lateral flow test | 0/4 | 0/7 | 5/5 |

Table 3.1 Characteristics of participating study population.

a; Not applicable. b; All individuals had direct contact with leprosy patients and were involved in sampling the blood/or skin slit smears. Two individuals physically examined leprosy patients on a regular basis. Average contact was one or two active leprosy cases per week. c; The mycobacterial products were native antigens from *M. tuberculosis* H37Rv, *M. bovis* BCG. Two of seven individuals had additionally worked with *M. leprae.* d; Most of the individuals had received a single dose of the BCG vaccine but one individual had 3 doses of the BCG vaccine at different times.

The participants in this study were categorized into 3 groups (Table 3.1): 1) Nonexposed group (NonExp group, n= 4), i.e. those who were born in the U.S.A., were both TB skin test and PGL-I test negative, and had worked for less than 4 months in a mycobacterial laboratory; 2) Laboratory nonviable mycobacteria exposed group (labMycoExp group, n= 7), i.e. those who were born in leprosy/tuberculosis endemic countries and/or had experience in handling nonviable laboratory mycobacteria or *M*. *leprae* in mycobacterial laboratories for 3-40 years. Six of seven individuals in this group were TB skin test positive but PGL-I test negative; and 3) Leprosy patient exposed group (LepExp group, n=5), i.e. those who were PGL-I test positive, born in leprosy endemic countries/regions and had extensively worked with leprosy patients but not significantly with TB patients. However, three of individuals in this group were TB skin test negative (Table 3.1). Seroreactivity of those individuals to the PGL-I test appeared to be a good correlation with exposure time to leprosy patients.

3.4.3 Production of IFN- γ by PBMCs in response to MLMA derived antigens

PBMCs from the NonExp group did not produce IFN- γ in response to any of the membrane derived antigens (MLMA, MLMA-SP and MLMA-InsP) or PPD, but responded well to the positive mitogen control (PHA) (Figures 3.3 and 3.4). In the NonExp group, MLMA-SP induced a low level of IFN- γ production (mean=112.1 pg/ml), whereas MLMA-InsP induced the highest level (mean=358.2 pg/ml) among the four test antigens. The low level of IFN- γ production in non-exposed individuals tends to confirm that both MLMA-SP and MLMA show little or no response to an immunocompetent population selected for minimal mycobacterial exposure (Figure 3.4).

The MLMA and the two subfractions showed differences in their ability to induce IFN- γ secretion between the labMycoExp and LepExp groups. The LepExp group showed the highest responses to MLMA and MLMA-SP (Figures 3.3 and 3.4). Individuals in the labMycExp group born in endemic regions exhibited higher CMI responses to all tested antigens as compared to the NonExp group due perhaps to low but continuous exposure to other mycobacterial antigens. One individual in the labMycoExp group out of 7 individuals received three doses of BCG vaccine. This individual consistently had higher levels of IFN- γ production to all antigens (PPD, MLMA and MLMA-P) except MLMA-SP.



Figure 3.3 IFN- γ **production by PBMCs induced in response to PHA and PPD.** PPD (1µg/ml) and PHA (0.5 µg/ml) stimulated the PBMCs from all experimental groups which were to NonExp group (n=4), the labMycoExp group (n=7) and the LepExp group (n=5). The levels of released IFN- γ in cultured supernants of PBMCs were measured by IFN- γ ELISA. All values were corrected by subtraction of mean negative control values. The vertical bars indicate 0.95 confidence intervals. One way Turkey-Kramer tests were assessed for significance at p=0.05 (JMP 8.0 Statistical Discovery Software). The right side of graphical presentation in each graph shows the statistical difference between groups. The less overlap between circles indicates significant differences.



Figure 3.4 IFN- γ production by PBMCs induced in response to three of the MLMA derived subfractions. MLMA, MLMA-SP and MLMA-InsP (2µg/ml) were used to stimulate the PBMCs from the same three test groups for Figure 3.2. Analysis and evaluation are described in Figure 3.3.

Variance in MLMA-SP responses was in part attributed to the reduced immune response of three individuals, all of whom were exposed to leprosy patients with positivity of PGL-I lateral flow test but negativity of TB skin test. Also these three individuals showed relatively low levels of IFN-γ production in response to PPD which induced high level of IFN-γ production from all individuals in the labMycoExp group and other members of the LepExp group. However, MLMA-SP compared to either MLMA-InsP or PPD was capable of inducing relatively higher levels of IFN-γ production from three of these low responders (Figures 3.3 and 3.4).

3.5 Discussion

The aims of this work were: 1) to enrich the immunodominant proteins from the native cytosolic membrane fraction (MLMA) of *M. leprae* to be used eventually, presumably as recombinant proteins, for CMI based assays; and 2) to evaluate this fraction of *M. leprae* to detect individuals exposed to/ infected by *M. leprae* prior to development of clinical symptoms of leprosy, as a prelude to application for CMI based test for *M. leprae* infection in leprosy endemic regions.

Mycobacterial membrane fractions contain substantial amounts of LMs and LAMs which are exceptionally abundant in the membrane fractions of *M. leprae*, as well as the more conventional phospholipids (6,26,34). LMs and LAMs, immune modulators, induce proinflammatory cytokines and eventually promote the differentiation of naïve $CD4^+$ T cells into Th1 IFN- γ producing cells via TLR-2 or/and TLR-4 signaling (28,40). Additionally, ManLAMs from LAMs suppress immune response to mycobacteria (40). These heterogeneous glycolipids, LMs LAMs and PIMs, are structurally very similar across *Mycobacterium* spp. (40). In mycobacterial infections, IFN- γ is a potent proinflammatory mediator often induced in the course of the CMI response (18,31). IFN- γ secretion of PBMCs has been utilized as a readout of CMI based assays to mycobacterial antigens as a consequence of mycobaterial infection or exposure (10,38,42). Therefore, LM and LAMs are primarily responsible for cross-reactivity with related mycobacteria when applied in immune based diagnostic assay formats.

In an earlier study of our laboratory, a single Triton X-114 phase separation of the *M. leprae* subcellular membrane fraction dominantly enriched two major membrane proteins (MMP-I and MMP-II) and partially excluded LMs, LAMs and PIMs (27). However, the resulting membrane proteins in the aqueous phase were recovered with a great loss of protein quantities due to the precipitation of most proteins and subsequent mandatory detergent removal procedures (27). Here, we showed for the first time that simple treatment with an alkaline sodium carbonate solution successfully excluded most of the LM, LAM, and PIM content from the highly hydrophobic integral and peripheral protein antigens in MLMA without use of detergent and enriched a majority of membrane proteins typically obscured by enormous amount of LMs and LAMs (Figure 3.2). The protein recovery rate of the MLMA-SP was sufficient to pursue further immunological studies using a CMI based assay format.

Previously, *M. leprae* cell wall and soluble cytosolic fractions, MLSA-LAM and MLCwA, were shown to induce slightly higher levels of IFN- γ production from PBMCs of house hold contacts (HHCs) and TT patients but less cross-reactivity to PPD-positive healthy individuals, as compared to *M. leprae* crude lysates (12,33,53). The health-care workers and HHCs developed strong Th1 responses to the *M. leprae* derived antigens

presumably by continuous contacts with MB and PB patients (17,21,47,51). Levels of IFN- γ production to the mixed proteins antigen, MLSA or MLCwA, were quite varied from one IU/ ml to near 500 IU/ml in HHCs though they could discriminate well between leprosy patients and endemic healthy controls (33,53). The MLMA-SP, similar to these mixed protein antigens, produced relatively higher amounts of IFN- γ from the PBMCs of individuals in the LepExp group than MLMA, MLMA-InsP or PPD (Figures 3.3 and 3.4) while MLMA-InsP generated similar levels of IFN- γ from the PBMCs of individuals in both labMycoExp and LepExp groups. Therefore, the results suggest that the enhanced reduction of LMs and LAMs in MLMA-SP can improve the specificity of IGRA to detect exposure of leprosy.

Recently, several studies revealed that a few of HHCs were positive by the PGL-I rapid lateral flow test (13,22,23). In this present study, all individuals in the LepExp group were seropositive in the PGL-I rapid lateral flow test depending on their exposure time to leprosy patients (Table 3.1). Three individuals in this group showed low levels of IFN- γ response to MLMA-SP and even PPD (MLMA-InsP and PPD). Some individuals exposed to/infected by *M. leprae* induce no/low CMI response to crude mycobacterial antigens (PPD) or *M. leprae* antigens (33). Recent studies showed that host genetic polymorphisms in different components of the type 1 cytokine pathway failed to induce CMI responses and affect human resistance to mycobacterial diseases including leprosy (1,24,46) (described in section 1.9). It is possible that the low Th1 immunity of the three individuals in the LepExp group to mycobacterial antigens (PPD and MLMA-InsP) is associated with genetic susceptibility to leprosy (Figures 3.3 and 3.4). However, MLMA-SP caused relatively higher amounts of IFN- γ production than either MLMA-

InsP or PPD even in these same low T cell responders, implying that the MLMA-SP might contain the immunogenic molecules necessary to develop Th1 immune response.

Taken together, a single sodium carbonate treatment successfully enriched most of the membrane associated proteins from crude MLMA and simultaneously excluded most of the LAMs and LMs (Figure 3.2). The immune response to the resultant MLMA-SP seemingly discriminated individuals with leprosy exposure from those with other forms of mycobacterial exposure, implying that the MLMA-SP subfraction contains more *M. leprae* specific immunogenic antigens than MLMA or MLMA-InsP. Although, this initial work is promising, further evaluation of the MLMA-SP in TT patients and HHCs of MB patients is required prior to application in leprosy endemic clinical settings.

In the development of CMI based diagnostic tests using *M. leprae* specific antigens, the ultimate goal is to detect *M.leprae* infected individuals with asymptomatic leprosy as the biological reservoirs and to support the decision for early treatment of those affected individuals. CMI based IGRAs in combination with highly selected proteins or peptides were shown to identify the occupational contacts (LepExp group in the present study) or HHCs, who are exposed to/infected by *M. leprae* (5,20,21) but do not exhibit the obvious clinical symptoms as well as TT/PB patients. Prospective cohort studies on leprosy patients in chemoprophylaxis and their families revealed that the HHCs of PB and MB patients have a high risk of developing disease (14,37). The majority of these HHCs with asymptomatic infection and appreciable CMI reactivity are protected by maintaining an effective CMI response to continuous exposure (21), but a few of these individuals may be carriers of contagious *M. leprae* with/without developing leprosy (14,37). Therefore, follow-up studies should be performed to address the question of whether the HHCs

screened by IGRA using *M. leprae* specific antigens are resistant/susceptible to the development of clinical leprosy. Since leprosy is a slow onset disease with low incidence rate, such studies should be conducted with a large number of contacts over long periods of time in order to obtain conclusive results. The extensive literature shows a distinct cytokine profile between TT and LL patients. Therefore, additional analysis of other cytokine profiles (i.e. IL-2, TNF-, TGF- β , IL-4, and IL-10) during these clinical follow-up studies will allow for the discovery of new biomarkers in T cell assays of *M. leprae* antigens which can be used to identify protection against leprosy/disease progress. This type of prospective research will allow for the further understanding of leprosy pathogenesis and provide the evident indicator to initiate chemotherapy of individuals with true subclinical leprosy, preventing disease progression and transmission of *M. leprae* to uninfected individuals.

3.6 References

- 1. Alter, A., A. Grant, L. Abel, A. Alcais, and E. Schurr. 2010. Leprosy as a genetic disease. Mamm.Genome. 22:19-31.
- Araoz, R., N. Honore, S. Banu, C. Demangel, Y. Cissoko, C. Arama, M. K. Uddin, S. K. Hadi, M. Monot, S. N. Cho, B. Ji, P. J. Brennan, S. Sow, and S. T. Cole. 2006. Towards an immunodiagnostic test for leprosy. Microbes.Infect. 8:2270-2276.
- Araoz, R., N. Honore, S. Cho, J. P. Kim, S. N. Cho, M. Monot, C. Demangel, P. J. Brennan, and S. T. Cole. 2006. Antigen discovery: a postgenomic approach to leprosy diagnosis. Infect.Immun. 74:175-182.
- 4. **Bordier, C.** 1981. Phase separation of integral membrane proteins in Triton X-114 solution. J.Biol.Chem. **256**:1604-1607.
- 5. Brennan, P. J. 2000. Skin test development in leprosy: progress with first-generation skin test antigens, and an approach to the second generation. Lepr.Rev. 71 Suppl:S50-S54.
- 6. Brennan, P. J. and H. Nikaido. 1995. The envelope of mycobacteria. Annu.Rev.Biochem. 64:29-63.
- Buhrer-Sekula, S., H. L. Smits, G. C. Gussenhoven, L. J. van, S. Amador, T. Fujiwara, P. R. Klatser, and L. Oskam. 2003. Simple and fast lateral flow test for classification of leprosy patients and identification of contacts with high risk of developing leprosy. J.Clin.Microbiol. 41:1991-1995.
- Cole, S. T., R. Brosch, J. Parkhill, T. Garnier, C. Churcher, D. Harris, S. V. Gordon, K. Eiglmeier, S. Gas, C. E. Barry, III, F. Tekaia, K. Badcock, D. Basham, D. Brown, T. Chillingworth, R. Connor, R. Davies, K. Devlin, T. Feltwell, S. Gentles, N. Hamlin, S. Holroyd, T. Hornsby, K. Jagels, A. Krogh, J. McLean, S. Moule, L. Murphy, K. Oliver, J. Osborne, M. A. Quail, M. A. Rajandream, J. Rogers, S. Rutter, K. Seeger, J. Skelton, R. Squares, S. Squares, J. E. Sulston, K. Taylor, S. Whitehead, and B. G. Barrell. 1998. Deciphering the biology of *Mycobacterium tuberculosis* from the complete genome sequence. Nature 393:537-544.
- Cole, S. T., K. Eiglmeier, J. Parkhill, K. D. James, N. R. Thomson, P. R. Wheeler, N. Honore, T. Garnier, C. Churcher, D. Harris, K. Mungall, D. Basham, D. Brown, T. Chillingworth, R. Connor, R. M. Davies, K. Devlin, S. Duthoy, T. Feltwell, A. Fraser, N. Hamlin, S. Holroyd, T. Hornsby, K. Jagels, C. Lacroix, J. Maclean, S. Moule, L. Murphy, K. Oliver, M. A. Quail, M. A. Rajandream, K. M. Rutherford, S. Rutter, K. Seeger, S. Simon, M. Simmonds, J. Skelton, R. Squares, S. Squares, K. Stevens, K. Taylor, S. Whitehead, J. R. Woodward, and B. G. Barrell. 2001. Massive gene decay in the leprosy bacillus. Nature 409:1007-1011.
- Corstjens, P. L., M. Zuiderwijk, H. J. Tanke, van der Ploeg-van Schip JJ, T. H. Ottenhoff, and A. Geluk. 2008. A user-friendly, highly sensitive assay to detect the IFN-gamma secretion by T cells. Clin.Biochem. 41:440-444.
- 11. **Deshpande, R. G., M. B. Khan, D. A. Bhat, and R. G. Navalkar**. 1995. Immunoaffinity chromatographic isolation of a high molecular weight seroreactive protein from *Mycobacterium leprae* cell sonicate. FEMS Immunol.Med Microbiol. **11**:163-169.
- Dockrell, H. M., S. K. Young, K. Britton, P. J. Brennan, B. Rivoire, M. F. Waters, S. B. Lucas, F. Shahid, M. Dojki, T. J. Chiang, Q. Ehsan, K. P. McAdam, and R. Hussain. 1996. Induction of Th1 cytokine responses by mycobacterial antigens in leprosy. Infect.Immun. 64:4385-4389.
- Duthie, M. S., G. C. Ireton, G. V. Kanaujia, W. Goto, H. Liang, A. Bhatia, J. M. Busceti, M. MacDonald, K. D. Neupane, C. Ranjit, B. R. Sapkota, M. Balagon, J. Esfandiari, D. Carter, and S. G. Reed. 2008. Selection of antigens and development of prototype tests for point-of-care leprosy diagnosis. Clin.Vaccine Immunol. 15:1590-1597.
- Fischer, E., V. S. De, A. Meima, D. Habbema, and J. Richardus. 2010. Different mechanisms for heterogeneity in leprosy susceptibility can explain disease clustering within households. PLoS.One. 5:e14061.

- 15. **Fischer, F., D. Wolters, M. Rogner, and A. Poetsch**. 2006. Toward the complete membrane proteome: high coverage of integral membrane proteins through transmembrane peptide detection. Mol.Cell Proteomics. **5**:444-453.
- Fujiki, Y., A. L. Hubbard, S. Fowler, and P. B. Lazarow. 1982. Isolation of intracellular membranes by means of sodium carbonate treatment: application to endoplasmic reticulum. J.Cell Biol. 93:97-102.
- Geluk, A., M. R. Klein, K. L. Franken, K. E. van Meijgaarden, B. Wieles, K. C. Pereira, S. Buhrer-Sekula, P. R. Klatser, P. J. Brennan, J. S. Spencer, D. L. Williams, M. C. Pessolani, E. P. Sampaio, and T. H. Ottenhoff. 2005. Postgenomic approach to identify novel *Mycobacterium leprae* antigens with potential to improve immunodiagnosis of infection. Infect.Immun. **73**:5636-5644.
- 18. Geluk, A. and T. H. Ottenhoff. 2006. HLA and leprosy in the pre and postgenomic eras. Hum.Immunol. 67:439-445.
- 19. Geluk, A., J. S. Spencer, K. Bobosha, M. C. Pessolani, G. M. Pereira, S. Banu, N. Honore, S. Reece, M. Macdonald, B. R. Sapkota, C. Ranjit, K. L. Franken, M. Zewdie, A. Aseffa, R. Hussain, M. M. Stefani, S. N. Cho, L. Oskam, P. J. Brennan, and H. M. Dockrell. 2009. From Genome to Leprosy Diagnostics: from *in silico* predictions to *ex vivo* verification. Clin.Vaccine Immunol.
- Geluk, A., van der Ploeg-van Schip JJ, K. E. van Meijgaarden, S. Commandeur, J. W. Drijfhout, W. E. Benckhuijsen, K. L. Franken, B. Naafs, and T. H. Ottenhoff. 2010. Enhancing sensitivity of detection of immune responses to *Mycobacterium leprae* peptides in whole-blood assays. Clin.Vaccine Immunol. 17:993-1004.
- Geluk, A., P. J. van der, R. O. Teles, K. L. Franken, C. Prins, J. W. Drijfhout, E. N. Sarno, E. P. Sampaio, and T. H. Ottenhoff. 2008. Rational combination of peptides derived from different *Mycobacterium leprae* proteins improves sensitivity for immunodiagnosis of *M. leprae* infection. Clin.Vaccine Immunol. 15:522-533.
- Goulart, I. M., D. O. Bernardes Souza, C. R. Marques, V. L. Pimenta, M. A. Goncalves, and L. R. Goulart. 2008. Risk and protective factors for leprosy development determined by epidemiological surveillance of household contacts. Clin.Vaccine Immunol. 15:101-105.
- 23. Goulart, I. M. and L. R. Goulart. 2008. Leprosy: diagnostic and control challenges for a worldwide disease. Arch.Dermatol Res. **300**:269-290.
- 24. Goulart, L. R. and I. M. Goulart. 2009. Leprosy pathogenetic background: a review and lessons from other mycobacterial diseases. Arch.Dermatol Res. **301**:123-137.
- Gu, S., J. Chen, K. M. Dobos, E. M. Bradbury, J. T. Belisle, and X. Chen. 2003. Comprehensive proteomic profiling of the membrane constituents of a Mycobacterium tuberculosis strain. Mol.Cell Proteomics. 2:1284-1296.
- 26. Hunter, S. W., H. Gaylord, and P. J. Brennan. 1986. Structure and antigenicity of the phosphorylated lipopolysaccharide antigens from the leprosy and tubercle bacilli. J.Biol.Chem. 261:12345-12351.
- 27. Hunter, S. W., B. Rivoire, V. Mehra, B. R. Bloom, and P. J. Brennan. 1990. The major native proteins of the leprosy bacillus. J.Biol.Chem. 265:14065-14068.
- 28. Jozefowski, S., A. Sobota, and K. Kwiatkowska. 2008. How *Mycobacterium tuberculosis* subverts host immune responses. Bioessays **30**:943-954.
- 29. Maeda, Y., M. Gidoh, N. Ishii, C. Mukai, and M. Makino. 2003. Assessment of cell mediated immunogenicity of *Mycobacterium leprae*-derived antigens. Cell Immunol. 222:69-77.
- Maeda, Y., T. Mukai, J. Spencer, and M. Makino. 2005. Identification of an Immunomodulating Agent from *Mycobacterium leprae*. Infect.Immun. 73:2744-2750.
- 31. Magnani, Z. I., C. Confetti, G. Besozzi, L. R. Codecasa, P. Panina-Bordignon, R. Lang, G. A. Rossi, R. Pardi, and S. E. Burastero. 2000. Circulating, *Mycobacterium tuberculosis*-specific lymphocytes from PPD skin test-negative patients with tuberculosis do not secrete interferon-

gamma (IFN-gamma) and lack the cutaneous lymphocyte antigen skin-selective homing receptor. Clin.Exp.Immunol. **119**:99-106.

- 32. Makino, M., Y. Maeda, and N. Ishii. 2005. Immunostimulatory activity of major membrane protein-II from *Mycobacterium leprae*. Cell Immunol. **233**:53-60.
- Manandhar, R., J. W. LeMaster, C. R. Butlin, P. J. Brennan, and P. W. Roche. 2000. Interferon-gamma responses to candidate leprosy skin-test reagents detect exposure to leprosy in an endemic population. Int.J.Lepr.Other Mycobact.Dis. 68:40-48.
- 34. Marques, M. A., S. Chitale, P. J. Brennan, and M. C. Pessolani. 1998. Mapping and identification of the major cell wall-associated components of *Mycobacterium leprae*. Infect.Immun. **66**:2625-2631.
- 35. Marques, M. A., A. G. Neves-Ferreira, E. K. da Silveira, R. H. Valente, A. Chapeaurouge, J. Perales, B. R. da Silva, K. M. Dobos, J. S. Spencer, P. J. Brennan, and M. C. Pessolani. 2008. Deciphering the proteomic profile of *Mycobacterium leprae* cell envelope. Proteomics. 8:2477-2491.
- Mattow, J., F. Siejak, K. Hagens, F. Schmidt, C. Koehler, A. Treumann, U. E. Schaible, and S. H. Kaufmann. 2007. An improved strategy for selective and efficient enrichment of integral plasma membrane proteins of mycobacteria. Proteomics. 7:1687-1701.
- Moet, F. J., D. Pahan, R. P. Schuring, L. Oskam, and J. H. Richardus. 2006. Physical distance, genetic relationship, age, and leprosy classification are independent risk factors for leprosy in contacts of patients with leprosy. J.Infect.Dis. 193:346-353.
- 38. Mori, T., M. Sakatani, F. Yamagishi, T. Takashima, Y. Kawabe, K. Nagao, E. Shigeto, N. Harada, S. Mitarai, M. Okada, K. Suzuki, Y. Inoue, K. Tsuyuguchi, Y. Sasaki, G. H. Mazurek, and I. Tsuyuguchi. 2004. Specific detection of tuberculosis infection: an interferon-gamma-based assay using new antigens. Am.J.Respir.Crit Care Med 170:59-64.
- Murray, R. A., M. R. Siddiqui, M. Mendillo, J. Krahenbuhl, and G. Kaplan. 2007. Mycobacterium leprae inhibits dendritic cell activation and maturation. J.Immunol. 178:338-344.
- 40. Nigou, J., M. Gilleron, and G. Puzo. 2003. Lipoarabinomannans: from structure to biosynthesis. Biochimie 85:153-166.
- 41. **Ohyama, H., F. Nishimura, M. Meguro, S. Takashiba, Y. Murayama, and S. Matsushita**. 2002. Counter-antigen presentation: fibroblasts produce cytokines by signalling through HLA class II molecules without inducing T-cell proliferation. Cytokine **17**:175-181.
- 42. P Murphy, P. T. M. W. 2007. Janeway's Immunobiology 7th ed.
- 43. **Pessolani, M. C. and P. J. Brennan**. 1996. Molecular definition and identification of new proteins of *Mycobacterium leprae*. Infect.Immun. **64**:5425-5427.
- Pessolani, M. C., D. R. Smith, B. Rivoire, J. McCormick, S. A. Hefta, S. T. Cole, and P. J. Brennan. 1994. Purification, characterization, gene sequence, and significance of a bacterioferritin from *Mycobacterium leprae*. J.Exp.Med. 180:319-327.
- Rivoire, B., M. C. Pessolani, C. M. Bozic, S. W. Hunter, S. A. Hefta, V. Mehra, and P. J. Brennan. 1994. Chemical definition, cloning, and expression of the major protein of the leprosy bacillus. Infect.Immun. 62:2417-2425.
- Scollard, D. M., L. B. Adams, T. P. Gillis, J. L. Krahenbuhl, R. W. Truman, and D. L. Williams. 2006. The continuing challenges of leprosy. Clin.Microbiol.Rev. 19:338-381.
- 47. Sekar, B. 2007. Recent advances in immunodiagnosis of leprosy. Indian J.Lepr. 79:85-106.
- 48. Shepard, C. C., P. Draper, R. J. Rees, and C. Lowe. 1980. Effect of purification steps on the immunogenicity of *Mycobacterium leprae*. Br.J.Exp.Pathol. **61**:376-379.
- 49. Shepard, C. C. and D. H. McRae. 1968. A method for counting acid-fast bacteria. Int.J.Lepr.Other Mycobact.Dis. 36:78-82.

- Sinha, S., K. Kosalai, S. Arora, A. Namane, P. Sharma, A. N. Gaikwad, P. Brodin, and S. T. Cole. 2005. Immunogenic membrane-associated proteins of *Mycobacterium tuberculosis* revealed by proteomics. Microbiology 151:2411-2419.
- 51. Spencer, J. S., H. M. Dockrell, H. J. Kim, M. A. Marques, D. L. Williams, M. V. Martins, M. L. Martins, M. C. Lima, E. N. Sarno, G. M. Pereira, H. Matos, L. S. Fonseca, E. P. Sampaio, T. H. Ottenhoff, A. Geluk, S. N. Cho, N. G. Stoker, S. T. Cole, P. J. Brennan, and M. C. Pessolani. 2005. Identification of specific proteins and peptides in *Mycobacterium leprae* suitable for the selective diagnosis of leprosy. J.Immunol. 175:7930-7938.
- 52. Triccas, J. A., P. W. Roche, N. Winter, C. G. Feng, C. R. Butlin, and W. J. Britton. 1996. A 35-kilodalton protein is a major target of the human immune response to *Mycobacterium leprae*. Infect.Immun. **64**:5171-5177.
- 53. Weir, R. E., P. J. Brennan, C. R. Butlin, and H. M. Dockrell. 1999. Use of a whole blood assay to evaluate *in vitro* T cell responses to new leprosy skin test antigens in leprosy patients and healthy subjects. Clin.Exp.Immunol. **116**:263-269.

CHAPTER 4 IDENTIFICATION AND BIOCHEMICAL DEFINITION OF NEW MYCOBACTERIUM LEPRAE ANTIGENS PRESENTED BY CD1a OF LANGERHANS CELLS

This work has been done in the collaboration with Dr. Peter Sieling and Dr. Robert Modlin, in University of California, Los Angeles. Protein production/fractionation at Colorado State University by Hee Jin Kim. T cell assays at UCLA by Dr. Peter Sieling

4.1 Introduction

CD1 proteins are expressed on the surface of B cells, myeloid dendritic cells, Langerhans cells and other antigen presenting cells (37). They are structurally related to the Major Histocompatibility Complex-I (MHC-I), a peptide antigen presenting molecule (4). A CD1 molecule consists of two chains: β 2 microglobulin (β 2m) and a heavy chain containing three extracellular domains (α 1- α 3) (Figure 4.1). On the basis of sequence comparisons, the molecules can be divided into two groups: Group 1 consists of CD1a, CD1b, CD1c, and CD1e, which are involved in antigen presentation of microbial lipids to clonal T cells; Group 2 contains only CD1d, presenting lipid antigens to natural killer T cells (2,4,32).

In antigen presentation, CD1 holds the alkyl chains of amphipathic lipids into a deep hydrophobic groove of CD1 heavy chain formed between the α 1 and α 2 helices (2,32) (Figure 4.1). The hydrophilic parts of these antigens are exposed on the surface of CD1 molecules and presented to T cell receptors (2,32) (Figure 4.1). Many of the unique

mycobacterial lipids found in the cell envelope can be presented by CD1a, CD1b and CD1c to activate a variety of T cell clones. These include mycolates (21,29), lipoglycans (8,41), sulfoglycolipid (14), mycobactin (31) and mannosyl β -1-phospho-mycoketides (27,30) (see Figures 1.7, 1.8 and 1.22).



Figure 4.1 Crystal structure of murine CD1d loaded with a synthetic α galactosylceramide (adapted from Barral and Brenner (2) with permission). The acyl chains of α -galactosylceramide (α -GalCer) (yellow) are inserted into the hydrophobic binding groove which is formed between α 1 (blue) and α 2 (light green) helices of the heavy chain of CD1. The hydrophilic oxygenated (red) sugar of this CD1d antigen is exposed to T cell receptors of a CD1⁺ T cell.

CD1-restricted T-cells (CD1⁺ T cells) induce protective Th1 immunity to

intracellular pathogens such as mycobacteria (see section 1.8.3). Particularly, the

mycobacteria specific CD1⁺ T cells secrete potent cytolytic molecules such as perforin,

granulysin and efficiently kill cells that are intracellularly infected by live virulent M.

tuberculosis (3,47,48). Importantly, CD1⁺ T cells appear frequently in patients infected

by *M. tuberculosis* or *M. leprae* (8,14).

In *M. leprae* infection, Toll like receptor-1 (TLR-1), TLR-2 and CD1s are more strongly expressed in skin lesions of tuberculoid leprosy (TT) patients, as compared to those of lepromatous leprosy (LL) patients (20,43). TLR-2 signaling by synthetic lipopeptides (derived from mycobacterial lipoproteins) efficiently mediates differentiation of CD1⁺ dendritic cells from peripheral blood mononuclear cells (PBMCs) of TT patients (20). Langerhans cells, as abundant dendritic cells in TT skin lesions, coexpress high levels of CD1a and langerin (CD207), as one of the C-type lectin receptors (52). Langerin can mediate phagocytosis and efficient innate immunity in skin lesions via interaction with carbohydrate containing cell surface molecules such as mannose, N-acetyl glucosamine and fucose (44). The CD1a⁺Langerhans cells efficiently present mycobacterial lipids to CD1a⁺ T cells which can produce high levels of IFN- γ and granulysin (34,44) suggesting that bacterial lipoprotein may serve as CD1a restricted antigens or could be involved in CD1a antigen presentation via Langerhans cells.

In this study, we hypothesized that Langerhans cells present a unique set of microbial antigens, bacterial lipoproteins, to CD1a restricted T cells. Therefore, the aim of this study was to identify and to biochemically characterize lipoprotein(s) as novel *M*. *leprae* specific antigens recognized by CD1a-restricted T cell clones from leprosy skin lesions.

4.2 Materials and methods

4.2.1 *T cell lines, proliferation and IFN-γ released assays*

T cell lines were derived from skin lesions of borderline tuberculoid leprosy (BT) patients according to Sieling *et al.* (44). Briefly, lymphocytes were obtained from patients' skin lesions with a tissue sieve and isolated by Ficoll-Hypaque (GE healthcare,

Pittsburgh, PA) gradient centrifugation. T cell lines were activated by adding recombinant IL-2 (rIL-2; 1nM; Chiron Diagnostics, Emeryville, CA) and irradiated PBMCs from the same donors. Subsequently, T cell lines were cultured with CD1⁺ antigen presenting cells which were stimulated by M. leprae lysates. The established T cell lines were maintained in rIL-2 containing media and propagated by adding heterologous PBMCs and PHA (Sigma, St. Louis, MO). In antigen specific T cell proliferation assays, T cells (1 \times 10⁴) and CD1⁺ antigen presenting cells (usually 1 \times 10⁴) were cultured with or without of bacterial antigens in microtiter wells (in triplicate) at 37 °C in a 7% CO₂ incubator. After 3 days, cells were pulsed with $[^{3}H]$ thymidine (1 µCi/well; ICN Biomedicals, Irvine, CA), harvested 4–6 hr later and subjected to liquid scintillation counting. For *in vitro* IFN- γ release assays, T cells (1 x 10⁴) were stimulated in the same manner as the proliferation assay but incubated only for 24 hr. The supernatants were collected for measuring IFN- γ levels by an ELISA assay kit (BD Pharmingen, San Jose, CA). All of these biological experiments were conducted by Dr. Sieling in UCLA.

4.2.2 Proteinase K treatment of M. leprae membrane antigen (MLMA)

The 50 µg of MLMA was treated with or without 2 unit/ml of proteinase K (Sigma, St. Louis, MO) in PBS. For reaction controls, proteinase K was heat-inactivated and added into MLMA. All reactions were incubated at 37 °C overnight and inactivated for T cell assays by heating at 100°C for 10 min.

4.2.3 Sub-fractionation of MLMA

The hydrophobic MLMA-P (Trion X-114 insoluble fraction of MLMA-InsP) was generated from MLMA by the methods described in section 3.3.2. Figure 4.2 summarized the experimental procedure.



Figure 4.2 Experimental scheme for the sub-fractionation of MLMA (modified from Figure 3.1)

Briefly, MLMA-InsP was subjected to 1% Triton X-114 solubilization and ultracentrifugation (100,000 \times g, 3 hr). The resultant Triton X-114 insoluble fraction (MLMA-P) in the pellets was subjected to cold acetone precipitation to remove residual detergent. The supernatant was subjected to phase separation which allowed enriching lipoproteins into the detergent phase. This detergent phase fraction was applied to cold acetone precipitation.

4.2.3.1 Anion exchange chromatography of MLMA-P for preparation of a lipoprotein enriched fraction

The protein pellet of the Triton X-114 (200 μ g) detergent phase fraction was subjected to solubilization with 1 ml of denaturing buffer (8M urea, 50 mM Tris (pH 11.0), 20 mM dithiothreitol) for 3 hr at 4 °C and was centrifuged to remove the insoluble materials for 30min at 27,000 × g at 4 °C. The supernatants were applied onto an anion exchange column (Q Sepharose; GE healthcare, Pittsburgh, PA) to remove LAMs (lipoarabinomannan), LMs (lipomannan) and PIMs (phosphatidylinositol mannosides), and washed with 5 × column volumes of denaturing buffer. The resulting lipoprotein enriched fraction (40 μ g) was recovered by adding 500 μ l of elution buffer (8M urea, 50 mM Tris-HCl (pH 4.0), 20 mM dithiothreitol) and was dialyzed against 10 mM ammonium bicarbonate prior to T cell simulation assays.

4.2.3.2 Size exclusion (SE) fractionation of MLMA-P

The MLMA-P was further fractionated on the basis of molecular mass of its constituents using the Whole Gel Eluter (Bio-Rad Life Science, Hercules, CA) according to the manufacturer's instructions. Briefly, the MLMA-P (1 mg) was dissolved in 1 ml of NuPAGE LDS sample buffer (Invitrogen Life Technologies, Carlsbad, CA) containing 50 mM dithiothreitol and was electrophoretically separated in a precast 4-12% NuPAGE Novex Bis-Tris mini preparative gel (Invitrogen Life Technologies, Carlsbad, CA) with MES Running Buffer (Invitrogen Life Technologies, Carlsbad, CA). The gel was placed in a Mini Whole Gel Eluter (Bio-Rad Life Science, Hercules, CA). Subsequently, an

electric current (75 mA, 30 min) was applied to recover the separated proteins into 14 fractions in an elution buffer (60 mM Tris 40 mM CAPS, pH 9.3).

4.2.3.3 Preparative isoelectric focusing (IEF) fractionation of MLMA

The MLMA (2 mg) was dialyzed against 10mM ammonium bicarbonate, lyophilized and resuspended with 2 ml of IEF buffer (7 M urea, 2 M thiourea (Invitrogen Life Technologies, Carlsbad, CA), 0.5% octyl β-D-glucopyranoside (Sigma, ST. Louis, MO), 0.5% ASB-14 (Sigma, ST. Louis, MO), 4% ampholyte (pH range 4 to 7; Invitrogen Life Technologies, Carlsbad, CA), 1% ampholyte (pH range 3 to 10; Invitrogen Life Technologies, Carlsbad, CA), 1% ampholyte (pH range 3 to 10; Invitrogen Life Technologies, Carlsbad, CA), 10 mM DTT). The protein solution was separated into six fractions by liquid-phase isoelectric focusing with a Mini-Rotofor apparatus (Bio-Rad Life Science, Hercules, CA) under constant power of 15 watt for 3 hr, according to the manufacturer's instructions. Each fraction was dialyzed and subjected to cold acetone precipitation.

4.2.4 Limulus amoebocyte lysate test, bicinchoninic acid assay and SDS-PAGE analysis of protein fractions

Protein quantities of all protein fractions and recombinant proteins were determined by the bicinchoninic acid assay. Individual fractions were analyzed by electrophoresis in custom made 15% SDS-PAGE or commercial 4-12% NuPAGE gels (Invitrogen Life Technologies, Carlsbad, CA). The gels were visualized using a periodic acid oxidation step for LMs, LAMs and PIMs, and silver staining for the proteins. All resultant protein fractions were dialyzed against 10 mM ammonium bicarbonate and

lyophilized until use. LPS contents of all protein fractions were determined by the Limulus amoebocyte lysate test and were less than 2 ng/mg of protein.

4.2.5 Butanol extraction of MLMA and MLMA-P

Both MLMA and MLMA-P (1mg per protein) were resuspended with 1ml of PBS and added to 1ml of PBS saturated butanol. Subsequently, the preparations were incubated on a rocker at 25 °C for 30 min and centrifuged at 27,000 × g for 10 min at 25 °C. The samples were separated into three phases. Delipidated proteins (devoid of noncovalently linked proteins) were located at both the interphase and the lower water phase while lipids were in the upper organic phase. The interphase and lower water phase were combined and were dialyzed against 10 mM ammonium bicarbonate. The lipids in upper phase were dried under N₂ gas. Fractions were analyzed by thin layer chromatography in CHCl₃:CH₃OH:H₂O (65:35:5) developed with 0.2% sulfuric acid spray, to assess the lipid content, and by SDS-PAGE to analyze the protein content.

4.2.6 In gel tryptic digestion and electrospray ionization-MS analysis

Protein fractions which induced IFN-γ production of LCD4.15, were applied to electrophoresis (4-12% NuPAGE gel; Invitrogen Life Technologies, Carlsbad, CA) and proteins stained using Coomassie Brilliant Blue R-250 (Invitrogen Life Technologies, Calsbad, CA). Coomassie stained proteins spots were excised from the gel and subjected to in gel digestion with trypsin (Roche Applied Sciences, Indianapolis, IN), according to Marques *et al.* (25). Briefly, protein gel pieces were destained three times with 25 mM ammonium bicarbonate (pH 8.0) in 50% acetonitrile and dried in a speed-vacuum concentrator (ThermoFisher Scientific, Hampton, NH). The gel pieces were reconstituted in 15 μ l of 200 mM ammonium bicarbonate (pH8.0) containing 20 ng/ μ l of trypsin and were incubated overnight at 37 °C. The resulting peptides were recovered by twice extracting with 100 μ l of 50% acetonitrile containing 5% trifluoroacetic acid with vortexing for 30 min and dried in a speed-vacuum concentrator. Dried peptides were resuspended with 20 μ l of 5% acetonitrile, 0.1% acetic acid and subjected to electrospray ionization-MS analysis at the PMF (CSU).

Peptides were purified and concentrated using an on-line enrichment Agilent Zorbax (Agilent, Santa Clara, CA) C_{18} column 5µm (5 × 0.3mm). Subsequent chromatographic separation was performed on a reverse phase nanospray column (Agilent 1100 nanoHPLC Zorbax C_{18} , 5µm, 75 µm ID × 150mm column; Agilent, Santa Clara, CA) by using a 42 min linear gradient from 25%-55% buffer B (90% acetonitrile, 0.1% formic acid) at a flow rate of 300 nl/min. Peptides were eluted directly into the mass spectrometer (Thermo Scientific LTQ Linear Ion Trap, ThermoFinnigan, San Jose, CA) and spectra were collected over a m/z range of 200-2000 Da using a dynamic exclusion limit of 2 MS/MS spectra of a given peptide mass. Compound lists of the resulting spectra were generated by using Bioworks 3.0 software (ThermoFinnigan, San Jose, CA) with an intensity threshold of 5,000 and 1 scan/group. MS/MS spectra were searched against the NCBI nonredundant protein database and the *M. leprae* database (1614 entries) (22) using the Mascot database search engine (version 2.3). The scaffold software 3.0 (Proteome Software, Portland, OR) was applied to verify peptide identifications by the use of MASCOT 2.2 and SEQUEST search engine probabilistically validating the peptide and protein identifications to all MS/MS sequencing results (25).
4.2.7 Analysis of lipoproteins by using a bacterial lipoprotein database and BLASTP algorithm

AA sequences of putative *M. leprae* lipoproteins were analyzed by a bacterial lipoprotein database (DOLOP, <u>http://www.mrc-mb.cam.ac.uk/genomes/dolop/)</u> (see section 1.7.1.1). The AA sequences of the N terminal regions (Cys at +1 to +17) from selective lipoproteins were compared to orthologues of *M. tuberculosis* (51) by BLASTP (http://blast.ncbi.nlm.nih.gov/Blast.cgi?PROGRAM=blastp&BLAST_PROGRAMS=blas tp&PAGE_TYPE=BlastSearch&SHOW_DEFAULTS=on&LINK_LOC=blasthome)

4.2.8 Cloning of M. leprae lipoproteins into M. smegmatis expression vector pVV16

Eight genes encoding putative lipoproteins were amplified from genomic DNA of *M. leprae* Thai-53 using the Touchdown PCR method (see section 2.3.4). The primers (IDT Technology, Denver, CO) contained the recognition sequences of unique restriction enzymes, *Nde*I, or *Bam*HI and *Hind*III respectively, and were specific to the upstream and the downstream of the ORFs (Table 4.1). All PCR products were directly digested with restriction enzymes and cloned into the *E. coli-M. smegmatis* shuttle expression vector pVV16, provided by Dr. Vissa (CSU), which allowed to produce the 6-Histidine tag at the C-terminus of the target protein. The DNA sequences of all recombinant clones were confirmed by automated nucleotide sequencing at the PMF, CSU. Each of recombinant clones was transformed into the expression host, *M. smegmatis* mc²155.

| Name | Sequence | Name | Sequence |
|----------------|----------------------------|----------------|-----------------------------|
| ML1086 forward | AACATATGGTGGTGAGTCGGCGC | ML0489 forward | AACATATGCGGTCCGGCTTATTCG |
| ML1086 reverse | AAAAGCTTCGGCAACAAGCCCTTC | ML0489 reverse | AAAAGCTTTCGTACTAACGCCCAT |
| ML0706 forward | AACATATGAAGGCTGCCGTGACAC | ML0303 forward | TGGATCCGATGACGCGCCAGTT |
| ML0706 reverse | AAAAGCTTCCACTGCTGCACCAAG | ML0303 reverse | AAAAGCTTATCCACATACCTCGGTGTG |
| ML2095 forward | AACATATGAAGCTCAACCAGTTTGGT | ML2593 forward | AACATATGAGGCGTGCAATGAGTG |
| ML2095 reverse | AAAAGCTTAGCGATGGCATTAGCC | ML2593 reverse | AAAAGCTTGCATCGCCTATCGG |
| ML2446 forward | AACATATGATCGCATCCACTCCGC | ML1699 forward | AAACATATGCGGATAGGGCGGTC |
| ML2446 reverse | AAAAGCTTACCACCGGGTCTAGCC | ML1699 reverse | AAGCTTGGTTTTGTCGTCGTTG |

Table 4.1 List of primers used in cloning of putative lipoprotein ORFs

Recognition sequence of the restriction enzyme is shaded. AAGCTT is the recognition sequence for *HindIII*, CATATG for *Nde*I and GGATCC for *Bam*HI

4.2.9 Subcellular fractionation of M. smegmatis and detection of the recombinant

lipoproteins

Each of *M. smegmatis* recombinant clones containing lipoprotein gene constructs, pVV16:*lpr1086*, pVV16:*ml2095* or pVV16:*ml2446*, was cultured in 2 L of Luria-Bertani (LB) broth or M7H9 broth (supplemented with 0.2% glycerol) with 50 µg/ml of kanamycin and 25 µg/ml of hygromycin to mid-log phase at 37 °C (42), centrifuged at 4 °C and frozen at -70°C. The cells (10 g, wet weight) were resuspended with lysis buffer (50mM Tris-HCl (pH8.0), 150mM NaCl) containing 10 µg/ml of DNase, 10 µg/ml of RNase and protease inhibitor cocktail (P8340, Sigma, St. Louis, MO), and disrupted by intermittent probe sonication with a Soniprep 150 sonicator (MSE Sonic Prep150, MSE-Sanyo, Palisades Park, NJ) for 10 cycles (60 sec bursts/90 sec of cooling on ice). The lysates were centrifuged at 27,000 × g for 5 min to remove unbroken cells and the supernatants were centrifuged at 27,000 × g for 30 min at 4 °C. The pellets contained insoluble cell walls. The supernatants were applied to ultracentrifugation at 100,000 × g for 3 hr at 4 °C and were separated into cytosol fraction in the supernatant and membrane fraction in the pellet. In order to verify the localization of recombinant proteins, all

fractions were analyzed by 15% SDS-PAGE and Western blot using anti-6-Histidine tag polyclonal antibody (1:3000 dilutions; Sigma, St. Louis, MO).

4.2.10 Purification of recombinant lipoproteins

The harvested recombinant *M. smegmatis* (80-100 g) from 20 L of culture were subjected to subcellular fractionations. Both resulting cell wall and membrane fractions were resuspended with denaturing buffer (50 mM Tris- HCl (pH 8.0), 8M urea), incubated on the rocker at 4°C for 1 hr and centrifuged at 27000 × g for 30 min at 4°C. The supernatant was applied onto Ni-NTA agarose column (Qiagen, Valencia, CA) and the column was washed with 20 × column volumes of the same buffer. The recombinant proteins were eluted with 1 × column volume of denaturing buffer containing stepwise gradients of imidazole (5, 10, 20, 50,100,150 and 200 mM) (46). The recombinant lipoproteins (200 µg, obtained from approximately 100 g of each *M. smegmatis* clone) were dialyzed against 10mM ammonium bicarbonate, lyophilized and kept at -70 °C

4.2.11 Construction the hybrid Lpr1086 by using the signal sequence of M. tuberculosis LprG (Rv1411c)

The DNA sequence coding for the Rv1411c signal sequence was amplified from the pVV16 construct with *rv1411c* provided by Dr. Belisle (CSU) using the two step PCR (72 °C for annealing and extension steps, and 94 °C for the denaturing step). In this reaction, the forward primer (5'-TGGCAGCGAGGACAACTTGA 3') was specific to the 135 base pair upstream region from the translational initiation site of pVV16: *rv1411c*, and reverse primer

(5' <u>GGTGGGAGCCTGCGTGCCCCCCGCTGCA</u>GCCGGCAACGACAGTGGCGGCGA

TGC 3') contained the nucleotide sequence encoding the signal peptide and Cys₊₁ of Rv1411c (shaded). This reverse primer also contained 24bps of nucleotide sequence encoding consecutive 6 AAs (Gly₊₂ to Gln₊₇) of the ML1086 (underlined). The resulting PCR product encoding the Rv1411c signal sequence and 6 AAs (Gly₊₂ to Gln₊₇) of ML1086, was then purified from agarose gel electrophoresis and used as a macroprimer in a second round of PCR. Subsequently, the hybrid *lpr1086* was generated using the macroprimer, ML1086 reverse primer (Table 4.1) and the pVV16 construct with *ml1086* as template (24). The PCR product, *lpr1086*, was cloned into the pVV16 vector. The DNA sequence of this recombinant clone was confirmed by automated nucleotide sequencing at the PMF (CSU). The pVV16: *lpr1086* was transformed into *M. smegmatis* mc²155 to express and purify the recombinant lipoprotein.

4.3 Results

4.3.1 Characterization of the LCD4.15 T-cell clone

Dr. Sieling in UCLA had established CD1a-restricted T-cell lines from BT patients. These T cell lines were mostly activated by Langerhans cells' presentation of the antigens in several mycobacterial extracts including *M. tuberculosis*, *M. smegmatis* and *M. leprae*. None of these clones was activated by the didehydroxy-mycobactin, which was proven as a CD1a restricted antigen in *M. tuberculosis* infection, or several of the known CD1 antigens including glucose monomycolate, mycolic acid, ManLAM and mannosyl β -1-phosphomycoketid (see section 1.8.3; data not shown). However, one T cell clone, LCD4.15, recognized solely *M. leprae* whole cell lysates (Figure 4.3.A).



Figure 4.3 Proliferative response and IFN-γ production of LCD4.15. A) Proliferation assay of LCD4.15 in response to lysates of *M. tuberculosis* (*M. tb*), *M. smegmatis* (*M. smeg*) and *M. leprae*. B) CD1 blocking assay by using IgGs of CD1 isoforms. mIgG1 indicates mouse IgG. C) T cell reactivity after proteinase K (ProK) treatment of MLMA.

Blocking of CD1a by neutralizing IgG abolished the response to *M. leprae* whole cell lysates of this T cell clone (Figure 4.3.B). In contrast to other CD1⁺ T cell clones recognized by mycobacterial glycolipids, this CD1a-restricted T cell line was unable to recognize proteinase K-treated *M. leprae* extracts (Figure 4.3.C). Since the *mbt* operon, which is responsible for the biosynthesis of mycobactin (nonribosomal lipopeptide), is absent in *M. leprae* (5), other biosynthetic intermediates related to mycobactin cannot be the antigen candidates recognized by LCD4.15. Collectively, the responses of LCD4.15 was restricted to CD1a antigen presentation by Langerhans cells and was protease-sensitive indicating that this line is capable of recognizing a *M. leprae* specific lipoprotein or a protein-lipid complex.

4.3.2 Activation of LCD4.15 in response to MLMA derived protein subfractions

Previous biochemical studies of CD1 antigens, supported by CD1crystallization studies, indicated that CD1 restricted antigens are generally amphipathic lipids (32). Therefore, we hypothesized that this new CD1a protein antigen would be covalently or noncovalently linked with lipids to retain the amphipathic feature of a CD1 antigen and would be enriched in hydrophobic protein pools, probably MLMA. Previously, as described in section 3.3.2, subfractionation of MLMA generated MLMA-SP (peripheral membrane protein of MLMA), MLMA-P (highly hydrophobic MLMA) and a lipoprotein enriched fraction (Figure 4.4.A and B).



Figure 4.4 SDS-PAGE analysis of MLMA derived fractions and IFN-γ production by LCD4.15 induced in response to antigens.

A) MLMA fractions were stained with silver nitrate. B) Identical gel stained with silver staining in combination with sodium periodate for LMs, LAMs and PIMs. 5 μ g of each protein fraction was run in individual lanes. Lane 1, MLMA; lane 2, MLMA-P; lane 3, MLMA-SP, lane 4, MLMA TX-114 detergent phase, and lane 5, lipoprotein enriched fraction. C) T cell reactivity to MLMA and each of subfractions derived from MLMA.

Overall, all MLMA derived fractions efficiently activated LCD4.15. T cell reactivity was especially strong on stimulation with both lipoprotein enriched fractions and MLMA-P which contained most of the lipoproteins and hydrophobic proteins but was devoid of LMs and LAMs (Figure 4.4.C). Thus, this novel antigen(s) might be a lipoprotein, or required a lipid/lipoprotein as carrier for lipidic antigen presentation.

In order to evaluate the role of lipoproteins in CD1a antigen presentation to LCD4.15, butanol extraction removed the noncovalently linked lipids from proteins proteins and possible nonribosomal lipopeptides (other than mycobactin) from MLMA-P and MLMA. Thin layer chromatography analysis showed that delipidated MLMA and MLMA-P proteins (devoid of noncovalently linked lipids) in the water phase did not contain free lipids, which were mostly extracted into the resulting organic phase (Figure 4.5.A).



Figure 4.5 Thin layer chromatography analysis and T cell reactivity of MLMA and MLMA-P before/after delipidation. A) Thin layer chromatography analysis. MLMA and MLMA-P were delipidated with butanol and subjected to Thin layer chromatography in CHCl₃:CH₃OH:H₂O (65:35:5). The plates were developed with a sulfuric acid spray. B) IFN-γ production from LCD4.15 in response to MLMA, MLMA-P, delipidated version of the fractions (protein fractions without covalently linked lipids) and the lipid fractions. Lipid ext, lipid extracts from delipidation of MLMA or MLMA-P.

T cell reactivities to both delipidated fractions on LCD4.15 were equivalent to those of the parent MLMA-P and MLMA, but extracted lipids did not induce any T cell reactivity on stimulation of LCD4.15 (Figure 4.5.B). The results indicated that a lipoprotein, which is covalently acylated, was the likely new CD1a restricted *M. leprae* specific antigen recognized by LCD4.15. This was the first evidence that a bacterial lipoprotein is involved in CD1 restricted antigen presentation.

4.3.3 Identification of a CD1a restricted M. leprae specific antigen by using mimic Multi-Dimensional Protein Identification Technology

In order to identify the precise CD1a antigens, the sequential multi-dimensional protein identification technology (11) might not be feasible with the very limited supply of *M. leprae* available. Therefore, the MLMA-P and MLMA were subjected to the mimic sequential multi-dimensional protein identification technology which consisted of two different separation methods. The first was size exclusion (SE) fractionation by SDS-PAGE gel elution and the other was a preparative IEF fractionation, combined with T cell reactivity assays.

The first size exclusion (SE) fractionation generated 14 fractions from MLMA-P (Figure 4.6.A). T cell reactivities of all fractions revealed that the SE fraction 8 strongly induced IFN-γ production on stimulation of LCD4.15 indicating that this T cell clone recognizes a protein antigen in a range of molecular weight 37 to 55 kDa (Figure 4.6.B). In order to indentify this new CD1a antigen, three consecutive fractions (SE fraction 7 to 9) were subjected to in gel tryptic digestion and MS based peptide identification.

Bioinformatics analyses of MS/MS data from all three fractions identified more than 100 proteins where two or more unique peptides were identified.



Figure 4.6 Size exclusion (SE) Fractionation of MLMA-P. A) SDS-PAGE analysis of fractions after size exclusion (SE) fractionation of MLMA-P; B) Production of IFN- γ in response to resultant fractions on LCD4.15.

Secondly, the preparative IEF fractionation generated six additional fractions. The T cell reactive assay of LCD4.15 revealed that a CD1a antigen was enriched in IEF fraction 4 which was in the range of pI 6 and 8 (Figure 4.7). This protein fraction was subjected to peptide identification by electrospray ionization-MS/MS analysis.



Figure 4.7 Preparative Isoelectric Focusing (IEF) fractionation of MLMA. A) SDS-PAGE analysis of fractions after IEF fractionation of MLMA-P; (a) silver stained gel for the proteins (b) sodium periodate stained gel for the carbohydrates: B) Production of IFN- γ in response to resultant fractions on LCD4.15.

A total of 24 proteins were identified by MS/MS analyses, common to both SE fraction 8 and IEF fraction 4, but were absent in SE fraction 7 and 9. Bioinformatic analysis screened five of the most plausible antigen candidates that could be covalently acylated. Among the five possible candidates (Table 4.2), two lipoproteins, ML1086 (a solute binding protein, LpqY) and ML2095 (a phosphate binding protein, Pst3), were identified as possible CD1a antigen candidates. In mass spectrometry analysis, the AA sequences extracted from unique peptides which allowed the identification of ML1086 or ML2095, corresponded to 8% or 33% of sequence coverage of the predicted AA sequence of ML1086 or ML2095 respectively (Table 4.2). Three other proteins identified in this manner were found to be involved in the fatty acid biosynthetic pathway and these proteins could be transiently acylated. Therefore, further study focused on the expression and evaluation of recombinant proteins, ML1086 and ML2095.

| ORF (Protein) | Function | MW (kDa) | pI | Peptide Sequence | Sequence Coverage (%) | Protein probability (%) |
|--------------------|--|-------------|-------|--|--------------------------|-------------------------------|
| ML2565 (FabG4) | 3-ketoacyl- (acyl-carrier- protein) reductase | 47.325 | 6.172 | LTEGLVGNGSISK VVVIGTTPDATTSTDER WTDSFGGLVFDATGITAPAELK LNSLLQGGLPVDVAETIAYFAIPASNAVTGNVIR | 19% | 95% |
| ML2639 | Aldehyde dehydrogenase | 54.090 | 9.642 | IQPTVIVDPDPEGPLMTK LPFGGVGESGIGAYHGK ALQALMVENEGAIAQALADDLDRNPVEAFIVDIAATVSEAK | 12% | 95% |
| ML2563c (FadE3) | Acyl-CoA dehydrogenase | 66.567 | 4.992 | IQPTVIVDPDPEGPLMTK IDSLYEGTTAIQAQDFFFR ALTDIQGMEASLTGYLMAAQQDVTSLYK FITSGDSDDLFENIFHLVLARPEGAGPGTK ALQALMVENEGAIAQALADDLDRNPVEAFIVDIAATVSEAK | 16% | 95% |
| ML1086 (LpqY) | Solute-binding transport lipoprotein | 50.402 | 5.322 | TLGDMSYTDIGR TALESFVAATQQHVTGEVR LSAQDVVADFSDNLAEELNFR | 8% | 95% |
| ML2095 (Pst3) | Periplasmic phosphate- binding lipoprotein | 37.714 | 5.418 | FVNAFER YPDAQVGR NDESGTTDNFQR IATSADPEPIAISVDSVGK AFLQSTIGGGQNGLGDNGYVPIPDSFK IFNGSIASWNDPAIQALNTGVALPAEPIHVVFR | 33% | 95% |

 Table 4.2 List of CD1a antigen candidates indentified by proteomic analysis of both IEF fraction 4 and SE fraction 8.

MS/MS data of the peptides were searched against the NCBI database using the Mascot or SEQUEST algorithms. The scaffold software 3.0 was used to validate MS/MS based peptide and protein identifications. Protein probabilities were assigned by the Protein Prophet algorithm.

4.3.4 Bioinformatics based identification of a novel CD1a restricted antigen

Since lipoproteins mostly exist as low abundant proteins in the mycobacterial native proteome, bioinformatics based analysis was applied to identify putative lipoproteins using the 'lipobox' which is the consensus sequence of lipoproteins (see section 1.7.1.1; Figure 1.13). T cell reactivities of SE fractions on LCD4.15 suggested that the new CD1a antigen could be a lipoprotein in the range of molecular mass 37 to 55 kDa (Figure 4.6). Besides ML1086 andML2095 which were identified by proteomics analysis, six putative lipoproteins in the molecular mass range were additionally identified as CD1a antigen candidate(s) according to their molecular mass (Table 4.3).

 Table 4.3 List of eight lipoprotein candidates selected by bioinformatics and proteomics analyses.

| Gene | Mass | Alignment of N-terminus with homology in M. tuberculosis |
|---------------|------|---|
| ML0489 | 58.9 | ML0489 LTAC SGSAAGE I VYVVDGAL Rv2585c LTAC SGSAAAQ I DYVVDGAL |
| ML1699, LppZ | 39.1 | ML1699 LSSC ARFNNTQSQPFTTKP Rv3006 SSGC ARFNDAQSQPFTTEP |
| ML1086 , LpqY | 50.4 | ML1086 VLAC GAGGDQLV ISFYTPA Rv1235 AAAC GADSQGLVVSFYTPA |
| ML2446c, LprQ | 46.9 | ML2446c FVAC AGSA IKPSEKKTTPAP Rv0483 LVAC AGKVTKLAEKRPPPAP |
| ML2593, LprK | 42.3 | ML2593 LSSC GWRGISNVSI PGGPG Rv0173 LSSC GWRGISNVAI PGGPG |
| ML0706, PmmB | 56.9 | ML0706 LAAC DPDELAARFTRALRFG Rv3308 LAAC GPDELKARFSRPLAF |
| ML2095, PstS3 | 37.7 | ML2095 LSGC GSDNN AAVGSARTGP Rv0928 LSAC GNDDNVTGGGATTGQA |
| ML0303, GInH | 35.7 | ML0303 LAAC GHAELLT VASMPTLPP Rv0411 LASC GHSETLGV EAT PTLPL |

Grey box shows the lipobox. Pink boxes shows difference between *M. leprae* and *M. tuberculosis* orthologues

4.3.5 Expression of putative lipoproteins

The literature shows that the fast growing avirulent strain, *M. smegmatis*, shares highly conserved phospholipids and the machinery of post translational modification with pathogenic mycobacteria including *M. tuberculosis* and *M. leprae* (13,16,17,36,39,50,54).

Therefore, all genes encoding the selected lipoproteins were cloned into a mycobacterial expression vector, pVV16, and every attempt was made to overexpress these mycobacterial recombinant lipoproteins. However most of these candidates were not expressed in *M. smegmatis* with the exception of recombinant lipoproteins ML2446 and ML2095. These recombinant proteins were expressed at low levels and solely enriched in the cell wall subcellular fraction of recombinant *M. smegmatis* (Figure 4.8).



Figure 4.8 Subcellular localization of recombinant lipoproteins by using Western blot with anti-6-Histidine tag antibody. Lane 1 is whole lysate from *M. smegmatis* containing pVV16. Lane 2, 3, 4 and 5 are cytosolic, membrane, cell wall, and culture supernatant respectively of *M. smegmatis* containing each lipoprotein clone. + is a non-lipoprotein with a 6-Histidine tag as a positive control. Arrows indicate the recombinant lipoproteins. Lpr1086 is the recombinant fusion protein which contains the signal sequence of LprG/Rv1411c and ML1086 without its own signal sequence.

The purpose of this study was to identify and biochemically define the CD1a antigenic portion which would be the N-terminal part of acylated lipoprotein. Therefore, several genetic engineering technologies were applied to express the N-terminal part of lipoproteins in the present study. The first attempt was to express truncated forms of the recombinant lipoproteins which contained an acylated N-terminal part but their transmembrane domains were excluded. Trans-membrane domains of lipoproteins or integral membrane proteins has been found to cause toxicity to the expression host when produced as recombinant protein (26). The second approach was to construct a fusion

gene containing the N-terminal portion of the gene of interest (signal sequence with +20AAs from Cys_{+1}) and the C-terminal protein of rv1411c (without its signal sequence). Several previous studies showed that Rv1411c was successfully over-expressed in M. smegmatis; we expected that the C-terminal protein of rv1411c would help to express the N-terminal parts of target lipoproteins (10, 12, 15). However both approaches were unsuccessful to express the recombinant lipoproteins. Unexpectedly, we observed that a fusion construct (encoding for signal sequence ML1086 and Rv1411c) was expressed with substantially lower level compared to the intact rv1411c construct (data not shown) suggesting that the foreign signal sequence of ML1086 may affect the expression level of recombinant Rv1411c. Therefore, a final attempt was made to construct a fusion gene consisting of *ml1086* (without its signal sequence) and only the signal sequence of rv1411c into the pVV16 expression vector. The resulting lpr1086 is a hybrid of ml1086 and signal sequence of rv1411c. Lpr1086 was successfully expressed in M. smegmatis and enriched into both cell wall and membrane subcellular fractions (Figure 4.8). This method was applied to the remaining lipoproteins but it was not successful.

4.3.6 Purification, characterization and evaluation of T cell reactivity of recombinant Lpr1086, ML2095 and ML2446

The purified recombinant ML2095, ML2446 and Lpr1086 were analyzed by SDS-PAGE and Western blot with anti-6-Histidine tag antibodies (Figure 4.9.A and B). All three recombinant proteins were subjected to in gel tryptic digestion. Subsequent MS/MS data verified that all recombinant proteins were derived from the ORFs of ML1086, ML2095 and ML2446, respectively.



Figure 4.9 SDS-PAGE analysis and CD1a-restricted T cell response to recombinant ML2446, ML2095 and Lpr1086 from *M. smegmatis*. A) Silver or Coomassie staining of SDS-PAGE; B) Western blot hybridized with anti-6-Histidine tag polyclonal antibody (1:5000). C) Proliferative response; D) IFN-γ production of LCD4.15 to rLpr1086, rML2095 and rML2446 Arrow indicates the recombinant protein (see section 4.2.11).

In order to analyze protein acylation, all three recombinant lipoproteins were treated with proteinase K followed by CHCl₃: CH₃OH: H₂O (10:10:3) extraction. The resulting lipopeptides in the organic solution were analyzed by Q-TOF LC/MS (see Chapter 5). Q-TOF LC/MS results verified that Lpr1086, ML 2095 and ML2446 contain di- or tri-acylated glycerylcysteine residues consisting of two or three of fatty acids (Table 5.1). However, all recombinant Lpr1086, ML 2095 and ML2446 expressed in *M. smegmatis* failed to stimulate the CD1a restricted T cell clone, LCD 4.15 (Figure 4.9.C and D).

4.4 Discussion

The major aim of this study was to identify a lipoprotein as a new class of CD1a antigens. This antigen would eventually allow understanding of the immune relationship between TLR activation and CD1a antigen presentation in cutaneous infections.

According to the crystal structures, the dimension of the antigen binding groove is 1300 Å for CD1a representing the smallest binding groove, as compared to 2200 Å for CD1b which forms the largest antigen binding pocket (32,56). While CD1b can load long alkyl chains that are C_{16} to C_{80} in length (2,32), CD1a presents relatively shorter alkyl chain of lipidic antigen but also can accept two short fatty acyl chains. CD1a restricted T cells recognize several T cell antigens including didehydroxymycobactin, and sulfatides (32). In *M. tuberculosis* infection, didehydroxy-mycobactin, a nonribosomal lipopeptide related to mycobactin siderophores, is the only known antigen recognized by mycobacterial specific CD1a restricted T cells clone. The structure of didehydroxymycobactin consists of a short acyl chain (C_{20}) and an invariant 5mer peptide which are linked via ester, oxazaline ring and amide bonds (53) (Figure 1.22). This broad structural similarity provided the hypothesis that bacterial lipoproteins which contain the N-acylated cysteine could be potential antigens for CD1 restricted T cells (31). Recently, Van Rhijn et al. (53) synthesized an N-acyl glycine dodecamer peptide similar to myristosylated peptide found in virus and mammalian cells. This synthetic lipopeptide was recognized by CD1c restricted T cells suggesting that bacterial lipoproteins could serve as CD1 restricted antigens and induce powerful protective immune response. Sieling *et al.* (42) demonstrated that both recombinant LprG/Rv1411c and LprG/ML0557 were recognized by CD4⁺ MHC-II restricted T cell clones from a leprosy lesion and patients' blood.

Therefore, it has been hypothesized that bacterial lipoproteins are the most powerful vaccine candidates which can induce strong innate immunity via TLR-2 signaling and directly cause potent adaptive immunity in response to intracellular pathogens by serving as both MHC-II and CD1 restricted T cell antigens (28,32).

In this dissertation, we found that a CD1a-restricted T-cell line, LCD4.15, from skin lesions of leprosy patients specifically recognized *M. leprae* lysates through antigen presentation of CD1a⁺ Langerhans cells (Figure 4.3.A and B). Unlike other CD1 restricted T cells, the response of LCD4.15 to *M. leprae* lysate was protease sensitive indicating that a bacterial lipoprotein could be an antigen for this clone (Figure 4.3.C). We developed simple proteomic and chemical methods to enrich a minor population of lipoproteins but excluded most of the lipid components which were often found to be the CD1 restricted antigens (Figures 4.4 and 4.5). Subsequent T cell reactivities of delipidated protein fractions revealed that a lipoprotein was possibly a new class of CD1a⁺ antigen recognized by LCD4.15. Here, we provided the tentative evidence that CD1a can present a lipopeptide derived from a novel bacterial lipoprotein in intracellular pathogens.

The three possible antigen candidates, ML1086, ML2095 and ML2446, were identified by proteomic and genomic analysis, were expressed in *M. smegmatis*. Q -TOF LC/MS analysis of all three recombinant proteins showed the presence of di- or tri-acylated glycerylcysteine residues at their N terminus (Table 5.1). However, recombinant ML1086, ML2095 and ML2446, at least expressed in *M. smegmatis* and presumably with the proper lipid function, failed to activate the CD1a restricted T cell clone, LCD4.15 (Figure 4.9). The possible reasons for this outcome are: 1) the antigen

recognized by LCD4.15 could be another lipoprotein, less abundant such that it could not be detected by current proteomic analysis; 2) the recombinant lipoproteins expressed in *in vitro* grown *M. smegmatis* could present different acylation patterns from those of *M. leprae*, and this foreign acylation pattern could disturb the presentation of this protein for CD1a; 3) a nonribosomal lipopeptide, which consists of a short acyl chain and a highly polar peptide such as to be water soluble and synthesized only in *M. leprae*, could be involved in the activation of LCD4.15 by Langerhans cells. The nonribosomal lipopeptides in procaryotes commonly contain D-amino acids rather than L-amino acids which are the building blocks of protein synthesis by ribosomes (40,49). These lipopeptides are mostly protease resistant and also extractable into organic solvent, suggesting that a nonribosomal lipopeptide may be the least possible CD1a restricted antigen candidates.

The knowledge that the activation of a single CD1a T cell clone in mycobacterial infection is restricted to both a dehydroxy unsaturated (C20:1) acyl chain and the peptide of didehydroxy-mycobactin (31), suggests the second hypothesis. The activation of CD1a restricted T cells may be uniquely specific to both lipid and peptide structures compared to that of other group 1 CD1s. *M. tuberculosis* in infected cells metabolize host lipid as a primary carbon source to generate energy via the glyoxylate shunt (18,38). Recently, Carvalho *et al.* (7) revealed that *M. tuberculosis* could simultaneously catabolize multiple carbon sources but other bacteria preferred one or two types of carbon sources to support their growth, providing the evidence that mycobacteria are highly capable of adapting to various environmental conditions, particularly in phagocytes.

availability, in vivo grown mycobacteria contain different lipid profiles from in vitro grown mycobacteria (1,9,18,23,33,35,38,45). *M. tuberculosis* grown *in vivo* were found to possess uncommon structural modifications on several mycobacterial lipids particularly PDIM, sulfolipid-1 and possibly phenolic glycolipid all involved in the pathogenesis of mycobacterial disease (19). Although *M. leprae* has undergone massive gene decay (5,55), genes involved in anabolic pathways and β -oxidation of fatty acids are conserved to favor intracellular survival of *M. leprae* (6). In addition, mycobacteria exogenously uptake oleic acid (C18:1) derived from Tween added to growth media and incorporate it into triacylglycerol under hypoxic conditions that are thought to be similar to conditions in persistent mycobacterial infections (23). Our data (Table 5.1) also showed that bacterial lipoproteins contain several isoforms depending on acylation patterns such as di- or tri- acylated forms with various combinations of fatty acids even in in vitro grown M. smegmatis (see Chapter 5). The data suggest that M. leprae, an obligatory intracellular pathogen, could differently modify its lipoproteins in comparison to *in vitro* grown mycobacteria. Consequently, these distinct acylation patterns of M. *leprae* lipoproteins could result in the species-specific T cell reactivity of LCD4.15. Therefore, further chemical analysis of lipoproteins should be performed on recombinant forms expressed in mycobacteria that are grown under immune pressure which mimic conditions such as nitric oxide or iron stress, hypoxic conditions etc. A major obstacle to performance of this type of study is that Lpr1086, ML2095 and ML2446 were expressed at low levels in *M. smegmatis*. Although it is difficult to obtain sufficient quantities of recombinant lipoproteins for chemical analysis, such trials could provide the general acylation profile of lipoproteins which presumably appear in *in vivo* grown mycobacteria.

Based on the defined lipid profiles of *in vivo* grown mycobacterial lipoproteins, the subsequent synthetic lipopeptides derived from ML1086, ML2095 or ML2446 could identify the CD1a antigens as an intracellular lipid antigen.

Pathogenic mycobacteria continuously develop molecular mechanisms to favor their growth in phagosomes where they can escape from the host immune system but where they obtain nutrients to grow and multiply. Several mycobacterial CD1 antigens are found to be either biosynthetic intermediates or metabolites (21,27,29-31). The structural variants of CD1 antigens in the mycobacterial cell envelope could be major virulence factors (21,53), suggesting that these intracellular lipid CD1 antigens could be naturally selected to aid the mycobacterial survival in phagocytes. Therefore, CD1⁺ T cell immune response to these mycobacterial lipids could be a key mechanism to eliminate intracellular pathogens. *M. leprae* uniquely evolves to survive in the human host implying that the structural modifications of pathogen associated molecular patterns (PAMPs) found in *M. leprae* (described in section 1.6.3) may play a unique biological role in host-M. leprae interaction which has never been fully examined yet. In this context, this novel M. leprae CD1a antigen may allow an understanding of the *M. leprae*-host interaction as well as in designing a novel vaccine against this intracellular pathogen.

4.5 References

- Bacon, J., L. G. Dover, K. A. Hatch, Y. Zhang, J. M. Gomes, S. Kendall, L. Wernisch, N. G. Stoker, P. D. Butcher, G. S. Besra, and P. D. Marsh. 2007. Lipid composition and transcriptional response of *Mycobacterium tuberculosis* grown under iron-limitation in continuous culture: identification of a novel wax ester. Microbiology 153:1435-1444.
- 2. Barral, D. C. and M. B. Brenner. 2007. CD1 antigen presentation: how it works. Nat.Rev.Immunol. 7:929-941.
- 3. Beckman, E. M., S. A. Porcelli, C. T. Morita, S. M. Behar, S. T. Furlong, and M. B. Brenner. 1994. Recognition of a lipid antigen by CD1-restricted ab+ T cells. Nature **372**:691-694.
- 4. **Brigl, M. and M. B. Brenner**. 2004. CD1: antigen presentation and T cell function. Annu.Rev.Immunol. **22**:817-890.
- Cole, S. T., K. Eiglmeier, J. Parkhill, K. D. James, N. R. Thomson, P. R. Wheeler, N. Honore, T. Garnier, C. Churcher, D. Harris, K. Mungall, D. Basham, D. Brown, T. Chillingworth, R. Connor, R. M. Davies, K. Devlin, S. Duthoy, T. Feltwell, A. Fraser, N. Hamlin, S. Holroyd, T. Hornsby, K. Jagels, C. Lacroix, J. Maclean, S. Moule, L. Murphy, K. Oliver, M. A. Quail, M. A. Rajandream, K. M. Rutherford, S. Rutter, K. Seeger, S. Simon, M. Simmonds, J. Skelton, R. Squares, S. Squares, K. Stevens, K. Taylor, S. Whitehead, J. R. Woodward, and B. G. Barrell. 2001. Massive gene decay in the leprosy bacillus. Nature 409:1007-1011.
- Cruz, D., A. D. Watson, C. S. Miller, D. Montoya, M. T. Ochoa, P. A. Sieling, M. A. Gutierrez, M. Navab, S. T. Reddy, J. L. Witztum, A. M. Fogelman, T. H. Rea, D. Eisenberg, J. Berliner, and R. L. Modlin. 2008. Host-derived oxidized phospholipids and HDL regulate innate immunity in human leprosy. J.Clin.Invest 118:2917-2928.
- de Carvalho, L. P., S. M. Fischer, J. Marrero, C. Nathan, S. Ehrt, and K. Y. Rhee. 2010. Metabolomics of *Mycobacterium tuberculosis* reveals compartmentalized co-catabolism of carbon substrates. Chem.Biol. 17:1122-1131.
- de la Salle, H., S. Mariotti, C. Angenieux, M. Gilleron, L. F. Garcia-Alles, D. Malm, T. Berg, S. Paoletti, B. Maitre, L. Mourey, J. Salamero, J. P. Cazenave, D. Hanau, L. Mori, G. Puzo, and L. G. De. 2005. Assistance of microbial glycolipid antigen processing by CD1e. Science 310:1321-1324.
- Domenech, P. and M. B. Reed. 2009. Rapid and spontaneous loss of phthiocerol dimycocerosate (PDIM) from *Mycobacterium tuberculosis* grown *in vitro*: implications for virulence studies. Microbiology 155:3532-3543.
- Drage, M. G., H. C. Tsai, N. D. Pecora, T. Y. Cheng, A. R. Arida, S. Shukla, R. E. Rojas, C. Seshadri, D. B. Moody, W. H. Boom, J. C. Sacchettini, and C. V. Harding. 2010. *Mycobacterium tuberculosis* lipoprotein LprG (Rv1411c) binds triacylated glycolipid agonists of Toll-like receptor 2. Nat.Struct.Mol.Biol. 17:1088-1095.
- 11. **Fischer, F., D. Wolters, M. Rogner, and A. Poetsch**. 2006. Toward the complete membrane proteome: high coverage of integral membrane proteins through transmembrane peptide detection. Mol.Cell Proteomics. **5**:444-453.
- 12. Gehring, A. J., K. M. Dobos, J. T. Belisle, C. V. Harding, and W. H. Boom. 2004. *Mycobacterium tuberculosis* LprG (Rv1411c): a novel TLR-2 ligand that inhibits human macrophage class II MHC antigen processing. J.Immunol. **173**:2660-2668.
- 13. Gibbons, H. S., F. Wolschendorf, M. Abshire, M. Niederweis, and M. Braunstein. 2007. Identification of two *Mycobacterium smegmatis* lipoproteins exported by a SecA2-dependent pathway. J.Bacteriol. **189**:5090-5100.
- Gilleron, M., S. Stenger, Z. Mazorra, F. Wittke, S. Mariotti, G. Bohmer, J. Prandi, L. Mori, G. Puzo, and L. G. De. 2004. Diacylated sulfoglycolipids are novel mycobacterial antigens stimulating CD1-restricted T cells during infection with *Mycobacterium tuberculosis*. J.Exp.Med 199:649-659.

- 15. Harding, C. V. and W. H. Boom. 2010. Regulation of antigen presentation by *Mycobacterium tuberculosis*: a role for Toll-like receptors. Nat.Rev.Microbiol. 8:296-307.
- Herrmann, J. L., R. Delahay, A. Gallagher, B. Robertson, and D. Young. 2000. Analysis of post-translational modification of mycobacterial proteins using a cassette expression system. FEBS Lett. 473:358-362.
- Herrmann, J. L., P. O'Gaora, A. Gallagher, J. E. Thole, and D. B. Young. 1996. Bacterial glycoproteins: a link between glycosylation and proteolytic cleavage of a 19 kDa antigen from *Mycobacterium tuberculosis*. EMBO J. 15:3547-3554.
- 18. Jain, M. and J. S. Cox. 2005. Interaction between polyketide synthase and transporter suggests coupled synthesis and export of virulence lipid in *M. tuberculosis*. PLoS.Pathog. 1:e2.
- 19. Jain, M., C. J. Petzold, M. W. Schelle, M. D. Leavell, J. D. Mougous, C. R. Bertozzi, J. A. Leary, and J. S. Cox. 2007. Lipidomics reveals control of *Mycobacterium tuberculosis* virulence lipids via metabolic coupling. Proc.Natl.Acad.Sci U.S.A **104**:5133-5138.
- Krutzik, S. R., B. Tan, H. Li, M. T. Ochoa, P. T. Liu, S. E. Sharfstein, T. G. Graeber, P. A. Sieling, Y. J. Liu, T. H. Rea, B. R. Bloom, and R. L. Modlin. 2005. TLR activation triggers the rapid differentiation of monocytes into macrophages and dendritic cells. Nat.Med 11:653-660.
- Layre, E., A. Collmann, M. Bastian, S. Mariotti, J. Czaplicki, J. Prandi, L. Mori, S. Stenger, L. G. De, G. Puzo, and M. Gilleron. 2009. Mycolic acids constitute a scaffold for mycobacterial lipid antigens stimulating CD1-restricted T cells. Chem.Biol. 16:82-92.
- 22. Leproma. http://genolist.pasteur.fr/Leproma/
- Low, K. L., P. S. Rao, G. Shui, A. K. Bendt, K. Pethe, T. Dick, and M. R. Wenk. 2009. Triacylglycerol utilization is required for regrowth of *in vitro* hypoxic nonreplicating *Mycobacterium bovis* bacillus Calmette-Guerin. J.Bacteriol. 191:5037-5043.
- 24. Mackler, J. M. and N. E. Reist. 2001. Mutations in the second C2 domain of synaptotagmin disrupt synaptic transmission at Drosophila neuromuscular junctions. J.Comp Neurol. 436:4-16.
- Marques, M. A., A. G. Neves-Ferreira, E. K. da Silveira, R. H. Valente, A. Chapeaurouge, J. Perales, B. R. da Silva, K. M. Dobos, J. S. Spencer, P. J. Brennan, and M. C. Pessolani. 2008. Deciphering the proteomic profile of *Mycobacterium leprae* cell envelope. Proteomics. 8:2477-2491.
- Massey-Gendel, E., A. Zhao, G. Boulting, H. Y. Kim, M. A. Balamotis, L. M. Seligman, R. K. Nakamoto, and J. U. Bowie. 2009. Genetic selection system for improving recombinant membrane protein expression in *E. coli*. Protein Sci 18:372-383.
- Matsunaga, I., A. Bhatt, D. C. Young, T. Y. Cheng, S. J. Eyles, G. S. Besra, V. Briken, S. A. Porcelli, C. E. Costello, W. R. Jacobs, Jr., and D. B. Moody. 2004. *Mycobacterium tuberculosis* pks12 produces a novel polyketide presented by CD1c to T cells. J.Exp.Med 200:1559-1569.
- 28. Moody, D. B. 2006. TLR gateways to CD1 function. Nat.Immunol. 7:811-817.
- Moody, D. B., B. B. Reinhold, M. R. Guy, E. M. Beckman, D. E. Frederique, S. T. Furlong, S. Ye, V. N. Reinhold, P. A. Sieling, R. L. Modlin, G. S. Besra, and S. A. Porcelli. 1997. Structural requirements for glycolipid antigen recognition by CD1b-restricted T cells. Science 278:283-286.
- Moody, D. B., T. Ulrichs, W. Muhlecker, D. C. Young, S. S. Gurcha, E. Grant, J. P. Rosat, M. B. Brenner, C. E. Costello, G. S. Besra, and S. A. Porcelli. 2000. CD1c-mediated T-cell recognition of isoprenoid glycolipids in *Mycobacterium tuberculosis* infection. Nature 404:884-888.
- Moody, D. B., D. C. Young, T. Y. Cheng, J. P. Rosat, C. Roura-Mir, P. B. O'Connor, D. M. Zajonc, A. Walz, M. J. Miller, S. B. Levery, I. A. Wilson, C. E. Costello, and M. B. Brenner. 2004. T cell activation by lipopeptide antigens. Science 303:527-531.
- 32. Moody, D. B., D. M. Zajonc, and I. A. Wilson. 2005. Anatomy of CD1-lipid antigen complexes. Nat.Rev.Immunol. 5:387-399.

- Munoz-Elias, E. J. and J. D. McKinney. 2006. Carbon metabolism of intracellular bacteria. Cell Microbiol. 8:10-22.
- Ochoa, M. T., S. Stenger, P. A. Sieling, S. Thoma-Uszynski, S. Sabet, S. Cho, A. M. Krensky, M. Rollinghoff, S. E. Nunes, A. E. Burdick, T. H. Rea, and R. L. Modlin. 2001. T-cell release of granulysin contributes to host defense in leprosy. Nat.Med 7:174-179.
- 35. Pandey, A. K. and C. M. Sassetti. 2008. Mycobacterial persistence requires the utilization of host cholesterol. Proc.Natl.Acad.Sci.U.S.A 105:4376-4380.
- Rezwan, M., T. Grau, A. Tschumi, and P. Sander. 2007. Lipoprotein synthesis in mycobacteria. Microbiology 153:652-658.
- Roura-Mir, C., L. Wang, T. Y. Cheng, I. Matsunaga, C. C. Dascher, S. L. Peng, M. J. Fenton, C. Kirschning, and D. B. Moody. 2005. *Mycobacterium tuberculosis* regulates CD1 antigen presentation pathways through TLR-2. J.Immunol. 175:1758-1766.
- 38. **Russell, D. G.** 2001. *Mycobacterium tuberculosis*: here today, and here tomorrow. Nat.Rev.Mol.Cell Biol. **2**:569-577.
- Sartain, M. J. and J. T. Belisle. 2009. N-Terminal clustering of the O-glycosylation sites in the Mycobacterium tuberculosis lipoprotein SodC. Glycobiology 19:38-51.
- 40. Schorey, J. S. and L. Sweet. 2008. The mycobacterial glycopeptidolipids: structure, function, and their role in pathogenesis. Glycobiology 18:832-841.
- Sieling, P. A., D. Chatterjee, S. A. Porcelli, T. I. Prigozy, R. J. Mazzaccaro, T. Soriano, B. R. Bloom, M. B. Brenner, M. Kronenberg, P. J. Brennan, and . 1995. CD1-restricted T cell recognition of microbial lipoglycan antigens. Science 269:227-230.
- 42. Sieling, P. A., P. J. Hill, K. M. Dobos, K. Brookman, A. M. Kuhlman, M. Fabri, S. R. Krutzik, T. H. Rea, D. G. Heaslip, J. T. Belisle, and R. L. Modlin. 2008. Conserved mycobacterial lipoglycoproteins activate TLR2 but also require glycosylation for MHC class II-restricted T cell activation. J.Immunol. 180:5833-5842.
- 43. Sieling, P. A., D. Jullien, M. Dahlem, T. F. Tedder, T. H. Rea, R. L. Modlin, and S. A. Porcelli. 1999. CD1 expression by dendritic cells in human leprosy lesions: correlation with effective host immunity. J.Immunol. 162:1851-1858.
- 44. Sieling, P. A., J. B. Torrelles, S. Stenger, W. Chung, A. E. Burdick, T. H. Rea, P. J. Brennan, J. T. Belisle, S. A. Porcelli, and R. L. Modlin. 2005. The human CD1-restricted T cell repertoire is limited to cross-reactive antigens: implications for host responses against immunologically related pathogens. J.Immunol. 174:2637-2644.
- Sirakova, T. D., V. S. Dubey, C. Deb, J. Daniel, T. A. Korotkova, B. Abomoelak, and P. E. Kolattukudy. 2006. Identification of a diacylglycerol acyltransferase gene involved in accumulation of triacylglycerol in *Mycobacterium tuberculosis* under stress. Microbiology 152:2717-2725.
- 46. Spencer, J. S., H. J. Kim, A. M. Marques, M. Gonzalez-Juarerro, M. C. Lima, V. D. Vissa, R. W. Truman, M. L. Gennaro, S. N. Cho, S. T. Cole, and P. J. Brennan. 2004. Comparative analysis of B- and T-cell epitopes of *Mycobacterium leprae* and *Mycobacterium tuberculosis* culture filtrate protein 10. Infect.Immun. 72:3161-3170.
- Stenger, S., D. A. Hanson, R. Teitelbaum, P. Dewan, K. R. Niazi, C. J. Froelich, T. Ganz, S. Thoma-Uszynski, A. Melian, C. Bogdan, S. A. Porcelli, B. R. Bloom, A. M. Krensky, and R. L. Modlin. 1998. An antimicrobial activity of cytolytic T cells mediated by granulysin. Science 282:121-125.
- Stenger, S., R. J. Mazzaccaro, K. Uyemura, S. Cho, P. F. Barnes, J. P. Rosat, A. Sette, M. B. Brenner, S. A. Porcelli, B. R. Bloom, and R. L. Modlin. 1997. Differential effects of cytolytic T cell subsets on intracellular infection. Science 276:1684-1687.
- 49. Strieker, M. and M. A. Marahiel. 2009. The structural diversity of acidic lipopeptide antibiotics. Chembiochem. 10:607-616.

- Tschumi, A., C. Nai, Y. Auchli, P. Hunziker, P. Gehrig, P. Keller, T. Grau, and P. Sander. 2009. Identification of apolipoprotein N-acyltransferase (Lnt) in mycobacteria. J.Biol.Chem. 284:27146-27156.
- 51. Tuberculolist. http://genolist.pasteur.fr/TubercuList/
- 52. Valladeau, J., V. Duvert-Frances, J. J. Pin, C. Dezutter-Dambuyant, C. Vincent, C. Massacrier, J. Vincent, K. Yoneda, J. Banchereau, C. Caux, J. Davoust, and S. Saeland. 1999. The monoclonal antibody DCGM4 recognizes Langerin, a protein specific of Langerhans cells, and is rapidly internalized from the cell surface. Eur.J.Immunol. 29:2695-2704.
- 53. Van, R., I, D. C. Young, J. A. De, J. Vazquez, T. Y. Cheng, R. Talekar, D. C. Barral, L. Leon, M. B. Brenner, J. T. Katz, R. Riese, R. M. Ruprecht, P. B. O'Connor, C. E. Costello, S. A. Porcelli, V. Briken, and D. B. Moody. 2009. CD1c bypasses lysosomes to present a lipopeptide antigen with 12 amino acids. J.Exp.Med. 206:1409-1422.
- 54. VanderVen, B. C., J. D. Harder, D. C. Crick, and J. T. Belisle. 2005. Export-mediated assembly of mycobacterial glycoproteins parallels eukaryotic pathways. Science **309**:941-943.
- 55. Vissa, V. D. and P. J. Brennan. 2001. The genome of *Mycobacterium leprae*: a minimal mycobacterial gene set. Genome Biol. 2:REVIEWS1023.
- 56. Zajonc, D. M., M. A. Elsliger, L. Teyton, and I. A. Wilson. 2003. Crystal structure of CD1a in complex with a sulfatide self antigen at a resolution of 2.15 A. Nat.Immunol. 4:808-815.

CHAPTER 5 STRUCTURAL ANALYSIS OF THE LIPID COMPONENTS OF MYCOBACTERIAL LIPOPROTEINS

From a manuscript form in preparation by Hee Jin Kim, John T. Belisle, Mark J. Sartain, Sebabrata Mahapatra and Patrick J. Brennan. Department of Microbiology, Immunology and Pathology, Colorado State University, Fort Collins, Colorado 80523-1677, USA;

5.1 Abstract

Bacterial lipoproteins, containing an N-acyl di-O-acylglyceryl-cysteine unit, have long been known to affect the immune system through macrophage stimulation, and, in general, the N-terminal triacylated region is important in the cell-mediated immune response to bacterial infections through Toll like receptor 2 (TLR-2). The presence of acylated cysteine at the N terminus of lipoprotein is primarily predicted by bioinformatic analysis but rarely verified by biochemical analysis. The aim of this study was to define the structure of the N-acylated mycobacterial lipoproteins. Lipopeptides derived from mycobacterial lipoproteins were prepared by proteolysis and organic extraction, and analyzed by gas chromatography-mass spectrometry (GC-MS) and quadrapole time of flight liquid chromatography/mass spectrometry (Q-TOF LC/MS). GC-MS results revealed that the major fatty acids of the glyceryl portion of the lipoproteins are tuberculostearic acid, stearic acid, oleic acid and palmitic acid, which are also the dominant fatty acids of mycobacterial phospholipids. The N-linked fatty acid of the cysteine residue is predominantly palmitic acid. Interestingly, Q-TOF LC/MS data showed that mycobacterial lipoproteins appeared as di- and tri-acylated mature forms, but

also had an array of fatty acids in various combinations. Di-acylated lipopeptides appeared more frequently than tri-acylated lipopeptides. The present study demonstrated a distinct fatty acid composition of the lipid portion of mycobacterial lipoproteins. Moreover, we defined the full spectrum of the acylation pattern in mycobacterial lipoproteins, the impact of which in host-bacteria responses is subject to further immunological investigation.

5.2 Introduction

Bacterial lipoproteins, those containing lipid covalently linked to an N-terminal cysteine residue, have been extensively studied in Gram-negative bacteria (4,15). The lipidated N-terminus is presumed to contribute several physiological features of bacteria such as cell wall integrity, transport systems, and signal transduction into the outer leaflet of the cytoplasmic membrane (16, 32, 38). Analysis of the signal sequence of the Nterminus of prolipoproteins has revealed the consensus sequence Leu-(Ala, Ser)-(Gly, Ala)-Cys at -3 to +1 positions (in relation of cysteine), known as the "lipobox". This motif representes the signal cleavage region of about three-fourths of all bacterial lipoproteins (15,37) (see section 1.7.1). The prolipoprotein undergoes sequential modification and processing catalyzed by diacylglyceryl transferase (Lgt), signal peptidase II (LspA), and N-acyl transferase (Lnt) to form mature lipoproteins, and the major glycerophospholipids such as phosphatidylglycerol, phosphatidylethanolamine and cardiolipin serve as fatty acyl donors for this reaction (15) (Figure 1.14). The cysteine (Cys) residue at the +1 site contains a sulfhydryl-linked diacylglycerol and an N-linked fatty acid (4,15) (Figure 1.13). The ester-linked fatty acids in lipoproteins are generally

similar in composition to those of bulk phospholipids, and 65% of the amide-linked fatty acids are palmitic acids (4).

About 99 putative lipoproteins with the consensus sequence for lipid modification have been identified in the *M. tuberculosis* genome and 31 in the case of *M. leprae* (21,24,38,42) (Table 1.3). The immunodominant antigens, 19kDa/LpqH and 38kDa/PhoS1 of *M. tuberculosis*, have been described as lipoprotein based on metabolic labeling with [³⁵S]-Met and [^{9,10}H]-palmitic acid) (48). Recently, MALDI-TOF analysis of in gel digested lipopeptides provided the evidence that the mycobacterial lipoproteins contain the species specific tuberculostearic acid derived from phospholipids as the acyl donor (41).

Several publications indicate that mycobacterial lipoproteins have affected the immune response to bacterial infections through macrophage activation and the release of certain cytokines, leading to the amplification of Th1 immunity (6,11,14,22,27). Although current research highlights the role of lipoproteins for the development of new immune therapies and vaccines, two major gaps in our knowledge of mycobacterial lipoproteins are the fact that the evidence for *M. leprae* protein acylation has largely relied on bioinformatic analysis; and that the chemical structures of *M. leprae* lipoproteins are presumed to be identical in all mycobacteria. Therefore, the goal of this present study was to define the full spectrum of acylation pattern in mycobacterial lipoproteins.

5.3 Materials and methods

5.3.1 Preparation of a hydrophobic protein fraction from Mycobacterium smegmatis through Triton X-114 phase separation

M. smegmatis was grown in a glycerol/alanine salts medium to mid-log phase at 37 °C, centrifuged at 4 °C (10 g, wet weight), washed with PBS, resuspended in the same buffer at 4 °C, and disrupted by intermittent probe sonication with a Soniprep 150 sonicator (MSE Sonic prep150, MSE-Sanyo, Palisades Park, NJ) for 30 min. The suspension of disrupted cells was diluted to a protein concentration of 1mg/ml with PBS followed by the addition of Triton X-114 (Sigma, St. Louis, MO) to a final concentration of 4%. The solution was stirred at 4 °C for 24 hr, and the cellular debris was removed by centrifugation at $27,000 \times g$ for 20 min at 4°C (3). The supernatant was maintained at 25°C for 2 hr until clouding occurred, then centrifuged at $1500 \times g$ for 10 min at 4 °C. The proteins in the resulting aqueous and detergent phases were recovered by precipitation with 5 volumes of chilled (-20 °C) acetone and centrifugation (3). The hydrophobic proteins arising from the detergent phase were extracted with $CHCl_3:CH_3OH(2:1, v/v)$ to remove free fatty acids, and a phenol-water partition step was applied to remove lipomannan/lipoarabinomannan (LMs/LAMs) (27). The hydrophobic proteins at the interphase of the phenol partition were lyophilized and the concentration of the proteins was determined by bicinchoninic acid assay (Pierce, Rockford, IL). The hydrophobic and hydrophilic protein populations were analyzed by SDS-PAGE and 2-D IEF gel electrophoresis (26,27).

5.3.2 Proteolysis of M. smegmatis hydrophobic proteins and release of the esterlinked fatty acids

The hydrophobic protein fraction was suspended in 200 ml of PBS and incubated with 10 unit/ml proteinase K (Sigma, St. Louis, MO) overnight at 37°C. The reaction was stopped by the addition of 200 ml CHCl₃:CH₃OH (2:1, v/v) and the lipopeptides extracted into CHCl₃-CH₃OH (2:1, v/v). These were subjected to thin layer chromatography (TLC) in CHCl₃:CH₃OH:H₂O (65:35:5), and the plates developed with 0.2% ninhydrin in 95% C₂H₅OH. The ester-linked fatty acids of the lipopeptides were released by mild alkali treatment with CHCl₃:CH₃OH:0.8N NaOH (10:10:3, v/v/v) at 50 °C for 30 min (7). Free fatty acids were separated from the deacylated peptides by partitioning between hexane and water (6:1, v/v) (25). The deacylated peptides in the water phase were subjected to TLC in CHCl₃:CH₃OH:H₂O (65:35:5) and the plates developed with ninhydrin.

5.3.3 Purification and analysis of lipopeptides from M. smegmatis hydrophobic proteins

The lipopeptides from preparative TLC plates were solubilized in 1 ml of 25 mM octyl glucopyranoside in 0.1 M ammonium acetate (pH 7.0) and 50 mM CaCl₂ (22). Insoluble materials were removed by centrifugation and the soluble lipopeptides were applied to a preparative (10×250 mm) Macherey and Nagel HPLC column with Nucleosil 120-7C₁₈ as stationary phase and eluted at 40 °C with a water/2-propanol gradient (29,30). The amide-linked fatty acids were prepared by hydrolysis of the deacylated lipopeptides with 6 N HCl at 110 °C for 18 hr. The free fatty acids were

methyl esterified with 3M methanolic HCl (Supelco Inc, Bellefonte, PA) at 80 °C for 16 hr (1) recovered by partitioning between hexane and water and analyzed by gas chromatography on a BPX-70 column (Supelco Inc, Bellefonte, PA) in a model 5890 Gas Chromatogram (Hewlett-Packard) with flame ionization detector. For analysis of amino acids, the lipopeptides were hydrolyzed with 6 N HCl (Thermo Fisher Scientific, Rockford, IL) at 110 °C for 18 hr under N₂ gas and sequentially derivatized with 3N isobutanolic HCl and heptafluorobutyric anhydride, at 120 °C for 20 min and 150 °C for 5 min, respectively (25). The N-(O)-heptafluorobutyryl isobutyl amino acid esters of the lipopeptides were analyzed by gas chromatography/electron impact mass spectrometry (GC/EIMS) (Finnigan Polanis Ion Trap Mass Spectrometer). L- α -aminoadipic acid (10nmol) was added as internal standard.

5.3.4 Quadrapole time of flight (Q-TOF) LC/MS analysis of lipopeptides from recombinant lipoproteins Lpr1086, ML2095 and ML2446

The recombinant lipoproteins Lpr1086, ML2095 and ML2446 (see chapter 6) were suspended in 200 μ l of water and incubated with 0.5 unit/ μ l of proteinase K (Sigma, St. Louis, MO) for overnight at 37 °C. A negative control reaction was included with the same treatment to exclude false value for MS data analysis. The reactions were stopped by heat inactivation (100 °C for 10 min). The resulting lipopeptides were extracted into CHCl₃-CH₃OH (10:10:3, v/v/v). The samples were clarified by centrifugation and subjected to liquid chromatography-mass spectrometry (LC-MS). Ten microliters of lipopeptides were applied either to a Waters X-Bridge (Waters, Milford, MA,) C₁₈ column 3.5 μ m (2.1 × 150 mm) that was connected to an Agilent 1200 series HPLC

system. The lipopeptides were eluted with 99% CH_3OH , 5 mM CH_3COONH_4 to 100% linear gradient of 79% n-propyl alcohol, 20% hexane, 5 mM CH₃COONH₄ at a flow rate of 320 µl/min. The eluant was introduced directly into either an Agilent 6220 Accurate-Mass TOF or Agilent 6250 Q-TOF mass spectrometer (Agilent, Santa Clara, CA,) quipped with an Agilent multimode source with simultaneous ESI and atmospheric pressure chemical ionization capability. The positive ion MS data were collected using Agilent Mass Hunter work station software (Agilent, Santa Clara, CA). MS data obtained from each sample were processed using the Molecular Feature Extractor of the Mass Hunter Qualitative Analysis software to generate a file of molecular mass features, their retention times, and their relative abundance. A preset minimum 100 counts as cut off was used to filter out extremely low abundance molecular features and were searched against a custom database containing a list of all possible lipopeptides generated from each lipoprotein. The resulting MS data were processed with Agilent MassHunter Qualitative Analysis software to extract molecular features; i.e. obtain a defined exact mass and retention times of compounds in each sample. The relative abundance of these filtered molecular features were compared with the negative reaction control and indentified true lipopeptides which are significantly higher (at least 5 fold) in relative abundance.

5.4 Results

5.4.1 Preparation and analysis of lipoproteins of M. smegmatis

Phase separation of the lysate of *M. smegmatis* in Triton X-114 allowed a clear differentiation of proteins into the hydrophilic population found exclusively in the

aqueous phase and the hydrophobic proteins in the detergent phase (3,27). The hydrophobic protein fraction (Figure 5.1.A) was subjected to proteolysis to generate the lipopeptides from the inherent lipoproteins extracted with $CHCl_3:CH_3OH$ (2:1, v/v) and analyzed by TLC. Ninhydrin showed violet spots for the resolved lipopeptides, whereas the undigested proteins stayed at the origin (Figure 5.1.B). The lipopeptides were isolated by preparative TLC and subjected to mild alkali-treatment to remove the ester-linked fatty acids, which applied to TLC in the same solvent resulted in a marked change in polarity in that the now deacylated lipopeptides remained at the origin and the free fatty acids migrated in the plate (Figure 5.1.C).



Figure 5.1 Proteolysis of lipoproteins and thin layer chromatography analysis of the alkali-treated lipopeptides. The lipoproteins from *M. smegmatis* were subjected to proteolysis with 10 unit/ml proteinase K for 2 hr at 37° C. The lipopeptides were extracted with CHCl₃:CH₃OH (2:1, v/v) and subjected to TLC in CHCl₃:CH₃OH:H₂O (65:35:5), and the plates were developed with 0.2% ninhydrin. A) SDS-PAGE, Lane 1, standard molecular weight maker; lane 2, lipoproteins; lane 3, lipoproteins after proteolysis: B) TLC of lipopeptides. Lane 1, lipoproteins; lane 2, lipopeptides in organic phase after extraction with organic solvent; lane 3, peptides in the water phase after extraction with organic solvent; lane 3, peptides in the water phase after extraction with organic solvent; lane 1, lipoproteins; lane 2, lipopeptides; lane 3, deacylated peptides after mild alkali-treatment.

5.4.2 Purification and analysis of lipopeptides

Four products isolated by HPLC (Figure 5.2.A) were subjected to TLC analysis. Only P3 reacted strongly to both ninhydrin and charing reagents (Figure 5.2.B and C) but was phosphorus-negative (data not shown). To identify the fatty acids of the lipopeptides of the *M. smegmatis* lipoproteins, the HPLC-purified peak P3 was subjected to mild alkali treatment followed by acid hydrolysis with 6N HCl at 110°C for 18 hr and the alkali- and acid-released fatty acids were separately analyzed by GC-MS.



Figure 5.2 Purification of lipopeptides by reverse-phase HPLC. Lipopeptides were reconstituted with octyl glucopyranoside and applied to a 10×250 mm reverse-phase C₁₈ column. Elution was at 40°C with a water/2-propanol gradient. A) HPLC profile of lipopeptide: B) TLC of HPLC purified peaks stained by ninhydrin ; Lane 1, P1; lane 2, P2; lane 3, P3; lane 4, P4: C) TLC of the HPLC purified peaks stained by 10% cupric sulfate.

The alkali-labile fatty acids from the diacylglycerol portion of the lipopeptides were identified as tuberculostearic acid (C19:0) (m/z 312), stearic acid (C18:0) (m/z 298) and palmitic acid (C16:0) (m/z 270) (Figure 5.3.A) (33), whereas the alkali-stable acid-labile fatty acids were all palmitic acid (C16:0) (Figure 5.3.B). In some other preparations of lipopeptides, small quantities of oleic acid were observed.



Figure 5.3 Analysis of the fatty acids of lipopeptides by GC-MS. The ester-linked fatty acids of the lipopeptides had been released by mild alkali treatment. The amide-liked fatty acids were prepared by hydrolysis of the deacylated lipopeptide with 6N HCl and hexane extraction. A) GC-MS profile of ester-linked fatty acids of the lipopeptide. B) GC-MS profile of amide-linked fatty acids of the lipopeptide.

After 6N HCl hydrolysis and derivatization of the lipopeptide to identify the amino acid composition, the N(O)-heptafluorobutyryl isobutyl amino acid esters were analyzed by GC-MS, which demonstrated the presence of lysine (m/z 280, 50%), arginine (m/z 504, 6%), leucine (m/z 240, 36.7%; m/z 282, 13.2%), threonine (m/z 252, 43.6%; m/z 253, 46%) (27). Lysine, arginine, leucine, and threonine were in the ratio of about 2:1:1:1 (Figure 5.4). The high content of basic amino acids explains the ninhydrin reactivity of the lipopeptides of the lipoproteins. In order to define further chemical structure, this lipopeptide was applied to subsequent MALDI-TOF MS analysis but no

identifiable fractions were obtained. The last step, unveiled for obligatory definition of the functionally crucial lipopeptdes of mycobacterial lipoproteins will be repeated.



Figure 5.4 Amino acid analysis of lipopeptides by GC-MS. The lipopeptides were hydrolyzed with 6N HCl and dervatized into N(O)-heptafluorobutyryl amino acid esters. Amino acid identification was established by their retention time and typical ion mass fragmentation patterns.

5.4.3 Identification of lipopeptides from recombinant Lpr1086, ML2095 and ML2446 by Q-TOF LC/MS analysis

Purified recombinant Lpr1086, ML2095 and ML2446 (see Chapter 4) were treated with proteinase K and subjected to organic extraction. Subsequently, the extracted lipopeptides were subjected to Q-TOF LC/MS analysis. In order to experimentally identify lipopeptides, the custom lipopeptide databases of Lpr 1086, ML2095 and ML2446 were constructed by calculating exact mass of hypothetical lipopeptides in combination of fatty acids including palmitic acid (C16), pamitoleic acid (C16:1), stearic acid (C18), oleic acid (C18:1) and tuberculostearic acid (C19). Each database contained theoretical masses of all hypothetical lipopeptides containing di-and tri-acylated glycerylcysteines, to di- and tri-acylated 20mer peptide. Since some of lipoproteins were found to be glycosylated, the custom database also contained theoretical masses of mono- to penta-glycosylated units with di- and tri-acylated 20mer

peptide. The resulting MS data of each lipopeptide pool were searched against this custom database and were analyzed to evaluate the chemical structure based on the retention time and the exact mass of each compound. Lipopeptides experimentally found were summarized and compared with theoretical mass in Table 5.1. The amino acid sequences of these peptides are shown in Figure 5.5. Q-TOF LC/MS data analysis additionally identified 18, 25 and 12 possible lipopeptides from custom databases of ML2095, ML2446 and Lpr1086 respectively but was unable to create their molecular formula from experimental masses (data not shown); the masses of these unverified lipopeptides were greater than 1500 resulting in an inability to generate molecular formula. Present Q-TOF LC/MS analysis revealed that ML2095, ML2446 and Lpr1086 were found as both di-acylated and tri-acylated lipoproteins. Palmitic acid, oleic acid, stearic acid and tuberculostearic acid were heterogeneously found in glycerylcysteine of each lipopeptides but palmitoleic acid was not found. Interestingly, the di-acylated forms were found more frequently than tri-acylated forms of recombinant lipoproteins expressed in *M. smegmatis*.

ML2095 <u>GlyCys</u>GS DNN AA VGSAR TGPSSGQVSCGGKPTLKASGSTAQAN

ML2446 <u>GlyCys</u>AGS AI KP SE KKTTPAPHLTFQPATA TDDVIPVAPISVQIADG

Lpr1086
<u>GlvCys</u>GAGG DQ LVI SFYTPASEDATFTEVARRCTEQFDG

Figure 5.5 N- terminal amino acid sequence of mature ML2095, ML2446 and Lpr1086 and a map of lipopeptides observed by Q-TOF LC/MS analysis. GlyCys represents glycerylcysteine. Each line indicates peptide sequences that were found by Q-TOF LC/MS analysis.
| Source of Lipopeptides | Observed Lipopeptides | Theoretical Mass | Experimental Mass | Retention Time (min) | Molecular Formula |
|---------------------------|-------------------------------|---------------------|----------------------|-------------------------|---|
| Lipopopulatio | | 111105 | 111055 | | 1 official |
| All of three | C16,C16-GlyCys | 671.5159 | 671.5176 | 1.88 | $C_{38}H_{73}NO_6S$ |
| recombinant | C16,C16,C19-GlyCys | 951.7925 | 951.7909 | 12.16 | $C_{57}H_{109}NO_7S$ |
| lipoproteins | C16C,16,C18:1-GlyCys | 935.7609 | 935.7607 | 11.49 | $C_{56}H_{105}NO_7S$ |
| ML2095 | C18,C18-GlyCysGSDNN | 1214.7447 | 1214.7476 | 3.61 | $C_{59}H_{108}N_8O_{16}S$ |
| | C19,C19-GlyCysGSDNN | 1242.7760 | 1242.7822 | 4.5 | $C_{61}H_{114}N_8O_{16}S$ |
| | | | 1342.8080 | 5.02 | $C_{64}H_{116}N_{10}O_{18}$ |
| | C16,C19-GlyCysGSDNNAA | 1342.8034 | | | S |
| | | | 1368.8135 | 5.78 | $C_{66}H_{120}N_{10}O_{18}$ |
| | C18,C19-GlyCysGSDNNAA | 1368.814 | | | S |
| | | | 1354.7998 | 5.85 | $C_{65}H_{116}N_{10}O_{18}$ |
| | C18,C18:1-GlyCysGSDNNAA | 1354.803 | | | S |
| | | | 1827.0830 | 4.62 | $C_{80}H_{144}N_{18}O_{25}$ |
| | C16,C16-GlyCysGSDNNAAVGSAR | 1827.0792 | | | S |
| | C16,C18,C18-GlyCysGS | 1109.8615 | 1109.8616 | 14.61 | $C_{63}H_{119}N_{3}O_{11}S$ |
| ML2446 | C16,C18:1-GlyCysAGSAIKP | 1250.8539 | 1250.8504 | 5.51 | C ₆₉ H ₁₂₇ N ₉ O ₁₄ S |
| | C19,C18:1-GlyCysAGSAIKP | 1292.9032 | 1292.9054 | 7.34 | $C_{72}H_{133}N_9O_{14}S$ |
| | C16, C18, C18-GlyCysAGS | 1109.8616 | 1106.8592 | 14.37 | $C_{66}H_{124}N_4O_{11}S$ |
| | C16, C18, C19-GlyCysAGS | 1123.8772 | 1123.8757 | 14.87 | C ₆₇ H ₁₂₆ N ₄ O ₁₁ S |
| | C16,C18,C18:1-GlyCysAGSAI | 1289.9501 | 1289.9468 | 12.29 | C ₇₆ H ₁₄₂ N ₆ O ₁₄ S |
| | | | 1775.2357 | 16.23 | $C_{98}H_{179}N_{11}O_{20}$ |
| | C18,C18:1,C19-GlyCysAGSAIKPSE | 1775.2364 | | | S |
| Lpr1086 | C16,C18:1-GlyCysGAGG | 939.6330 | 939.6307 | 3.42 | $C_{49}H_{80}N_5O_{10}S$ |
| r | C18:1.C19-GlyCysGAGG | 981.6799 | 981.6763 | 3.57 | C52H95N5O10S |
| | , | | 1549,9987 | 4.83 | C78H141N11O18 |
| | C18.C19-GlyCysGAGGDOLVI | 1550.0020 | | | S |
| | C16.C16.C19-GlvCvsGAGGDO | 1436.9795 | 1436.9811 | 14.26 | C75H136N8O17S |
| | C16,C16,C19-GlyCysGAGGDOLVI | | 1762.2132 | 15.34 | $C_{92}H_{167}N_{11}O_{19}$ |
| | , , | 1762.2160 | | | S |

Table 5.1 List of lipopeptides observed by Q-TOF LC/MS analysis.

C16, C18, C18:1 and C19 represent palmitic acid, stearic acid, oleic acid and tuberculostearic acid, respectively.

5.5 Discussion

The aim of this work was to develop methods for the study of mycobacterial lipoproteins and, particularly, to identify the fatty acid composition of the N-terminal acylated region and to explain the biological roles of protein acylation in host pathogen interaction.

Although MALDI-TOF analysis of in gel digested lipopeptide provided the

biochemical evidence of mycobacterial protein acylation (41), the existence of

lipoproteins in *Mycobacterium* is still supported predominantly by the presence of lipoprotein consensus sequences in the genome databases and the report of Young and Garbe (48) in demonstrating that metabolic labeling of *M. tuberculosis* with [³H]-palmitic acid and L-[³⁵S]-cysteine followed by phase separation demonstrated that four major antigens, 19 kDa, 26 kDa, 27 kDa, and 38 kDa, were labeled (48).

In the present study, proteolysis of the hydrophobic protein fraction followed by lipid extraction was the single most important step in allowing identification of the fatty acid composition of mycobacterial lipoproteins. The mycobacterial lipoproteins are predominantly substituted with palmitic acid, stearic acid, and tuberculostearic acid in the glyceryl portion and palmitic acid in amide-linkage to the cysteine residue (Figure 5.3 and Table 5.1). The acylation pattern of other bacterial lipoproteins is usually dominated by palmitic acid (13), oleic acid (28,49), or hydroxystearic acid (44). Although chemical composition of the lipid portion of bacterial lipoproteins is rarely characterized, it is believed that the acylation profiles of bacterial lipoproteins are largely influenced by the distribution of fatty acids on the cellular lipids, particularly phosphoglycerides (1,12,15). The chemical composition of these phospholipids is varied depending on culture conditions (1,13). Our results (Figure 5.3) and those of Tschumi et al. (41) indicate that the mycobacterial lipoproteins have an array of fatty acids similar to those of mycobacterial phospholipids (5,19,41). The presence of tuberculostearic acid is noteworthy, since it is mostly associated with phosphatidyl inositol, the phosphatidyl inositol mannosides, LM and LAM (19), implying that turnover of these characteristic mycobacterial phospholipids may contribute to the acylation pattern of mycobacterial lipoprotein.

However, MALDI-TOF analysis of the lipopeptides derived from *M. smegmatis* native lipoproteins failed to elucidate the exact biochemical structure. This inability of MS based analysis may have been due to the long hydrocarbon chains of the lipid portion resulting in low recovery into the analytic instrument as well as low ionization efficiency during MS (43). This problem is accentuated when analytic materials are heterogeneous and low abundant similar to the lipopeptides derived from *M. smegmatis* native lipoproteins.

In order to define the chemical structure of mycobacterial lipoproteins, we identified eight ORFs coding for proteins containing the lipobox (Table 4.3) from the *M. leprae* genome and expressed in *M. smegmatis*. However, attempts to express many of the recombinant equivalents were mostly unsuccessful, since, presumably, the recombinant lipoproteins were highly toxic for the host cell and expressed at low level. The *advancement of analytical techniques has* made it possible to identify and quantify low abundant compounds from complex biological samples such as urine and serum (2,8,9). We developed a strategy based on an HPLC separation system with high mass accurate Q-TOF LC/MS detection of mycobacterial lipids. In order to overcome difficulties on MS based analysis of lipopeptide, we utilized this platform to directly further analyze low abundant lipopeptides.

The biosynthetic machinery of bacterial lipoproteins was first identified and fully characterized in Gram-negative bacteria (15,31,35,36,46). The lipoprotein biosynthetic machinery consists of three enzymes. The diacylglyceryl transferase (Lgt) covalently links the diacylglycerol residue at the thiol group of Cys in the lipobox (16,35). Subsequently, signal peptidase II (LspA) cleaves the signal peptide of diacylglyceryl

prolipoprotein (31,47) and the N-acyl transferase (Lnt) (16) adds an amide-linked fatty acid (Figure 1.14). Lgt and LspA are conserved in all bacteria but Lnt is predominantly found in Gram-negative bacteria (31). Chemical analysis showed that several lipoproteins exist as the tri-acylated form in Gram-negative bacteria (E. coli, Borrelia burgdorferi, Neisseria gonorrhoeae and Porphyromonas gingivalis) but those of mycoplasma were the di-acylated forms (39,40). Although Bacillus subtilis and Staphlycoccus aureus were reported to contain Lnt, there is little known of protein acylation in Gram-positive bacteria (39-41). The three essential enzymes, Lgt, LspA and Lnt (1,15,31), have been identified in the genomes of *M. smegmatis*, *M. tuberculosis* and *M. leprae* by homology alignment (31,32,34,41). The enzymatic activity of *M*. *tuberculosis* Lnt has been characterized by genetic deletion mutation and its complementation (41). Our results showed that mycobacterial lipoproteins have an array of fatty acids with different combinations. Table 5.1 revealed that di-acylated lipopeptides appeared more repeatedly than tri-acylated lipopeptides in *M. leprae* recombinant lipoproteins expressed in *M. smegmatis*. Recent findings showed that *S*. *aureus* containing *lnt* can produce both di- and tri-acylated forms of lipoprotein and the di-acylated forms of lipoproteins are more potent immune modulators (39,40). Despite the fact that mycobacteria possess functionally active Lnt, our data showed that mycobacterial lipoproteins exist as both di-and tri-acylated forms similar to the acylation pattern of lipoproteins in S. aureus (Table 5.1). Drage et al. (10) showed that rLprA, rLprG. rLpqH and rPhoS1 induce high levels of TNF-α production in TLR-2 signaling dependent manner on stimulation of TLR2-/-, TLR1-/- and TLR6-/- macrophages. In that study, the TLR response to rLprG or rLpqH was restricted to TLR-1 while that of LprA

or Phos1 was not required for either TLR-1 or TLR-6. TLR responses of all lipoproteins have varied depending on the involvement of CD14 and CD36 (10). Since all lipoproteins were not biochemically defined and expressed in *M. smegmatis* or *M. tuberculosis* (H37Ra; avirulent strain), it is still obscure how these mycobacterial lipoproteins influence TLR signaling in mycobacterial infection but implies that mycobacterial lipoproteins exist as several isoforms such as di-, tri or non-acylated forms.

Recently, Leng *et al.* (23) revealed that lipoproteins containing unsaturated fatty acid induces higher gene expression of the Th1/Th17-associated cytokines in bone marrow derived macrophages than synthetic lipopeptide containing saturated fatty acid via TLR-2 signaling mechanism, suggesting that different acylation pattern of lipoprotein can induce different immune responses. Interestingly, Q-TOF LC/MS analysis in the present study also showed that some lipopeptides contained the unsaturated fatty acid oleic acid (C18:1) (Table 5.1).

Overall, we concluded that mycobacterial lipoproteins (ML1086, ML2095 and ML2446) expressed in *M. smegmatis* are present as di- or tri-acylated isoforms and also contain a set of fatty acids (palmitic acids, stearic acids, oleic acids and tuberculostearic acids) at the N terminal cysteine. This new information raises two major questions. The first is whether the protein acylation pattern of the virulent mycobacteria, *M. leprae* and *M. tuberculosis*, in the host is identical to that of *M. smegmatis* characterized in the present study or whether one di-/tri acylated isoform is predominant in the *in vivo* grown virulent mycobacteria. The great challenge in addressing this question is that bacterial lipoproteins are generally of low abundance and hydrophobic in the native proteome, making it difficult to define their chemical structures even in the *in vitro* grown

mycobacteria. As previously mentioned in Chapter 4, the most plausible approach to resolve this problem would be the use of recombinant *M. smegmatis* expressing the lipoproteins of *M. tuberculosis* or *M. leprae*, to define the acylation pattern of mycobacterial lipoproteins generated under the conditions that mimic in vivo growth. The second question is whether two different acylated isoforms of lipoproteins with various combinations of fatty acids play different biological roles in the pathogenesis of mycobacterial disease by influencing TLR-2/1 and/or TLR-2/6 signaling. Recently, genetic polymorphisms studies in the TLR-2 gene and TLR-2 related genes showed that the tri-acylated lipoproteins may contribute to the pathogenesis of leprosy but di-acylated lipoproteins play an important role in development of tuberculosis by modulating TLR-2 responses (17,18,20,45). Therefore, it would be of interest to address whether the proportional difference between di- and tri-acylated mycobacterial lipoproteins influences the TLR-2 signaling by manipulating the involvement of TLR-1or TLR-6 and/or other costimulatory molecules and whether these differences result in different disease outcomes. Future work using synthetic di- or tri-acylated lipopeptides containing various combinations of fatty acids should be performed to understand the biological relevance of distinct mycobacterial lipopeptides in terms of the pathogenesis of mycobaterial infections. Indeed, the methodologies and knowledge presented in this dissertation will help to define the immunological role of lipoprotein in the mycobacterial pathogens, M. tuberculosis and M. leprae.

5.6 References

- 1. Belisle, J. T., M. E. Brandt, J. D. Radolf, and M. V. Norgard. 1994. Fatty acids of *Treponema pallidum* and *Borrelia burgdorferi* lipoproteins. J.Bacteriol. **176**:2151-2157.
- 2. Bogialli, S. and C. A. Di. 2009. Recent applications of liquid chromatography-mass spectrometry to residue analysis of antimicrobials in food of animal origin. Anal.Bioanal.Chem. **395**:947-966.
- 3. Bordier, C. 1981. Phase separation of integral membrane proteins in Triton X-114 solution. J.Biol.Chem. 256:1604-1607.
- 4. **Braun, V.** 1975. Covalent lipoprotein from the outer membrane of *Escherichia coli*. Biochim.Biophys.Acta **415**:335-377.
- 5. Brennan, P. J. and H. Nikaido. 1995. The envelope of mycobacteria. Annu.Rev.Biochem. 64:29-63..
- Brightbill, H. D., D. H. Libraty, S. R. Krutzik, R. B. Yang, J. T. Belisle, J. R. Bleharski, M. Maitland, M. V. Norgard, S. E. Plevy, S. T. Smale, P. J. Brennan, B. R. Bloom, P. J. Godowski, and R. L. Modlin. 1999. Host defense mechanisms triggered by microbial lipoproteins through toll-like receptors. Science 285:732-736.
- Chatterjee, D., C. M. Bozic, M. McNeil, and P. J. Brennan. 1991. Structural features of the arabinan component of the lipoarabinomannan of *Mycobacterium tuberculosis*. J.Biol.Chem. 266:9652-9660.
- Danaceau, J. P., M. M. Scott, and M. H. Slawson. 2008. Quantitative confirmation of testosterone and epitestosterone in human urine by LC/Q-ToF mass spectrometry for doping control. J.Mass Spectrom. 43:993-1000.
- 9. Dettmer, K., P. A. Aronov, and B. D. Hammock. 2007. Mass spectrometry-based metabolomics. Mass Spectrom.Rev. 26:51-78.
- Drage, M. G., N. D. Pecora, A. G. Hise, M. Febbraio, R. L. Silverstein, D. T. Golenbock, W. H. Boom, and C. V. Harding. 2009. TLR2 and its co-receptors determine responses of macrophages and dendritic cells to lipoproteins of *Mycobacterium tuberculosis*. Cell Immunol. 258:29-37.
- 11. Gehring, A. J., K. M. Dobos, J. T. Belisle, C. V. Harding, and W. H. Boom. 2004. *Mycobacterium tuberculosis* LprG (Rv1411c): a novel TLR-2 ligand that inhibits human macrophage class II MHC antigen processing. J.Immunol. **173**:2660-2668.
- Gupta, S. D., W. Dowhan, and H. C. Wu. 1991. Phosphatidylethanolamine is not essential for the N-acylation of apolipoprotein in *Escherichia coli*. J.Biol.Chem. 266:9983-9986.
- Hantke, K. and V. Braun. 1973. Covalent binding of lipid to protein. Diglyceride and amidelinked fatty acid at the N-terminal end of the murein-lipoprotein of the *Escherichia coli* outer membrane. Eur.J.Biochem. 34:284-296.
- 14. Harding, C. V. and W. H. Boom. 2010. Regulation of antigen presentation by *Mycobacterium tuberculosis*: a role for Toll-like receptors. Nat.Rev.Microbiol. 8:296-307.
- 15. Hayashi, S. and H. C. Wu. 1990. Lipoproteins in bacteria. J.Bioenerg.Biomembr. 22:451-471.
- Hutchings, M. I., T. Palmer, D. J. Harrington, and I. C. Sutcliffe. 2009. Lipoprotein biogenesis in Gram-positive bacteria: knowing when to hold 'em, knowing when to fold 'em. Trends Microbiol. 17:13-21.
- Johnson, C. M., E. A. Lyle, K. O. Omueti, V. A. Stepensky, O. Yegin, E. Alpsoy, L. Hamann, R. R. Schumann, and R. I. Tapping. 2007. Cutting edge: A common polymorphism impairs cell surface trafficking and functional responses of TLR1 but protects against leprosy. J.Immunol. 178:7520-7524.
- Kang, T. J. and G. T. Chae. 2001. Detection of Toll-like receptor 2 (TLR2) mutation in the lepromatous leprosy patients. FEMS Immunol.Med Microbiol. 31:53-58..
- 19. Khoo, K. H., A. Dell, H. R. Morris, P. J. Brennan, and D. Chatterjee. 1995. Structural definition of acylated phosphatidylinositol mannosides from *Mycobacterium tuberculosis*:

definition of a common anchor for lipomannan and lipoarabinomannan. Glycobiology 5:117-127.

- Khor, C. C., S. J. Chapman, F. O. Vannberg, A. Dunne, C. Murphy, E. Y. Ling, A. J. Frodsham, A. J. Walley, O. Kyrieleis, A. Khan, C. Aucan, S. Segal, C. E. Moore, K. Knox, S. J. Campbell, C. Lienhardt, A. Scott, P. Aaby, O. Y. Sow, R. T. Grignani, J. Sillah, G. Sirugo, N. Peshu, T. N. Williams, K. Maitland, R. J. Davies, D. P. Kwiatkowski, N. P. Day, D. Yala, D. W. Crook, K. Marsh, J. A. Berkley, L. A. O'Neill, and A. V. Hill. 2007. A Mal functional variant is associated with protection against invasive pneumococcal disease, bacteremia, malaria and tuberculosis. Nat.Genet. 39:523-528.
- Krutzik, S. R., M. T. Ochoa, P. A. Sieling, S. Uematsu, Y. W. Ng, A. Legaspi, P. T. Liu, S. T. Cole, P. J. Godowski, Y. Maeda, E. N. Sarno, M. V. Norgard, P. J. Brennan, S. Akira, T. H. Rea, and R. L. Modlin. 2003. Activation and regulation of Toll-like receptors 2 and 1 in human leprosy. Nat.Med 9:525-532.
- 22. Lancioni, C. L., Q. Li, J. J. Thomas, X. Ding, B. Thiel, M. G. Drage, N. D. Pecora, A. G. Ziady, S. Shank, C. V. Harding, W. H. Boom, and R. E. Rojas. 2010. *Mycobacterium tuberculosis* lipoproteins directly regulate human memory CD4+ T cell activation via TLR2/1. Infect.Immun..
- 23. Leng, C. H., H. W. Chen, L. S. Chang, H. H. Liu, H. Y. Liu, Y. P. Sher, Y. W. Chang, S. P. Lien, T. Y. Huang, M. Y. Chen, A. H. Chou, P. Chong, and S. J. Liu. 2010. A recombinant lipoprotein containing an unsaturated fatty acid activates NF-kappaB through the TLR2 signaling pathway and induces a differential gene profile from a synthetic lipopeptide. Mol.Immunol. 47:2015-2021.
- 24. Leproma. http://genolist.pasteur.fr/Leproma/
- MacKenzie, S. L. and L. R. Hogge. 1977. Gas chromatography-mass spectrometry of the N(O)heptafluorobutyryl isobutyl esters of the protein amino acids using electron impact ionisation. J.Chromatogr. 132:485-493.
- Marques, M. A., S. Chitale, P. J. Brennan, and M. C. Pessolani. 1998. Mapping and identification of the major cell wall-associated components of *Mycobacterium leprae*. Infect.Immun. 66:2625-2631.
- Mehrotra, J., A. Mittal, A. K. Rastogi, A. K. Jaiswal, N. K. Bhandari, and S. Sinha. 1999. Antigenic definition of plasma membrane proteins of Bacillus Calmette-Guerin: predominant activation of human T cells by low-molecular-mass integral proteins. Scand.J.Immunol. 50:411-419.
- Mizuno, T. 1979. A novel peptidoglycan-associated lipoprotein found in the cell envelope of Pseudomonas aeruginosa and Escherichia coli. J.Biochem. 86:991-1000.
- Muhlradt, P. F., M. Kiess, H. Meyer, R. Sussmuth, and G. Jung. 1998. Structure and specific activity of macrophage-stimulating lipopeptides from *Mycoplasma hyorhinis*. Infect.Immun. 66:4804-4810.
- Muhlradt, P. F., H. Meyer, and R. Jansen. 1996. Identification of S-(2,3dihydroxypropyl)cystein in a macrophage-activating lipopeptide from *Mycoplasma fermentans*. Biochemistry 35:7781-7786.
- Rezwan, M., T. Grau, A. Tschumi, and P. Sander. 2007. Lipoprotein synthesis in mycobacteria. Microbiology 153:652-658.
- 32. **Rigel, N. W. and M. Braunstein**. 2008. A new twist on an old pathway-accessory Sec systems. Mol.Microbiol. **69**:291-302.
- 33. **Ryhage, R. and Stenhagen E.** 1963. Massspectrometry of long-chain esters in Mass Spectrometry of Organic Ions(McLufferty, F.W., Ed.). Academic Press Inc.399-452.
- Sander, P., M. Rezwan, B. Walker, S. K. Rampini, R. M. Kroppenstedt, S. Ehlers, C. Keller, J. R. Keeble, M. Hagemeier, M. J. Colston, B. Springer, and E. C. Bottger. 2004. Lipoprotein processing is required for virulence of *Mycobacterium tuberculosis*. Mol.Microbiol. 52:1543-1552.

- 35. Sankaran, K., S. D. Gupta, and H. C. Wu. 1995. Modification of bacterial lipoproteins. Methods Enzymol. 250:683-697.
- Sankaran, K. and H. C. Wu. 1994. Lipid modification of bacterial prolipoprotein. Transfer of diacylglyceryl moiety from phosphatidylglycerol. J.Biol.Chem. 269:19701-19706.
- Sankaran, K. and H. C. Wu. 1995. Bacterial prolipoprotein signal peptidase. Methods Enzymol. 248:169-180.
- Sutcliffe, I. C. and D. J. Harrington. 2004. Lipoproteins of *Mycobacterium tuberculosis*: an abundant and functionally diverse class of cell envelope components. FEMS Microbiol.Rev. 28:645-659.
- Sutcliffe, I. C. and R. R. Russell. 1995. Lipoproteins of gram-positive bacteria. J.Bacteriol. 177:1123-1128.
- Tawaratsumida, K., M. Furuyashiki, M. Katsumoto, Y. Fujimoto, K. Fukase, Y. Suda, and M. Hashimoto. 2009. Characterization of N-terminal structure of TLR2-activating lipoprotein in *Staphylococcus aureus*. J.Biol.Chem. 284:9147-9152.
- Tschumi, A., C. Nai, Y. Auchli, P. Hunziker, P. Gehrig, P. Keller, T. Grau, and P. Sander. 2009. Identification of apolipoprotein N-acyltransferase (Lnt) in mycobacteria. J.Biol.Chem. 284:27146-27156.
- 42. Tuberculolist. http://genolist.pasteur.fr/TubercuList/
- 43. Ujihara, T., I. Sakurai, N. Mizusawa, and H. Wada. 2008. A method for analyzing lipidmodified proteins with mass spectrometry. Anal.Biochem. **374**:429-431.
- 44. Wheeler, P. R. and C. Ratledge. 1994. Metabolism of *M. tuberculosis*. In Tuberculosis: Pathogenesis, Protection and Control (Bloom, B.R.) ASM Press353-383.
- 45. Wong, S. H., S. Gochhait, D. Malhotra, F. H. Pettersson, Y. Y. Teo, C. C. Khor, A. Rautanen, S. J. Chapman, T. C. Mills, A. Srivastava, A. Rudko, M. B. Freidin, V. P. Puzyrev, S. Ali, S. Aggarwal, R. Chopra, B. S. Reddy, V. K. Garg, S. Roy, S. Meisner, S. K. Hazra, B. Saha, S. Floyd, B. J. Keating, C. Kim, B. P. Fairfax, J. C. Knight, P. C. Hill, R. A. Adegbola, H. Hakonarson, P. E. Fine, R. M. Pitchappan, R. N. Bamezai, A. V. Hill, and F. O. Vannberg. 2010. Leprosy and the adaptation of human toll-like receptor 1. PLoS.Pathog. 6: e1000979.
- 46. Wu, H. C. and J. J. Lin. 1976. *Escherichia coli* mutants altered in murein lipoprotein. J.Bacteriol. **126**:147-156.
- 47. Yamagata, H., N. Taguchi, K. Daishima, and S. Mizushima. 1983. Genetic characterization of a gene for prolipoprotein signal peptidase in *Escherichia coli*. Mol.Gen.Genet. **192**:10-14.
- 48. Young, D. B. and T. R. Garbe. 1991. Lipoprotein antigens of *Mycobacterium tuberculosis*. Res.Microbiol. 142:55-65.
- Zlotnick, G. W., V. T. Sanfilippo, J. A. Mattler, D. H. Kirkley, R. A. Boykins, and R. C. Seid, Jr. 1988. Purification and characterization of a peptidoglycan-associated lipoprotein from *Haemophilus influenzae*. J.Biol.Chem. 263:9790-9794.

CHAPTER 6 CONCLUSIONS AND FUTURE DIRECTIONS

The WHO-directed MDT strategy introduced in 1982 efficiently sterilized patients and removed much of the biological reservoir of *M. leprae*. However, the number of new cases has been sustained indicating that current diagnosis, the basis of the MDT intervention strategy, is not sufficient to detect *M. leprae* infected individuals. The early diagnosis of leprosy is a foremost priority for more effective control of leprosy and to interrupt *M. leprae* transmission in endemic regions.

M. leprae persist in the host for an extremely long time without presenting the obvious clinical symptoms. The disease progression of leprosy is highly governed by the host immune status and exhibits two contrasting disease spectra (strong CMI in the form of Th1 immunity to humoral Th2 immunity) in affected individuals. However, the physiology, pathogenicity and immunopathogenesis of *M. leprae* in human hosts is far from understood. These knowledge gaps obstruct the development of sensitive and specific diagnosis of leprosy that can distinguish *M. leprae* infection from other mycobacterial infections. Importantly, the long incubation time of *M. leprae* in most affected individuals makes it a challenge to detect asymptomatic leprosy particularly that which could be found in house hold contacts (HHCs) of multibacillary leprosy (MB) patients but mostly in individuals without a known history of close contact with leprosy patients (6-8,38). During the last decade, vigorous research efforts of our own and others

have agreed that CMI-based *in vitro* IFN- γ release assays (IGRA) to *M. leprae* specific antigens are the most promising and pragmatic diagnostic tool to detect individuals in the early stages of *M. leprae* infection.

In general, the identification of applicable antigens has been achieved through numerous chromatographic approaches in order to reduce the complexity of native protein pools and to enrich certain key proteins into isolated fractions. Due to the inability to culture *M. leprae* in the laboratory, this type of proteomic approach becomes impractical as applied to *M. leprae* derived proteins. *M. leprae* specific antigens have been identified from both native *M. leprae* fractions and recombinant proteins or synthetic peptides using simple proteomic approaches combined with *in silico* bioinformatic analyses of mycobacterial genomes. However, the physiological functions of these antigens during disease progression are rarely characterized. In this dissertation, we have identified and characterized a panel of new *M. leprae* derived protein antigens using advanced genomic techniques and developed simple and novel proteomic techniques that allow for the enrichment of antigens present in low abundance and to a very limited amount, applied these in diagnostic settings.

6.1 Leprosy diagnostic antigens

By using postgenomic approaches (see section 1.10), 31 ORFs and hundreds of peptides were identified from 142 "hypothetical unknowns" as *M. leprae* specific antigens; these hypothetical unknowns had not been annotated in the original genome definition. Some of hypothetical unknowns were shown to improve the specificity of IGRA for leprosy diagnosis (5,6,8,38). However, these recombinant hypothetical

unknown proteins and peptides were recognized by a significant number of endemic controls but the *M. leprae*-derived peptides showed considerable variation in reactivity to racially/geographically different study populations (7,8). Most of the problems with these IGRAs originated from low levels of IFN- γ response to hypothetical unknown proteins and particularly peptides derived from hypothetical unknowns.

For the first time, we established the expression status of 136 M. leprae unique proteins (hypothetical unknowns) to address correlation between antigen availability and immunogenicity and eventually to enhance the screening process of novel antigens using the target based qRT-PCR assays (Chapter 2). In our analysis, 60% of the hypothetical unknowns expressed less than 100pg of mRNA and thus were found to be transcriptionally inactive including hypothetical unknowns already studied in the literature (gene expression level was (Table 2.2). Our analysis showed that the low sensitivity of IGRA to previously studied hypothetical unknowns was attributable to low antigen availability, leading to poor T cell recognition of parent proteins by the host immune system. In this dissertation, twenty-six antigen candidates that showed a detectable level of gene expression were identified as the most promising antigens to improve the sensitivity and specificity of IGRA diagnostic tests (Table 2.3). Subsequent seroreactivity to these selective recombinant antigens showed that 9 of the selected antigen candidates were recognized by circulating antibodies in sera of lepromatous leprosy (LL) patients but not in sera of tuberculosis (TB) patients (Figure 2.2). This result suggests that these antigens might be processed and accommodated on the MHC molecule to present to T helper cells. Therefore, these novel recombinant antigens have great potential to improve the sensitivity and specificity of CMI-based diagnosis in the

detection of *M. leprae*-infected individuals with asymptomatic leprosy. All of these purified recombinant antigens are now being evaluated against PBMCs of leprosy patients and HHCs in endemic regions.

The *M. leprae* membrane antigen (MLMA) studied in Chapter 3 induced IFN- γ production of both naïve and memory CD4⁺ T cells and CD8⁺ T cells on stimulation of dendritic cells (23) suggesting that MLMA may contain strong immunogens to improve the sensitivity of IGRA for leprosy diagnosis. However, this MLMA also contains large amounts of lipomannans/lipoarabinomannan (LMs/LAMs) making it difficult to isolate low abundant membrane associated proteins and also causing cross-reactivity with other mycobacteria in CMI based assays. Therefore, the identification and characterization of T cell antigens associated with native membrane proteins had never been fully explored. In this dissertation, we developed simple and novel proteomic and biochemical techniques to create the first generation of membrane associated diagnostic protein antigens (MLMA-SP, alkali soluble proteins of MLMA) from which most of the LAMs/LMs were simultaneously excluded (Figure 3.2). Additionally, these techniques allowed the identification and characterization of lipoproteins as CD1a restricted antigens from this membrane fraction (Chapter 4). IGRA of 16 volunteers having worked in mycobacterial laboratories/leprosy clinics indicated that the MLMA-SP induced higher levels of IFN-y production from peripheral blood mononuclear cells (PBMCs) of the group exposed to leprosy patients, as compared to those individuals with possible environmental mycobacterial exposure (Figure 3.4). This result indicated that this protein pool may contain the immunogenic molecules to mount Th1 immune response but obviously much more work is required. The massive gene decay and reduction in the

M. leprae genome creates a unique protein profile in MLMA. Earlier studies from our laboratory have provided evidence that the unique proteins appearing in MLMA can achieve a distinct antigenic profile different from that caused by other mycobacteria. Therefore, further identification of the individual antigens within MLMA-SP may improve the specificity as well as sensitivity of IGRA. Though this introductory work was seemingly promising, further evaluation of the MLMA-SP in tuberculoid leprosy (TT) patients and HHCs of MB patients is required prior to the application in leprosy endemic clinical settings.

Until now, we and others intensively explored *M. leprae* antigens to be utilized for the early diagnostics of leprosy by proteomic and bioinformatic approaches. In order to utilize CMI based diagnostic tests for routine clinical practice, future follow-up studies of large numbers of HHCs over long periods of time should address the question of whether the individuals indentified by IGRA using highly selected *M. leprae* proteins or peptides, are resistant/susceptible to the development of clinical leprosy. This type of longitudinal study will also allow for understanding the pathogenesis of leprosy and dictate decisions to introduce the chemotherapy to asymptomatic leprosy patients. Although two promising antigens, MLSA-LAM and MLCwA, had been translated into clinical products, major remaining challenges related with these CMI-based assays are to ensure affordability to patients and the easiness of performance combined with affordability in poor resource endemic settings where leprosy is most prevalent. Therefore, the future research to achieve "elimination of global leprosy" should be directed to development of a cost effective and user friendly IGRA applicable in leprosy endemic settings.

6.2 Identification and characterization of lipoproteins as a CD1a restricted antigen

Lipoproteins in bacterial cell surfaces contribute to a variety of physiological features (section 1.7.1) but appear to be low abundant proteins mostly within membranes. However, the N-terminal acylated moiety of lipoproteins is primarily responsible for the activation of innate immunity as well as the expression of MHC-I, II and CD1 on the cell surface, leading to adaptive immunity (9,18,19,24). Currently a few studies have revealed that mycobacterial lipoproteins are directly involved in adaptive immunity by serving as MHC-II restricted T cell antigens and enhancing memory T cell immunity (20,35). Therefore, mycobacterial lipoproteins containing an N-acyl di-O-acylglyceryl-cysteine unit have been long highlighted in terms of developing new vaccines or immunomodulatory therapies for controlling intracellular infectious diseases.

As previously discussed (Chapters 1 and 4), CD1-restricted T cells in response to mycobacterial lipids efficiently and rapidly induced Th1 immunity by secreting high levels of IFN- γ and proliferating cytolytic T cells to kill intracellular pathogens such as *M. tuberculosis* and *M. leprae* (1,10,26,39). Toll like receptor-1 (TLR-1), TLR-2 and CD1s in skin lesions of tuberculoid leprosy (TT) patients are more strongly expressed than in those of LL patients (19,36). TLR-2 signaling on stimulation with synthetic bacterial lipopeptides efficiently mediates to differentiate CD1⁺dendritic cells from PBMCs (19). The CD1a restricted Langerhans cells, an abundant dendritic cell in TT skin lesions, efficiently present mycobacterial lipids to CD1a⁺ T cells which can produce high levels of IFN- γ and granulysin (31,37), suggesting that bacterial lipoprotein may serve as CD1a restricted antigens or could be involved in CD1a antigen presentation via Langerhans cells.

Didehydroxymycobactin, a nonribosomal lipopeptide related to the mycobacterial mycobactin siderophores, is the sole known antigen presented by CD1a molecules in *M. tuberculosis* infection. Bacterial lipoproteins share structural similarity with didehydroxy-mycobactin. Therefore, bacterial lipoproteins containing lipids covalently linked to cysteine have been proposed as CD 1 restricted antigens and are believed to be important in the context of vaccines for controlling intracellular pathogens; however, this proposition has not been proven yet.

The proteomic approach to the identification of native proteins based on 2 D gel analysis or LC-MS/MS mainly estimates protein abundance based on peptide identification. However many hydrophobic proteins in low abundance are difficult to identify due to inability to enrich and/or resistance to common trypsin digestion. Therefore, these low abundant proteins have been neglected in terms of their functional role in host pathogen interactions. Growing evidence is showing that post-translationally modified mycobacterial proteins play an important role in the pathogenesis of mycobacterial disease as well as induction of protective Th1 immunity (4,14,30,32-34). However, many of these proteins, particularly lipoproteins, are or low abundance and are enriched in the hydrophobic membrane fractions.

This present work developed simple proteomic and chemical methodologies to enrich lipoproteins of low abundance from MLMA but excluding all lipid components that themselves were often found to be CD1 restricted antigens (Chapters 4 and 5). Subsequent T cell reactivity to delipidated *M. leprae* native proteins (devoid of noncovalently linked lipids) combined with proteolysis revealed that bacterial lipoprotein(s) were a new class of CD1a restricted antigens recognized by the CD1a

restricted T cell line, LCD4.15, derived from leprosy patients (Figures 4.3, 4.4 and 4.5). This was the first provisional evidence that CD1 could present a lipopeptide derived from natural lipoproteins in an intracellular bacteria. However, recombinant ML1086, ML2095 and ML2446 which were identified as CD1a restricted antigens by proteomic and genomic analyses, failed to activate LCD4.15 suggesting that these three lipoproteins expressed in *M. smegmatis* could not be the antigen to activate the LCD4.15 T cell clone via CD1a presentation (Figure 4.9). Since the identification of these lipoproteins was highly dependent on bioinformatic analysis, it raised the question of whether these lipoproteins are truly acylated in the mycobacterial proteome.

There are convincing data to suggest that polymorphisms in the TLR-2 gene and TLR-2 related genes may contribute to the pathogenesis of mycobacterial disease by modulating TLR-2 responses to di- or tri-acylated lipoproteins (13,15,17,45). Especially, the TLR-1 I602S SNP was associated with a decreased incidence of leprosy (13) but TIRAP S180L SNP related to TLR-2/6 signaling protected against pneumococcal disease, bacteremia, malaria and tuberculosis (17). The results indicated that different TLR-2/1 or TLR-2/6 responses by sensing di- or tri-acylated lipoproteins may differently affect the progression of mycobacterial diseases (4,12,14,18,30,32). Prolonged TLR-2 signaling by lipoproteins are known to inhibit the expression of MHC class II molecules (4,14,30,32) contributing to the pathogenesis of mycobacterial disease by the regulation of anti-mycobacterial activities including phagolysosomal maturation, apoptosis and antigen presentation (12). Several mycobacterial protein, MPT-32, was found to be capable of capturing pulmonary C-type lectin receptor, which is often used for immune

suppression by pathogens (34) (see section 1.7.2). This information implies that the glycosylated parts of mycobacterial lipoproteins also contribute to immune evasion mediated by prolonged TLR-2 response to their acylated functions. Since mycobacterial lipoproteins which were used in previous immunological studies have never been structurally characterized, the immunological roles of lipoproteins in mycobacterial infection are still unclear.

In order to encourage immune mechanistic study of TLR-2 responses to mycobacterial lipoproteins, we established several methodologies to define the chemical structure of N-terminal acylated portion of lipoproteins (Chapter 5). The major challenges of MS based analysis of lipoproteins are the low recovery and low ionization efficiency of lipopeptides due to the long hydrocarbon chains of lipoproteins (44). Here, we showed that proteolysis of the lipoproteins followed by lipid extraction allowed enhanced recovery of the resulting lipopeptides. The high mass accurate Q-TOF LC/MS detection made it possible to identify the full spectrum of lipopeptides derived from mycobacterial lipoproteins. Our study (Figure 5.3) and that of Tschumi et al. (43) revealed that mycobacterial lipoproteins have an array of fatty acids similar to those of mycobacterial phospholipids (2,16,43). The tuberculostearic acid is predominantly associated with phosphatidyl inositol, the phosphatidyl inositol mannosides and LM and LAM (16), suggesting that these characteristic mycobacterial phospholipids could be the source of acyl functions of mycobacterial lipoproteins. Interestingly, our results showed that mycobacterial lipoproteins appeared as di- and tri-acylated isoforms, and had an array of fatty acids including palmitic acid, stearic acid, oleic acid or tuberculosteraric acid (Table 5.1). The di-acylated lipopeptides appeared more frequently than tri-acylated

lipopeptides. Currently, some literature has revealed that the di-acylated forms of lipoprotein are more potent immune modulators as compared to tri-acylated forms (40,41). Therefore, it will be important to study whether the coapperance of di- and tri-acylated isoforms of lipoproteins with various proportions in *M. leprae* or *M. tuberculosis* could play different biological roles in the pathogenesis of mycobacterial disease by affecting TLR-2/1 and/or TLR-2/6 signaling.

The lipoproteins acylated with unsaturated fatty acids induced higher gene expression level of the Th1/Th17-associated cytokines compared to that of synthetic lipopeptides containing saturated fatty acids (22). This new information suggests that different acylation pattern of lipoproteins can induce different immune responses via TLR-2 signaling. Th17 cells particularly produce high levels of IL-17 to promote the recruitment of neutrophils and to result in tissue damage, but efficiently control *M*. *tuberculosis* in the early stages of infection (42). However, IL-17 in chronic infection mediates a shift of Th1 immunity to Th2 or T regulatory immunity which favors the pathogenesis of mycobacterial lipoproteins, which we showed in this study, may differentially influence TLR-2 signaling of phagocytes and mediate protective Th1 immunity or pathogenic Th2 immunity to *M. leprae* infection. Therefore, future work should focus on understanding the biological relevance of these biochemically defined lipoproteins in the pathogenesis of leprosy and tuberculosis.

An unsolved question from our current studies is the nature of CD1a restricted antigens recognized by LCD 4.15. As discussed in Chapter 4, the activation of this T cell clone was highly specific to *M. leprae*, which is obligatorily capable of growing *in vivo*.

Recent mycobacterial lipidomics studies have shown that lipids of *in vivo* grown mycobacteria were differently modified from those of *in vitro* grown mycobacteria such as in PDIM, sulfolipid-1, and phenolic glycolipids (11), which are abundantly found in the cell envelope of mycobacteria but are also known as virulence factors. Since the chemical structures of mycobacterial CD1 antigens have rarely been defined or even found in the case of *in vitro* grown mycobacteria (21,25,27-29), we speculated that M. *leprae* solely grown under pressure of the host immune system may result in a unique acylation pattern of lipoproteins leading to *M. leprae* specific activation of LCD4.15. Therefore, further biochemical analysis of mycobacterial lipoproteins from in vivo versus in vitro grown mycobacteria could allow identifying the CD1a antigens as intracellular lipid antigens. Since many of the mycobacterial cell wall components are important contributors to mycobaterial pathogenesis, CD1a restricted antigens may further allow an understanding of host-M. leprae interaction. Finally, the biochemical, genetic and proteomic methodologies and knowledge presented in this dissertation will allow further understanding the biological roles of mycobacterial lipoproteins in the pathogenesis of mycobacterial disease such as tuberculosis and leprosy.

6.3 References

- 1. Beckman, E. M., S. A. Porcelli, C. T. Morita, S. M. Behar, S. T. Furlong, and M. B. Brenner. 1994. Recognition of a lipid antigen by CD1-restricted ab+ T cells. Nature **372**:691-694.
- Brennan, P. J. and H. Nikaido. 1995. The envelope of mycobacteria. Annu.Rev.Biochem. 64:29-63.
- Espitia, C., I. Cervera, R. Gonzalez, and R. Mancilla. 1989. A 38-kD Mycobacterium tuberculosis antigen associated with infection. Its isolation and serologic evaluation. Clin.Exp.Immunol. 77:373-377.
- Gehring, A. J., K. M. Dobos, J. T. Belisle, C. V. Harding, and W. H. Boom. 2004. *Mycobacterium tuberculosis* LprG (Rv1411c): a novel TLR-2 ligand that inhibits human macrophage class II MHC antigen processing. J.Immunol. 173:2660-2668. doi:173/4/2660 [pii].
- Geluk, A., M. R. Klein, K. L. Franken, K. E. van Meijgaarden, B. Wieles, K. C. Pereira, S. Buhrer-Sekula, P. R. Klatser, P. J. Brennan, J. S. Spencer, D. L. Williams, M. C. Pessolani, E. P. Sampaio, and T. H. Ottenhoff. 2005. Postgenomic approach to identify novel *Mycobacterium leprae* antigens with potential to improve immunodiagnosis of infection. Infect.Immun. 73:5636-5644.
- 6. Geluk, A. and T. H. Ottenhoff. 2006. HLA and leprosy in the pre and postgenomic eras. Hum.Immunol. 67:439-445.
- Geluk, A., J. S. Spencer, K. Bobosha, M. C. Pessolani, G. M. Pereira, S. Banu, N. Honore, S. T. Reece, M. MacDonald, B. R. Sapkota, C. Ranjit, K. L. Franken, M. Zewdie, A. Aseffa, R. Hussain, M. M. Stefani, S. N. Cho, L. Oskam, P. J. Brennan, and H. M. Dockrell. 2009. From genome-based *in silico* predictions to *ex vivo* verification of leprosy diagnosis. Clin.Vaccine Immunol. 16:352-359.
- Geluk, A., J. van der Ploeg, R. O. Teles, K. L. Franken, C. Prins, J. W. Drijfhout, E. N. Sarno, E. P. Sampaio, and T. H. Ottenhoff. 2008. Rational combination of peptides derived from different *Mycobacterium leprae* proteins improves sensitivity for immunodiagnosis of M. leprae infection. Clin.Vaccine Immunol. 15:522-533.
- 9. Harding, C. V. and W. H. Boom. 2010. Regulation of antigen presentation by *Mycobacterium tuberculosis*: a role for Toll-like receptors. Nat.Rev.Microbiol. 8:296-307.
- Hava, D. L., N. van der Wel, N. Cohen, C. C. Dascher, D. Houben, L. Leon, S. Agarwal, M. Sugita, Z. M. van, S. C. Kent, H. Shams, P. J. Peters, and M. B. Brenner. 2008. Evasion of peptide, but not lipid antigen presentation, through pathogen-induced dendritic cell maturation. Proc.Natl.Acad.Sci U.S.A 105:11281-11286.
- 11. Jain, M. and J. S. Cox. 2005. Interaction between polyketide synthase and transporter suggests coupled synthesis and export of virulence lipid in *M. tuberculosis*. PLoS.Pathog. 1:e2.
- 12. Jo, E. K., C. S. Yang, C. H. Choi, and C. V. Harding. 2007. Intracellular signalling cascades regulating innate immune responses to Mycobacteria: branching out from Toll-like receptors. Cell Microbiol. 9:1087-1098.
- Johnson, C. M., E. A. Lyle, K. O. Omueti, V. A. Stepensky, O. Yegin, E. Alpsoy, L. Hamann, R. R. Schumann, and R. I. Tapping. 2007. Cutting edge: A common polymorphism impairs cell surface trafficking and functional responses of TLR1 but protects against leprosy. J.Immunol. 178:7520-7524.
- Jung, S. B., C. S. Yang, J. S. Lee, A. R. Shin, S. S. Jung, J. W. Son, C. V. Harding, H. J. Kim, J. K. Park, T. H. Paik, C. H. Song, and E. K. Jo. 2006. The mycobacterial 38-kilodalton glycolipoprotein antigen activates the mitogen-activated protein kinase pathway and release of proinflammatory cytokines through Toll-like receptors 2 and 4 in human monocytes. Infect.Immun. 74:2686-2696.
- 15. Kang, T. J. and G. T. Chae. 2001. Detection of Toll-like receptor 2 (TLR2) mutation in the lepromatous leprosy patients. FEMS Immunol.Med Microbiol. **31**:53-58.

- 16. **Khoo, K. H., A. Dell, H. R. Morris, P. J. Brennan, and D. Chatterjee**. 1995. Structural definition of acylated phosphatidylinositol mannosides from *Mycobacterium tuberculosis*: definition of a common anchor for lipomannan and lipoarabinomannan. Glycobiology **5**:117-127.
- Khor, C. C., S. J. Chapman, F. O. Vannberg, A. Dunne, C. Murphy, E. Y. Ling, A. J. Frodsham, A. J. Walley, O. Kyrieleis, A. Khan, C. Aucan, S. Segal, C. E. Moore, K. Knox, S. J. Campbell, C. Lienhardt, A. Scott, P. Aaby, O. Y. Sow, R. T. Grignani, J. Sillah, G. Sirugo, N. Peshu, T. N. Williams, K. Maitland, R. J. Davies, D. P. Kwiatkowski, N. P. Day, D. Yala, D. W. Crook, K. Marsh, J. A. Berkley, L. A. O'Neill, and A. V. Hill. 2007. A Mal functional variant is associated with protection against invasive pneumococcal disease, bacteremia, malaria and tuberculosis. Nat.Genet. 39:523-528.
- Krutzik, S. R., M. T. Ochoa, P. A. Sieling, S. Uematsu, Y. W. Ng, A. Legaspi, P. T. Liu, S. T. Cole, P. J. Godowski, Y. Maeda, E. N. Sarno, M. V. Norgard, P. J. Brennan, S. Akira, T. H. Rea, and R. L. Modlin. 2003. Activation and regulation of Toll-like receptors 2 and 1 in human leprosy. Nat.Med 9:525-532.
- 19. Krutzik, S. R., B. Tan, H. Li, M. T. Ochoa, P. T. Liu, S. E. Sharfstein, T. G. Graeber, P. A. Sieling, Y. J. Liu, T. H. Rea, B. R. Bloom, and R. L. Modlin. 2005. TLR activation triggers the rapid differentiation of monocytes into macrophages and dendritic cells. Nat.Med 11:653-660.
- Lancioni, C. L., Q. Li, J. J. Thomas, X. Ding, B. Thiel, M. G. Drage, N. D. Pecora, A. G. Ziady, S. Shank, C. V. Harding, W. H. Boom, and R. E. Rojas. 2010. *Mycobacterium tuberculosis* lipoproteins directly regulate human memory CD4+ T cell activation via TLR2/1. Infect.Immun.
- Layre, E., A. Collmann, M. Bastian, S. Mariotti, J. Czaplicki, J. Prandi, L. Mori, S. Stenger, L. G. De, G. Puzo, and M. Gilleron. 2009. Mycolic acids constitute a scaffold for mycobacterial lipid antigens stimulating CD1-restricted T cells. Chem.Biol. 16:82-92.
- 22. Leng, C. H., H. W. Chen, L. S. Chang, H. H. Liu, H. Y. Liu, Y. P. Sher, Y. W. Chang, S. P. Lien, T. Y. Huang, M. Y. Chen, A. H. Chou, P. Chong, and S. J. Liu. 2010. A recombinant lipoprotein containing an unsaturated fatty acid activates NF-kappaB through the TLR2 signaling pathway and induces a differential gene profile from a synthetic lipopeptide. Mol.Immunol. 47:2015-2021.
- 23. Maeda, Y., M. Gidoh, N. Ishii, C. Mukai, and M. Makino. 2003. Assessment of cell mediated immunogenicity of *Mycobacterium leprae*-derived antigens. Cell Immunol. 222:69-77.
- Maeda, Y., M. Makino, D. C. Crick, S. Mahapatra, S. Srisungnam, T. Takii, Y. Kashiwabara, and P. J. Brennan. 2002. Novel 33-kilodalton lipoprotein from *Mycobacterium leprae*. Infect.Immun. 70:4106-4111.
- 25. Matsunaga, I., A. Bhatt, D. C. Young, T. Y. Cheng, S. J. Eyles, G. S. Besra, V. Briken, S. A. Porcelli, C. E. Costello, W. R. Jacobs, Jr., and D. B. Moody. 2004. *Mycobacterium tuberculosis* pks12 produces a novel polyketide presented by CD1c to T cells. J.Exp.Med **200**:1559-1569.
- 26. Montoya, D. and R. L. Modlin. 2010. Learning from leprosy: insight into the human innate immune response. Adv.Immunol. 105:1-24.
- Moody, D. B., B. B. Reinhold, M. R. Guy, E. M. Beckman, D. E. Frederique, S. T. Furlong, S. Ye, V. N. Reinhold, P. A. Sieling, R. L. Modlin, G. S. Besra, and S. A. Porcelli. 1997. Structural requirements for glycolipid antigen recognition by CD1b-restricted T cells. Science 278:283-286.
- Moody, D. B., T. Ulrichs, W. Muhlecker, D. C. Young, S. S. Gurcha, E. Grant, J. P. Rosat, M. B. Brenner, C. E. Costello, G. S. Besra, and S. A. Porcelli. 2000. CD1c-mediated T-cell recognition of isoprenoid glycolipids in *Mycobacterium tuberculosis* infection. Nature 404:884-888.
- Moody, D. B., D. C. Young, T. Y. Cheng, J. P. Rosat, C. Roura-Mir, P. B. O'Connor, D. M. Zajonc, A. Walz, M. J. Miller, S. B. Levery, I. A. Wilson, C. E. Costello, and M. B. Brenner. 2004. T cell activation by lipopeptide antigens. Science 303:527-531.
- 30. Noss, E. H., R. K. Pai, T. J. Sellati, J. D. Radolf, J. Belisle, D. T. Golenbock, W. H. Boom, and C. V. Harding. 2001. Toll-like receptor 2-dependent inhibition of macrophage class II MHC

expression and antigen processing by 19-kDa lipoprotein of *Mycobacterium tuberculosis*. J.Immunol. **167**:910-918.

- 31. Ochoa, M. T., S. Stenger, P. A. Sieling, S. Thoma-Uszynski, S. Sabet, S. Cho, A. M. Krensky, M. Rollinghoff, S. E. Nunes, A. E. Burdick, T. H. Rea, and R. L. Modlin. 2001. T-cell release of granulysin contributes to host defense in leprosy. Nat.Med 7:174-179. d
- 32. Pecora, N. D., A. J. Gehring, D. H. Canaday, W. H. Boom, and C. V. Harding. 2006. Mycobacterium tuberculosis LprA is a lipoprotein agonist of TLR2 that regulates innate immunity and APC function. J.Immunol. 177:422-429.
- Pennini, M. E., R. K. Pai, D. C. Schultz, W. H. Boom, and C. V. Harding. 2006. *Mycobacterium tuberculosis* 19-kDa lipoprotein inhibits IFN-gamma-induced chromatin remodeling of MHC2TA by TLR2 and MAPK signaling. J.Immunol. 176:4323-4330.
- 34. Ragas, A., L. Roussel, G. Puzo, and M. Riviere. 2007. The *Mycobacterium tuberculosis* cellsurface glycoprotein apa as a potential adhesin to colonize target cells via the innate immune system pulmonary C-type lectin surfactant protein A. J.Biol.Chem. **282**:5133-5142.
- 35. Sieling, P. A., P. J. Hill, K. M. Dobos, K. Brookman, A. M. Kuhlman, M. Fabri, S. R. Krutzik, T. H. Rea, D. G. Heaslip, J. T. Belisle, and R. L. Modlin. 2008. Conserved mycobacterial lipoglycoproteins activate TLR2 but also require glycosylation for MHC class II-restricted T cell activation. J.Immunol. 180:5833-5842.
- Sieling, P. A., D. Jullien, M. Dahlem, T. F. Tedder, T. H. Rea, R. L. Modlin, and S. A. Porcelli. 1999. CD1 expression by dendritic cells in human leprosy lesions: correlation with effective host immunity. J.Immunol. 162:1851-1858.
- 37. Sieling, P. A., J. B. Torrelles, S. Stenger, W. Chung, A. E. Burdick, T. H. Rea, P. J. Brennan, J. T. Belisle, S. A. Porcelli, and R. L. Modlin. 2005. The human CD1-restricted T cell repertoire is limited to cross-reactive antigens: implications for host responses against immunologically related pathogens. J.Immunol. 174:2637-2644. d
- Spencer, J. S., H. M. Dockrell, H. J. Kim, M. A. Marques, D. L. Williams, M. V. Martins, M. L. Martins, M. C. Lima, E. N. Sarno, G. M. Pereira, H. Matos, L. S. Fonseca, E. P. Sampaio, T. H. Ottenhoff, A. Geluk, S. N. Cho, N. G. Stoker, S. T. Cole, P. J. Brennan, and M. C. Pessolani. 2005. Identification of specific proteins and peptides in *Mycobacterium leprae* suitable for the selective diagnosis of leprosy. J.Immunol. 175:7930-7938.
- Stenger, S., D. A. Hanson, R. Teitelbaum, P. Dewan, K. R. Niazi, C. J. Froelich, T. Ganz, S. Thoma-Uszynski, A. Melian, C. Bogdan, S. A. Porcelli, B. R. Bloom, A. M. Krensky, and R. L. Modlin. 1998. An antimicrobial activity of cytolytic T cells mediated by granulysin. Science 282:121-125.
- Sutcliffe, I. C. and R. R. Russell. 1995. Lipoproteins of gram-positive bacteria. J.Bacteriol. 177:1123-1128.
- 41. Tawaratsumida, K., M. Furuyashiki, M. Katsumoto, Y. Fujimoto, K. Fukase, Y. Suda, and M. Hashimoto. 2009. Characterization of N-terminal structure of TLR2-activating lipoprotein in *Staphylococcus aureus*. J.Biol.Chem. **284**:9147-9152.
- 42. Torrado, E. and A. M. Cooper. 2010. IL-17 and Th17 cells in tuberculosis. Cytokine Growth Factor Rev. 21:455-462.
- Tschumi, A., C. Nai, Y. Auchli, P. Hunziker, P. Gehrig, P. Keller, T. Grau, and P. Sander. 2009. Identification of apolipoprotein N-acyltransferase (Lnt) in mycobacteria. J.Biol.Chem. 284:27146-27156.
- 44. Ujihara, T., I. Sakurai, N. Mizusawa, and H. Wada. 2008. A method for analyzing lipidmodified proteins with mass spectrometry. Anal.Biochem. **374**:429-431.
- 45. Wong, S. H., S. Gochhait, D. Malhotra, F. H. Pettersson, Y. Y. Teo, C. C. Khor, A. Rautanen, S. J. Chapman, T. C. Mills, A. Srivastava, A. Rudko, M. B. Freidin, V. P. Puzyrev, S. Ali, S. Aggarwal, R. Chopra, B. S. Reddy, V. K. Garg, S. Roy, S. Meisner, S. K. Hazra, B. Saha, S. Floyd, B. J. Keating, C. Kim, B. P. Fairfax, J. C. Knight, P. C. Hill, R. A. Adegbola, H. Hakonarson, P. E. Fine, R. M. Pitchappan, R. N. Bamezai, A. V. Hill, and F. O. Vannberg. 2010. Leprosy and the adaptation of human toll-like receptor 1. PLoS.Pathog. 6:e1000979.